LATE METABOLIC CONSEQUENCES OF CHILDHOOD CANCER

Vincent G. Pluimakers

LATE METABOLIC CONSEQUENCES OF CHILDHOOD CANCER

Vincent G. Pluimakers

© copyright Vincent Pluimakers, 2023

Lay-out and printing: ProefschriftMaken || www.proefschriftmaken.nl Cover design: Midjourney Bot

DOI: https://doi.org/10.33540/1778

All rights reserved. No part of this publication may be reproduced, stored in a retrieval system or transmitted, in any form or by any means, electronic, mechanical, photocopying, recording or otherwise, without prior permission of the author or the copyright-owning journals for previous published chapters.

Late metabolic consequences of childhood cancer

Metabole lange-termijnbijwerkingen van kinderkanker

(met een samenvatting in het Nederlands)

Proefschrift

ter verkrijging van de graad van doctor aan de Universiteit Utrecht op gezag van de rector magnificus, prof. dr. H.R.B.M. Kummeling, ingevolge het besluit van het college voor promoties in het openbaar te verdedigen op dinsdag 24 oktober 2023 des middags te 4.15 uur

door

Vincent George Pluimakers Geboren op 25 februari 1991

te Utrecht

Promotor:

Prof. dr. M.M. van den Heuvel-Eibrink

Copromotoren:

Dr. S.J.C.M.M. Neggers Dr. G.O.R. Janssens

Beoordelingscommissie:

Prof. dr. A.J. van der Lelij Prof. dr. M.C. Minnema Prof. dr. M.M. van Noesel Prof. dr. A.G. Uitterlinden Prof. dr. F.L.J. Visseren (voorzitter)

TABLE OF CONTENTS

	Introduction	9
Chapter 1.	Metabolic syndrome as cardiovascular risk factor in childhood cancer survivors <i>Critical Reviews in Oncology/Hematology, 2019 Jan;133:129-141</i>	23
Chapter 2.	Metabolic syndrome detection with biomarkers in childhood cancer survivors <i>Endocrine Connections, 2020 Jun;9:1-11</i>	61
Chapter 3.	Can biomarkers be used to improve diagnosis and prediction of metabolic syndrome in childhood cancer survivors? A systematic review <i>Obesity Reviews, 2021 Nov;22(11):e13312.</i>	87
Chapter 4.	Metabolic syndrome parameters, determinants, and biomarkers in adult survivors of childhood cancer: protocol for the Dutch Childhood Cancer Survivor Study on Metabolic Syndrome (Dutch LATER METS study) <i>JMIR Research Protocols, 2021 Jan;10(1):e21256</i>	137
Chapter 5.	Prevalence, risk factors and optimal way to determine overweight, obesity and morbid obesity, in the first Dutch cohort of 2,338 very long-term survivors of childhood cancer: a DCCSS-LATER study <i>Submitted</i>	157
Chapter 6.	Genetic susceptibility to treatment related dyslipidemia in adult long-term survivors of childhood cancer: a report from the Childhood Cancer Survivor Study and the St Jude Lifetime cohorts <i>In preparation</i>	201
Chapter 7.	Perspectives on follow-up care and research for childhood cancer survivors: results from an international SIOP meet-the-expert questionnaire in Kyoto, 2018 Japanese Journal of Clinical Oncology, 2021 Oct 5;51(10):1554-1560	239
Chapter 8.	Discussion and future perspectives	261

Summary	279
Nederlandse samenvatting	283
Curriculum vitae	287
PhD portfolio	288
List of publications	289
Dankwoord	291

INTRODUCTION

CHILDHOOD CANCER AND SURVIVAL

Annually, more than 400,000 children worldwide are diagnosed with cancer⁽¹⁾. Forty to 45% of these diagnoses include hematologic malignancies (e.g., several types of leukemias and lymphomas), 30-35% are solid tumors (e.g., kidney, bone and muscle tumors) and 20-25% are brain and peripheral nervous system tumors⁽¹⁾. Until the 1960's most childhood cancer diagnoses would almost certainly lead to death. The development of treatment strategies, including chemotherapy, radiotherapy, stem cell transplantation and, more recently, immunotherapy, made it possible to cure most childhood cancer patients. Advances in supportive care and better risk group stratification over time further increased survival rates for most of the pediatric cancer types. These treatments in young and developing children are often intensive and can take several years, but the results are impressive. To date, more than 80% of children with cancer in high-income countries survive^(2, 3).

In The Netherlands, 600 childhood cancer patients are diagnosed annually⁽⁴⁾. Pediatric oncology centers started treating these children in the 1960s, and protocols were developed in clinical trials on a national level. From 1972, this started with protocols for leukemia treatment, coordinated by the Dutch Childhood Leukemia Group, and in 2002 this was expanded to brain and solid tumors by the Dutch Childhood Oncology Group. In 2014, centralization of pediatric oncology care on a national level started by centralizing care for children with solid tumors in the Princess Maxima Center for Pediatric Oncology care and research became centralized in this hospital. Rising survival rates have led to an increasing cohort of childhood cancer survivors. Currently, more than 15,000 childhood cancer survivors are alive in The Netherlands⁽⁴⁾, and a dedicated outpatient clinic has been set up in the Princess Maxima Center.

LONG-TERM TOXICITY OF CHILDHOOD CANCER TREATMENT

These childhood cancer survival rates, however, reflect only the *five-year* survival. Recent studies showed that in the 25 years beyond that moment, an additional 20% of childhood cancer survivors die⁽⁵⁾. This indicates that one in five survivors will die at a substantially younger age than their peers in the general population. In the first years after the five-year survival moment, this is often caused by cancer relapse. Moreover, severe adverse effects of their cancer treatment are responsible for this excess mortality. The two most important late side-effects leading to early death are secondary tumors

in radiation exposed tissues and cardiovascular disease due to damage caused by chemo- and radiotherapy $^{(6,7)}$.

Additionally, significant morbidity, that can affect all organ systems, occurs. This includes pulmonary disease, impaired motor performance, for instance after amputation or due to paralysis, ototoxicity including deafness and tinnitus, cardiotoxicity, and kidney failure. Among the most common late effects are several endocrine sequelae such as growth hormone deficiency, infertility, impaired bone health and body composition sequelae including sarcopenia and adiposity. More general consequences include fatigue, neuropsychological adverse effects, accelerated aging and frailty. Previous studies showed that 75% of survivors develops at least one adverse event, which was classified as severe in more than half of these, and 50% developed three or more adverse events^(8, 9). This seems to be a continuously increasing risk, as illustrated by a study from the United States which revealed that, by the age of 50 years, a survivor had on average *seventeen* chronic health conditions, of which *five severe*⁽¹⁰⁾. These numbers were twice as high as in the general population, clearly illustrating the excess morbidity among long-term childhood cancer survivors.

METABOLIC SYNDROME

One of the conditions that needs attention in childhood cancer survivors is metabolic syndrome. It is a cluster of conditions that occur together and aggravate each other: overweight, dyslipidemia, insulin resistance and hypertension.

Overweight is an unhealthy weight caused by the excess accumulation of fat. It is usually described as weight in relation to height (body mass index (BMI) in kg/m²), with overweight defined as BMI ≥ 25 kg/m² and obesity ≥ 30 kg/m²⁽¹¹⁾. BMI can however underestimate the true adiposity status or overestimate overweight in muscular people. Therefore, another overweight measurement commonly used is waist circumference (WC), which is specifically directed at measuring abdominal, unhealthy fat⁽¹²⁾. An unhealthy WC is usually described per population, e.g., ≥ 94 cm for Caucasian men and ≥ 80 cm for Caucasian women⁽¹³⁾, or as a single, higher threshold of ≥ 102 cm in men and ≥ 88 cm in women⁽¹⁴⁾.

Dyslipidemia refers to unhealthy levels of one or more blood lipids. Routinely used parameters are high low-density-lipoprotein (LDL) cholesterol (\geq 4.1mmol/L), high triglycerides (\geq 1.7mmol/L), high total cholesterol (\geq 6.2mmol/L), and low high-density-lipoprotein (HDL) cholesterol (\leq 1.0mmol/L in men and \leq 1.3mmol/L in women)^(15, 16).

Insulin resistance is a condition in which muscles, fat and liver respond less to insulin. It causes the pancreas to produce more insulin to maintain glucose uptake in the cell. Insulin resistance is usually defined as a fasting plasma glucose between 5.6 and 7 mmol/L, indicating that the pancreas is unable to produce enough insulin to maintain healthy glucose levels⁽¹⁷⁾. This is also called prediabetes, whereas diabetes mellitus is defined as fasting glucose $\geq 7 \text{mmol/L}^{(17)}$.

In patients with arterial hypertension the systolic and/or diastolic blood pressure are consistently too high. Depending on location (office or ambulatory) and time (during the day or 24 hours) of measurement commonly used thresholds for hypertension are 130, 135 or 140mmHg systolic and 80, 85 or 90mmHg diastolic⁽¹⁸⁾.

The central pathophysiologic mechanism of metabolic syndrome is a vicious circle of overweight leading to insulin resistance and vice versa, thereby also causing dyslipidemia and hypertension (Figure 1). Excess adipose tissue secretes an overabundance of free fatty acids, which increase gluconeogenesis in the liver and make muscles more resistant to insulin, both of which cause the pancreas to produce more insulin. Hyperinsulinemia and insulin resistance lead to lipogenesis, thus weight gain, as well as increased release of free fatty acids from this adipose tissue⁽¹⁹⁻²¹⁾. Additionally, dyslipidemia is caused by the accumulation of abundant free fatty acids in the liver, leading to steatosis hepatis, overproduction of very-low density lipoprotein cholesterol and consequent other lipid abnormalities including increased LDL cholesterol and triglycerides and decreased HDL cholesterol⁽²²⁾. Also, hyperinsulinemia increases sodium reabsorption in the kidney and sympathetic nervous system activity, adipose tissue secretes more leptin which also activates the sympathetic nervous system, and adipocytes have an aldosterone releasing effect on the adrenal gland which also increases sodium reabsorption. All these mechanisms contribute to the development of hypertension. Overeating and a sedentary lifestyle have led to a drastic worldwide increase in the prevalence of metabolic syndrome. In a large study in The Netherlands ten years ago the prevalence was 25%⁽²³⁾, and this is also the expected current worldwide prevalence⁽²⁴⁾.



Figure 1. Pathophysiology of metabolic syndrome

Since its inception, multiple definitions have been used to classify the metabolic syndrome. These definitions used different methods and thresholds to measure the four components overweight, dyslipidemia, insulin resistance, and hypertension. Also, some definitions used overweight as an essential criterion, whereas others regarded all components equally important. One definition that is frequently used in clinical practice and research was released by the National Cholesterol Education Program Adult Treatment Panel III (NCEP ATP III) in 2001 (Table 1)⁽¹⁶⁾. In 2009, in an attempt to align the different perspectives on metabolic syndrome classification, a collaboration between the International Diabetes Federation, the National Heart, Lung, and Blood Institute and a number of other health associations developed the Harmonized Metabolic Syndrome Definition, also known as Joint Interim Statement⁽²⁴⁾. These two definitions, that require any three components for diagnosis, are currently most in use, and they only differ in overweight threshold. In addition to the components comprising the classifications, other components have been observed to be involved in metabolic syndrome pathophysiology, including hyperuricemia, inflammation, adipokine signaling and pro-thrombotic factors^(21, 25).

	Joint Interim Statement*	NCEP ATP III	
Required for diagnosis	3 or more	criteria	
Overweight	Waist circumference with ethnicity specific thresholds	Waist circumference >102/88cm (men/women)	
Insulin resistance	Fasting plasma glucose ≥5.	6 mmol/L or treatment	
Drolinidamia	Triglycerides ≥1.7mmol/L or treatment		
Dyshpideinia	HDL cholesterol <1/1.3mmol/L (men/women) or treatment		
Hypertension	≥130/85mmHg	or treatment	

Table 1. Metabolic syndrome definitions commonly used in clinical practice and research

* Involved organizations are the International Diabetes Federation Task Force on Epidemiology and Prevention; National Heart, Lung, and Blood Institute; American Heart Association; World Heart Federation; International Atherosclerosis Society; and International Association for the Study of Obesity

Metabolic syndrome leads to an increased risk of diabetes and atherosclerotic disease and, hence, mortality. Previous studies have shown that in the general population metabolic syndrome is associated with a three-fold and five-fold increased risk of developing subsequent diabetes and atherosclerotic disease, respectively, that diabetes mellitus itself further increases atherosclerotic disease risk by three-fold, and that consequently patients with metabolic syndrome carry an up to doubled risk of dying from atherosclerotic disease⁽²⁶⁻²⁹⁾.

METABOLIC SYNDROME IN CHILDHOOD CANCER SURVIVORS

Some previous studies have observed mechanisms that can induce metabolic syndrome and separate components on the long-term after childhood cancer treatment. One important mechanism is damage to the hypothalamus and pituitary gland, which leads to several hormonal deficiencies – growth hormone is the most important one, followed by sex hormone and thyroid hormone – that contribute to metabolic syndrome risk^(30, 31). This damage can be caused by brain tumors in this region, as well as local surgery and radiotherapy. Furthermore, it can be caused by cranial radiotherapy as administered in the first childhood leukemia treatment protocols and by total body irradiation as myeloablative conditioning regimen for stem cell transplantation. A second important mechanism for metabolic syndrome is primary hypogonadism caused by ovarian and testicular tumors, resections, abdominal and pelvic radiotherapy, and alkylating chemotherapy^(31, 32). Thirdly, abdominal radiotherapy can damage the pancreas, leading to reduced insulin-secretion and hence faster deterioration of insulin resistance into diabetes^(33, 34). Finally, nephrectomy and nephrotoxic chemotherapy have been shown to increase risk of hypertension^(35, 36). Altogether, these treatment-related risk factors, and additional comorbidities that can further aggravate the situation, lead to a higher metabolic syndrome risk in childhood cancer survivors, even at a very young age⁽³⁷⁻⁴⁰⁾.

These insights have made clinicians aware of the importance to include metabolic syndrome screening in the follow-up of survivors. Metabolic syndrome is often a subclinical condition, i.e., it tends to develop without apparent symptoms for years. Hence, it is important to screen for the components in childhood cancer survivors, to be able to start early lifestyle and medical interventions, and thereby prevent further derangement into life-threatening diabetes and cardio- and cerebrovascular disease. However, there are challenges in identifying metabolic syndrome in survivors early and effectively.

First, there is a risk of underdiagnosis of metabolic syndrome with the standard criteria. After abdominal radiation the abdominal wall can be damaged, and therefore waist circumference may not reflect total body fat percentage. This leads to underestimation of overweight, thereby underdiagnosing metabolic syndrome^(41, 42). Diagnosis of metabolic syndrome could be improved by including other diagnostic markers of metabolic syndrome in surveillance. Potential alternatives include additional serum biomarkers, vascular ultrasonography, other anthropometric measurements, and more detailed body composition assessment with dual-energy X-ray absorptiometry. Although studies in the general population have indicated the potential value of these alternative methods, studies in survivors are scarce and adequate overweight and metabolic syndrome assessment remains challenging in subgroups of survivors.

Also, knowledge on national prevalence and risk factors of metabolic syndrome and its components remains incomplete. In addition to patient and treatment related risk factors, this also includes genetic risk, which is suspected because similarly treated survivors can have different occurrence of metabolic syndrome. This knowledge on prevalence and determinants is lacking because studies so far are either based on small and biased survivor cohorts, have insufficient treatment data, are questionnaire-based, or have short follow-up. Large cohort studies on metabolic syndrome that actively recruited childhood cancer survivors are scarce and for some subtopics unavailable. Therefore, there remain steps to be taken towards personalized metabolic syndrome risk assessment and effective follow-up.

AIMS OF THIS THESIS

This thesis attempts to contribute knowledge based on large, representative national cohorts of childhood cancer survivors. The aims of this thesis were to identify additional (bio)markers to improve metabolic syndrome diagnosis, and to describe prevalence as well as clinical and genetic determinants of metabolic syndrome, with a specific focus on the components overweight and dyslipidemia.

Chapter 1 describes a literature review on the occurrence of metabolic syndrome in childhood cancer survivors and established and possible essential links with treatment. In Chapter 2 a single-center recruitment study among survivors of childhood nephroblastoma and neuroblastoma is described, that aimed to show the potential value of additional serum biomarkers and vascular ultrasonography for diagnosing metabolic syndrome. Chapter 3 includes a systematic literature review and metaanalysis on the diagnostic and predictive value of nine novel metabolic syndrome related biomarkers that have been described in the normal population as well as in survivors. Chapter 4 describes the methodology of our study on prevalence and determinants of metabolic syndrome in survivors based on the nationwide Dutch LATER cohort, using various assessment modalities. In Chapter 5 the results on prevalence and determinants of overweight and obesity in this national cohort, and the optimal way to determine overweight, are described. *Chapter 6* reports the first genome-wide association study on the metabolic syndrome component dyslipidemia in childhood cancer survivors, in three large survivor cohorts in the United States. Chapter 7 describes perspectives on optimal survivor follow-up from an international panel of healthcare workers who attended a meet-the-expert session on survivorship at the 2018 conference of the International Society of Pediatric Oncology. In Chapter 8 the results of this thesis are further discussed including future perspectives.

REFERENCES

- 1. Steliarova-Foucher E, Colombet M, Ries LAG, et al. International incidence of childhood cancer, 2001-10: a population-based registry study. Lancet Oncol. 2017;18(6):719-31.
- Howlader N NA, Krapcho M, Miller D, Brest A, Yu M, Ruhl J, Tatalovich Z, Mariotto A, Lewis DR, Chen HS, Feuer EJ, Cronin KA (eds). SEER Cancer Statistics Review, 1975-2018. National Cancer Institute, Bethesda, MD.
- 3. Gatta G, Botta L, Rossi S, et al. Childhood cancer survival in Europe 1999–2007: results of EUROCARE-5—a population-based study. The Lancet Oncology. 2014;15(1):35-47.
- 4. Dutch Childhood Oncology Group. SKION Basisregistratie [Aug 2021]. Available from: https:// www.skion.nl/info/4208/kinderoncologie-in-cijfers/4209/skion-basisregistratie/.
- 5. Armstrong GT, Liu Q, Yasui Y, et al. Late mortality among 5-year survivors of childhood cancer: a summary from the Childhood Cancer Survivor Study. J Clin Oncol. 2009;27(14):2328-38.
- 6. Mertens AC, Yasui Y, Neglia JP, et al. Late mortality experience in five-year survivors of childhood and adolescent cancer: the Childhood Cancer Survivor Study. J Clin Oncol. 2001;19(13):3163-72.
- 7. Bagnasco F, Caruso S, Andreano A, et al. Late mortality and causes of death among 5-year survivors of childhood cancer diagnosed in the period 1960-1999 and registered in the Italian Off-Therapy Registry. Eur J Cancer. 2019;110:86-97.
- 8. Geenen MM, Cardous-Ubbink MC, Kremer LCM, et al. Medical Assessment of Adverse Health Outcomes in Long-term Survivors of Childhood Cancer. JAMA. 2007;297(24):2705-15.
- 9. Oeffinger KC, Mertens AC, Sklar CA, et al. Chronic Health Conditions in Adult Survivors of Childhood Cancer. N Engl J Med. 2006;355(15):1572-82.
- 10. Bhakta N, Liu Q, Ness KK, et al. The cumulative burden of surviving childhood cancer: an initial report from the St Jude Lifetime Cohort Study (SJLIFE). Lancet. 2017;390(10112):2569-82.
- 11. Obesity and overweight fact sheet [Internet]. 2017 [cited November 2017].
- 12. World Health Organization. Waist circumference and waist-hip ratio: report of a WHO expert consultation, Geneva, 8-11 December 2008. 2011.
- 13. Grundy SM, Cleeman JI, Daniels SR, et al. Diagnosis and management of the metabolic syndrome: an American Heart Association/National Heart, Lung, and Blood Institute Scientific Statement. Circulation. 2005;112(17):2735-52.
- 14. Graham I, Atar D, Borch-Johnsen K, et al. European guidelines on cardiovascular disease prevention in clinical practice: executive summary. Atherosclerosis. 2007;194(1):1-45.
- Grundy SM, Stone NJ, Bailey AL, et al. 2018 AHA/ACC/AACVPR/AAPA/ABC/ACPM/ADA/ AGS/APhA/ASPC/NLA/PCNA Guideline on the Management of Blood Cholesterol: A Report of the American College of Cardiology/American Heart Association Task Force on Clinical Practice Guidelines. Circulation. 2019;139(25):e1082-e143.
- 16. National Cholesterol Education Program Expert Panel on Detection E, and Treatment of High Blood Cholesterol in Adults (Adult Treatment Panel III). Third Report of the National Cholesterol Education Program (NCEP) Expert Panel on Detection, Evaluation, and Treatment of High Blood Cholesterol in Adults (Adult Treatment Panel III). 2002.
- 17. American Diabetes Association Professional Practice C. 2. Classification and Diagnosis of Diabetes: Standards of Medical Care in Diabetes-2022. Diabetes Care. 2022;45(Suppl 1):S17-S38.
- Whelton PK, Carey RM, Aronow WS, et al. 2017 ACC/AHA/AAPA/ABC/ACPM/AGS/ APhA/ASH/ASPC/NMA/PCNA Guideline for the Prevention, Detection, Evaluation, and Management of High Blood Pressure in Adults: A Report of the American College of

Cardiology/American Heart Association Task Force on Clinical Practice Guidelines. Circulation. 2018;138(17):e484-e594.

- 19. Reaven GM. Banting lecture 1988. Role of insulin resistance in human disease. Diabetes. 1988;37(12):1595-607.
- 20. Kaplan NM. The deadly quartet. Upper-body obesity, glucose intolerance, hypertriglyceridemia, and hypertension. Arch Intern Med. 1989;149(7):1514-20.
- 21. Eckel RH, Grundy SM, Zimmet PZ. The metabolic syndrome. The Lancet. 2005;365(9468):1415-28.
- 22. Adiels M, Olofsson SO, Taskinen MR, Boren J. Overproduction of very low-density lipoproteins is the hallmark of the dyslipidemia in the metabolic syndrome. Arterioscler Thromb Vasc Biol. 2008;28(7):1225-36.
- 23. Blokstra A, Vissink P, Venmans LMAJ, et al. Measuring The Netherlands. A monitoring study of risk factors in the general population, 2009-2010. RIVM Rapport 260152001/2011. 2011.
- 24. Alberti KG, Eckel RH, Grundy SM, et al. Harmonizing the metabolic syndrome: a joint interim statement of the International Diabetes Federation Task Force on Epidemiology and Prevention; National Heart, Lung, and Blood Institute; American Heart Association; World Heart Federation; International Atherosclerosis Society; and International Association for the Study of Obesity. Circulation. 2009;120(16):1640-5.
- 25. Moller DE, Kaufman KD. Metabolic syndrome: a clinical and molecular perspective. Annu Rev Med. 2005;56:45-62.
- 26. Alberti G, Zimmet P, Shaw J, Grundy SM. The IDF consensus worldwide definition of the metabolic syndrome. 2006.
- 27. Wilson PW, D'Agostino RB, Parise H, Sullivan L, Meigs JB. Metabolic syndrome as a precursor of cardiovascular disease and type 2 diabetes mellitus. Circulation. 2005;112(20):3066-72.
- 28. Ford ES. The metabolic syndrome and mortality from cardiovascular disease and all-causes: findings from the National Health and Nutrition Examination Survey II Mortality Study. Atherosclerosis. 2004;173(2):309-14.
- 29. Ju SY, Lee JY, Kim DH. Association of metabolic syndrome and its components with all-cause and cardiovascular mortality in the elderly: A meta-analysis of prospective cohort studies. Medicine (Baltimore). 2017;96(45):e8491.
- 30. Darzy KH, Shalet SM. Hypopituitarism following Radiotherapy Revisited. Endocr Dev. 2009;15:1-24.
- 31. Chemaitilly W, Cohen LE. Diagnosis of Endocrine Disease: Endocrine late-effects of childhood cancer and its treatments. Eur J Endocrinol. 2017;176(4):R183-R203.
- 32. van Dorp W, Blijdorp K, Laven JS, et al. Decreased ovarian function is associated with obesity in very long-term female survivors of childhood cancer. Eur J Endocrinol. 2013;168(6):905-12.
- 33. de Vathaire F, El-Fayech C, Ben Ayed FF, et al. Radiation dose to the pancreas and risk of diabetes mellitus in childhood cancer survivors: a retrospective cohort study. The Lancet Oncology. 2012;13(10):1002-10.
- 34. van Nimwegen FA, Schaapveld M, Janus CP, et al. Risk of diabetes mellitus in long-term survivors of Hodgkin lymphoma. J Clin Oncol. 2014;32(29):3257-63.
- 35. Dekkers IA, Blijdorp K, Cransberg K, et al. Long-term nephrotoxicity in adult survivors of childhood cancer. Clin J Am Soc Nephrol. 2013;8(6):922-9.
- Gibson TM, Li Z, Green DM, et al. Blood Pressure Status in Adult Survivors of Childhood Cancer: A Report from the St. Jude Lifetime Cohort Study. Cancer Epidemiol Biomarkers Prev. 2017;26(12):1705-13.

- Talvensaari KK, Lanning M, Tapanainen P, Knip M. Long-Term Survivors of Childhood Cancer Have an Increased Risk of Manifesting the Metabolic Syndrome. J Clin Endocrinol Metab. 1996;81(8):3051-5.
- 38. van Waas M, Neggers SJ, Pieters R, van den Heuvel-Eibrink MM. Components of the metabolic syndrome in 500 adult long-term survivors of childhood cancer. Ann Oncol. 2010;21(5):1121-6.
- Mostoufi-Moab S, Seidel K, Leisenring WM, et al. Endocrine Abnormalities in Aging Survivors of Childhood Cancer: A Report From the Childhood Cancer Survivor Study. J Clin Oncol. 2016;34(27):3240-7.
- Nottage KA, Ness KK, Li C, Srivastava D, Robison LL, Hudson MM. Metabolic syndrome and cardiovascular risk among long-term survivors of acute lymphoblastic leukaemia - From the St. Jude Lifetime Cohort. Br J Haematol. 2014;165(3):364-74.
- 41. van Waas M, Neggers SJ, Raat H, van Rij CM, Pieters R, van den Heuvel-Eibrink MM. Abdominal radiotherapy: a major determinant of metabolic syndrome in nephroblastoma and neuroblastoma survivors. PLoS One. 2012;7(12):e52237.
- 42. Blijdorp K, van den Heuvel-Eibrink MM, Pieters R, et al. Obesity is underestimated using body mass index and waist-hip ratio in long-term adult survivors of childhood cancer. PLoS One. 2012;7(8):e43269.

Chapter 1

Metabolic syndrome as cardiovascular risk factor in childhood cancer survivors

V.G. Pluimakers*, M. van Waas*, S.J.C.M.M. Neggers, M.M. van den Heuvel-Eibrink

Critical Reviews in Oncology/Hematology, 2019 Jan;133:129-141

ABSTRACT

Over the past decades, survival rates of childhood cancer have increased considerably from 5-30% in the early seventies to current rates exceeding 80%. This is due to the development of effective chemotherapy, surgery, radiotherapy and stem cell transplantation, combined with an optimized stratification of therapy and better supportive care regimens. As a consequence, active surveillance strategies of late sequelae have been developed to improve the quality of survival. Several epidemiological studies have reported an increased incidence of (components of) metabolic syndrome (MetS) and cardiovascular disease in childhood cancer survivors (CCS). Growth hormone deficiency (GHD) after cranial radiotherapy (CRT) has been previously described as an important cause of MetS. New insights suggest a role for abdominal radiotherapy as a determinant for MetS as well. The role of other risk factors, such as specific chemotherapeutic agents, steroids, gonadal impairment, thyroid morbidity and genetics, warrants further investigation. This knowledge is important to define subgroups of CCS that are at risk to develop (subclinical) MetS features. These survivors might benefit from standard surveillance and early interventions, for example lifestyle and diet advice and medical treatment, thereby preventing the development of cardiovascular disease.

INTRODUCTION

Over the past decades, survival rates of childhood cancer have increased considerably from 5-30% in the early seventies to current rates exceeding 80%⁽¹⁾. This is due to the development of effective chemotherapy, surgery, radiotherapy and stem cell transplantation (SCT), combined with an optimized stratification of therapy and better supportive care regimens. These improved survival rates currently result in an ongoing increasing number of survivors⁽²⁾, which in turn resulted in increased awareness of late side effects of treatment for childhood cancer, and research investigating these late sequelae.

Several epidemiological studies have reported an increased incidence of cardiovascular disease in survivors of childhood cancer (Supplemental table 1). Standardized mortality risk, e.g. due to stroke and coronary heart disease, ranges from 1.9 to 12.7, with higher risk for specific subgroups with regard to diagnosis, administered treatment and follow-up time⁽³⁻¹⁶⁾.

The pathophysiology of the development of cardiovascular disease in childhood cancer survivors is a multifactorial process as in the normal population, but with additional treatment and disease specific modulators. Frequently reported risk factors for cardiovascular sequelae are adiposity, hypertension, diabetes mellitus and dyslipidemia, which cluster as the entity "metabolic syndrome"⁽¹⁷⁻¹⁹⁾. This narrative review summarizes existing literature on the frequency and determinants of metabolic syndrome and its components in childhood cancer survivors (CCS).

METHODS

We searched PubMed and Embase for the following terms and synonyms: "childhood cancer survivor", "metabolic syndrome", "obesity", "insulin resistance", "diabetes", "dyslipidemia" and "hypertension".

COMPONENTS OF THE METABOLIC SYNDROME IN CHILDHOOD CANCER SURVIVORS

Overweight and adiposity

Overweight and obesity are frequently described phenomena in CCS. Overweight is defined as body mass index (BMI) \geq 25 and <30 kg/m2, obesity as BMI \geq 30 kg/m2. Population based, the prevalence of overweight has increased enormously over the past decades, especially in developed countries. In 2014, an estimated 1.9 billion

adults (i.e. 39% of the adult population worldwide), suffered from overweight, of which a third was $obese^{(20)}$. Overweight has a negative influence on blood pressure, lipid metabolism and insulin resistance. A five kg/m² BMI increase has been described to be associated with a 1.5- or 2-fold risk increase for coronary heart disease, and 4- or 8-fold for diabetes mellitus⁽²¹⁾. Also, overweight enhances the risk of stroke (1.3-fold⁽²²⁾) and of several types of cancer, e.g. postmenopausal breast, colon, thyroid, renal, endometrium and esophageal, with a relative risk of 1.12-1.59 per 5 points BMI increase⁽²³⁾.

Adiposity is a broader term including more accurate measurements of adipose tissue accumulation, such as waist circumference, waist/hip ratio and sometimes fat percentage or body composition (assessed by Dual-energy X-ray Absorptiometry [DXA])⁽²⁴⁻³⁰⁾. There is increasing evidence that BMI values reflect underestimations of adiposity, and that the accumulation of visceral fat as well as body composition as measured by DXA are more reliable measures for overweight to predict the development of cardiovascular disease⁽³⁰⁻³⁶⁾. However, since DXA is a time consuming, financially less attractive diagnostic test which, in addition, requires low dose radiation in children who have often already been exposed to teratogenic treatments, BMI is the most commonly used tool to study overweight.

The first reports on obesity risk after childhood cancer were published in the eighties, initiated by the impression that many survivors of childhood leukemia were overweight or obese⁽³⁷⁾. A correlation with CRT, often associated with growth hormone deficiency (GHD), was reported, which was confirmed in consecutive studies thereafter^(24, 30, 38, 39). Subsequently, further detailed studies pointed out that the risk of overweight was especially high among female survivors and survivors diagnosed at younger age and was radiation dose- and site-dependent⁽⁴⁰⁻⁴³⁾. On the other hand, a recent meta-analysis in 1742 ALL survivors reported a high prevalence of overweight -80^{th} BMI percentile –, independent of patient and treatment characteristics⁽⁴⁴⁾. Nine recently published studies performed multivariable analysis to describe independent risk factors for overweight, six of which had a cross-sectional design, and three were retrospective studies^(42, 43, 45-50). The largest is a report from the Childhood Cancer Survivor Study (CCSS), comparing self-reported overweight between 13000 survivors, after median 24 years' follow-up, and 4000 siblings in 27 participating centers in the United States and Canada⁽⁴²⁾. Overweight rate was the same in both study groups (RR 1.0, 95% CI 0.9-1.1). Among survivors, CRT >18Gy, total body irradiation (TBI) and abdominal radiotherapy were independent risk factors for overweight. After a follow up of 24.6 years, the St. Jude Lifetime cohort, consisting of ~2000 patients that underwent late effect surveillance in the After Completion of Therapy (ACT) Clinic, showed a prevalence of obesity of 36%, with a standardized morbidity ratio of 1.14

when compared to matched controls⁽⁴³⁾. CRT (OR 1.66) and previous glucocorticoids treatment (OR 1.37) as well as older age at evaluation were independent risk factors of becoming obese, whereas previous chest/abdominal/pelvic radiation (OR 0.48) was associated with lower obesity prevalence among survivors. In the Swiss Childhood Cancer Survivor Study, the prevalence of self-reported overweight in 2400 CCS was similar to siblings and the general population, and CRT >20Gy was an independent risk factor for overweight among survivors⁽⁵¹⁾. The three other studies with a cross-sectional design comprised between 330 and 900 survivors, and reported the following independent risk factors: brain tumor, CRT, anthracyclines, high BMI at diagnosis and Hispanic race⁽⁴⁵⁻⁴⁷⁾. In summary, in studies of highest quality, CRT is the most frequently reported independent risk factor of overweight in CCS (Table 1a).

Insulin resistance and type II diabetes mellitus

Diabetes mellitus (DM) gives rise to the risk of micro- and macrovascular damage^(52, 53) (Figure 1). Type II DM (DM2) is thought to be the result of insulin resistance (IR) and (visceral) adiposity-associated chronic inflammation and, ultimately, pancreatic β -cell dysfunction^(54, 55). It is estimated that worldwide 422 million people suffer from DM. As with obesity, the prevalence of DM – especially DM2 – in the general population has increased substantially over the past decades, from 4.7% in 1980 to 8.5% in 2014⁽⁵⁶⁾. As adiposity is highly associated with the development of fatty liver disease, IR and DM2⁽⁵⁷⁻⁶⁰⁾, it is anticipated that adipose survivors more frequently suffer from diabetes than non-adipose survivors. In addition, some studies suggest an increased prevalence of diabetes after adjusting for obesity, e.g. due to radiotherapy^(27, 61-64).

Table 1a. Obe	sity in ch	ildhood car	ncer survivors						
Author	Year	z	Population	Design	Follow-up	% Overweight	% Obese	MVA	Prognostic variable
Zhang	2014	1742	ALL	1. Meta-analysis of literature 20 studies	<10y	80th percentile	n.a.	Meta- analysis	None, high prevalence of obesity among survivors independent of patient- and treatment characteristics
Moustoufi	2016	13203	CCS	2. Cross-sectional	24y Mdn (5-39y)	RR 1.0 (95% CI 0.9-1.1)	n.a.	Yes	CRT>18Gy, TBI, Abd Rtx greater risk compared with other survivors; CRT<18Gy lower risk compared to siblings; No Abd Rtx or TBI greater risk than siblings
Wilson	2015	1996	CCS	2. Cross-sectional	24.6y Mdn (10.7-48.3y)	27.9	36.2	Yes	CRT, glucocorticoids, older age at evaluation; Abdominal, chest, pelvic radiation less obesity
Van Santen	2015	893	CCS	2. Cross-sectional	14.9y M (4.7-36.2y)	23.3	n.a.	Yes	CRT, younger age at diagnosis, high BMI at diagnosis
Brown	2016	406	ALL	2. Cross-sectional	11.39y M (±5.33y)	27.6	22.2	Yes	CRT, Hispanic
Felicetti	2015	330	CCS	2. Cross-sectional	16.1y Mdn (5.1-33.0y)	n.a.	×	Yes	Brain tumor (HR 10.0), anthracyclines (HR 1.3), age at diagnosis (HR 1.12)
Prasad	2015	648 (471 <18y)	ccs	3. Retrospective	6y Mdn (2-16y) (<18y); 11.5y Mdn (2-41y) (≥18y)	10.8 (<18y); 8.5 (≥18y)	2.7 / 0	Yes	ALL, brain tumor
Gunn	2016	276	CCS	3. Retrospective	n.a.	32.3	n.a.	Yes	CRT
Lindemulder	2015	269	ALL CRT-	3. Retrospective	9.1y Mdn (4.8-13.7y)	18.1	20.9	Yes	n.a.
Essig	2014	556	ALL	2. Cross-sectional	18.4y Mdn (0.0-33.0y)	n.a.	21	No	n.a.
Stolley	2015	452	CCS	2. Cross-sectional	$18.4y (\pm 9.3)/16.7y$ (± 6.8)/20.2 (± 7.9) M	n.a.	32 / 42 / 23	No	Hispanic, Afr-American
Berdan	2014	413	CCS	2. Cross-sectional	18.5y M (±8.1y)	28.9	32.4	No	Hispanic
Brouwer	2013	277	CCS	2. Cross-sectional	18y Mdn (5-31y)	33	n.a.	No	n.a.
Nayjager	2017	75	ALL	2. Cross-sectional	15.07y Mdn (10.22-26.30y)	25.3	8	No	n.a.
Latoch	2016	75	CCS	2. Cross-sectional	12.15y Mdn (1-23.5y)	29.3	n.a.	No	n.a.

Author	Year	Z	Population	Design	Follow-up	% Overweight	% Obese	MVA	Prognostic variable
Siviero	2013	56	ALL	2. Cross-sectional	8.5y M (±3.9y)	n.a.	3.6	No	n.a.
Murphy	2015	53	CCS	2. Cross-sectional	Range 3.2-14.4y	0	n.a.	No	n.a.
Jahnukainen	2015	49	ALL male	2. Cross-sectional	20y Mdn (10-29y)	n.a.	15	No	Cranial/testicular radiation (NS)
Van Dorp	2013	191	CCS female	3. Retrospective	18.8y Mdn (2.3-48.8y)	13	10	No	n.a.
Shalitin	2014	139	Non-brain	3. Retrospective	9y Mdn (1.2-29.5y)	n.a.	1.4	No	n.a.
			solid						
Harper	2013	27	ALL RTx-	3. Retrospective	6y	40.7	n.a.	No	n.a.
Highest quality (heterogeneous	y evidenc ; group); .	ce above da: ALL = acut	shed line, based e lymphoblastic	on design (1 = meta : leukemia; RTx = rad	-analysis, 2 = cross-sectional, 3 = .iotherapy; RR = relative risk; CR	= retrospective) and : (T = cranial irradiation)	multivariable on	analysis.	CCS = (childhood) cancer survivors

Table 1b provides an overview of recent literature on IR and DM in CCS. Twenty years ago, the first reports on an increased risk of DM after abdominal radiation in survivors of Wilms tumor were published^(65, 66), suggesting a damaging effect of radiation to the pancreas. From a cross-sectional study in ~8600 survivors by Meacham, the prevalence of self-reported DM after 23.5 years of follow-up was 2.5% in survivors and 1.7% in siblings (p<0.01). Among survivors, this was explained in particular by TBI (OR 7.2), abdominal radiotherapy (OR 2.7), alkylating agents (OR 1.7) and younger age at diagnosis (OR 2.4). No association was found with CRT and corticosteroids⁽⁶²⁾. Holmqvist retrospectively reported hospitalizations for DM in a large cohort of ~33000 survivors, ten years after diagnosis. The observed hospitalization rate was 1.6 times higher than expected and especially high in survivors treated with radiotherapy, i.e. Wilms tumor (OR 2.9), leukemia (2.0), CNS tumor (1.8), germ-cell tumor (1.7) and bone tumor $(1.7)^{(67)}$. A large cross-sectional study in ~1000 adult survivors treated with HSCT also revealed TBI as an independent risk factor for DM (OR 3.42)⁽⁶¹⁾. A study in 750 pediatric HSCT treated survivors added asparaginase toxicity, defined as hyperglycemia and/or pancreatitis, as an independent risk factor⁽⁶⁸⁾, and a prospective study in 250 CCS reported TBI and hypogonadism as independent risk factors⁽²⁷⁾. Chao found no significant increase in DM frequency in 650 survivors compared to 6520 non-cancer controls⁽⁶⁹⁾. In summary, several studies investigated DM in large cohorts of cancer survivors, and radiotherapy - total body as well as abdominal seems to be the most frequently reported independent risk factor.

The link with damage to the pancreas by radiotherapy was closely investigated by De Vathaire⁽⁶³⁾. Radiation to the pancreatic tail, where the majority of insulin-secreting Langerhans islets is located, increased the risk of diabetes in a dose-dependent way (RR at 1 Gy 1.61), whereas the radiation dose to the head or body had no significant effect. A similar dose-dependent relation between radiation to the pancreatic tail and the occurrence of DM was found in adult Hodgkin lymphoma survivors⁽⁷⁰⁾. In our study in nephro- and neuroblastoma survivors, radiotherapy to the whole pancreas increased the risk of IR, compared to controls and to radiation to parts of the pancreas⁽³⁵⁾. A study in ALL survivors reported lower pancreatic volume and insulin secretion after TBI, suggesting a reduced beta cell reserve⁽⁷¹⁾. Apart from pancreatic radiation damage impairing insulin secretion, it might be that radiotherapy impairs fat cell expansion, which increases liver steatosis and circulation of free fatty acids (FFA), subsequently causing IR and DM. In mice, it has been shown that adipose tissue fibrosis restricts adipocyte enlargement and is associated with local inflammation and systemic IR^(72,73). Whether these biological mechanisms determine the higher MetS risk in abdominally irradiated cancer survivors as well, needs to be investigated.

Table 1b. Insu	lin resisté	ance and	diabetes mellitu	s in childhood cance	survivors				
Author	Year	z	Population	Design	Follow-up	% IR	% DM	MVA	Prognostic variable
Holmqvist	2014	32903	CCS	3. Retrospective	10y Mdn (0-42y)	n.a.	1.5, SHRR 1.6	Yes	Wilms', leukemia, CNS, germ-cell, bone, HL
Meacham	2009	8599	CCS	2. Cross-sectional	23.5y M (16.0- 35.2y)	n.a.	2.5	Yes	TBI (OR 7.2), Abd Rtx (OR 2.7), alkylating agent (OR 1.7), young age at diagnosis (OR 2.4); no association: CRT, corticosteroids, asparaginase
Baker	2007	1089	HSCT (adult)	2. Cross-sectional	8.6y M (±5.1y)	n.a.	7.6	Yes	TBI (OR 3.42)
Hoffmeister	2004	748	HSCT	3. Retrospective	11y Mdn (2.0- 30.0y)	n.a.	4.5	Yes	CML (HR 85.0), AML/ALL (HR 13.0), asparaginase toxicity
Chao	2016	652	CCS	2. Cross-sectional	6.2y M (±4.1y)	n.a.	1.1 vs 0.05;	Yes	n.a.
							1.8/1000p-yr; adjusted IRR 1.6 (diabetes) (NS)		
Neville	2006	248	CCS	1. Prospective	12.9y Mdn (2.3- 33.6y)	6.9 (IGT)	9.7	Yes	TBI, hypogonadism
Latoch	2016	75	CCS	2. Cross-sectional	11.8y M (±5.2y)	1.33	n.a.	Yes	(BMI)
Moustoufi	2016	14290	CCS	2. Cross-sectional	24y Mdn (5-39y)	n.a.	RR 1.8 (95% CI 1.4-2.3)	No	TBI, CRT
Kero	2016	2530	CCS	3. Retrospective	10.4y Mdn (0-18y)	n.a.	HR 3.0	No	ALL, AML, CNS
De Vathaire	2012	2520	CCS	3. Retrospective	۸.	n.a.	2.6	No	Pancreatic tail radiation
Van Waas	2013	532	CCS	3. Retrospective	17.9y median (5.0- 48.8y)	n.a.	0.9	No	Wilms', Abd Rtx
Felicetti	2015	330	CCS	2. Cross-sectional	16.1y Mdn (5.1- 33.0y)	n.a.	1.5	No	n.a.
Wilhelmsson	2014	204	HSCT	3. Retrospective	12y Mdn (4-28y)	n.a.	6	No	n.a.

Metabolic syndrome as cardiovascular risk factor in childhood cancer survivors

Iable 1D. C	ontinued								
Author	Year	z	Population	Design	Follow-up	% IR	% DM	MVA	Prognostic variable
Shalitin	2014	139	Non-brain solid	3. Retrospective	9y Mdn (1.2-29.5y)	1.4 (IGT)	n.a.	No	n.a.
Gunn	2016	62	CCS	3. Retrospective	18.0y M (6.8-37.9y)	32.3	n.a.	No	n.a.
Wei	2016	35	A(L/M)L +	2. Cross-sectional	12.5y Mdn (3.2-	34.2	17.1	No	n.a.
			HSCT		18.2y)	(IGT)			
Cohen	2014	24	HR NBL +	3. Retrospective	6.1y Mdn (1.0-	45.9	4.2	No	n.a.
			SCT		15.2y)	(HbA1c)			
Hichaet anali	tu andanca		hed line based	al an multivariable and	cie notient number ond	decian (1 - nro		ectional 3.	- H) SCT - (hemitralouic)

(nematologic) stem cell transplantation; A(L/M)L = acute lymphoblastic/myeloid leukemia; NBL = neuroblastoma; IGT = impaired glucose tolerance; HR = hazard ratio; TBI = total body irradiation; cuuspectives. (11)301 Highest quality evidence above dashed line, based on multivariable analysis, patient number and design (1 = prospective, 2 = cross-sectional, 5 = HL = Hodgkin lymphoma; Abd Rtx = abdominal radiotherapy.

Dyslipidemia

Classic parameters of dyslipidemia include elevated fasting levels of total cholesterol and low-density lipoprotein cholesterol and triglycerides, and low levels of high-density lipoprotein cholesterol. These alterations in lipid metabolism are associated with cardiovascular disease⁽⁷⁴⁻⁷⁶⁾. Adipose tissue plays an important causal role in the occurrence of dyslipidemia through the release of FFA, which leads to increased triglyceride and very low-density lipoprotein cholesterol production in the liver⁽⁷⁷⁾. Hence, cancer survivors with an increased risk of overweight carry an increased risk of dyslipidemia as well. Hypogonadism following cancer therapy can cause dyslipidemia directly as well; this was observed in survivors of adult testicular cancer^(78, 79), breast cancer treated with aromatase inhibitors⁽⁸⁰⁾ and prostate cancer treated with LHRH-agonists⁽⁸¹⁾.

As depicted in Table 1c, the rate of dyslipidemia in CCS varied greatly and different outcome measures are reported. Only one study reported independent risk factors for dyslipidemia in CCS. In 330 survivors, after 16.1 years of follow-up, age at diagnosis (HR 1.1), TBI (2.7), GHD (2.3) and autologous SCT (3.2) were independent risk factors for hypercholesterolemia, and TBI (6.5) and GHD (7.2) were also independent risk factors for hypertriglyceridemia⁽⁴⁷⁾. Chao studied dyslipidemia in 650 survivors and reported a higher risk (incidence rate ratio 1.9) compared to controls, but no specific prognostic variables were identified in multivariable analysis⁽⁶⁹⁾. In the CCSS the incidence of dyslipidemia was 8.9%, compared to 6.0% in siblings; this increased to a significant difference at age 50 (23.0 vs 13.6%), whereas the obesity rate at older age in this cohort was significantly higher among siblings⁽⁷⁾. In a large Finnish cohort of ~2500 survivors, the rate of dyslipidemia, defined as the purchase of lipid-lowering drugs, was 4.3 times higher than in siblings⁽⁸²⁾.

Highest quality evidence iabove dashed line, based on design (2 = cross-sectional, 3 = retrospective) and multivariable analysis. GHD = growth hormone deficiency.
Hypertension

Arterial hypertension is a condition in which blood pressure is persistently raised, defined as ≥ 140 mmHg systolic or ≥ 90 diastolic. Globally, the overall prevalence of hypertension in the general population aged 25 and over has been reported to be around $40\%^{(83)}$. The availability of low-cost medication has significantly decreased the occurrence of hypertension to e.g. 18% in the USA^(83, 84). Hypertension is a major risk factor for coronary heart disease and ischemic as well as hemorrhagic stroke, being responsible for ~50% of deaths due to these diseases⁽⁸⁵⁾. In addition, blood pressure level as continuous variable has been shown to be related to the risk of stroke, coronary heart disease, heart failure, peripheral vascular disease, renal impairment and retinal hemorrhage⁽⁸⁶⁻⁹⁰⁾.

Already in 1989, Kantor described hypertension in 20% of long-term survivors of childhood renal cancer⁽⁹¹⁾. According to a Cochrane review by Knijnenburg, prevalence of hypertension in childhood cancer survivors ranges from 0% to 18.2%⁽⁹²⁾. Three reports thereafter showed even higher prevalence, and one of these observed a sharp increase with age, exceeding 70% by age 50^(50, 93, 94) (Table 1d). In most case control studies, survivors reveal relatively high hypertension rates^(7, 10, 82, 94). A study in ~650 survivors found no significant difference between survivors and controls, but this was a study with a rather short follow-up time of 6 years⁽⁶⁹⁾. In the CCSS, the presence of hypertension significantly increased the risk of major cardiac events and cardiac-specific mortality⁽⁷⁾. The aforementioned Cochrane review included 24 studies with ~4000 survivors in total, and a high BMI was the only consistent independent risk factor for hypertension reported in multiple studies. Other reported independent risk factors are the use of total body or abdominal irradiation, nephrectomy, acute kidney injury, SCT, growth hormone therapy, older age at screening and male sex^(50, 92, 94).

Hypertension in CCS may be caused by direct kidney damage through irradiation⁽⁹⁵⁾. Unilateral nephrectomy is known to induce hyperfiltration in the remaining kidney, which may give rise to hypertension⁽⁹⁶⁾. Ifosfamide and cisplatin have nephrotoxic side effects^(93, 97, 98), but hypertension is not reported as a consequence of these agents; one study reported a non-significant risk increase⁽⁹⁹⁾. In the general adult population, it is known that treatment of hypertension towards below 140/90mmHg is associated with a reduction in cardiovascular complications⁽¹⁰⁰⁾. This suggests that identification and treatment of subclinical hypertension in childhood cancer survivors by standard surveillance may decrease morbidity and mortality^(93, 94).

Table 1d. Hyp	ertensio	n in child	thood cancer surviv	vors				
Author	Year	z	Population	Design	Follow-up	% Hypertension	MVA	Prognostic variable
Knijnenburg	2013	4073	SOC	1. Cochrane review	Various	0-18.2%	Yes	Higher BMI, TBI, abdominal radiation, acute kidney injury, stem cell transplantation, growth hormone therapy, older age at screening and male gender
Gibson	2017	3016	CCS	2. Cross-sectional	>10y	22.4	Yes	Older age, nephrectomy (OR 1.68)
Baker	2007	1089	HSCT (adult)	2. Cross-sectional	8.6y M (±5.1y)	18.5	Yes	Allogeneic transplant (OR 2.31)
Chao	2016	652	CCS	2. Cross-sectional	6.2y M (±4.1y)	1.5 vs 1.0; 2.5/1000p-yr; adjusted IRR 1.2 (NS)	Yes	n.a.
Gunn	2016	269	CCS	3. Retrospective	n.a.	19.0	Yes	Male sex, older age, overweight/obesity
Armstrong	2013	10724	CCS	3. Retrospective	25.6y Mdn (7.4-39.3y)	14.9	No	n.a.
Van Laar	2014	3247	CCS	3. Retrospective	>5y	7.75 vs 2.57 / 10.000p.y.	No	n.a.
Kero	2016	2530	CCS	3. Retrospective	10.4y Mdn (0-18y)	HR 4.6	No	ALL, bone tumor
Dekkers	2013	763	CCS	2. Cross-sectional	18.3y Mdn (5.0-58.2)	23.4	No	Renal tumor (31.4%), Abdominal radiation
Essig	2014	556	ALL	3. Retrospective	18.4y Mdn (0.0-33.0)	13	No	n.a.
Felicetti	2015	330	CCS	2. Cross-sectional	16.1y Mdn (5.1-33.0y)	5.3	No	n.a.
Brouwer	2013	277	CCS	2. Cross-sectional	18y Mdn (5-31y)	14	No	n.a.
Wilhelmsson	2014	204	HSCT	3. Retrospective	12y Mdn (4-28y)	7	No	n.a.
Shalitin	2014	139	Non-brain solid	3. Retrospective	9y Mdn (1.2-29.5y)	8.6	No	n.a.
Interiano	2015	75	Wilms	2. Cross-sectional	19.6y Mdn (10.0-32.8)	6.7	No	n.a.
Wei	2016	35	A(L/M)L + HSCT	2. Cross-sectional	12.5y Mdn (3.2-18.2y)	17	No	п.а.
	-	-			-		-	

Highest quality evidence above dashed line, based on design (1 = Cochrane review, 2 = cross-sectional, 3 = retrospective) and multivariable analysis. IRR = incidence rate ratio.

THE METABOLIC SYNDROME IN CHILDHOOD CANCER SURVIVORS

Definition

Metabolic syndrome is a cluster of adiposity, IR, dyslipidemia and hypertension⁽¹⁰¹⁻¹⁰³⁾. It was first described by Reaven in 1988, who found a clustering of symptoms in patients and called this Syndrome X⁽¹⁰⁴⁾. The symptoms of this cluster are related and interacting in various ways. In general, imbalance in energy intake and consumption results in increased (visceral) adiposity. Secondary effects of adiposity include increased circulating FFA and reduced adiponectin – thus, an increase in IR factors – and increased pro-inflammatory and pro-thrombotic mediators such as IL-6, TNF-alpha and PAI-1. Increased lipid flux into the liver can result in steatosis, which also mediates IR. The liver also produces fibringen, enhancing the pro-thrombotic state. IR in liver and muscle leads to hyperinsulinemia, with a result of adipose tissue growth and tissue resistance to insulin. Hyperinsulinemia also contributes to hypertension through enhanced sodium resorption and sympathetic nervous system activation^(105, 106). It is estimated that 20-25% of the world's adult population suffers from MetS⁽¹⁰²⁾ and, consequently, are three times more likely to have a heart attack or stroke and twice as likely to die from cardio- and cerebrovascular disease, compared to people without MetS. In addition, patients with MetS are five times more likely to develop DM2 and people with diabetes are three times more likely to develop cardiovascular disease^(102, 107, 108) (Figure 1). Metabolic syndrome is also associated with fatty liver disease, gallstones, hepatocellular carcinoma, chronic kidney disease and polycystic ovary syndrome⁽¹⁰⁹⁻¹¹⁵⁾.

Currently, three definitions of metabolic syndrome are commonly used: those created by the World Health organization (WHO)⁽¹⁰³⁾, National Cholesterol Education Program – Third Adult Treatment Panel (NCEP/ATPIII)⁽¹⁰¹⁾ and the International Diabetes Foundation (IDF)⁽¹⁰²⁾ (Supplemental table 2). Although the definition of MetS is based on the principle of clustered components, these components themselves are also independent risk factors for the development of cardiovascular disease⁽¹⁰⁵⁾. The prevalence of MetS can vary, depending on which definition is used. In young adults, who less frequently meet all MetS criteria, partial clustering of risk factors should be examined. The MetS definitions provide useful guidelines to identify those individuals at risk for development of DM2, atherosclerotic cardiovascular disease, and cardiovascular death. MetS is a "disguised" syndrome; without measurement of blood pressure and lipids, metabolic sequelae can develop for years. This underlines the need for active surveillance.



Figure 1. Metabolic syndrome in childhood cancer survivors and the risk of cardiovascular disease The components of the metabolic syndrome, risk factors for developing the syndrome and the risk for metabolic syndrome patients to develop cardiovascular disease and type 2 diabetes.

Risk and determinants

Several studies have focused on the development of MetS in CCS. Comparison of these studies is hampered by the fact that often small patient groups are analyzed, the heterogeneity of malignancies as well as therapies and the different definitions of MetS that are used. An overview of existing literature on the frequency of MetS and prognostic factors in CCS is presented in Table 2. The first study on this subject was by Talvensaari, reporting a prevalence of 16% in 50 survivors, compared to none of the controls⁽¹¹⁶⁾. Since then, reported frequencies of MetS in CCS vary between zero and 39 percent^(17, 35, 48, 116-134).

Our literature search retrieved twenty-two studies, six of which performed multivariable analyses in search of risk factors for developing MetS. Only three out of six had a prospective study-design. These were all reports from the French LEA program, a cohort of acute leukemia survivors⁽¹¹⁷⁻¹¹⁹⁾. MetS occurred in 6.9-17.1% of the survivors. In the first study, HSCT with TBI as conditioning regimen was the

only risk factor for metabolic syndrome (OR 3.9). In the second study, TBI was not a risk factor for MetS, nor were gender, total post-transplant steroid dose and follow-up duration. The only risk factor was higher BMI at time of transplantation (OR 1.57). In the third study, male sex (OR 2.64), older age at evaluation and higher BMI at diagnosis were risk factors for MetS, whereas CNS irradiation was not. The three other studies with multivariable analyses had a cross-sectional or retrospective study design. The largest investigated MetS in 784 ALL survivors in the St. Jude Lifetime Cohort, compared to 777 healthy controls⁽¹²⁰⁾. Metabolic syndrome was present in 33.6 percent of survivors (RR 1.43). Risk factors in multivariable analyses were CRT, especially with craniospinal radiation (RR 1.88), and older age at evaluation. Steroid dose was not a risk factor. A smaller study in 74 ALL survivors also revealed HSCT (OR 22.99) as risk factor for MetS⁽¹²¹⁾. In a large, retrospective study in 648 Indian childhood cancer survivors, not one patient fully met all the criteria for MetS⁽⁴⁸⁾. Only when overweight patients were included (next to obese patients), prevalence was 2.4% for underage survivors and 9.6% for survivors aged 18 years and older. It should be mentioned that follow-up in this study was short (6 and 11.5 years median for survivors below and over 18 years, respectively).

Of the remaining sixteen studies that our search yielded, three had a prospective design. The largest described MetS in a single center cohort of 103 nephro- and neuroblastoma survivors⁽³⁵⁾. Survivors had more components of MetS than healthy controls (OR 5.2 in nephroblastoma, 6.5 in neuroblastoma) and frequency was three times higher in patients who received abdominal irradiation (28% vs 9%). A small study in 21 AML survivors reported SCT as risk factor for having more MetS components than healthy controls (OR 24.1), whereas chemotherapy only was not a risk factor⁽¹²³⁾. In the third study, none of the 45 survivors of a hematological malignancy treated with HSCT had MetS⁽¹²²⁾, but this was also a study with a short follow-up time. Risk factors described in the other studies include cranial^(17, 125, 132) and abdominal radiation⁽¹³²⁾, while the other studies found no significant prognostic variables or did not perform this analysis.

Summarizing the studies with the highest quality of data, the following prognostic variables were risk factors for developing the MetS in CCS: the treatment components HSCT, CRT, TBI (although not all studies support this finding) and abdominal radiation, and the patient characteristics male sex (not in all studies), higher BMI at diagnosis or time of transplantation, and older age at evaluation.

Table 2. Meti	abolic syı	ndrome in	childhood cancer s	survivors						
Author	Year	z	Population	Design	Follow-up	MetS definition	% MetS	MVA	Prognostic variable	Control group
Saultier	2016	650	A(L/M)L HSCF-	1. Prospective	16y mean (±6.79y)	NCEP ATP III	6.9	Yes	Male, older age and high BMI at diagnosis	n.a.
Oudin	2011	184	A(L/M)L	1. Prospective	15.4y median (3.4- 30.2v)	NCEP ATP III	9.2	Yes	HSCT-TBI	n.a.
Oudin	2015	170	A(L/M)L HSCT+	1. Prospective	14.5y mean (±6.1y)	NCEP ATP III	17.1	Yes	BMI at time of	n.a.
Nottage	2014	784	ALL	2. Cross-sectional	26.1y median (11-	NCEP ATP III	33.6, RR	Yes	transplantation CRT	777
Chow	2010	74	ALL	2. Cross-sectional	45.3y) 10y/10.5y median (1-18y)	IDF / NCEP ATP III	$1.43 \\ 10.8$	Yes	TBI-HSCT, CRT, positive family history	NHANES n.a.
Prasad	2015	648 (471	CCS	3. Retrospective	6y median (2-16y)	IDF	0	Yes	n.a.	
		<18y)			(<18y); 11.5y median (2-41y) (≥18y)					
Van Waas	2012	103	Nephro-/ neuroblast.	1. Prospective	26.2y median (6.4- 48.9y)	NCEP ATP III	10 vs 5 (?)	No	Abd RTx	61 HC
Bizzarri	2015	45	Hemat. malig. w. HSCT	1. Prospective	4y mean (±3.2y)/6.9y (±3.1y)	(modified) NCEP ATP III	0	No	n.a.	90 HC
Blijdorp	2013	21	AML	1. Prospective	20y median (9-31y)	NCEP ATP III	10 vs 6.3	No	SCT	48 HC
Smith	2014	1598	CCS	2. Cross-sectional	25.6y mean (±7.6y)	NCEP ATP III	31.8	No	n.a.	n.a.
Van Waas	2010	500	CCS	2. Cross-sectional	19y median (6-49y)	Modified NCEP ATP III	13	No	CRT, ALL, male	MORGEN
Trimis	2007	80	ALL	2. Cross-sectional	6.37 median/5.9y mean (1.1-12.2y)	NCEP ATP III / WHO	11	No	Cranial radiation	n.a.
Mohapatra	2016	76	ALL	2. Cross-sectional	3y median (IQR 2.3-5y)	IDF / NCEP ATP III	5.2	No	n.a.	n.a.
Gurney	2006	75	ALL	2. Cross-sectional	24.6y mean (±4.8y)	NCEP ATP III	16.6 vs 17.5 (NS)	No	n.a.	730 Healthy NHANES
Aldhafiri	2012	56	ALL	2. Cross-sectional	6.2y mean (±3.9y)	IDF / NCEP ATP III	7.1 / 5.4	No	n.a.	n.a.
Talvensaari	1996	50	CCS	2. Cross-sectional	12.6y ave (7.9-21.3y)	Obesity, ↑insulin, ↓HDL	16 vs 0	No	n.a.	50 HC

Table 2. Co	ntinued									
Author	Year	z	Population	Design	Follow-up	MetS definition	% MetS	MVA	Prognostic variable	Control
Karakurt	2013	44	ALL	2. Cross-sectional	5.4y mean (3-10y)	IDF	6.8	Ν	n.a.	Family
Hoffman	2008	32	Sarcoma	2. Cross-sectional	17.3y median (2.9-	NCEP ATP III	33	No	n.a.	U.S.
					32.6y)					population data
Taskinen	2007	31	SCT	2. Cross-sectional	6y median (1-20y)	OHW	39	No	n.a.	n.a.
Van Waas	2013	532	CCS	3. Retrospective	17.9y median (5.0- 48.8y)	Modified NCEP ATP III	15	No	CRT, Abd RT _x	n.a.
Ness	2005	486	Adult CS	3. Retrospective	n.a.	NCEP ATP III	25.8 vs 18.4	No	Breast cancer history	12,526 HC
Kojima	2013	49	CCS	3. Retrospective	5.1y median (3.0-14.6y)) Japanese criteria	6.1	No	n.a.	n.a.
Highest qual	ity evidenc	e above e	dashed line, based o	n design (1 = prospect	ive, 2 = cross-sectional, 3	3 = retrospective) and	multivariable ana	lysis.		

Metabolic syndrome as cardiovascular risk factor in childhood cancer survivors

PATHOPHYSIOLOGY OF THE METABOLIC SYNDROME IN CHILDHOOD CANCER SURVIVORS

Growth hormone deficiency

Disease as well as treatment, i.e., respectively, brain tumors, $CRT^{(135-139)}$ and brain surgery^(140, 141), but also $TBI^{(135, 138, 142)}$ and chemotherapy^(143, 144) can damage the hypothalamus and pituitary gland, which leads to several endocrine disorders, the most common being $GHD^{(136, 145, 146)}$. GHD induces the components of the metabolic syndrome, as shown in several studies: adiposity⁽¹⁴⁷⁻¹⁵⁰⁾, insulin resistance^(151, 152), dyslipidemia^(149, 153, 154) and hypertension^(149, 154, 155). A recent study in CCS associated GHD with the development of clusters of three or more cardiovascular risk factors⁽¹⁵⁵⁾. GHD has also been linked to endothelial dysfunction and atherosclerosis^(156, 157) and to decreased left ventricular ejection fraction⁽¹⁵⁸⁾, further increasing the risk of cardiovascular complications. Schneider et al. reported an increased ten-year risk of cardiovascular events in ~350 GHD patients compared to healthy controls (4.6% vs. 3.7%)⁽¹⁵⁹⁾.

The hypothalamus, rather than the pituitary gland, is regarded as the primary site of radiation damage^(136, 160, 161). The somatotropic axis is affected first, followed by the gonadal axis, and, least sensitive, the thyroid and adrenal axis^(136, 137, 162). After radiotherapy growth hormone secretion may gradually and irreversibly decrease over the course of years in a dose-dependent manner; at 16 Gy the risk of developing GHD five year off treatment is 50%⁽¹⁶³⁾. The most relevant radiotherapy threshold is not clear: other reported thresholds are 18 Gy⁽⁴²⁾, 22 Gy⁽¹³⁷⁾ and 30 Gy⁽¹⁶²⁾. In a meta-analysis by Mulder the pooled prevalence of GHD after cranial radiation was $35.6\%^{(164)}$.

In non-cancer survivors with GHD, it has been shown that growth hormone replacement has positive effects on cardiac function, cardiovascular risk factors such as body composition, lipid levels and blood pressure, and on the occurrence of cardiovascular events^(159, 165-169). On the other hand, a large study in ~2500 growth hormone deficient adults found no decrease in the prevalence of MetS after three years of replacement⁽¹⁷⁰⁾, and Claessen even reported a substantial increase in MetS after ten years of treatment in 98 patients, from 32.7% to 57.1%⁽¹⁷¹⁾. Unfortunately, there is only scarce literature on growth hormone replacement in CCS. Furthermore, clinical interpretation of these studies is commonly complicated by methodologic shortcomings such as lack of a control group and the use of surrogate markers instead of cardiovascular morbidity and mortality. A small study in eighteen ALL survivors with GHD on two years' replacement therapy reported improved cardiac systolic function and reduced incidence of metabolic syndrome⁽¹⁷²⁾. Another small

study with eleven ALL survivors on twelve months' growth hormone replacement reported positive effects on fat mass and fat free mass, but hyperleptinemia and insulin resistance remained unaffected⁽¹⁷³⁾. Van den Heijkant found higher lean mass and lower percentage fat after two years of therapy in 14 ALL survivors⁽¹⁷⁴⁾, and Murray reported beneficial effects on waist-hip ratio, cholesterol and triglycerides in 27 ALL and brain tumor survivors after twelve months' therapy⁽¹⁷⁵⁾.

Gonadal impairment, thyroid morbidity and adrenal insufficiency

The production of other pituitary hormones, and damage to other endocrine end organs seems to be less frequently affected after childhood cancer and therapies. CRT >30Gy causes long-term central hypogonadism in 20-30% of survivors⁽¹⁶²⁾, and regimens harming the gonads can be causative factors as well⁽¹⁷⁶⁻¹⁷⁹⁾. Mainly tested in men, hypogonadism is reported to contribute to MetS and vice versa⁽¹⁸⁰⁻¹⁸⁴⁾. A few studies associated gonadal impairment with MetS traits in CCS^(27, 130, 185-187). A recent meta-analysis reported that testosterone supplementation in men with testosterone deficiency syndrome had positive effects on body weight and composition and glucose and lipid metabolism⁽¹⁸⁸⁾. There are no studies available that investigated the effect of sex hormone therapy on the metabolic syndrome in CCS, so far.

Cranial radiation doses of 30Gy and higher cause central hypothyroidism in 3-9% of survivors⁽¹⁶²⁾. Thyroid malignancies (although rare in children) or neck and mantle radiation for other cancer types damage the thyroid and lead to primary hypothyroidism^(189, 190). Metabolic manifestations of hypothyroidism include adiposity, hypertension (due to an increase in peripheral vascular resistance) and dyslipidemia⁽¹⁹¹⁻¹⁹⁴⁾. As in the normal population, hypothyroidism after childhood cancer is treated with levothyroxine. Although it is anticipated that levothyroxine treatment has positive effects on the metabolic profile of CCS, no studies have investigated this yet.

Adrenal insufficiency occurs in 3-6% of patients receiving >30Gy CRT⁽¹⁶²⁾ and can also temporally be caused by high-dose steroid treatment. Hypocortisolism in itself is not associated with MetS features, but treatment with corticosteroids – especially dexamethasone – is notorious for causing short term adiposity, IR and diabetes^(195, 196). It is conceivable that the use of glucocorticoids in childhood cancer treatment can have these consequences on long term as well. Van Beek showed that treatment with prednisone or dexamethasone is associated with long term increases in BMI and body fat in ALL and Hodgkin lymphoma survivors^(26, 197).

General fitness

Another potential mechanism for the development of MetS in CCS is physical inactivity, as this promotes obesity and IR. In the St. Jude Lifetime cohort, 28% of survivors were found not to adhere to lifestyle guidelines. Males and females who did not follow these guidelines were approximately twice more likely to have MetS⁽¹²⁴⁾. Similarly, Warner reported total energy expenditure and physical activity to be lower in 34 ALL survivors compared to 21 survivors of other childhood malignancies, and to healthy controls. This was negatively associated with percentage body fat, but it remains the question whether this is either a cause or a consequence⁽¹⁹⁸⁾. Additionally, we showed that especially male neuroblastoma survivors might be at risk for reduced physical activity⁽¹⁹⁹⁾. Visual impairment after certain brain tumors may enhance MetS risk, due to the reduced ability to perform physical activity. For example, in a study in 178 childhood- and adult-onset craniopharyngioma survivors, visual impairment was a borderline significant independent risk factor for MetS⁽²⁰⁰⁾. To date, it is not entirely clear yet whether reduced physical activity and sedentary lifestyle play a causative role in development of MetS. However, as it is one of the few modifiable factors that might decrease MetS, it is of great value to initiate intervention studies with regard to physical activity in CCS.

Genetic susceptibility

The role of genetic susceptibility in the development of MetS and cardiovascular disease in childhood cancer survivors has not been extensively studied yet. In our cohort of 532 survivors, we used a candidate gene approach, containing genes previously associated with components of the metabolic syndrome, i.e. JAZF1, THADA, IRS1, TFAP2B, MSRA and ATP2B1⁽¹³²⁾. None of the allelic variants was associated with metabolic syndrome, indicating that treatment factors were more dominant than genetic variation. England et al. performed whole-exome sequencing in 209 ALL survivors and reported that variants in BAD and FCRL3 genes were associated with a phenotype of three or more cardiometabolic risk factors⁽²⁰¹⁾. In the St. Jude Lifetime cohort, a genome-wide association study (GWAS) identified single nucleotide polymorphisms associated with obesity in the following genes: FAM155A, which is expressed in the hypothalamus and pituitary, and GLRA3, SOX11 and CDH18, which are involved in neural growth, repair and connectivity⁽⁴³⁾. To date, these findings have not been validated, nor has GWAS been performed to identify genetic variants associated with diabetes, dyslipidemia, hypertension and MetS in CCS⁽²⁰²⁾.

SUMMARY AND FUTURE PERSPECTIVE

After almost 25 years of research on childhood cancer survivors, we have gained knowledge on potential late effects, of which the metabolic syndrome so far has been rather disguised. Many CCS are already at risk for cardiovascular disease, for example, due to anthracycline- or radiation-induced cardiotoxicity⁽²⁰³⁻²⁰⁵⁾. Additionally, they face an additive risk after CRT, causing GHD and MetS. The role between MetS and other risk factors, such as abdominal radiation, specific chemotherapeutic agents, steroids, gonadal impairment, thyroid morbidity and genetics, warrants further investigation. It is however clear that specific groups of CCS are at higher risk of developing components of the MetS (Figure 1), which underlines the need for close monitoring. These survivors might benefit from early interventions targeting overweight, hypertension and dyslipidemia, for instance lifestyle and diet advice and medication⁽¹⁰⁵⁾. Since MetS is a cluster of symptoms with heterogeneous presentation among individuals, medical treatment requires a personalized approach⁽¹⁰⁵⁾.

In our opinion, future research may focus on the following three topics: 1) Unravelling the pathophysiologic mechanism of the development of the MetS in specific CCS subgroups, 2) Determining which subgroups of CCS are at risk to develop (components of) MetS by using prediction models, and 3) Determining which preventive and therapeutic interventions are successful in targeting the MetS in CCS – favourably multiple components with the same intervention. As childhood cancer is relatively rare, research will benefit from collaborations between (inter)national cohorts, to enhance effect size and for replication purposes.

Funding

This work was funded by Foundation KiKa (Children Cancerfree) project number 171/ODAS.

Author	Year	Ν	Population	Outcome	Result
Reulen	2010	17.981	CCS	Death due to circulatory disease	SMR 4.0 (95% CI 3.4-4.6)
Tukenova	2010	4.122	CCS	Death due to cardiovascular disease	SMR 5.0 (95% CI 3.3-6.7)
Castellino	2011	2.742	Hodgkin	Death due to cardiovascular disease	EAR 13.1/10,000py
Prasad	2012	9.245	CCS	Death due to cardiovascular disease	SMR 1.9 (95% CI 1.5-2.3)
Armstrong	2013	10.724	CCS	Coronary artery disease	Cum. inc. 5.3% (95% CI 4.4-6.1%)
Perkins	2013	3.627	CNS tumor	Death due to cardiovascular disease	SMR 2.5 (95% CI 1.2-4.8)
Kero	2014	13.860	CCS	Cardiovascular hospitalizations	HR 3.4 (95% CI 2.3-5.1) / 3.3 (95% CI 1.7-6.5)
Van Laar	2014	3.247	CCS	Cardiovascular hospitalizations	RR 2.6 (95% CI 1.9-3.6) / 1.2 (95% CI 0.9-1.5)
Olsen	2014	2.243	CCS	Cardiovascular hospitalizations	HR 2.5 (95% CI 2.1 - 2.9)
Kero	2015	16.769	CCS	Death due to cardiovascular disease	SMR 1.9 (95% CI 1.7-2.1)
Gudmundsdottir	2015	32.308	CCS	Cardiovascular hospitalizations	RR 2.1 (95% CI 2.0-2.2), EAR 324/100,000py
Bhakta	2016	348	Hodgkin	Cardiovascular disease	Cum. burden 100.8 vs 17.0/100 survivors
Schindler	2016	3.965	CCS	Death due to circulatory disease	SMR 12.7 (95% CI 7.8-20.7)
Kero	2016	8.197	CCS	Cardiovascular medication	HR 7.2 (95% CI 5.1-10.1)

Supplemental table 1. Epidemiological studies on cardiovascular disease in childhood cancer survivors

SMR = standardized mortality ratio; EAR = excess adverse risk.

Supplemental table 2. WHO, NCEP-ATP III and IDF criteria for metabolic syndrome

		WHO	NCEP-ATP III	IDF
Required for diagnosis		IR + ≥ 2 others	≥3	Adiposity + ≥2 others
Adiposity	BMI (kg/m ²)	>30	-	-
	Waist (cm)	-	>102*/88**	≥94*/80**
	Waist/hip	>0.90*/0.85**	-	-
Insulin resistance	Fasting glucose (mmol/L)	≥5.5 or DM2	≥5.5 or Rx	≥5.5 or DM2
	IGT (mmol/L)	6.1	-	-
Dyslipidemia	Triglycerides (mmol/L)	≥1.7	≥1.7 or Rx	≥1.7 or Rx
	HDL cholesterol (mmol/L)	<0.9*/1.0**	$<1.03^{*}/1.3^{**}$ or Rx	$<1.03^{*}/1.3^{**}$ or Rx
Blood pressure (mmHg)		≥140/90	≥130/85 or Rx	≥130/85 or Rx
Other		Microalbuminuria	-	-

REFERENCES

- 1. SEER Cancer Statistics Review, 1975-2014, National Cancer Institutie [Internet]. 2016 [cited November 2017].
- 2. Meadows AT. Pediatric cancer survivors: past history and future challenges. Current Problems in Cancer. 2003;27(3):112-26.
- 3. Reulen RC, Winter DL, Frobisher C, Lancashire ER, Stiller CA, Jenney ME, et al. Long-term Cause-Specific Mortality Among Survivors of Childhood Cancer. JAMA. 2010;304(2):172-9.
- Tukenova M, Guibout C, Oberlin O, Doyon F, Mousannif A, Haddy N, et al. Role of cancer treatment in long-term overall and cardiovascular mortality after childhood cancer. J Clin Oncol. 2010;28(8):1308-15.
- 5. Castellino SM, Geiger AM, Mertens AC, Leisenring WM, Tooze JA, Goodman P, et al. Morbidity and mortality in long-term survivors of Hodgkin lymphoma: a report from the Childhood Cancer Survivor Study. Blood. 2011;117(6):1806-16.
- Prasad PK, Signorello LB, Friedman DL, Boice JD, Jr., Pukkala E. Long-term non-cancer mortality in pediatric and young adult cancer survivors in Finland. Pediatr Blood Cancer. 2012;58(3):421-7.
- Armstrong GT, Oeffinger KC, Chen Y, Kawashima T, Yasui Y, Leisenring W, et al. Modifiable risk factors and major cardiac events among adult survivors of childhood cancer. J Clin Oncol. 2013;31(29):3673-80.
- 8. Perkins SM, Fei W, Mitra N, Shinohara ET. Late causes of death in children treated for CNS malignancies. J Neurooncol. 2013;115(1):79-85.
- Kero AE, Jarvela LS, Arola M, Malila N, Madanat-Harjuoja LM, Matomaki J, et al. Cardiovascular morbidity in long-term survivors of early-onset cancer: a population-based study. Int J Cancer. 2014;134(3):664-73.
- van Laar M, Feltbower RG, Gale CP, Bowen DT, Oliver SE, Glaser A. Cardiovascular sequelae in long-term survivors of young peoples' cancer: a linked cohort study. Br J Cancer. 2014;110(5):1338-41.
- 11. Olsen M, Schmidt M, Lash TL, Sørensen K, Pedersen L, Sørensen HT. Cardiovascular disease risk in childhood cancer survivors. Am J Epidemiol. 2014;180(1):120-3.
- 12. Kero AE, Jarvela LS, Arola M, Malila N, Madanat-Harjuoja LM, Matomaki J, et al. Late mortality among 5-year survivors of early onset cancer: a population-based register study. Int J Cancer. 2015;136(7):1655-64.
- Gudmundsdottir T, Winther JF, de Fine Licht S, Bonnesen TG, Asdahl PH, Tryggvadottir L, et al. Cardiovascular disease in Adult Life after Childhood Cancer in Scandinavia: A populationbased cohort study of 32,308 one-year survivors. Int J Cancer. 2015;137(5):1176-86.
- Bhakta N, Liu Q, Yeo F, Baassiri M, Ehrhardt MJ, Srivastava DK, et al. Cumulative burden of cardiovascular morbidity in paediatric, adolescent, and young adult survivors of Hodgkin's lymphoma: an analysis from the St Jude Lifetime Cohort Study. The Lancet Oncology. 2016;17(9):1325-34.
- Schindler M, Spycher BD, Ammann RA, Ansari M, Michel G, Kuehni CE, et al. Cause-specific long-term mortality in survivors of childhood cancer in Switzerland: A population-based study. Int J Cancer. 2016;139(2):322-33.
- Kero AE, Madanat-Harjuoja LM, Jarvela LS, Malila N, Matomaki J, Lahteenmaki PM. Cardiovascular medication after cancer at a young age in Finland: A nationwide registry linkage study. Int J Cancer. 2016;139(3):683-90.

- 17. van Waas M, Neggers SJ, Pieters R, van den Heuvel-Eibrink MM. Components of the metabolic syndrome in 500 adult long-term survivors of childhood cancer. Ann Oncol. 2010;21(5):1121-6.
- 18. Rosen GP, Nguyen HT, Shaibi GQ. Metabolic syndrome in pediatric cancer survivors: a mechanistic review. Pediatr Blood Cancer. 2013;60(12):1922-8.
- 19. Casco S, Soto-Vega E. Development of Metabolic Syndrome Associated to Cancer Therapy: Review. Horm Cancer. 2016;7(5-6):289-95.
- 20. Obesity and overweight fact sheet [Internet]. 2017 [cited November 2017].
- 21. Willett WC, Dietz WH, Colditz GA. Guidelines for Healthy Weight. NEJM. 1999;341(6):427-34.
- 22. Walker SP, Rimm EB, Ascheno A, Kawachi I, Stampfer MJ, Willett WC. Body Size and Fat Distribution as Predictors of Stroke among US Men. Am J Epidemiol. 1996;144(12):1143-50.
- 23. Renehan AG, Tyson M, Egger M, Heller RF, Zwahlen M. Body-mass index and incidence of cancer: a systematic review and meta-analysis of prospective observational studies. The Lancet. 2008;371(9612):569-78.
- 24. Nysom K, Holm K, Fleischer Michaelsen K, Hertz H, Müller J, Mølgaard C. Degree of Fatness after Treatment for Acute Lymphoblastic Leukemia in Childhood. J Clin Endocrinol Metab. 1999;84(12):4591-6.
- 25. Nysom K, Holm K, Michaelsen KF, Hertz H, Jacobsen N, Müller J, et al. Degree of fatness after allogeneic BMT for childhood leukaemia or lymphoma. Bone Marrow Transplant. 2001;27:817-20.
- 26. van Beek RD, de Muinck Keizer-Schrama SM, Hakvoort-Cammel FG, van der Sluis IM, Krenning EP, Pieters R, et al. No difference between prednisolone and dexamethasone treatment in bone mineral density and growth in long term survivors of childhood acute lymphoblastic leukemia. Pediatr Blood Cancer. 2006;46(1):88-93.
- 27. Neville KA, Cohn RJ, Steinbeck KS, Johnston K, Walker JL. Hyperinsulinemia, impaired glucose tolerance, and diabetes mellitus in survivors of childhood cancer: prevalence and risk factors. J Clin Endocrinol Metab. 2006;91(11):4401-7.
- 28. Jarfelt M, Lannering B, Bosaeus I, Johannsson G, Bjarnason R. Body composition in young adult survivors of childhood acute lymphoblastic leukaemia. Eur J Endocrinol. 2005;153(1):81-9.
- 29. Janiszewski PM, Janssen I, Ross R. Does Waist Circumference Predict Diabetes and Cardiovascular Disease Beyond Commonly Evaluated Cardiometabolic Risk Factors? Diabetes Cares. 2007;30:3105-9.
- 30. Blijdorp K, van den Heuvel-Eibrink MM, Pieters R, Boot AM, Delhanty PJ, van der Lely AJ, et al. Obesity is underestimated using body mass index and waist-hip ratio in long-term adult survivors of childhood cancer. PLoS One. 2012;7(8):e43269.
- 31. Shuster A, Patlas M, Pinthus JH, Mourtzakis M. The clinical importance of visceral adiposity: a critical review of methods for visceral adipose tissue analysis. Br J Radiol. 2012;85(1009):1-10.
- 32. Fox CS, Massaro JM, Hoffmann U, Pou KM, Maurovich-Horvat P, Liu CY, et al. Abdominal visceral and subcutaneous adipose tissue compartments: association with metabolic risk factors in the Framingham Heart Study. Circulation. 2007;116(1):39-48.
- Mostoufi-Moab S, Ginsberg JP, Bunin N, Zemel BS, Shults J, Thayu M, et al. Body composition abnormalities in long-term survivors of pediatric hematopoietic stem cell transplantation. J Pediatr. 2012;160(1):122-8.
- 34. Siviero-Miachon AA, Spinola-Castro AM, de Martino Lee ML, Andreonie S, Geloneze B, Lederman H, et al. Cranial radiotherapy predisposes to abdominal adiposity in survivors of childhood acute lymphocytic leukemia. Radiat Oncol. 2013;8(39):1-9.
- 35. van Waas M, Neggers SJ, Raat H, van Rij CM, Pieters R, van den Heuvel-Eibrink MM. Abdominal radiotherapy: a major determinant of metabolic syndrome in nephroblastoma and neuroblastoma survivors. PLoS One. 2012;7(12):e52237.

- 36. Marriott CJC, Beaumont LF, Farncombe TH, Cranston AN, Athale UH, Yakemchuk VN, et al. Body composition in long-term survivors of acute lymphoblastic leukemia diagnosed in childhood and adolescence: A focus on sarcopenic obesity. Cancer. 2017.
- 37. Zee P, Chen C-H. Prevalence of Obesity in Children After Therapy for Acute Lymphoblastic Leukemia. Am J Pediatr Hematol Oncol. 1986;8(4):294-9.
- 38. Odame I, Reilly JJ, Gibson BES, Donaldson MDC. Patterns of obesity in boys and girls after treatment for acute lymphoblastic leukaemia. Arch Dis Child. 1994;71:147-9.
- Sklar CA, Mertens AC, Walter A, Mitchell D, Nesbit ME, O', et al. Changes in Body Mass Index and Prevalence of Overweight in Survivors of Childhood Acute Lymphoblastic Leukemia: Role of Cranial Irradiation. Med Pediatr Oncol. 2000;35:91-5.
- Oeffinger KC, Mertens AC, Sklar CA, Yasui Y, Fears T, Stovall M, et al. Obesity in adult survivors of childhood acute lymphoblastic leukemia: a report from the Childhood Cancer Survivor Study. J Clin Oncol. 2003;21(7):1359-65.
- Garmey EG, Liu Q, Sklar CA, Meacham LR, Mertens AC, Stovall MA, et al. Longitudinal changes in obesity and body mass index among adult survivors of childhood acute lymphoblastic leukemia: a report from the Childhood Cancer Survivor Study. J Clin Oncol. 2008;26(28):4639-45.
- 42. Mostoufi-Moab S, Seidel K, Leisenring WM, Armstrong GT, Oeffinger KC, Stovall M, et al. Endocrine Abnormalities in Aging Survivors of Childhood Cancer: A Report From the Childhood Cancer Survivor Study. J Clin Oncol. 2016;34(27):3240-7.
- 43. Wilson CL, Liu W, Yang JJ, Kang G, Ojha RP, Neale GA, et al. Genetic and clinical factors associated with obesity among adult survivors of childhood cancer: A report from the St. Jude Lifetime Cohort. Cancer. 2015;121(13):2262-70.
- 44. Zhang FF, Kelly MJ, Saltzman E, Must A, Roberts SB, Parsons SK. Obesity in pediatric ALL survivors: a meta-analysis. Pediatrics. 2014;133(3):e704-15.
- 45. van Santen HM, Geskus RB, Raemaekers S, van Trotsenburg AS, Vulsma T, van der Pal HJ, et al. Changes in body mass index in long-term childhood cancer survivors. Cancer. 2015;121(23):4197-204.
- Brown AL, Lupo PJ, Danysh HE, Okcu MF, Scheurer ME, Kamdar KY. Prevalence and Predictors of Overweight and Obesity Among a Multiethnic Population of Pediatric Acute Lymphoblastic Leukemia Survivors: A Cross-Sectional Assessment. J Pediatr Hematol Oncol. 2016;38(6):429-36.
- 47. Felicetti F, D'Ascenzo F, Moretti C, Corrias A, Omede P, Marra WG, et al. Prevalence of cardiovascular risk factors in long-term survivors of childhood cancer: 16 years follow up from a prospective registry. Eur J Prev Cardiol. 2015;22(6):762-70.
- Prasad M, Arora B, Chinnaswamy G, Vora T, Narula G, Banavali S, et al. Nutritional status in survivors of childhood cancer: Experience from Tata Memorial Hospital, Mumbai. Indian J Cancer. 2015;52(2):219-23.
- 49. Lindemulder SJ, Stork LC, Bostrom B, Lu X, Devidas M, Hunger S, et al. Survivors of standard risk acute lymphoblastic leukemia do not have increased risk for overweight and obesity compared to non-cancer peers: a report from the Children's Oncology Group. Pediatr Blood Cancer. 2015;62(6):1035-41.
- Gunn HM, Emilsson H, Gabriel M, Maguire AM, Steinbeck KS. Metabolic Health in Childhood Cancer Survivors: A Longitudinal Study in a Long-Term Follow-Up Clinic. J Adolesc Young Adult Oncol. 2016;5(1):24-30.
- 51. Belle FN, Weiss A, Schindler M, Goutaki M, Bochud M, Zimmermann K, et al. Overweight in childhood cancer survivors: the Swiss Childhood Cancer Survivor Study. Am J Clin Nutr. 2018;107:3-11.

- 52. Collaboration TERF. Diabetes mellitus, fasting blood glucose concentration, and risk of vascular disease: a collaborative meta-analysis of 102 prospective studies. Lancet. 2010;375:2215-22.
- 53. WHO, IDF. Definition and Diagnosis of Diabetes Mellitus and Intermediate Hyperglycemia. http://www.who.int/diabetes/publications/diagnosis_diabetes2006/en/; 2006.
- 54. Donath MY, Shoelson SE. Type 2 diabetes as an inflammatory disease. Nat Rev Immunol. 2011;11(2):98-107.
- 55. Tack CJ, Stienstra R, Joosten LAB, Netea MG. Inflammation links excess fat to insulin resistance: the role of the interleukin-1 family. Immunol Rev. 2012;249:239-52.
- 56. Diabetes fact sheet [Internet]. 2017 [cited November 2017].
- 57. Colditz GA, Willett WC, Rotnitzky A, Manson JE. Weight Gain as a Risk Factor for Clinical Diabetes Mellitus in Women. Ann Intern Med. 1995;122(7):481-6.
- 58. Mokdad AH, Ford ES, Bowman BA, Dietz WH, Vinicor F, Bales VS, et al. Prevalence of Obesity, Diabetes, and Obesity-Related Health Risk Factors, 2001. JAMA. 2003;289:76-9.
- Nguyen NT, Nguyen XM, Lane J, Wang P. Relationship between obesity and diabetes in a US adult population: findings from the National Health and Nutrition Examination Survey, 1999-2006. Obes Surg. 2011;21(3):351-5.
- 60. Menke A, Rust KF, Fradkin J, Cheng YJ, Cowie CC. Associations between trends in race/ ethnicity, aging, and body mass index with diabetes prevalence in the United States: a series of cross-sectional studies. Ann Intern Med. 2014;161(5):328-35.
- 61. Baker KS, Ness KK, Steinberger J, Carter A, Francisco L, Burns LJ, et al. Diabetes, hypertension, and cardiovascular events in survivors of hematopoietic cell transplantation: a report from the bone marrow transplantation survivor study. Blood. 2007;109(4):1765-72.
- 62. Meacham LR, Sklar CA, Li S, Liu Q, Gimpel N, Yasui Y, et al. Diabetes mellitus in long-term survivors of childhood cancer. Increased risk associated with radiation therapy: a report for the childhood cancer survivor study. Arch Intern Med. 2009;169(15):1381-8.
- 63. de Vathaire F, El-Fayech C, Ben Ayed FF, Haddy N, Guibout C, Winter D, et al. Radiation dose to the pancreas and risk of diabetes mellitus in childhood cancer survivors: a retrospective cohort study. The Lancet Oncology. 2012;13(10):1002-10.
- 64. Cohen LE, Gordon JH, Popovsky EY, Gunawardene S, Duffey-Lind E, Lehmann LE, et al. Late effects in children treated with intensive multimodal therapy for high-risk neuroblastoma: high incidence of endocrine and growth problems. Bone Marrow Transplant. 2014;49(4):502-8.
- 65. Teinturier C, Tournade M-F, Caillat-Zucman S, Boitard C, Amoura Z, Bougneres P-F, et al. Diabetes mellitus after abdominal radiation therapy. Lancet. 1995;346:633-4.
- 66. Cicognani A, Cacciari E, Mancini AF, Pasini A, Salardi S, Salmi S, et al. Abnormal insulin response to glucose following treatment for Wilms' tumor in childhood. Eur J Pediatr. 1997;156:371-5.
- 67. Holmqvist AS, Olsen JH, Andersen KK, de Fine Licht S, Hjorth L, Garwicz S, et al. Adult life after childhood cancer in Scandinavia: diabetes mellitus following treatment for cancer in childhood. Eur J Cancer. 2014;50(6):1169-75.
- 68. Hoffmeister PA, Storer BE, Sanders JE. Diabetes Mellitus in Long-Term Survivors of Pediatric Hematopoietic Cell Transplantation. J Pediatr Hematol Oncol. 2004;26:81-90.
- 69. Chao C, Xu L, Bell E, Cooper R, Mueller L. Long-term Health Outcomes in Survivors of Childhood Cancer Diagnosed Between 1990 and 2000 in a Large US Integrated Health Care System. J Pediatr Hematol Oncol. 2016;38:123-30.
- van Nimwegen FA, Schaapveld M, Janus CP, Krol AD, Raemaekers JM, Kremer LC, et al. Risk of diabetes mellitus in long-term survivors of Hodgkin lymphoma. J Clin Oncol. 2014;32(29):3257-63.

- 71. Wei C, Thyagiarajan M, Hunt L, Cox R, Bradley K, Elson R, et al. Reduced beta-cell reserve and pancreatic volume in survivors of childhood acute lymphoblastic leukaemia treated with bone marrow transplantation and total body irradiation. Clin Endocrinol (Oxf). 2015;82(1):59-67.
- 72. Berryman DE, Henry B, Hjortebjerg R, List EO, Kopchick JJ. Developments in our understanding of the effects of growth hormone on white adipose tissue from mice: implications to the clinic. Expert Rev Endocrinol Metab. 2016;11(2):197-207.
- 73. Sun K, Tordjman J, Clement K, Scherer PE. Fibrosis and adipose tissue dysfunction. Cell Metab. 2013;18(4):470-7.
- 74. Genest JJ, Martin-Munley SS, McNamara JR, Ordovas JM, Jenner J, Myers RH, et al. Familial Lipoprotein Disorders in Patients With Premature Coronary Artery Disease. Circulation. 1992;85:2025-33.
- 75. Mora S, Buring JE, Ridker PM, Cui Y. Association of HDL Cholesterol with Incident Cardiovascular Events by LDL Cholesterol and Apolipoprotein B100 in Women. Ann Intern Med. 2011;155(11):742-50.
- 76. Parish S, Offer A, Clarke R, Hopewell JC, Hill MR, Otvos JD, et al. Lipids and lipoproteins and risk of different vascular events in the MRC/BHF Heart Protection Study. Circulation. 2012;125(20):2469-78.
- 77. Jung UJ, Choi MS. Obesity and its metabolic complications: the role of adipokines and the relationship between obesity, inflammation, insulin resistance, dyslipidemia and nonalcoholic fatty liver disease. Int J Mol Sci. 2014;15(4):6184-223.
- Haugnes HS, Wethal T, Aass N, Dahl O, Klepp O, Langberg CW, et al. Cardiovascular risk factors and morbidity in long-term survivors of testicular cancer: a 20-year follow-up study. J Clin Oncol. 2010;28(30):4649-57.
- 79. Hashibe M, Abdelaziz S, Al-Temimi M, Fraser A, Boucher KM, Smith K, et al. Long-term health effects among testicular cancer survivors. J Cancer Surviv. 2016;10(6):1051-7.
- Amir E, Seruga B, Niraula S, Carlsson L, Ocana A. Toxicity of adjuvant endocrine therapy in postmenopausal breast cancer patients: a systematic review and meta-analysis. J Natl Cancer Inst. 2011;103(17):1299-309.
- Kintzel PE, Chase SL, Schultz LM, O'Rourke TJ. Increased Risk of Metabolic Syndrome, Diabetes Mellitus, and Cardiovascular Disease in Men Receiving Androgen Deprivation Therapy for Prostate Cancer. Pharmacotherapy. 2008;28(12):1511-22.
- 82. Kero AE, Madanat-Harjuoja LM, Jarvela LS, Malila N, Matomaki J, Lahteenmaki PM. Health conditions associated with metabolic syndrome after cancer at a young age: A nationwide register-based study. Cancer Epidemiol. 2016;41:42-9.
- 83. Raised blood pressure situation and trends [Internet]. 2017 [cited November 2017].
- 84. Q&As on hypertension [Internet]. 2017 [cited November 2017].
- 85. WHO. A global brief on hypertension. http://www.who.int/cardiovascular_diseases/publications/global_brief_hypertension/en/; 2013.
- 86. Levy D, Larson MG, Vasan RS, Kannel WB, Ho KK. The progression from hypertension to congestive heart failure. JAMA. 1996;275(20):1557-62.
- 87. Thrift AG, McNeil JJ, Forbes A, Donnan GA. Risk Factors for Cerebral Hemorrhage in the Era of Well-Controlled Hypertension. Stroke. 1996;27(11):2020-5.
- Staessen JA, Fagard R, Thijs L, Celis H, Arabidze GG, Birkenhäger WH, et al. Randomised double-blind comparison of placebo and active treatment for older patients with isolated systolic hypertension. The Lancet. 1997;350(9080):757-64.

- Coresh J, Wei GL, McQuillan G, Brancati FL, Levey AS, Jones C, et al. Prevalence of High Blood Pressure and Elevated Serum Creatinine Level in the United States. Arch Intern Med. 2001;161:1207-16.
- Emdin CA, Anderson SG, Callender T, Conrad N, Salimi-Khorshidi G, Mohseni H, et al. Usual blood pressure, peripheral arterial disease, and vascular risk: cohort study of 4.2 million adults. BMJ. 2015;351:h4865.
- 91. Kantor AF, Li FP, Janov AJJ, Tarbell NJ, Sallan SE. Hypertension in Long-Term Survivors of Childhood Renal Cancers. J Clin Oncol. 1989;7:912-5.
- 92. Knijnenburg SL, Mulder RL, Schouten-Van Meeteren AY, Bokenkamp A, Blufpand H, van Dulmen-den Broeder E, et al. Early and late renal adverse effects after potentially nephrotoxic treatment for childhood cancer. Cochrane Database Syst Rev. 2013(10):CD008944.
- 93. Dekkers IA, Blijdorp K, Cransberg K, Pluijm SM, Pieters R, Neggers SJ, et al. Long-term nephrotoxicity in adult survivors of childhood cancer. Clin J Am Soc Nephrol. 2013;8(6):922-9.
- 94. Gibson TM, Li Z, Green DM, Armstrong GT, Mulrooney DA, Srivastava D, et al. Blood Pressure Status in Adult Survivors of Childhood Cancer: A Report from the St. Jude Lifetime Cohort Study. Cancer Epidemiol Biomarkers Prev. 2017;26(12):1705-13.
- 95. Breitz H. Clinical aspects of radiation nephropathy. Cancer Biother Radiopharm. 2004;19(3):359-62.
- Schell M, Cochat P, Hadj-Aïssa A, Bouffet E, Dubourg L, Brunat-Mentigny M. Renal function following unilateral nephrectomy for neuroblastoma and Wilms' tumour. Pediatr Nephrol. 1995;9(5):579-82.
- 97. Oberlin O, Fawaz O, Rey A, Niaudet P, Ridola V, Orbach D, et al. Long-term evaluation of Ifosfamide-related nephrotoxicity in children. J Clin Oncol. 2009;27(32):5350-5.
- Skinner R, Parry A, Price L, Cole M, Craft AW, Pearson AD. Persistent nephrotoxicity during 10-year follow-up after cisplatin or carboplatin treatment in childhood: relevance of age and dose as risk factors. Eur J Cancer. 2009;45(18):3213-9.
- Cardous-Ubbink MC, Geenen MM, Schade KJ, Heinen RC, Caron HN, Kremer LC, et al. Hypertension in long-term survivors of childhood cancer: a nested case-control study. Eur J Cancer. 2010;46(4):782-90.
- 100. Blood Pressure Lowering Treatment Trialists C, Turnbull F, Neal B, Ninomiya T, Algert C, Arima H, et al. Effects of different regimens to lower blood pressure on major cardiovascular events in older and younger adults: meta-analysis of randomised trials. BMJ. 2008;336(7653):1121-3.
- 101. National Cholesterol Education Program Expert Panel on Detection E, and Treatment of High Blood Cholesterol in Adults (Adult Treatment Panel III). Third Report of the National Cholesterol Education Program (NCEP) Expert Panel on Detection, Evaluation, and Treatment of High Blood Cholesterol in Adults (Adult Treatment Panel III). 2002.
- 102. Alberti G, Zimmet P, Shaw J, Grundy SM. The IDF consensus worldwide definition of the metabolic syndrome. 2006.
- Alberti KGMM, Zimmet PZ. Definition, Diagnosis and Classification of Diabetes Mellitus and its Complications Part 1: Diagnosis and Classification of Diabetes Mellitus Provisional Report of a WHO Consultation. Diabet Med. 1998;15:539-53.
- Reaven GM. Banting lecture 1988. Role of insulin resistance in human disease. Diabetes. 1988;37(12):1595-607.
- Moller DE, Kaufman KD. Metabolic syndrome: a clinical and molecular perspective. Annu Rev Med. 2005;56:45-62.
- 106. Eckel RH, Grundy SM, Zimmet PZ. The metabolic syndrome. The Lancet. 2005;365(9468):1415-28.
- 107. Wilson PW, D'Agostino RB, Parise H, Sullivan L, Meigs JB. Metabolic syndrome as a precursor of cardiovascular disease and type 2 diabetes mellitus. Circulation. 2005;112(20):3066-72.

- 108. Ford ES. The metabolic syndrome and mortality from cardiovascular disease and all-causes: findings from the National Health and Nutrition Examination Survey II Mortality Study. Atherosclerosis. 2004;173(2):309-14.
- 109. Marceau P, Biron S, Hould F-S, Marceau S, Simard S, Thung SN, et al. Liver Pathology and the Metabolic Syndrome X in Severe Obesity. J Clin Endocrinol Metab. 1999;84(5):1513-7.
- 110. Hamaguchi M, Kojima T, Takeda N, Nakagawa T, Taniguchi H, Fujii K, et al. The Metabolic Syndrome as a Predictor of Nonalcoholic Fatty Liver Disease. Ann Intern Med. 2005;143(10):722-8.
- 111. Chen J, Muntner P, Hamm L, Jones DW, Batuman V, Fonseca V, et al. The Metabolic Syndrome and Chronic Kidney Disease in U.S. Adults. Ann Intern Med. 2004;140(3):167-74.
- 112. Kurella M, Lo JC, Chertow GM. Metabolic syndrome and the risk for chronic kidney disease among nondiabetic adults. J Am Soc Nephrol. 2005;16(7):2134-40.
- 113. Pasquali R, Gambineri A, Anconetani B, Vicennati V, Colitta D, Caramelli E, et al. The natural history of the metabolic syndrome in young women with the polycystic ovary syndrome and the effect of long-term oestrogen–progestagen treatment. Clin Endocrinol (Oxf). 1999;50:517-27.
- 114. Siegel AB, Zhu AX. Metabolic syndrome and hepatocellular carcinoma: two growing epidemics with a potential link. Cancer. 2009;115(24):5651-61.
- 115. Akanji AO, Smith RJ. The insulin-like growth factor system, metabolic syndrome, and cardiovascular disease risk. Metab Syndr Relat Disord. 2012;10(1):3-13.
- Talvensaari KK, Lanning M, Tapanainen P, Knip M. Long-Term Survivors of Childhood Cancer Have an Increased Risk of Manifesting the Metabolic Syndrome. J Clin Endocrinol Metab. 1996;81(8):3051-5.
- 117. Saultier P, Auquier P, Bertrand Y, Vercasson C, Oudin C, Contet A, et al. Metabolic syndrome in long-term survivors of childhood acute leukemia treated without hematopoietic stem cell transplantation: an L.E.A. study. Haematologica. 2016;101(12):1603-10.
- 118. Oudin C, Auquier P, Bertrand Y, Contet A, Kanold J, Sirvent N, et al. Metabolic syndrome in adults who received hematopoietic stem cell transplantation for acute childhood leukemia: an LEA study. Bone Marrow Transplant. 2015;50(11):1438-44.
- Oudin C, Simeoni M-C, Sirvent N, Contet A, Begu-Le Coroller A, Bordignoni P, et al. Prevalence and risk factors of the metabolic syndrome in adult survivors of childhood leukemia. Blood. 2011;117(17):4442-8.
- Nottage KA, Ness KK, Li C, Srivastava D, Robison LL, Hudson MM. Metabolic syndrome and cardiovascular risk among long-term survivors of acute lymphoblastic leukaemia - From the St. Jude Lifetime Cohort. Br J Haematol. 2014;165(3):364-74.
- 121. Chow EJ, Simmons JH, Roth CL, Baker KS, Hoffmeister PA, Sanders JE, et al. Increased cardiometabolic traits in pediatric survivors of acute lymphoblastic leukemia treated with total body irradiation. Biol Blood Marrow Transplant. 2010;16(12):1674-81.
- 122. Bizzarri C, Pinto RM, Ciccone S, Brescia LP, Locatelli F, Cappa M. Early and progressive insulin resistance in young, non-obese cancer survivors treated with hematopoietic stem cell transplantation. Pediatr Blood Cancer. 2015;62(9):1650-5.
- 123. Blijdorp K, van Waas M, van der Lely AJ, Pieters R, van den Heuvel-Eibrink M, Neggers S. Endocrine sequelae and metabolic syndrome in adult long-term survivors of childhood acute myeloid leukemia. Leuk Res. 2013;37(4):367-71.
- 124. Smith WA, Li C, Nottage KA, Mulrooney DA, Armstrong GT, Lanctot JQ, et al. Lifestyle and metabolic syndrome in adult survivors of childhood cancer: a report from the St. Jude Lifetime Cohort Study. Cancer. 2014;120(17):2742-50.

- 125. Trimis G, Moschovi M, Papassotiriou I, Chrousos G, Tzortzatou-Stathopoulou F. Early Indicators of Dysmetabolic Syndrome in Young Survivors of Acute Lymphoblastic Leukemia in Childhood as a Target for Preventing Disease. J Pediatr Hematol Oncol. 2007;29:309-14.
- 126. Mohapatra S, Bansal D, Bhalla AK, Verma Attri S, Sachdeva N, Trehan A, et al. Is there an increased risk of metabolic syndrome among childhood acute lymphoblastic leukemia survivors? A developing country experience. Pediatr Hematol Oncol. 2016;33(2):136-49.
- 127. Gurney JG, Ness KK, Sibley SD, O'Leary M, Dengel DR, Lee JM, et al. Metabolic syndrome and growth hormone deficiency in adult survivors of childhood acute lymphoblastic leukemia. Cancer. 2006;107(6):1303-12.
- 128. Aldhafiri F, Al-Nasser A, Al-Sugair A, Al-Mutairi H, Young D, Reilly JJ. Obesity and metabolic syndrome in adolescent survivors of standard risk childhood acute lymphoblastic leukemia in Saudi Arabia. Pediatr Blood Cancer. 2012;59(1):133-7.
- 129. Karakurt H, Sarper N, Kilic SC, Gelen SA, Zengin E. Screening survivors of childhood acute lymphoblastic leukemia for obesity, metabolic syndrome, and insulin resistance. Pediatr Hematol Oncol. 2012;29(6):551-61.
- 130. Hoffman KE, Derdak J, Bernstein D, Reynolds JC, Avila NA, Gerber L, et al. Metabolic syndrome traits in long-term survivors of pediatric sarcoma. Pediatr Blood Cancer. 2008;50(2):341-6.
- 131. Taskinen M, Lipsanen-Nyman M, Tiitinen A, Hovi L, Saarinen-Pihkala UM. Insufficient Growth Hormone Secretion is Associated With Metabolic Syndrome After Allogeneic Stem Cell Transplantation in Childhood. J Pediatr Hematol Oncol. 2007;29:529-34.
- 132. van Waas M, Neggers SJ, Uitterlinden AG, Blijdorp K, van der Geest IM, Pieters R, et al. Treatment factors rather than genetic variation determine metabolic syndrome in childhood cancer survivors. Eur J Cancer. 2013;49(3):668-75.
- 133. Ness KK, Oakes JM, Punyko JA, Baker KS, Gurney JG. Prevalence of the metabolic syndrome in relation to self-reported cancer history. Ann Epidemiol. 2005;15(3):202-6.
- 134. Kojima C, Kubota M, Nagai A, Adachi S, Watanabe K, Nakahata T. Adipocytokines in childhood cancer survivors and correlation with metabolic syndrome components. Pediatr Int. 2013;55(4):438-42.
- 135. Huma Z, Boulad F, Black P, Heller G, Sklar C. Growth in children after bone marrow transplantation for acute leukemia. Blood. 1995;86(2):819-24.
- 136. Darzy KH, Shalet SM. Radiation-induced growth hormone deficiency. Horm Res. 2003;59 Suppl 1:1-11.
- 137. Chemaitilly W, Li Z, Huang S, Ness KK, Clark KL, Green DM, et al. Anterior hypopituitarism in adult survivors of childhood cancers treated with cranial radiotherapy: a report from the St Jude Lifetime Cohort study. J Clin Oncol. 2015;33(5):492-500.
- 138. Davis NL, Stewart CE, Moss AD, Woltersdorf WW, Hunt LP, Elson RA, et al. Growth hormone deficiency after childhood bone marrow transplantation with total body irradiation: interaction with adiposity and age. Clin Endocrinol (Oxf). 2015;83(4):508-17.
- 139. Follin C, Gabery S, Petersen A, Sundgren PC, Bjorkman-Burtcher I, Latt J, et al. Associations between Metabolic Risk Factors and the Hypothalamic Volume in Childhood Leukemia Survivors Treated with Cranial Radiotherapy. PLoS One. 2016;11(1):e0147575.
- Thomsett MJ, Conte FA, Kaplan SL, Grumbach MM. Endocrine and neurologic outcome in childhood craniopharyngioma: Review of effect of treatment in 42 patients. J Pediatr. 1980;97(5):728-35.
- 141. Stahnke N, Grubel G, Lagenstein I, Willig RP. Long-term follow-up of children with craniopharyngioma. Eur J Pediatr. 1984;142(3):179-85.

- 142. Felicetti F, Manicone R, Corrias A, Manieri C, Biasin E, Bini I, et al. Endocrine late effects after total body irradiation in patients who received hematopoietic cell transplantation during childhood: a retrospective study from a single institution. J Cancer Res Clin Oncol. 2011;137(9):1343-8.
- 143. Gleeson HK, Gattamaneni HR, Smethurst L, Brennan BM, Shalet SM. Reassessment of growth hormone status is required at final height in children treated with growth hormone replacement after radiation therapy. J Clin Endocrinol Metab. 2004;89(2):662-6.
- Haddy TB, Mosher RB, Nunez SB, Reaman GH. Growth hormone deficiency after chemotherapy for acute lymphoblastic leukemia in children who have not received cranial radiation. Pediatr Blood Cancer. 2006;46(2):258-61.
- 145. Rose SR, Horne VE, Howell J, Lawson SA, Rutter MM, Trotman GE, et al. Late endocrine effects of childhood cancer. Nat Rev Endocrinol. 2016;12(6):319-36.
- 146. Chemaitilly W, Cohen LE. Diagnosis of Endocrine Disease: Endocrine late-effects of childhood cancer and its treatments. Eur J Endocrinol. 2017;176(4):R183-R203.
- 147. de Boer H, Blok GJ, Voerman HJ, de Vries PM, van der Veen EA. Body composition in adult growth hormone-deficient men, assessed by anthropometry and bioimpedance analysis. J Clin Endocrinol Metab. 1992;75(3):833-7.
- 148. Binnerts A, Deurenberg P, Swart GR, Wilson JH, Lamberts SW. Body composition in growth hormone-deficient adults. Am J Clin Nutr. 1992;55(5):981-23.
- 149. Rosén T, Edén S, Larson G, Wilhelmsen L, Bengtsson B-Å. Cardiovascular risk factors in adult patients with growht hormone disease. Acta Endocrinologica. 1993;129:195-200.
- 150. Franco C, Bengtsson B-Å, Johansson G. The GH/IGF-1 Axis in Obesity: Physiological and Pathological Aspects. Metab Syndr Relat Disord. 2006;4(1):51-6.
- 151. Hew FL, Koschmann M, Christopher M, Rantzau C, Vaag A, Ward G, et al. Insulin Resistance in Growth Hormone-Deficient Adults: Defects in Glucose Utilization and Glycogen Synthase Activity. J Clin Endocrinol Metab. 1996;81(2):555-64.
- 152. Johansson J-O, Fowelin J, Landin K, Lager I, Bengtsson B-Å. Growth Hormone-Deficient Adults Are Insulin-Resistant. Metabolism. 1995;44(9):1126-9.
- 153. de Boer H, Blok GJ, Voerman HJ, Phillips M, Schouten JA. Serum Lipid Levels in Growth Hormone-Deficient Men. Metabolism. 1994;43(2):199-203.
- 154. Bengtsson B-Å, Abs R, Bennmarker H, Monson JP, Feldt-Rasmussen U, Hernberg-Stahl E, et al. The effects of treatment and the individual responsiveness to growth hormone (GH) replacement therapy in 665 GH-deficient adults. KIMS Study Group and the KIMS International Board. J Clin Endocrinol Metab. 1999;84(11):3929.
- 155. Friedman DN, Hilden P, Moskowitz CS, Suzuki M, Boulad F, Kernan NA, et al. Cardiovascular Risk Factors in Survivors of Childhood Hematopoietic Cell Transplantation Treated with Total Body Irradiation: A Longitudinal Analysis. Biol Blood Marrow Transplant. 2017;23(3):475-82.
- 156. Elhadd TA, Abdu TA, Oxtoby J, Kennedy G, McLaren M, Neary R, et al. Biochemical and Biophysical Markers of Endothelial Dysfunction in Adults with Hypopituitarism and Severe GH Deficiency. J Clin Endocrinol Metab. 2001;86(9):4223-32.
- 157. Cannavo S, Marini F, Curto L, Torre ML, de Gregorio C, Salamone I, et al. High prevalence of coronary calcifications and increased risk for coronary heart disease in adults with growth hormone deficiency. J Endocrinol Invest. 2011;34(1):32-7.
- 158. Colao A, Di Somma C, Cuocolo A, Filippella M, Rota F, Acampa W, et al. The severity of growth hormone deficiency correlates with the severity of cardiac impairment in 100 adult patients with hypopituitarism: an observational, case-control study. J Clin Endocrinol Metab. 2004;89(12):5998-6004.

- 159. Schneider HJ, Klotsche J, Wittchen HU, Stalla GK, Schopohl J, Kann PH, et al. Effects of growth hormone replacement within the KIMS survey on estimated cardiovascular risk and predictors of risk reduction in patients with growth hormone deficiency. Clin Endocrinol (Oxf). 2011;75(6):825-30.
- 160. Crowne E, Gleeson H, Benghiat H, Sanghera P, Toogood A. Effect of cancer treatment on hypothalamic–pituitary function. The Lancet Diabetes & Endocrinology. 2015;3(7):568-76.
- Follin C, Erfurth EM. Long-Term Effect of Cranial Radiotherapy on Pituitary-Hypothalamus Area in Childhood Acute Lymphoblastic Leukemia Survivors. Curr Treat Options Oncol. 2016;17(9):50.
- 162. Darzy KH, Shalet SM. Hypopituitarism following Radiotherapy Revisited. Endocr Dev. 2009;15:1-24.
- Merchant TE, Rose SR, Bosley C, Wu S, Xiong X, Lustig RH. Growth hormone secretion after conformal radiation therapy in pediatric patients with localized brain tumors. J Clin Oncol. 2011;29(36):4776-80.
- 164. Mulder RL, Kremer LC, van Santen HM, Ket JL, van Trotsenburg AS, Koning CC, et al. Prevalence and risk factors of radiation-induced growth hormone deficiency in childhood cancer survivors: a systematic review. Cancer Treat Rev. 2009;35(7):616-32.
- 165. Baum H, Biller B, Finkelstein J, Baker Cannistraro K, Oppenheim D, Schoenfeld D, et al. Effects of Physiologic Growth Hormone Therapy on Bone Density and Body Composition in Patients with Adult-Onset Growth Hormone Deficiency. Ann Intern Med. 1996;125:883-90.
- 166. Bengtsson B-Å, Edén S, Lönn L, Kvist H, Stokland A, Lindstedt G, et al. Treatment of Adults with Growth Hormone (GH) Deficiency with Recombinant Human GH. J Clin Endocrinol Metab. 1993;76:309-17.
- 167. Maison P, Chanson P. Cardiac effects of growth hormone in adults with growth hormone deficiency: a meta-analysis. Circulation. 2003;108(21):2648-52.
- 168. Maison P, Griffin S, Nicoue-Beglah M, Haddad N, Balkau B, Chanson P, et al. Impact of growth hormone (GH) treatment on cardiovascular risk factors in GH-deficient adults: a Metaanalysis of Blinded, Randomized, Placebo-Controlled Trials. J Clin Endocrinol Metab. 2004;89(5):2192-9.
- 169. van Bunderen CC, van Varsseveld NC, Erfurth EM, Ket JC, Drent ML. Efficacy and safety of growth hormone treatment in adults with growth hormone deficiency: a systematic review of studies on morbidity. Clin Endocrinol (Oxf). 2014;81(1):1-14.
- 170. Attanasio AF, Mo D, Erfurth EM, Tan M, Ho KY, Kleinberg D, et al. Prevalence of metabolic syndrome in adult hypopituitary growth hormone (GH)-deficient patients before and after GH replacement. J Clin Endocrinol Metab. 2010;95(1):74-81.
- 171. Claessen KM, Appelman-Dijkstra NM, Adoptie DM, Roelfsema F, Smit JW, Biermasz NR, et al. Metabolic profile in growth hormone-deficient (GHD) adults after long-term recombinant human growth hormone (rhGH) therapy. J Clin Endocrinol Metab. 2013;98(1):352-61.
- 172. Follin C, Thilen U, Ahren B, Erfurth EM. Improvement in cardiac systolic function and reduced prevalence of metabolic syndrome after two years of growth hormone (GH) treatment in GHdeficient adult survivors of childhood-onset acute lymphoblastic leukemia. J Clin Endocrinol Metab. 2006;91(5):1872-5.
- 173. Bulow B, Link K, Ahren B, Nilsson AS, Erfurth EM. Survivors of childhood acute lymphoblastic leukaemia, with radiation-induced GH deficiency, exhibit hyperleptinaemia and impaired insulin sensitivity, unaffected by 12 months of GH treatment. Clin Endocrinol (Oxf). 2004;61(6):683-91.
- 174. van den Heijkant S, Hoorweg-Nijman G, Huisman J, Drent M, van der Pal H, Kaspers GJ, et al. Effects of growth hormone therapy on bone mass, metabolic balance, and well-being in

young adult survivors of childhood acute lymphoblastic leukemia. J Pediatr Hematol Oncol. 2011;33(6):231-8.

- 175. Murray RD, Darzy KH, Gleeson HK, Shalet SM. GH-deficient survivors of childhood cancer: GH replacement during adult life. J Clin Endocrinol Metab. 2002;87(1):129-35.
- 176. Nord C, Bjøro T, Ellingsen D, Mykletun A, Dahl O, Klepp O, et al. Gonadal Hormones in Long-Term Survivors 10 Years after Treatment for Unilateral Testicular Cancer. European Urology. 2003;44(3):322-8.
- 177. Wallace WH, Thomson AB, Saran F, Kelsey TW. Predicting age of ovarian failure after radiation to a field that includes the ovaries. Int J Radiat Oncol Biol Phys. 2005;62(3):738-44.
- 178. Huddart RA, Norman A, Moynihan C, Horwich A, Parker C, Nicholls E, et al. Fertility, gonadal and sexual function in survivors of testicular cancer. Br J Cancer. 2005;93(2):200-7.
- 179. Sprauten M, Brydoy M, Haugnes HS, Cvancarova M, Bjoro T, Bjerner J, et al. Longitudinal serum testosterone, luteinizing hormone, and follicle-stimulating hormone levels in a population-based sample of long-term testicular cancer survivors. J Clin Oncol. 2014;32(6):571-8.
- 180. Zitzmann M. Testosterone deficiency, insulin resistance and the metabolic syndrome. Nat Rev Endocrinol. 2009;5(12):673-81.
- 181. Brand JS, van der Tweel I, Grobbee DE, Emmelot-Vonk MH, van der Schouw YT. Testosterone, sex hormone-binding globulin and the metabolic syndrome: a systematic review and metaanalysis of observational studies. Int J Epidemiol. 2011;40(1):189-207.
- 182. Akinloye O, Blessing Popoola B, Bolanle Ajadi M, Gregory Uchechukwu J, Pius Oparinde D. Hypogonadism and metabolic syndrome in nigerian male patients with both type 2 diabetes and hypertension. Int J Endocrinol Metab. 2014;12(1):e10749.
- 183. Naifar M, Rekik N, Messedi M, Chaabouni K, Lahiani A, Turki M, et al. Male hypogonadism and metabolic syndrome. Andrologia. 2015;47(5):579-86.
- 184. Ventimiglia E, Capogrosso P, Colicchia M, Boeri L, Serino A, Castagna G, et al. Metabolic syndrome in white European men presenting for primary couple's infertility: investigation of the clinical and reproductive burden. Andrology. 2016;4(5):944-51.
- 185. van Dorp W, Blijdorp K, Laven JS, Pieters R, Visser JA, van der Lely AJ, et al. Decreased ovarian function is associated with obesity in very long-term female survivors of childhood cancer. Eur J Endocrinol. 2013;168(6):905-12.
- 186. Blijdorp K, van Dorp W, Laven JS, Pieters R, de Jong FH, Pluijm SM, et al. Obesity independently influences gonadal function in very long-term adult male survivors of childhood cancer. Obesity (Silver Spring). 2014;22(8):1896-903.
- 187. Jahnukainen K, Heikkinen R, Henriksson M, Andersson S, Ivaska KK, Puukko-Viertomies LR, et al. Increased Body Adiposity and Serum Leptin Concentrations in Very Long-Term Adult Male Survivors of Childhood Acute Lymphoblastic Leukemia. Horm Res Paediatr. 2015;84(2):108-15.
- Corona G, Giagulli VA, Maseroli E, Vignozzi L, Aversa A, Zitzmann M, et al. Testosterone supplementation and body composition: results from a meta-analysis of observational studies. J Endocrinol Invest. 2016;39(9):967-81.
- Jereczek-Fossa BA, Alterio D, Jassem J, Gibelli B, Tradati N, Orecchia R. Radiotherapy-induced thyroid disorders. Cancer Treat Rev. 2004;30(4):369-84.
- 190. Hancock SL, McDougall IR, Constine LS. Thyroid abnormalities after therapeutic external radiation. Int J Radiat Oncol Biol Phys. 1995;31(5):1165-70.
- 191. Taddei S, Caraccio N, Virdis A, Dardano A, Versari D, Ghiadoni L, et al. Impaired endotheliumdependent vasodilatation in subclinical hypothyroidism: beneficial effect of levothyroxine therapy. J Clin Endocrinol Metab. 2003;88(8):3731-7.

- 192. Duntas LH, Brenta G. The effect of thyroid disorders on lipid levels and metabolism. Med Clin North Am. 2012;96(2):269-81.
- 193. Knudsen N, Laurberg P, Rasmussen LB, Bulow I, Perrild H, Ovesen L, et al. Small differences in thyroid function may be important for body mass index and the occurrence of obesity in the population. J Clin Endocrinol Metab. 2005;90(7):4019-24.
- 194. Waring AC, Rodondi N, Harrison S, Kanaya AM, Simonsick EM, Miljkovic I, et al. Thyroid function and prevalent and incident metabolic syndrome in older adults: the Health, Ageing and Body Composition Study. Clin Endocrinol (Oxf). 2012;76(6):911-8.
- 195. Gurwitz JH, Bohn RL, Glynn RJ, Monane M, Mogun H, Avorn J. Glucocorticoids and the risk for initiation of hypoglycemic therapy. Arch Intern Med. 1994;154(1):97-101.
- 196. Da Silva JA, Jacobs JW, Kirwan JR, Boers M, Saag KG, Ines LB, et al. Safety of low dose glucocorticoid treatment in rheumatoid arthritis: published evidence and prospective trial data. Ann Rheum Dis. 2006;65(3):285-93.
- 197. van Beek RD, van den Heuvel-Eibrink MM, Hakvoort-Cammel FG, van den Bos C, van der Pal HJ, Krenning EP, et al. Bone mineral density, growth, and thyroid function in long-term survivors of pediatric Hodgkin's lymphoma treated with chemotherapy only. J Clin Endocrinol Metab. 2009;94(6):1904-9.
- 198. Warner JT, Bell W, Webb DK, Gregory JW. Daily energy expenditure and physical activity in survivors of childhood malignancy. Pediatr Res. 1998;43(5):607-13.
- 199. van Waas M, Wijnen M, Hartman A, de Vries AC, Pieters R, Neggers SJ, et al. Daily life physical activity in long-term survivors of nephroblastoma and neuroblastoma. J Pediatr Hematol Oncol. 2013;35(5):361-5.
- 200. Wijnen M, Olsson DS, van den Heuvel-Eibrink MM, Hammarstrand C, Janssen J, van der Lely AJ, et al. The metabolic syndrome and its components in 178 patients treated for craniopharyngioma after 16 years of follow-up. Eur J Endocrinol. 2018;178(1):11-22.
- 201. England J, Drouin S, Beaulieu P, St-Onge P, Krajinovic M, Laverdiere C, et al. Genomic determinants of long-term cardiometabolic complications in childhood acute lymphoblastic leukemia survivors. BMC Cancer. 2017;17(1):751.
- 202. Clemens E, van der Kooi ALF, Broer L, van Dulmen-den Broeder E, Visscher H, Kremer L, et al. The influence of genetic variation on late toxicities in childhood cancer survivors: A review. Critical Reviews in Oncology/Hematology. 2018;126:154-67.
- 203. Kremer LCM, van der Pal HJH, Offringa M, van Dalen EC, Voute PA. Frequency and risk factors of subclinical cardiotoxicity after anthracycline therapy in children: a systematic review. Annals of Oncology. 2002;13(6):819-29.
- 204. van Dalen EC, van den Brug M, Caron HN, Kremer LC. Anthracycline-induced cardiotoxicity: comparison of recommendations for monitoring cardiac function during therapy in paediatric oncology trials. Eur J Cancer. 2006;42(18):3199-205.
- 205. van der Pal HJ, van Dalen EC, van Delden E, van Dijk IW, Kok WE, Geskus RB, et al. High risk of symptomatic cardiac events in childhood cancer survivors. J Clin Oncol. 2012;30(13):1429-37.



Metabolic syndrome detection with biomarkers in childhood cancer survivors

V.G. Pluimakers, M. van Waas, C.W.N. Looman, M.P. de Maat, R. de Jonge, P. Delhanty, M. Huisman, F.U.S. Mattace-Raso, M.M. van den Heuvel-Eibrink, S.J.C.M.M. Neggers

Endocrine Connections, 2020 Jun;9:1-11

ABSTRACT

Purpose:

Augmented childhood nephroblastoma and neuroblastoma survival has increased long-term side effects as metabolic syndrome (MetS). Risk stratification is difficult after abdominal radiation because waist circumference underestimates adiposity. We aimed to develop a strategy for determining MetS in irradiated survivors using an integrated biomarker profile and vascular ultrasonography.

Methods:

The NCEP-ATPIII MetS-components, 14 additional serum biomarkers and 9 vascular measurements were assessed in a single-centre cohort of childhood nephroblastoma (n=67) and neuroblastoma (n=36) survivors and controls (n=61). Multivariable regression models were used to study treatment effects. Principal component analysis (PCA) was used to study all biomarkers in a combined analysis, to identify patterns and correlations.

Results:

After 27.5 years of follow-up, MetS occurred more often in survivors (14%) than controls (3%). Abdominal radiotherapy, and nephrectomy to a lesser extent, were associated with MetS and separate components, and with several biomarker abnormalities. PCA of biomarkers revealed a pattern on PC1 from favourable lipid markers (HDL-cholesterol, adiponectin) towards unfavourable markers (triglycerides, LDL-cholesterol, apo-B, uric acid). Abdominal radiotherapy was associated with the unfavourable biomarker profile (β =1.45, p=0.001). Vascular measurements were not of added diagnostic value.

Conclusions:

Long-term childhood nephro- and neuroblastoma survivors frequently develop MetS. Additional assessment of biomarkers identified in PCA – adiponectin, LDL, apo-B, uric acid – may be used especially in abdominally irradiated survivors, to classify MetS as alternative for waist circumference. Vascular ultrasonography was not of added value.

INTRODUCTION

Over the past decades, survival rates of childhood nephroblastoma and neuroblastoma have increased to respectively ~90% and ~40-95% (strongly dependent on stadium) [1, 2]. These tumours are of embryonic origin, with a peak incidence under the age of five years, and a presentation predominantly in the abdomen. Treatment often consists of a combination of surgery (nephrectomy and/or adrenalectomy), radiotherapy and/ or intensive chemotherapy. Because of increased survival rates, long-term side effects, such as adiposity, insulin resistance, dyslipidaemia and hypertension, have become more prominent, particularly after treatment with abdominal radiotherapy [3-5]. These risk factors for diabetes mellitus and cardiovascular disease interact, and cluster together as metabolic syndrome [6-8].

In order to prevent the development of diabetes mellitus and cardiovascular disease, it is important to identify survivors at risk of developing (components of) MetS and to diagnose and treat them in a timely fashion [9]. Risk stratification in childhood cancer survivors (CCS) with the classic criteria for MetS components can be difficult. This is due to the underestimation of adiposity by waist circumference, waist-hip ratio and body mass index (BMI), in particular after abdominal radiotherapy has been applied [3, 10]. Also, because CCS are relatively young, absolute occurrence rates of cardio- and cerebrovascular events are low, even though they are at higher relative risk [11-14].

There is evidence that measurement of (visceral) fat by DXA-scan is a better indicator of adiposity, and hence a better predictor for cardiovascular disease [3, 10], but this is a costly and time consuming test. Therefore, in addition to the serum biomarkers triglycerides and HDL cholesterol that are already included in the definition of MetS, several other biomarkers have been suggested as surrogate markers for development of MetS and cardiovascular disease. These additional biomarkers include lowdensity lipoprotein (LDL) cholesterol, adiponectin, uric acid, C-reactive protein and cystatin C [15-17]. Also, it has been proposed that metabolic biomarkers are more clinically useful for risk prediction of diabetes and cardiovascular disease when analysed as a combination reflecting different pathophysiologic pathways, to reveal underlying patterns or clusters of dysmetabolic development [18]. In addition, vascular ultrasound measurements, such as carotid intima media thickness (CIMT), pulse pressure amplification (PPA) and pulse wave velocity (PWV), have also been proposed as surrogate markers for cardiovascular disease [19, 20]. So far, no studies have reported on the value of these additional biomarkers and vascular ultrasound measurements in CCS.

The aim of this cross-sectional study was to develop a strategy for determining MetS even in abdominally irradiated long-term survivors of childhood nephro- and neuroblastoma, using an integrated biomarker profile, based on principal component analysis, and vascular ultrasound measurements.

PATIENTS AND METHODS

Patients

Patients were actively recruited as described before [3]. Briefly, all long-term (five or more years after treatment) adult survivors of childhood nephro- and neuroblastoma (except for survivors of neuroblastoma stage 4s who did not receive surgery, radiotherapy or chemotherapy), treated between 1961 and 2004 in the Erasmus MC/ Sophia Children's Hospital, Rotterdam, The Netherlands, that visited the late effects outpatient clinic regularly were invited to participate in this cross-sectional study. The study was approved by the MREC Erasmus MC Rotterdam (trial NL2685, study period 2009-2012). Survivors were asked to invite potential control subjects such as siblings, friends or neighbours, preferably of the same sex and within an age range of five years. Written informed consent was obtained from all participants.

Data collection

Disease and treatment data were obtained from the medical records. Detailed data regarding surgery were confirmed from the original surgical and pathological reports. Information on medication use (statins, antidiabetic, antihypertensive), smoking and socio-economic status was collected using a self-designed questionnaire. Weight was measured with underwear only to the nearest 0.1kg with a standard clinical balance. Height was measured to the nearest millimetre using a Harpenden Stadiometer. BMI was calculated (weight(kg)/height(cm)²). Waist circumference was measured between lower rib and iliac crest to the nearest centimetre. Blood pressure was measured with the subject in sitting position after an hour of rest on the right arm with the Dinamap[®] Procare and was defined as the mean of three measurements. Components of MetS were defined using the NCEP-ATPIII classification: waist ≥102(men)/88(women)cm, triglycerides \geq 1.7 mmol/L or use of statins, HDL cholesterol \leq 1.03(men)/1.29(women) mmol/L or use of statins, blood pressure $\geq 130/\geq 85$ mmHg or use of antihypertensives, fasting glucose ≥5.6mmol/L or antidiabetic treatment, with three or more criteria required for the diagnosis MetS [6]. The occurrence of MetS and components in the current study population have been previously published [3].

Laboratory measurements

Fasting venous blood samples were taken before 10am. In addition to the biomarkers in the NCEP-ATPIII classification, thirteen biomarkers were assessed: free fatty acids (FFA), apolipoprotein(apo)-A1, apo-B, LDL cholesterol (measured, not calculated), leptin, adiponectin, lipoprotein(a) (Lp(a)), insulin, cystatin C, uric acid, urea, creatinine and hsCRP. Homeostasis model assessment (HOMA) was used as an estimate of insulin resistance and beta-cell function calculated from glucose and insulin concentrations [21]. Also, antithrombin, protein C, protein S, diluted Russell's viper venom time and von Willebrandfactor antigen were measured to exclude subjects with possible non-cancer-therapy-related coagulation problems (results not reported as outcome variables).

Vascular ultrasound measurements

Central systolic and diastolic blood pressure were assessed with the SphygmoCor (AtCor Medical, Sydney, Australia), which calculates aortic blood pressure from brachial pulse wave. Brachial and central pulse pressure (PP, the difference between systolic and diastolic blood pressure) were calculated, as well as pulse pressure amplification (PPA, brachial divided by central PP). Measurements of the carotid artery were performed with the subject in supine position, the head tilted slightly towards the contralateral side. After five minutes of rest, diameter of the common carotid artery (CCA), carotid intima media thickness (CIMT) and distensibility were measured with a duplex scanner (operating frequency 7.5MHz, Pie Medical Imaging, Maastricht, The Netherlands) during six non-consecutive heartbeats and reported as mean values. Distensibility coefficient (DC) was calculated using the following formula: ((2000*distensibility)/diameter)/PP*133.22) [22]. Pulse wave velocity (PWV) was also measured with the subject in supine position, with the Complior (Alam Medical, Saint-Quentin-Fallavier, France), which simultaneously records pulse waves at the carotid and femoral arteries (PWV = carotid-femoral distance/time delay).

Statistical analysis

Statistical analysis was performed using R version 3.5.1 (R Foundation for Statistical Computing, Vienna, Austria). Baseline characteristics were compared between survivors and controls as well as nephroblastoma survivors compared to neuroblastoma survivors, using Fisher's exact test and Chi-squared test for categorical variables and by bootstrapping the difference in median values for continuous variables.

Occurrence of MetS and MetS components was compared between survivors and controls using Fisher's exact test and the Chi-squared test, respectively. Serum biomarkers and vascular ultrasound measurements were compared between survivors and controls by bootstrapping the difference in median values, for which the 95% confidence interval was calculated with percentiles.

The effect of abdominal radiotherapy and nephrectomy on (components of) MetS, the serum biomarkers and the vascular parameters was tested with univariable logistic and bootstrap linear regression models, and, when significant, also in multivariable regression models, adjusting for age, sex, smoking and socio-economic status. There was no need to adjust for use of steroids, as these are not administered in treatment protocols of these malignancies, nor for adrenalectomy, because this would lead to overcompensation, as we previously published [23].

The serum biomarkers were also analysed by principal component analysis (PCA), to identify correlations and discriminative patterns, and to reduce the effects of multiple testing. PCA is an unsupervised, combined analysis of all biomarkers, that explains most of the variance in two principal components (PC1 and PC2) and the relative contribution of each biomarker to these principal components. With this method, the individual, unbiased, contribution of each biomarker is calculated. PCA was performed on the correlation matrix, which means that all variables are standardized to Z-scores. Missing values were imputed with the median (except for cystatin C, which was missing in 15% of participants and therefore was predicted with R package *mice* based on age, sex and the other kidney function variables). The effect of abdominal radiotherapy and nephrectomy on PC1 and PC2 was tested with linear regression models.

For the analyses of serum biomarkers, vascular ultrasound measurements and PCA, participants using relevant medication were excluded. A p-value <0.05 was considered as statistically significant.

RESULTS

Study population

Eighty-eight nephroblastoma survivors were invited to participate in the study, of whom 67 (39 males) agreed (76%). Fifty-five neuroblastoma survivors were invited, of whom 36 (15 males) agreed to participate (65%). Survivors who did not participate were similar to participating survivors with respect to baseline characteristics. In total, sixty-one controls were included (33 males), 37 of whom were siblings and 24 were partner or friend. Baseline and treatment characteristics are depicted in Table 1. Median age was 30 and 31.8 years for survivors and controls, respectively,

median follow-up time of survivors was 27.5 years (range 6.4 - 48.9 years). Systolic and diastolic blood pressure were higher among survivors, whereas physical activity, smoking behaviour and socio-economic status were not significantly different between survivors and controls. Within survivors, nephroblastoma survivors were older at diagnosis and had been treated more often with nephrectomy and abdominal radiotherapy (Supplemental Table 1). None of the study participants had experienced a cardiac event or stroke at time of inclusion in the study.

Classic MetS components, biomarkers and vascular ultrasound measurements, as compared to controls

MetS, as defined by the presence of at least three of the NCEP-ATPIII criteria, was present in 14 survivors (14%) and 2 controls (3%, p=0.032), as previously described [3]. Thirty-four survivors (33%) revealed at least two MetS criteria, compared to 12 controls (20%, p=0.074). Hypertension and treatment for hypertension occurred significantly more often in survivors than controls, whereas the other MetS components did not differ significantly between groups (Table 2).

Triglycerides (Δ =0.17mmol/L, p=0.036), cystatin C (Δ =0.06mg/L, p=0.002) and creatinine levels (Δ =5mg/mmol, p=0.014) were significantly higher in survivors compared to controls (Table 3). The other additional biomarkers were not different between survivors and controls. All coagulation markers were within the reference range in all participants (data not shown).

All vascular measurements were similar between survivors and controls.

Influence of abdominal radiotherapy and nephrectomy on classic MetS components, biomarkers and vascular ultrasound measurements

Using univariable logistic regression, abdominal radiotherapy was associated with occurrence of MetS (odds ratio (OR)=6.04, 95% CI=2.04-17.89, p=0.001), presence of two or more MetS components (OR=3.36, 95% CI=1.59-7.07, p=0.001), as well as with all separate components of MetS (Table 4). Using multivariable regression analysis, adjusting for age, sex, smoking and socio-economic status, abdominal radiotherapy remained an independent risk factor for MetS occurrence (OR=15.3, 95% CI=3.21-73.36, p<0.001), occurrence of two or more MetS components (OR=3.23, 95% CI=1.35-7.73, p=0.009) as well as the MetS components high triglycerides, low HDL cholesterol and hypertension. Nephrectomy was not a risk factor for MetS occurrence (OR=2.97, 95% CI=0.98-8.97, p=0.054). Using multivariable regression, nephrectomy was a risk factor for having two or more MetS components (OR=2.78, 95% CI=1.26-6.17, p=0.012), high triglycerides or treatment (OR=4.68,

95% CI=1.66-13.19, p=0.004) as well as hypertension or treatment (OR=4.82, 95% CI=2.05-11.29, p<0.001).

With regard to the biomarkers, abdominally irradiated subjects had higher triglycerides, FFA, apo-B, LDL, cystatin C and urea levels (Supplemental Table 2a). Using multivariable linear regression analysis, abdominal radiotherapy remained an independent risk factor for higher triglycerides (β =0.57, p=0.002), higher FFA (β =0.15, p=0.008) and higher cystatin C (β =0.08, p=0.039) (Supplemental Table 2b). Nephrectomy was associated with higher cystatin C, uric acid, urea and creatinine levels (Supplemental Table 2c). Cystatin C (β =0.12, p<0.001), uric acid (β =0.05, p=0.006) and creatinine (β =6.95, p=0.042) remained significantly associated with nephrectomy in multivariable analysis (Supplemental Table 2d).

Ultrasonography revealed that abdominally irradiated survivors had significantly higher central systolic and diastolic blood pressure and PWV and lower DC (Supplemental Table 2a). In multivariable analysis, the association between abdominal radiation and higher central diastolic blood pressure remained significant (β =5.39, p=0.023) (Supplemental Table 2b). After nephrectomy, survivors had higher central systolic blood pressure, but this association was not significant in linear regression analysis (Supplemental Tables 2c,d). As peripheral blood pressure was higher as well, there was no clear added value of the vascular ultrasound measurements.

Principal component analysis of biomarkers

Principal component analysis of the panel of 17 serum biomarkers in survivors yielded principal component 1 with explained variance of 24.2% and a pattern of "favourable lipids" (high negative loading for HDL and adiponectin) towards "unfavourable lipids" (high positive loading of triglycerides, apo-B and LDL, as well as uric acid). Principal component 2 (PC2, orthogonal on PC1) explained 14.3% variance, with high negative loading reflected by "impaired glucose metabolism" (HOMA and glucose (as well as leptin)), and high positive loading reflected by "kidney disease" (creatinine and cystatin C (as well as HDL)). As all vectors in the biplot originate from the centre, high positive and negative loading on PC2 are two separate entities, i.e. kidney disease is not associated with favourable glucose metabolism and neither is impaired glucose metabolism with good kidney function. This is only applicable to PC1 since one pattern – favourable to unfavourable lipids – can be distinguished along the whole axis.

The effect of abdominal radiotherapy and nephrectomy on the two principal components is depicted in Figure 1. Survivors who received abdominal radiotherapy had a higher positive loading on PC1, constituting the unfavourable profile (β =1.45,

p=0.001). There was no significant influence of abdominal radiation on PC2 (β =0.14, p=0.68). Nephrectomy was associated with both higher positive loading on PC1 (unfavourable lipids, β =1.13, p=0.015) and PC2 (kidney disease, β =0.75, p=0.037).

DISCUSSION

This is the first report that describes the value of an integrated biomarker profile, defined by principal component analysis, and vascular ultrasound measurements, to estimate metabolic syndrome in long-term survivors of childhood nephroblastoma and neuroblastoma, in addition to classic parameters. We show that survivors more frequently develop MetS, and they have an unfavourable constitution of biomarkers in principal component 1 (PC1), particularly after abdominal radiotherapy. By using a principal component analysis, we could explore new variables better, since this analysis identifies individual contribution with no bias of multiple testing. This would not have been possible with another analysis, such as multiple correlation or regression. This unfavourable constitution of biomarkers consisted of a cluster of low HDL cholesterol and adiponectin and high triglycerides, LDL cholesterol, apo-B and uric acid.

Low HDL and high triglycerides are already classic components of MetS in the NCEP-ATPIII classification; the other biomarkers have been reported as risk predictors for MetS and cardiovascular disease in the general population as well as in CCS [16, 17, 24-26]. Therefore, we propose the addition of adiponectin, LDL, apo-B and uric acid in a surveillance setting, particularly in abdominally irradiated survivors, to classify MetS as alternative for waist circumference.

Our suggestion to add biomarkers to the classical components of MetS is in line with the recently updated dyslipidaemia management guideline from the European Society of Cardiology (ESC) and the European Atherosclerosis Society (EAS), in which apo-B analysis is recommended for cardiovascular risk assessment, particularly in people with high triglycerides, obesity, MetS and diabetes [27]. Apo-B has been reported as a more sensitive marker of atherogenicity of cholesterol particles, in particular in insulin resistant patients. In those subjects, a relative abundance of dense, more atherogenic LDL particles can be present, which would be reflected by higher apo-B levels [28]. Uric acid is linked to metabolic syndrome in several ways: hyperuricemia contributes to the development of hypertension, insulin resistance and obesity [24]. The observed inverse relation between adiponectin and abdominal radiotherapy is of interest, as it may suggest that local radiation damage leads to decreased endocrine function of the adipose tissue or a lower number of fat cells. Our finding that abdominal radiotherapy is strongly associated with the development of MetS components in CCS is consistent with other studies, and, more specifically, caused by radiation damage to the pancreas [29-32]. Additionally, abdominal radiotherapy can lead to underdevelopment of belly fat and musculature, and to scoliosis. Hence, measurement of waist circumference underestimates adiposity. Previously, we reported that body composition is more accurately measured in these CCS by DXA-scan [3]. The proposed use of additional biomarkers has the added advantage that this may be a cheaper and less burdensome diagnostic tool.

We found a moderate correlation between PC1 score and waist circumference in non-abdominally irradiated survivors (Pearson's r = 0.64, substantially higher than the correlations of the separate biomarkers) (Supplemental Figure 1), which supports the feasibility of this screening strategy. As a next step, to prove accuracy, sensitivity and cost-effectiveness of this strategy, replication in larger and independent cohorts is needed. Ultimately, for determining how the PC1 biomarkers could be incorporated in MetS classification, longitudinally collected information on solid endpoints (diabetes mellitus, cardio- and cerebrovascular morbidity and mortality) is needed.

Another finding was that principal component 2 was not of added value in determining MetS in abdominally irradiated survivors. By definition, PC2 explains less of the variance. Furthermore, there was no single discriminative pattern reflecting PC2 score and no difference in PC2 constitution was observed between abdominally irradiated and non-irradiated survivors.

Unexpectedly, the vascular ultrasound measurements were neither of evident added value in estimating MetS. We did observe some alterations suggestive of central arterial stiffness after abdominal radiotherapy: elevated central blood pressure and pulse wave velocity (PWV), and lower distensibility coefficient (DC), but after adjustment for potential covariates, only central blood pressure remained significantly associated with abdominal radiotherapy. Although central blood pressure is thought to better reflect cardiovascular risk as this represents the blood pressure in the coronary and cerebral arteries [33], we do not estimate this measurement of substantially added value, with peripherally measured blood pressure already being a classic MetS component. Although vascular abnormalities as observed by ultrasound can be early signs of MetS and its consequences, there can be variation in the development of these consequences, and the type of ultrasound patterns can vary as well. Therefore, in the aforementioned ESC/EAS dyslipidaemia guideline, it is postulated that assessment of arterial plaque burden can be considered as a risk modifier in individuals at low or moderate cardiovascular risk, in addition to standard cardiovascular risk assessment [27]. We think that it is conceivable that this variation, as well as the relatively young age of
our study cohort, may contribute to this unexpected finding. It could be, that these vascular ultrasound measurements will be useful at an older age for early detection of atherosclerosis, so it would be useful to have longitudinal data. The question remains whether asymptomatic atherosclerosis detection would have implications compared to interventions for the other MetS components.

In the separate analysis of the biomarkers, we observed elevated cystatine C in abdominally radiated survivors, even without elevation of creatinine. This discrepant finding may be due to underdeveloped abdominal musculature, and, if so, suggests that cystatine C is a more sensitive marker for assessing renal function in abdominally irradiated survivors [34].

The occurrence of MetS in our control group (3%) was relatively low, as MetS prevalence in The Netherlands at age 30-39 years has been reported as 10-20% [35]. We confirmed the representativeness of our controls by comparing their metabolic profile with other published, similar aged, Dutch reference cohorts [36, 37].

Some limitations of the current study merit consideration. This was a cross-sectional study, providing information at one time point only. As the study population was still relatively young, it is anticipated that the prevalence of MetS will increase when the survivors age. The advantage of diagnosis at younger age is the opportunity to intervene timely, to prevent diabetes and cardiovascular disease. This is particularly beneficial in childhood cancer survivors who received other, direct cardiotoxic treatment, such as anthracyclines and radiotherapy. Furthermore, we did not have information about daily calorie intake and family history of metabolic syndrome, diabetes and cardiovascular disease in our study population. However, we did take siblings (60% of the control group) and partners as control, and took the assumption that the calorie intake would be rather similar since they have a similar background. Family history is also most often similar between survivors and sibling controls.

Future research may focus on the validation of the use of adiponectin, LDL, apo-B and uric acid in larger, independent cohorts of survivors, with longitudinal follow-up. It may also be of interest to study ratios of biomarkers that provide additional diagnostic accuracy of MetS in the general population, such as triglycerides/HDL-ratio, and apoB-/apo-A1-ratio.

In conclusion, (young-)adult long-term survivors of childhood nephroblastoma and neuroblastoma, in particular after abdominal radiotherapy, frequently have MetS, defined by classic components, but also a novel, unfavourable integrated metabolic biomarker profile. This is important as the standard measurement of waist circumference after abdominal radiation is often infeasible in adult CCS. Our findings suggest that integrating the additional biomarkers identified in PCA – adiponectin, LDL, apo-B and uric acid – may be useful to assess MetS, particularly in abdominally irradiated survivors. In contrast, vascular ultrasound measurements do not seem to be of additional value in estimating MetS at this relatively young age. Validation of our proposed screening strategy will be of importance to elucidate the higher risk of MetS, diabetes mellitus and cardiovascular disease in CCS, after previous intensive cancer treatment, which is still relatively disguised at young age, and to identify subgroups at greater risk at an early stage.

ACKNOWLEDGEMENTS

We thank the following people for their contributions to this study: Rob Pieters, Manita van Baalen, Karin Blijdorp, Joke Beukers-Blom, Judith van Eck, Sjef van Asten, Moniek de Waart, Serieta Mohkamsing, Marjolein Gerrits-Boeye, Evelien Jager-van Beugen, Prof. van der Spek, Sigrid Swagemakers.

DECLARATIONS

Funding

KiKa 2009-030, KOCR Foundation and Princess Máxima Centre core funding.

Conflicts of interest

The authors have no conflicts of interest to declare with regard to this manuscript.

	Survivors	Controls	Bootstrap 95% CI	P-value
Number	103 (67 nephro-, 36 neuroblastoma)	61		
Male sex	54 (52.4%)	33 (54.1%)	n.a.	0.96 ¹
Age at follow-up (years) #	30.0 [25.2 - 37.9]	31.8 [23.3 - 40.0]	[-7.2;2.4]	0.33^{2}
Age at diagnosis (years) #	2.3 [0.8 – 5.0]	n.a.		
Follow-up time^ (years) #	27.5 [20.1 – 31.6]	n.a.		
BMI (kg/m^2) #	24.3 [21.3 – 26.3]	24.2 [22.1 – 27.2]	[-1.8;1.6]	0.84^{2}
Systolic BP (mmHg) #	124 [117 – 133]	118 [111 – 126]	[0.3:10.0]	0.026^{2*}
Diastolic BP (mmHg) #	76 [72 – 83]	72 [66 – 78]	[0.7;7.8]	0.012^{2*}
Medication use	, . [,]	, _ [, .]	[00,), 00]	
Lipid-lowering	4 (3.9%)	0	n.a.	0.30^{3}
Diabetes	6 (5.8%)	0	n.a.	0.085^{3}
Antihypertensive	6 (5.8%)	2 (3.3%)	n.a.	0.71^{3}
Physical activity score #	7695 [6390 - 10890]	8080 [6465 - 12278]	[-2947;1264]	0.71^{2}
Smoking			n.a.	0.62^{1}
Non-smoker	62 (60%)	32 (53%)		
Former smoker	15 (14.6%)	10 (16.4%)		
Smoker	26 (25%)	19 (31%)		
Socio-economic status			n.a.	0.31^{3}
Low	22 (21.4%)	10 (16.4%)		
Medium	36 (35.0%)	29 (47.5%)		
High	45 (43.7%)	22 (36.1%)		
Nephrectomy	74 (71.8%)	n.a.		
Adrenalectomy	47 (45.6%)	n.a.		
Abdominal radiotherapy	42 (40.8%)	n.a.		
Pancreas	32 (31.1%)			
Flank	17 (17.0%)			
Cumulative dose radiotherapy (Gy) #	21 [20 - 30]	n.a.		
Chemotherapy	90 (87 4%)	na		
Vincristine	65 (63.1%)	11.4.		
Actinomycine	48 (46.6%)			
Anthracyclines	30 (29.1%)			
Cyclofosfamide	31 (30.1%)			
Cisplatin	7 (6.8%)			
Teniposide	6 (5.8%)			
Dacarbazine	2 (1.9%)			
Ifosfamide	2 (1.9%)			
Corticosteroids	2 (1.9%)	n.a.		

Table 1. Baseline characteristics of included survivors and controls

BP = blood pressure; n.a. = not applicable.

Presented as median [IQR]; ^ Time after cessation of treatment.

Significance codes: 0 *** 0.001 ** 0.01 * 0.05

¹ Chi-squared test, ² Bootstrapped difference in medians, ³ Fisher's exact test

	Survivors (n=103)	Controls (n=61)	p-value
Metabolic syndrome (≥3 components)	14%	3%	0.032*1
≥2 MetS components	33%	20%	0.074^{1}
Abdominal obesity (waist circumference ≥102(men)/88(women)cm)	8%	11%	0.61 ²
High triglycerides (≥1.7mmol/L) or treatment	23%	10%	0.052^{2}
Low HDL cholesterol (≤1.03(men)/1.29(women)mmol/L or treatment	29%	18%	0.16 ²
High blood pressure (≥130/≥85mmHg) or treatment	35%	15%	0.007**2
High glucose (≥5.6mmol/L) or treatment	22%	11%	0.20 ²

Table 2. Occurrence of MetS and components in survivors and controls

Significance codes: 0 *** 0.001 ** 0.01 * 0.05 ¹ Fisher's exact test, ² Chi-squared test

Table 5. Comparison of serum biomarkers and vascular parameters between survivors and controls
--

Variable	Survivors (n=103) #	Controls (n=61) #	95% CI ^	P-value ^
Biomarkers				
Lipid metabolism ¹				
Triglycerides (mmol/L)	0.96 [0.72 – 1.41]	0.79 [0.63 – 1.20]	[0.01;0.30]	0.036*
HDL (mmol/L)	1.35 [1.11 – 1.52]	1.33 [1.14 – 1.54]	[-0.10;0.12]	0.88
FFA (mmol/L)	0.53 [0.42 – 0.69]	0.49 [0.35 – 0.63]	[-0.05;0.11]	0.36
ApoA1 (g/L)	1.35 [1.23 – 1.53]	1.36 [1.25 – 1.49]	[-0.06;0.07]	0.65
ApoB (g/L)	0.85 [0.71 – 1.06]	0.80 [0.68 - 0.94]	[-0.03;0.16]	0.17
LDL (mmol/L)	2.84 [2.28 - 3.57]	2.83 [2.25 - 3.20]	[-0.26;0.40]	0.84
Leptin (ng/mL)	8.10 [4.23 – 16.15]	7.69 [2.76 – 13.79]	[-2.61;4.33]	0.66
Adiponectin (µg/mL)	2.75 [1.01 – 4.29]	2.74 [1.88 - 4.16]	[-1.02;7.49]	0.91
Lpa (g/L)	0.13 [0.05 - 0.37]	0.11 [0.06 - 0.30]	[-0.03;0.06]	0.45
Glucose metabolism ²				
Glucose (mmol/L)	5.0 [4.6 - 8.7]	4.9 [4.7 – 5.2]	[-0.1;0.3]	0.42
Insulin (pmol/L)	21.5 [13.0 - 55.0]	25.0 [13.0 - 34.0]	[-12.0;13.0]	0.86
HOMA	$0.4 \ [0.4 - 0.8]$	0.5 [0.4 - 0.6]	[-0.1; 0.2]	0.41
Other MetS-associated biomark	ers			
Cystatin C (mg/L)	0.86 [0.81 - 0.94]	0.80 [0.74 - 0.86]	[0.03;0.11]	0.002**
Uric acid (mmol/L)	0.31 [0.25 - 0.40]	0.31 [0.26 - 0.34]	[-0.02; 0.05]	0.35
Urea (mmol/L)	5.3 [4.6 - 6.5]	5.1 [4.3 – 5.8]	[-0.2;0.8]	0.21
Creatinine (mg/mmol)	74 [67 – 85]	69 [63 – 78]	[1;10]	0.014^{*}
hsCRP (mg/L)	1.42 [0.55 – 3.47]	1.27 [0.52 – 3.57]	[-0.62;0.97]	0.46
Vascular parameters ³			-	
Central SBP (mmHg)	115 [105 – 126]	110 [101 – 122]	[-4;14]	0.28
Central DBP (mmHg)	76 [71 – 84]	77 [70 – 85]	[-5;4]	0.68
Central PP (mmHg)	38 [31 - 45]	33 [28 - 44]	[-1;10]	0.068
PP (mmHg)	46 [40 – 51]	46 [42 – 50]	[-4;2]	0.64
PPA	1.23 [1.04 – 1.44]	1.33 [1.07 – 1.69]	[-0.26;0.09]	0.40
Diameter CCA (mm)	6.38 [5.92 - 6.83]	6.40 [5.82 - 6.68]	[-0.25;0.33]	0.83
CIMT (µm)	523 [477 – 581]	531 [481 – 586]	[-40;21]	0.44
DC	25.6 [18.4 - 32.3]	28.3 [19.6 - 37.6]	[-7.2;3.7]	0.27
PWV (m/s)	6.9 [6.0 - 8.0]	7.0 [6.3 – 7.8]	[-0.7;0.4]	0.39

S/DBP = systolic/diastolic blood pressure; PP = pulse pressure; PPA = pulse pressure amplification; CCA = common carotid artery; CIMT = carotid intima media thickness; DC = distensibility coefficient; PWV = pulse wave velocity.

Presented as median [IQR]; ^ Bootstrapped difference in medians

Significance codes: 0 *** 0.001 ** 0.01 * 0.05

¹ Subjects using lipid-lowering medication excluded (n=4 survivors)

² Subjects with diabetes excluded (n=6 survivors)

³ Subjects using antihypertensive medication excluded (n=6 survivors, n=2 controls)

Table 4. Uni- and multivariable regression of the effect of abdominal radiotherapy and nephrectomy on MetS and components.

The influence of abdominal radiotherapy							
	Univa	riable analysis	5	Multivariable analysis ¹			
	OR (s.e.)	95% CI	p-value	OR (s.e.)	95% CI	p-value	
MetS	6.04 (0.554)	2.04;17.89	0.001**	15.3 (0.799)	3.21;73.36	< 0.001***	
≥2 MetS components	3.36 (0.380)	1.59;7.07	0.001**	3.23 (0.445)	1.35;7.73	0.009**	
Abdominal obesity	<0.0001 (1659)	-	0.99	-	-	-	
High triglycerides or treatment	5.70 (0.430)	2.45;13.24	<0.001***	7.01 (0.548)	2.39;20.52	<0.001 ***	
Low HDL cholesterol or treatment	2.39 (0.389)	1.11;5.12	0.025*	2.94 (0.447)	1.23;7.07	0.016*	
High blood pressure or treatment	4.24 (0.387)	1.99;9.06	<0.001***	5.11 (0.478)	2.00;13.02	<0.001***	
High glucose or treatment	2.38 (0.431)	1.02;5.53	0.044*	2.53 (0.514)	0.92;6.93	0.071	

Significance codes: 0 *** 0.001 ** 0.01 * 0.05

¹Corrected for age, sex, smoking and socio-economic status

The influence of nephrectomy

)					
	Univ	ariable analys	is	Multi	ivariable analy	sis ²
	OR (s.e.)	95%CI	p-value	OR (s.e.)	95%CI	p-value
MetS	2.97 (0.564)	0.98;8.97	0.054	-	-	-
≥2 MetS components	2.15 (0.354)	1.07;4.29	0.031*	2.78 (0.406)	1.26;6.17	0.012*
Abdominal obesity	0.41 (0.607)	0.13;1.35	0.14	-		-
High triglycerides or	2.96 (0.426)	1.29;6.82	0.011*	4.68 (0.528)	1.66;13.19	0.004**
treatment						
Low HDL cholesterol or	1.22 (0.361)	0.60;2.47	0.59	-	-	-
treatment						
High blood pressure or	3.95 (0.375)	1.89;8.25	< 0.001***	4.82 (0.435)	2.05;11.29	< 0.001***
treatment						
High glucose or	1.55 (0.413)	0.69;3.48	0.29	-		-
treatment						

Significance codes: 0 *** 0.001 ** 0.01 * 0.05

² Corrected for age, sex, smoking and socio-economic status



Figure 1. Biplots of PCA (principal components 1 and 2) of serum biomarkers, with the effect of abdominal radiotherapy (A) and nephrectomy (B). Score on PC1 and PC2 is a Z-score, based on loadings and Z-scores of biomarkers. ApoA1 = apolipoprotein-A1; ApoB = apolipoprotein-B; FFA = free fatty acids; HDL = high density lipoprotein cholesterol; HOMA = homeostasis model assessment; LDL = low density lipoprotein cholesterol; Lpa = lipoprotein(a).

SUPPLEMENTARY MATERIAL

	Nephroblastoma	Neuroblastoma	Bootstrap 95% CI	P-value
Number	67	36		
Male sex	39 (58.2%)	15 (41.7%)	n.a.	0.16 ¹
Age at follow-up (years) #	30.2 [25.2 - 39.4]	29.6 [25.1 - 33.4]	[-4.0;5.9]	0.85^{2}
Age at diagnosis (years) #	3.3 [1.8 – 5.2]	0.8 [0.3 – 1.9]	[1.4;3.8]	< 0.001 ² ***
Follow-up time^ (years) #	26.3 [19.3 – 32.7]	27.8 [21.7 - 30.9]	[-6.3;3.1]	0.54^{2}
BMI $(kg/m^2) \#$	24.6 [21.7 – 27.1]	24.1 [21.1 – 26.0]	[-1.7;2.6]	0.72^{2}
Systolic BP (mmHg) #	123 [118 – 133]	124 [115 – 129]	[-6;6]	0.62^{2}
Diastolic BP (mmHg) #	77 [73 – 83]	75 [72 – 84]	[-4;5]	0.20^{2}
Medication use				
Lipid-lowering	3 (4.5%)	1 (2.8%)	n.a.	13
Diabetes	5 (7.5%)	1 (2.8%)	n.a.	0.66^{3}
Antihypertensive	5 (7.5%)	1 (2.8%)	n.a.	0.66^{3}
Physical activity score #	8140 [6634 - 12070]	6685 [5805 - 8423]	[-248;2560]	0.10^{2}
Smoking			n.a.	0.29^{1}
Non-smoker	44 (65.7%)	18 (50.0%)		
Former smoker	8 (11.9%)	7 (19.4%)		
Smoker	15 (22.4%)	11 (30.6%)		
Socio-economic status			n.a.	13
Low	14 (20.9%)	8 (22.2%)		
Medium	24 (35.8%)	12 (33.3%)		
High	29 (43.3%)	16 (44.4%)		
Nephrectomy	67 (100%)	7 (19.4%)	n.a.	< 0.001 ****
Adrenalectomy	33 (49.3%)	14 (38.9%)	n.a.	0.42^{1}
Abdominal radiotherapy	35 (52.2%)	7 (19.4%)	n.a.	0.0031**
Pancreas	24 (35.8%)	7 (19.4%)		0.23 ¹
Flank	17 (26.6%)	0		0.002^{1**}
Cumulative dose	21 [20 - 30]	20 [19.6 - 20.5]	[-0.7;8.6]	0.064^{2}
radiotherapy (Gy) #				
Chemotherapy	59 (88.1%)	31 (86.1%)	n.a.	0.76^{3}
Vincristine	50 (74.6%)	15 (41.7%)		0.002^{1**}
Actinomycine	48 (71.6%)	0		< 0.001 ****
Anthracyclines	18 (26.9%)	12 (33.3%)		0.64^{1}
Cyclofosfamide	2 (3.0%)	29 (80.6%)		< 0.001 1***
Cisplatin	0	7 (19.4%)		< 0.001 ³
Teniposide	0	6 (16.7%)		0.001^{3}
Dacarbazine	2 (3.0%)	0		0.54^{3}
Ifosfamide	2 (3.0%)	0		0.54^{3}
Corticosteroids	0	2 (5.6%)	n.a.	0.12^{3}

Supplemental Table 1. Baseline characteristics of nephro- and neuroblastoma survivors separated.

BP = blood pressure; n.a. = not applicable.

Presented as median [IQR]; ^ Time after cessation of treatment.

Significance codes: 0 *** 0.001 ** 0.01 * 0.05

¹ Chi-squared test, ² Bootstrapped difference in medians, ³ Fisher's exact test

**				
Variable	Abdominal radiotherapy (n=42) #	No abdominal radiotherapy (n=61) #	95% CI ^	P-value ^
Biomarkers	^ .			
Lipid metabolism ¹				
Triglycerides (mmol/L)	1.35 [0.81 – 2.09]	0.91 [0.70 - 1.08]	[0.06;0.88]	0.024*
HDL (mmol/L)	1.32 [1.08 – 1.52]	1.35 [1.12 – 1.52]	[-0.19;0.13]	0.74
FFA (mmol/L)	0.57 [0.44 – 0.74]	0.50 [0.38 - 0.64]	[0.01;0.17]	0.026*
ApoA1 (g/L)	1.35 [1.21 – 1.58]	1.35 [1.25 – 1.49]	[-0.10;0.16]	0.75
ApoB (g/L)	0.99 [0.81 – 1.18]	0.81 [0.65 - 0.97]	[0.07;0.30]	0.004**
LDL (mmol/L)	3.19 [2.71 – 3.79]	2.55 [2.15 - 3.18]	[0.15;1.09]	0.004**
Leptin (ng/mL)	9.00 [4.98 - 18.38]	7.60 [3.67 – 12.90]	[-2.95;6.81]	0.51
Adiponectin (µg/mL)	2.30 [0.56 - 4.31]	2.83 [1.53 - 4.15]	[-2.09;1.22]	0.64
Lpa (g/L)	0.09 [0.03 - 0.45]	0.13 [0.05 - 0.35]	[-0.10;0.17]	0.65
Glucose metabolism ²				
Glucose (mmol/L)	5.0 [4.8 – 5.5]	4.9 [4.6 – 5.3]	[-0.2;0.4]	0.36
Insulin (pmol/L)	18.0 [13.0 - 44.0]	24.0 [13.0 - 58.0]	[-22.0;13.0]	0.48
HOMA	$0.4 \ [0.4 - 0.8]$	$0.4 \ [0.4 - 0.8]$	[-0.3; 0.3]	0.46
Other MetS-associated biom	narkers			
Cystatin C (mg/L)	0.88 [0.83 - 0.97]	0.85 [0.77 - 0.90]	[0.00;0.09]	0.048*
Uric acid (mmol/L)	0.35 [0.29 – 0.44]	0.30 [0.24 - 0.38]	[-0.01;0.10]	0.094
Urea (mmol/L)	5.7 [5.1 – 6.8]	5.0 [4.5 – 5.9]	[0.1;1.5]	0.006**
Creatinine (mg/mmol)	75 [68 - 84]	74 [66 – 85]	[-7;9]	0.83
hsCRP (mg/L)	1.62 [0.77 – 3.35]	1.42 [0.39 – 3.79]	[-0.90;1.61]	0.65
Vascular parameters ³				
Central SBP (mmHg)	124 [114 – 132]	109 [101 – 122]	[8;20]	< 0.001***
Central DBP (mmHg)	84 [76 - 80]	74 [69 – 77]	[4;15]	< 0.001***
Central PP (mmHg)	40 [34 - 47]	37 [30 - 44]	[-2;9]	0.19
PP (mmHg)	45 [41 – 53]	46 [40 – 49]	[-4;6]	0.95
PPA	1.23 [1.06 – 1.40]	1.30 [1.03 – 1.48]	[-0.21;0.15]	0.52
Diameter CCA (mm)	6.46 [5.93 – 6.90]	6.32 [5.92 - 6.83]	[-0.30;0.53]	0.47
CIMT (µm)	552 [482 - 595]	509 [458 – 569]	[-2;71]	0.052
DC	20.6 [15.7 – 26.1]	30.3 [21.3 - 39.8]	[-13.9;3.1]	0.004**
PWV (m/s)	8.0[6.1-8.8]	6.6 [6.0 – 7.3]	[0.2;1.9]	0.016*

Supplemental Table	2a.	The effect	of abdominal	radiotherapy or	biomarkers and	vascular r	parameters
Supplemental lable	La.	The encer	or abuomman	radiotificially of	i Diomarkers and	vasculai p	Jarameters.

S/DBP = systolic/diastolic blood pressure; PP = pulse pressure; PPA = pulse pressure amplification; CCA = common carotid artery; CIMT = carotid intima media thickness; DC = distensibility coefficient; PWV = pulse wave velocity.
Presented as median [IQR]; ^ Bootstrapped difference in medians

Significance codes: 0 *** 0.001 ** 0.01 * 0.05

¹ Subjects using lipid-lowering medication excluded (n=3 abdominal radiotherapy, n=1 no abdominal radiotherapy)

² Subjects with diabetes excluded (n=5 abdominal radiotherapy, n=1 no abdominal radiotherapy)

 3 Subjects using antihypertensive medication excluded (n=4 abdominal radiotherapy, n=2 no abdominal radiotherapy)

		Univariable	analysis	Multivariable analysis ¹	
Variable	P-value bootstrap difference medians	Beta (s.e.)	P-value	Beta (s.e.)	P-value
Triglycerides	0.024	0.614 (0.165)	< 0.001***	0.572 (0.154)	0.002**
FFA	0.026	0.123 (0.049)	0.008**	0.151 (0.059)	0.008**
АроВ	0.004	0.168 (0.054)	0.002**	0.083 (0.068)	0.19
LDL	0.004	0.469 (0.170)	0.008**	0.043 (0.235)	0.83
Cystatin C	0.048	0.065 (0.032)	0.041*	0.076 (0.039)	0.039*
Urea	0.006	0.895 (0.373)	0.011*	0.690 (0.465)	0.10
Central SBP	< 0.001	10.226 (3.687)	0.008**	6.029 (5.033)	0.19
Central DBP	< 0.001	9.122 (2.024)	< 0.001***	5.385 (2.929)	0.023*
DC	0.004	-8.578 (2.810)	0.004**	-2.672 (3.279)	0.44
PWV	0.016	0.935 (0.451)	0.025*	0.294 (0.456)	0.53

Supplemental Table 2b. Uni- and multivariable bootstrap linear regression analysis of the influence of abdominal radiotherapy on biomarkers and vascular parameters.

Significance codes: 0 *** 0.001 ** 0.01 * 0.05

¹ Corrected for age, sex, smoking and socio-economic status.

Supplemental Table 2c. The effect of nephrectomy on biomarkers and vascular parameters.

Variable	Nephrectomy (n=74) #	No nephrectomy (n=29) #	95% CI ^	P-value ^
Biomarkers				
Lipid metabolism ¹				
Triglycerides (mmol/L)	0.98 [0.77 – 1.58]	0.90 [0.70 - 1.19]	[-0.16;0.37]	0.43
HDL (mmol/L)	1.35 [1.09 – 1.55]	1.32 [1.16 – 1.48]	[-0.13;0.16]	0.72
FFA (mmol/L)	0.50 [0.40 - 0.61]	0.65 [0.49 – 0.77]	[-0.26;0.01]	0.054
ApoA1 (g/L)	1.38 [1.24 – 1.54]	1.31 [1.20 – 1.51]	[-0.04;0.16]	0.18
ApoB (g/L)	0.85 [0.70 - 1.05]	0.86 [0.74 - 1.08]	[-0.17;0.13]	0.79
LDL (mmol/L)	2.83 [2.20 - 3.54]	2.92 [2.45 - 3.63]	[-0.69;0.28]	0.43
Leptin (ng/mL)	8.37 [3.64 - 18.38]	8.10 [5.44 – 12.90]	[-5.47;3.66]	0.86
Adiponectin (µg/mL)	2.47 [0.98 - 4.31]	3.02 [1.81 - 4.13]	[-1.58;0.55]	0.26
Lpa (g/L)	0.13 [0.05 – 0.35]	0.11 [0.05 – 0.39]	[-0.11;0.08]	0.80
Glucose metabolism ²				
Glucose (mmol/L)	5.0 [4.7 – 5.3]	4.8 [4.5 – 5.4]	[-0.1;0.5]	0.086
Insulin (pmol/L)	29.0 [13.0 - 55.0]	13.0 [13.0 – 49.0]	[-6.0;23.0]	0.088
HOMA	0.6 [0.4 - 0.9]	$0.4 \; [0.4 - 0.8]$	[-0.1; 0.3]	0.060
Other MetS-associated bioma	arkers			
Cystatin C (mg/L)	0.88 [0.84 - 0.99]	0.81 [0.72 – 0.86]	[0.03;0.16]	0.002**
Uric acid (mmol/L)	0.35 [0.27 - 0.43]	0.26 [0.21 - 0.31]	[0.02;0.13]	0.002**
Urea (mmol/L)	5.4 [4.8 - 6.6]	5.0 [4.2 – 5.7]	[-0.2;1.0]	0.084
Creatinine (mg/mmol)	79 [69 – 89]	69 [63 – 75]	[2;16]	0.004**
hsCRP (mg/L)	1.62 [0.72 - 3.48]	1.42 [0.39 – 3.09]	[-0.93;1.45]	0.46

Variable	Nephrectomy (n=74) #	No nephrectomy (n=29) #	95% CI ^	P-value ^
Vascular parameters ³				
Central SBP (mmHg)	118 [108 – 127]	108 [100 – 123]	[0;19]	0.050*
Central DBP (mmHg)	77 [72 – 85]	75 [71 – 79]	[-2;7]	0.17
Central PP (mmHg)	39 [34 - 47]	34 [30 - 43]	[-3;10]	0.14
PP (mmHg)	46 [40 – 53]	45 [41 - 48]	[-3;5]	0.62
PPA	1.23 [0.99 – 1.41]	1.31 [1.13 – 1.54]	[-0.30;0.09]	0.43
Diameter CCA (mm)	6.30 [5.87 – 6.82]	6.57 [6.13 – 6.93]	[-0.72;0.21]	0.32
CIMT (µm)	521 [479 – 582]	523 [461 - 570]	[-43;53]	0.81
DC	24.0 [15.8 - 39.8]	29.9 [24.0 - 31.9]	[-10.7;1.5]	0.13
PWV (m/s)	6.8 [6.0 - 8.0]	6.9 [6.1 – 7.5]	[-0.7;0.7]	0.79

Supplemental Table 2c. Continued

S/DBP = systolic/diastolic blood pressure; PP = pulse pressure; PPA = pulse pressure amplification; CCA = common carotid artery; CIMT = carotid intima media thickness; DC = distensibility coefficient; PWV = pulse wave velocity. # Presented as median [IQR]; ^ Bootstrapped difference in medians

Significance codes: 0 *** 0.001 ** 0.01 * 0.05

¹ Subjects using lipid-lowering medication excluded (n=3 nephrectomy, n=1 no nephrectomy)

² Subjects with diabetes excluded (n=5 nephrectomy, n=1 no nephrectomy)

³ Subjects using antihypertensive medication excluded (n=6 nephrectomy)

		Univariable analy	sis	Multivariable a	nalysis ¹
Variable	P-value bootstrap difference medians	Beta (s.e.)	P-value	Beta (s.e.)	P-value
Cystatin C	0.002	0.128 (0.027)	< 0.001***	0.115 (0.027)	< 0.001***
Uric acid	0.002	0.075 (0.017)	< 0.001***	0.045 (0.013)	0.006**
Creatinine	0.004	12.062 (2.899)	0.002**	6.954 (2.640)	0.042*
Central SBP	0.050	8.233 (4.089)	0.054	-	-

Supplemental Table 2d. Uni- and multivariable bootstrap linear regression analysis of the influence of nephrectomy on biomarkers and vascular parameters.

Significance codes: 0 *** 0.001 ** 0.01 * 0.05

¹ Corrected for age, sex, smoking and socio-economic status



Supplemental Figure 1. Correlation between waist and PC1 score in non-abdominally irradiated survivors. Pearson's correlation coefficient: 0.64

REFERENCES

- 1. ASCO. Wilms Tumor Childhood: Statistics. 2019-01.
- 2. ASCO. Neuroblastoma Childhood: Statistics. 2019-01.
- van Waas M, Neggers SJ, Raat H, van Rij CM, Pieters R, van den Heuvel-Eibrink MM. Abdominal radiotherapy: a major determinant of metabolic syndrome in nephroblastoma and neuroblastoma survivors. PLoS One. 2012;7(12):e52237.
- Geenen MM, Bakker PJ, Kremer LC, Kastelein JJ, van Leeuwen FE. Increased prevalence of risk factors for cardiovascular disease in long-term survivors of acute lymphoblastic leukemia and Wilms tumor treated with radiotherapy. Pediatr Blood Cancer. 2010;55(4):690-7.
- 5. Norsker FN, Rechnitzer C, Cederkvist L, Holmqvist AS, Tryggvadottir L, Madanat-Harjuoja LM, Ora I, Thorarinsdottir HK, Vettenranta K, Bautz A, et al. Somatic late effects in 5-year survivors of neuroblastoma: a population-based cohort study within the Adult Life after Childhood Cancer in Scandinavia study. Int J Cancer. 2018;143(12):3083-96.
- 6. National Cholesterol Education Program Expert Panel on Detection E, and Treatment of High Blood Cholesterol in Adults (Adult Treatment Panel III). Third Report of the National Cholesterol Education Program (NCEP) Expert Panel on Detection, Evaluation, and Treatment of High Blood Cholesterol in Adults (Adult Treatment Panel III). 2002.
- 7. Alberti G, Zimmet P, Shaw J, Grundy SM. The IDF consensus worldwide definition of the metabolic syndrome. 2006.
- Alberti KGMM, Zimmet PZ. Definition, Diagnosis and Classification of Diabetes Mellitus and its Complications Part 1: Diagnosis and Classification of Diabetes Mellitus Provisional Report of a WHO Consultation. Diabet Med. 1998;15:539-53.
- 9. Moller DE, Kaufman KD. Metabolic syndrome: a clinical and molecular perspective. Annu Rev Med. 2005;56:45-62.
- Blijdorp K, van den Heuvel-Eibrink MM, Pieters R, Boot AM, Delhanty PJ, van der Lely AJ, Neggers SJ. Obesity is underestimated using body mass index and waist-hip ratio in long-term adult survivors of childhood cancer. PLoS One. 2012;7(8):e43269.
- Reulen RC, Winter DL, Frobisher C, Lancashire ER, Stiller CA, Jenney ME, Skinner R, Stevens MC, Hawkins MM. Long-term Cause-Specific Mortality Among Survivors of Childhood Cancer. JAMA. 2010;304(2):172-9.
- Schindler M, Spycher BD, Ammann RA, Ansari M, Michel G, Kuehni CE, Swiss Paediatric Oncology G. Cause-specific long-term mortality in survivors of childhood cancer in Switzerland: A population-based study. Int J Cancer. 2016;139(2):322-33.
- Tukenova M, Guibout C, Oberlin O, Doyon F, Mousannif A, Haddy N, Guerin S, Pacquement H, Aouba A, Hawkins M, et al. Role of cancer treatment in long-term overall and cardiovascular mortality after childhood cancer. J Clin Oncol. 2010;28(8):1308-15.
- Armstrong GT, Oeffinger KC, Chen Y, Kawashima T, Yasui Y, Leisenring W, Stovall M, Chow EJ, Sklar CA, Mulrooney DA, et al. Modifiable risk factors and major cardiac events among adult survivors of childhood cancer. J Clin Oncol. 2013;31(29):3673-80.
- Shlipak MG, Sarnak MJ, Katz R, Fried LF, Seliger SL, Newman AB, Siscovick DS, Stehman-Breen C. Cystatin C and the risk of death and cardiovascular events among elderly persons. N Engl J Med. 2005;352(20):2049-60.
- 16. Im JA, Kim SH, Lee JW, Shim JY, Lee HR, Lee DC. Association between hypoadiponectinemia and cardiovascular risk factors in nonobese healthy adults. Metabolism. 2006;55(11):1546-50.

- 17. Cicero AFG, Fogacci F, Giovannini M, Grandi E, Rosticci M, D'Addato S, Borghi C. Serum uric acid predicts incident metabolic syndrome in the elderly in an analysis of the Brisighella Heart Study. Sci Rep. 2018;8(1):11529.
- 18. Herder C, Karakas M, Koenig W. Biomarkers for the prediction of type 2 diabetes and cardiovascular disease. Clin Pharmacol Ther. 2011;90(1):52-66.
- 19. Cao JJ, Arnold AM, Manolio TA, Polak JF, Psaty BM, Hirsch CH, Kuller LH, Cushman M. Association of carotid artery intima-media thickness, plaques, and C-reactive protein with future cardiovascular disease and all-cause mortality: the Cardiovascular Health Study. Circulation. 2007;116(1):32-8.
- Nijdam ME, Plantinga Y, Hulsen HT, Bos WJ, Grobbee DE, van der Schouw YT, Bots ML. Pulse pressure amplification and risk of cardiovascular disease. Am J Hypertens. 2008;21(4):388-92.
- 21. Matthews DR, Hosker JP, Rudenski AS, Naylor BA, Treacher DF, Turner RC. Homeostasis model assessment: insulin resistance and beta-cell function from fasting plasma glucose and insulin concentrations in man. Diabetologia. 1985;28(7):412-9.
- 22. Laurent S, Cockcroft J, Van Bortel L, Boutouyrie P, Giannattasio C, Hayoz D, Pannier B, Vlachopoulos C, Wilkinson I, Struijker-Boudier H, et al. Expert consensus document on arterial stiffness: methodological issues and clinical applications. Eur Heart J. 2006;27(21):2588-605.
- 23. van Waas M, Neggers SJ, van Eck JP, van Noesel MM, van der Lely AJ, de Jong FH, Pieters R, van den Heuvel-Eibrink MM. Adrenal function in adult long-term survivors of nephroblastoma and neuroblastoma. Eur J Cancer. 2012;48(8):1159-66.
- 24. Kanbay M, Jensen T, Solak Y, Le M, Roncal-Jimenez C, Rivard C, Lanaspa MA, Nakagawa T, Johnson RJ. Uric acid in metabolic syndrome: From an innocent bystander to a central player. Eur J Intern Med. 2016;29:3-8.
- Parish S, Offer A, Clarke R, Hopewell JC, Hill MR, Otvos JD, Armitage J, Collins R, Heart Protection Study Collaborative G. Lipids and lipoproteins and risk of different vascular events in the MRC/BHF Heart Protection Study. Circulation. 2012;125(20):2469-78.
- McQueen MJ, Hawken S, Wang X, Ounpuu S, Sniderman A, Probstfield J, Steyn K, Sanderson JE, Hasani M, Volkova E, et al. Lipids, lipoproteins, and apolipoproteins as risk markers of myocardial infarction in 52 countries (the INTERHEART study): a case-control study. Lancet. 2008;372(9634):224-33.
- 27. Mach F, Baigent C, Catapano AL, Koskinas KC, Casula M, Badimon L, Chapman MJ, De Backer GG, Delgado V, Ference BA, et al. 2019 ESC/EAS Guidelines for the management of dyslipidaemias: lipid modification to reduce cardiovascular risk. Eur Heart J. 2019.
- Davidson MH. Apolipoprotein measurements: is more widespread use clinically indicated? Clin Cardiol. 2009;32(9):482-6.
- Mostoufi-Moab S, Seidel K, Leisenring WM, Armstrong GT, Oeffinger KC, Stovall M, Meacham LR, Green DM, Weathers R, Ginsberg JP, et al. Endocrine Abnormalities in Aging Survivors of Childhood Cancer: A Report From the Childhood Cancer Survivor Study. J Clin Oncol. 2016;34(27):3240-7.
- 30. Meacham LR, Sklar CA, Li S, Liu Q, Gimpel N, Yasui Y, Whitton JA, Stovall M, Robison LL, Oeffinger KC. Diabetes mellitus in long-term survivors of childhood cancer. Increased risk associated with radiation therapy: a report for the childhood cancer survivor study. Arch Intern Med. 2009;169(15):1381-8.
- 31. de Vathaire F, El-Fayech C, Ben Ayed FF, Haddy N, Guibout C, Winter D, Thomas-Teinturier C, Veres C, Jackson A, Pacquement H, et al. Radiation dose to the pancreas and risk of diabetes

mellitus in childhood cancer survivors: a retrospective cohort study. The Lancet Oncology. 2012;13(10):1002-10.

- 32. van Waas M, Neggers SJ, Uitterlinden AG, Blijdorp K, van der Geest IM, Pieters R, van den Heuvel-Eibrink MM. Treatment factors rather than genetic variation determine metabolic syndrome in childhood cancer survivors. Eur J Cancer. 2013;49(3):668-75.
- 33. Roman MJ, Devereux RB, Kizer JR, Lee ET, Galloway JM, Ali T, Umans JG, Howard BV. Central pressure more strongly relates to vascular disease and outcome than does brachial pressure: the Strong Heart Study. Hypertension. 2007;50(1):197-203.
- 34. Deinum J, Derkx FH. Cystatin for estimation of glomerular filtration rate? Lancet. 2000;356(9242):1624-5.
- Blokstra A, Vissink P, Venmans LMAJ, Holleman P, van der Schouw YT, Smit HA, Verschuren WM. Measuring The Netherlands. A monitoring study of risk factors in hte general population, 2009-2010. RIVM Rapport 260152001/2011. 2011.
- Beulens JW, Monninkhof EM, Verschuren WM, van der Schouw YT, Smit J, Ocke MC, Jansen EH, van Dieren S, Grobbee DE, Peeters PH, et al. Cohort profile: the EPIC-NL study. Int J Epidemiol. 2010;39(5):1170-8.
- Balder JW, de Vries JK, Nolte IM, Lansberg PJ, Kuivenhoven JA, Kamphuisen PW. Lipid and lipoprotein reference values from 133,450 Dutch Lifelines participants: Age- and gender-specific baseline lipid values and percentiles. J Clin Lipidol. 2017;11(4):1055-64 e6.

Chapter 3

Can biomarkers be used to improve diagnosis and prediction of metabolic syndrome in childhood cancer survivors? A systematic review

V.G. Pluimakers*, S.S. van Santen*, M. Fiocco, M.E. Bakker, A.J. van der Lelij, M.M. van den Heuvel-Eibrink**, S.J.C.M.M. Neggers**

Obesity Reviews, 2021 Nov;22(11):e13312.

ABSTRACT

Background: Childhood cancer survivors (CCS) are at increased risk to develop metabolic syndrome (MetS), diabetes and cardiovascular disease. Common criteria underestimate adiposity and possibly underdiagnose MetS, particularly after abdominal radiotherapy.

Design: A systematic literature review and meta-analysis on the diagnostic and predictive value of nine newer MetS related biomarkers (adiponectin, leptin, uric acid, hsCRP, TNF-alpha, IL-1, IL-6, apolipoprotein B (apoB), and lipoprotein(a) [lp(a)]) in survivors and adult non-cancer survivors was performed by searching PubMed and Embase. Evidence was summarized with GRADE after risk of bias evaluation (QUADAS-2/QUIPS). Eligible studies on promising biomarkers were pooled.

Results: We identified 175 general population, and 5 CCS studies. In the general population, valuable predictive biomarkers are uric acid, adiponectin, hsCRP and apoB (high level of evidence) and leptin (moderate level of evidence). Valuable diagnostic biomarkers are hsCRP, adiponectin, uric acid and leptin (low, low, moderate and high level of evidence, respectively). Meta-analysis showed OR for hyperuricemia of 2.94 (age-/sex-adjusted), OR per unit uric acid increase of 1.086 (unadjusted), and AUC for hsCRP of 0.71 (unadjusted).

Conclusions: Uric acid, adiponectin, hsCRP, leptin, and apoB can be alternative biomarkers in the screening setting for MetS in survivors, to enhance early identification of those at high risk of subsequent complications.

Abbreviations: CVD = cardiovascular disease; T2DM = type 2 diabetes mellitus; MetS = metabolic syndrome; CCS = childhood cancer survivors; DXA = Dual-energy X-ray Absorptiometry; hsCRP = high sensitivity C-reactive protein; TNF-alpha = Tumor Necrosis Factor alpha; IL-1 = interleukin 1; IL-6 = interleukin 6; apoB = apolipoprotein B; Lp(a) = lipoprotein(a); AUC = area under the curve; ROC = receiver operating characteristic; OR = odds ratio; HR = hazard ratio; GRADE = Grading of Recommendations Assessment Development and Evaluation; BMI = body mass index; ALL = acute lymphoblastic leukemia; apoA1 = apolipoprotein A1; LDL = low density lipoproteins; HOMA-IR = Homeostatic Model Assessment of Insulin Resistance; HDL = high density lipoproteins.

GRAPHICAL ABSTRACT



Value of Metabolic Syndrome biomarkers

Graphical abstract of systematic review for novel biomarkers for diagnosis and prediction of the metabolic syndrome in childhood cancer survivors and a young general population. Conclusions per biomarker are categorized as valuable, not valuable or conflicting evidence, and the level of evidence is expressed per color from low to high (black-blue-greenyellow-red; see figure on the lower right).

Key words: childhood cancer survivors, the metabolic syndrome, biomarker, systematic review

INTRODUCTION

Childhood cancer 5-year survival rates have increased from 5-30% in early seventies to more than 80% in the present time.^{1, 2} Deployed therapies, such as chemotherapy, radiotherapy, and stem cell transplantation, better stratification and enhanced supportive care regimens, are responsible for increase in survival rates. However, intensification of treatment is also associated with long-term excess mortality and morbidities in survivors.³ Survivors have a high level of frailty, suggesting their biological age progresses faster than their actual age. Consequently, survivors with an actual mean age of 33 have a biological age of 65 if they are compared to the general population.⁴⁻⁹ At the age of 45-50 years, the prevalence of any chronic health condition is very high, from 95% up to 99%.^{3, 10} One of these severe conditions is represented by cardiovascular disease (CVD), which is an important cause of premature death beyond 5 years cancer survival; the standardized mortality risk for CVD ranges from 1.9 to 12.7.¹¹⁻²⁵

This high risk of cardiovascular death is not only due to treatment effects, such as anthracycline exposure and cardiac irradiation;²⁶ survivors are also at high risk of type II diabetes mellitus (T2DM) and the metabolic syndrome (MetS).¹¹ These diseases are independent predictors of CVD and associated with factors such as adiposity, dyslipidemia, insulin resistance, and hypertension. These factors cluster together and form the 'deadly quartet', a MetS concept developed by Reaven in 1988.²⁷ The MetS had many definitions ever since.^{11, 27-37} Patients with MetS carry a doubled risk of dying from cardio- and cerebrovascular disease.^{11, 38} In addition, patients with the MetS are five times more likely to develop T2DM, which subsequently triples the risk of CVD.^{11, 39-41}

As survivors develop cardiovascular complications at a relatively young age, there is a need for early diagnosis of MetS, to possibly prevent T2DM and CVD, and to improve long-term survival.¹¹ The occurrence of MetS may be underestimated especially in abdominally irradiated childhood cancer survivors (CCS), who have an unreliable waist circumference, while their MetS risk is even higher.^{11, 42-44} Body mass index and bioimpedance are alternative methods for body composition measurement, but do not specifically measure abdominal fat, rely on hydration status and often underestimate body fat.^{42, 45-47} Obviously, another alternative option to evaluate adiposity is measuring fat percentage by Dual-energy X-ray Absorptiometry (DXA) scan, which is the gold standard in case of suspected discordance of anthropomorphic measurements and adiposity.^{42, 48, 49} However, performing DXA scans in all survivors on a routine basis is time-consuming and costly.¹¹ Additionally, there is currently no consensus for the threshold of fat percentage for diagnosing obesity.⁵⁰ Newer serum biomarkers may serve as another alternative for accurate early diagnosis or prediction of (disguised) MetS in CCS. Adult cardiologists currently apply multiple biomarkers that have been shown to improve risk estimation for CVD.⁵¹

Therefore, our primary objectives were to evaluate the value of the use of these newer serum biomarkers as (1) diagnostic marker, and as (2) additional independent predictor for the occurrence of MetS later in life, in survivors of childhood cancer specifically, as well as in a relatively young general, non-cancer population (studies with >75% of participants below 65 years). By including this selection of general population studies as well, we aimed to cover all available literature applicable and generalizable to young-adult survivors. To accomplish this, we performed a systematic literature search on adipokines adiponectin and leptin, uric acid, the inflammatory markers high sensitivity C-reactive protein (hsCRP), Tumor Necrosis Factor alpha (TNF-alpha), interleukin 1 (IL-1), and interleukin 6 (IL-6), and the lipid markers apolipoprotein B (apoB) and lipoprotein(a) [Lp(a)], and performed a meta-analysis of these outcomes for relevant recurrently published biomarkers. As secondary purpose, we screened for other new biomarkers that are not enlisted above, in order to reveal additional, potentially useful biomarkers.

METHODS

The Systematic Search

A systematic literature review was performed in PubMed and Embase, to gather all published literature published between the first of October 2009 and September 3, 2020. Details of the search terms are available in Supplementary Table 1; in general, the search terms were related to adults/general population, as well as to (childhood) cancer survivors, and combined with all enlisted 9 separate biomarkers (adiponectin, leptin, uric acid, hsCRP, TNF-alpha, IL-1, IL-6, apoB and lp(a)) and the MetS. The AMSTAR checklist for systematic reviews was followed.⁵² All titles and abstracts were screened by two independent reviewers (VP and SSvS), who were blinded to each other's judgement. Studies were included if they had the MetS as outcome, and one or more newer biomarker(s) as independent variable included in the model in predictive studies, or as discriminative variable in diagnostic studies. For studies performed in CCS, no limits were set for sample size or age. General population studies were eligible if the sample size was roughly 250 or larger and if 75% or more of this population was below 65 years of age, as they have comparable levels of frailty to a young adult survivor population.^{5, 7, 8} We excluded studies with older adults since they are expected to have higher levels of frailty, comorbidities and aging factors, which may be confounders in the correlation between the newer biomarker and the metabolic syndrome. Multivariable analysis was mandatory for article inclusion of studies that investigated the prediction of MetS.

Studies were excluded if all included patients had an elevated biomarker; if all or none of the subjects had the MetS; if it was a selected cohort with pre-existing comorbidities (i.e. familial hypercholesterolemia, psoriasis, schizophrenia, polycystic ovary syndrome, obesity, hypertension); if all patients suffered from MetS or endpoint(s) such as T2DM, cardio- or cerebrovascular disease, or non-alcoholic fatty liver disease; if the article was a review, case study, expert opinion or conference abstract; if the article was written in a language other than English or Dutch, or if the full text was unavailable (see Appendix A for an overview of selection criteria). Studies were only included if the outcome was presence or absence of MetS; those with separate MetS components or MetS risk score as outcome were out of the scope of this review. After all articles were screened based on title and abstract, the judgements were unblinded. Discrepancies were discussed and resolved by the two reviewers (VP and SSvS) and where necessary, two senior experts were consulted (MMvdHE and SJCMMN). A cross-reference check was performed with Scopus, to screen all forward and backward citations of included studies. The articles found by the cross-reference check were screened likewise. A flow diagram with the number of in- and excluded articles and reasons for exclusion illustrates this process (Figure 1).

Risk of bias assessment

The QUIPS tool was applied for critical appraisal of predictor studies^{53, 54} (Supplementary Table 2) and QUADAS-2 tool for diagnostic studies (Supplementary Table 3). Definitions for low risk of bias judgement are shown in Appendix A. In case of doubt, the study was discussed with both reviewers and senior experts (VP, SSvS, MMvdHE, SJCMMN).

Data extraction enlisted novel biomarkers

Data of all included articles were extracted and summarized; the summaries of the enlisted newer biomarkers (adiponectin, leptin, uric acid, hsCRP, TNF-alpha, IL-1, IL-6, apoB and lp(a)) are depicted in Supplementary Table 4A-V. Data of interest are details regarding the size of the population and its type (survivors and their previous diagnosis, or general population), the study design (cross-sectional or longitudinal and retrospective or prospective), the biomarker (which and how it was measured), the exact outcome (MetS definition) and statistical analysis of choice. For studies investigating the diagnostic value of the biomarker for MetS, outcomes of interest were area under the curve (AUC) of receiver operating characteristic (ROC) curves, sensitivity and specificity. For the studies evaluating the predictive value of the

biomarker of later development of the MetS, odds ratios (OR's) or beta-coefficients of multivariable logistic regression models, or hazard ratios (HR's) from multivariable Cox Proportional Hazards analysis were extracted from the publications.

Summary of evidence

The Grading of Recommendations Assessment Development and Evaluation (GRADE) tool was applied to summarize the quality of the evidence for each biomarker, per clinical research question (diagnosing or predicting MetS) and per population (general population and CCS).⁵⁵ The level of evidence was classified as insufficient, very low, low, moderate and high (Supplementary Table 4).⁵⁵ The applied thresholds for biomarkers are shown in Supplementary Table 5. An overview was made for studies assessing the same independent variables and outcome (Supplementary Table 6).

Data extraction non-enlisted biomarkers

As secondary objective we screened all articles for other biomarkers than the above enlisted nine biomarkers of our main interest (non-enlisted biomarkers). Details are discussed in Part 2 of the Appendix . These non-enlisted biomarkers were evaluated for presence of an effect if there were 4 or more publications with this biomarker in our search. As we did not search for these biomarkers systematically, evidence quality was not assessed with GRADE.

Meta-analysis

A meta-analysis was performed of relevant enlisted biomarkers with at least three publications on the same outcome measures and, if applicable, adjusted for the same covariates. Dichotomous outcomes were considered as comparable if the applied threshold differed less than the intra- and inter-assay variability for the biomarker as reported in literature. A random effects model with inverse variance weighting was used to estimate a pooled overall outcome measure. Overall heterogeneity (I-squared) and between-study variance (tau-squared) were calculated.⁵⁶ Meta-analysis was performed with the package *meta* in R.⁵⁷

RESULTS

Study selection

As shown in the flow chart (Figure 1), the literature search in PubMed and Embase yielded a total of 4,510 unique records. After title and abstract screening, 650 full-text articles were reviewed, after which 162 relevant studies remained. Backward and

forward citation searching identified 18 additional studies. Hence, a total of 180 studies were identified that reported on the diagnostic and/or predictive value of one or more of the enlisted nine biomarkers of interest. Only five studies among the 180 were performed among a population of CCS.⁵⁸⁻⁶² All other studies were performed in the general population.

Among 180 studies which included data regarding the 9 enlisted biomarkers, 60 also reported the value of other, non-enlisted newer biomarkers. Furthermore, we identified 119 other studies that only investigated non-enlisted newer biomarkers (other than the nine of our main interest), yielding a total of 179 studies for our secondary objective.

A detailed description of the critical appraisal of each of the 180 included studies for the nine predefined biomarkers is provided in the supplementary material (Supplementary Table 2 and 3).

Used metabolic syndrome definitions

In the included studies, a variety of MetS definitions was used of which the most common are described in Table 1 and the applied definition per study is depicted in Supplementary Table 4. The applied biomarker thresholds are summarized in Supplementary Table 5.

Evidence for newer, enlisted biomarkers as (additional) diagnostic criterion for metabolic syndrome

Twenty-nine studies reported on the diagnostic value of one or more of the nine enlisted newer biomarkers. These were all performed in the general population without a history of cancer. Six studies had a Caucasian study population.⁶³⁻⁶⁸ The number of studies per biomarker ranged between zero [IL-1 and lp(a)] and twelve (adiponectin). The biomarker studied in the largest total number of participants was uric acid (73,190 participants). The relevant data extracted from each study, as well as the summary of evidence scored with the GRADE tool for each biomarker, are provided in the supplementary material (Supplementary Table 4). For each biomarker, a description of the number of studies and participants, and a summary of the several diagnostic outcomes, are provided in Table 2.

Whereas, ideally, the additional diagnostic value of a biomarker would be tested by comparing the AUC, sensitivity and specificity for a model containing only relevant covariates, versus a model containing covariates and the newer biomarker, this method was used in only two of the 29 studies.^{65, 69} One study compared the AUC of the biomarker with the AUC of waist circumference.⁶⁹ Most studies, however, only

reported the AUC of the biomarker, either unadjusted or adjusted for age, sex, and sometimes body mass index (BMI) or waist circumference. Therefore, interpretation of the additional value is limited by detection and confounding bias for most of the biomarkers.

The overall summary of our findings, with a conclusion about the diagnostic value of each biomarker in the general population and in survivors based on the GRADE assessment, is shown in Figure 2. Of the nine investigated biomarkers, four were identified as valuable diagnostic biomarkers for MetS: leptin (high quality of evidence), uric acid (moderate quality), adiponectin, and hsCRP (both low quality). In addition, apoB may be valuable, although based on only one study with moderate quality of evidence. TNF-alpha and IL-6 appeared to be unusable, based on one low quality study testing both biomarkers. For IL-1 and lp(a), no studies were found.

Evidence for newer, enlisted biomarkers as independent predictor of metabolic syndrome

In total, 162 general population studies, and 5 survivor studies [two in acute lymphoblastic leukemia (ALL) survivors, two in survivors of hematological malignancies, one in survivors of heterogeneous tumors],⁵⁸⁻⁶² investigated the role of one or more of the nine enlisted, newer biomarkers as independent predictors of MetS. Twenty-six of the general population studies had a Western/Caucasian study population.^{65, 67, 68, 70-92} The number of general population studies per biomarker ranged between 3 (TNF-alpha, 1,458 participants in total) and 78 (uric acid, 447,559 participants in total). Two of the survivors studies had a Western/Caucasian study population,^{58, 59} the others were performed in Japan,⁶⁰ Malaysia,⁶¹ and Mexico.⁶² The number of survivors studies per biomarker ranged between zero [IL-1, apoB, and lp(a)] and 3 (adiponectin and leptin). The biomarker studied in the largest total number of survivors was uric acid (390 survivors). The relevant data extracted from each study, as well as the summary of evidence scored with the GRADE tool for each biomarker, are provided in the supplementary material (Supplementary Table 4). For each biomarker, a description of the number of studies and participants, and a summary of the several prognostic outcomes, are provided in Table 2.

A common analysis strategy in these studies was to divide the biomarker value in quantiles, with thresholds that may differ per study. Not all participants in the highest or lowest quantile always had a biomarker value that would be classified as abnormal according to reference values. This may attenuate its value in predicting MetS. On the other hand, this bias towards the null hypothesis increases the effect of true positive findings. Also, several studies tested a dose-response effect by comparing the effect on MetS across the quantiles. Studies can be compared on whether a dose-response effect was observed or not.

Figure 2 shows the overall summary of our findings, consisting of a conclusion about the role of each biomarker as independent predictor of MetS in the general population and in survivors, after GRADE assessment. Five biomarkers were identified as independent predictors of MetS in the general population: uric acid, adiponectin, hsCRP, apoB (all high quality of evidence), and leptin (moderate quality). There is conflicting evidence for the value of TNF-alpha, IL-1, IL-6, and lp(a) (very low quality of evidence). Among survivors, uric acid and hsCRP may be valuable as prognostic biomarkers, based on two and one studies, respectively, with very low quality of evidence. There is conflicting evidence for the prognostic value of adiponectin and leptin (very low quality). TNF-alpha and IL-6 appear not to be independent predictors, based on one very low quality study testing both biomarkers. For IL-1, apoB, and lp(a), no studies were found.

Meta-analysis of most relevant findings of enlisted biomarkers

We aimed to perform a meta-analysis of the most promising biomarkers: uric acid, adiponectin, leptin, hsCRP, and apoB. For diagnostic studies, only the AUC is suitable for meta-analysis, due to different thresholds used for sensitivity and specificity (Supplementary Table 6). For predictor studies, only dichotomous and continuous (per unit, or per unit log-transformed) studies are useful. Many studies use quantiles but these are unsuited for meta-analysis: cut-offs between the quantiles depend on the range and distribution in each study population, and are therefore insufficiently comparable between studies to perform a meta-analysis.

A wide variety of outcome measures was used in the studies, and many studies performed an analysis that was unsuited for meta-analysis. Also, there was variance in thresholds used for dichotomous outcomes, as well as in covariates in multivariable models. Therefore, we were unable to retain at least three sufficiently comparable studies for most biomarkers, and for most outcomes, in order to perform a metaanalysis. For a few biomarkers, enough studies were eligible for meta-analysis, because the authors also published crude outcomes, and outcomes that were only age- and sex-adjusted (Supplementary Table 6).

We were able to perform a meta-analysis for the prognostic value of uric acid (hyperuricemia and continuous uric acid levels), and for the diagnostic value of hsCRP. We estimated the pooled OR for the association between hyperuricemia and MetS, adjusted for age and sex (four studies,⁹³⁻⁹⁶ with threshold variability accepted of 10%,⁹⁷ OR 2.94, 95%CI 2.08-4.15), the pooled OR per unit increase in uric acid,

unadjusted (three studies,^{90, 98, 99} OR 1.086, 95%CI 1.066-1.106), and the pooled AUC for hsCRP, also unadjusted (three studies,^{91, 100, 101} AUC 0.71, 95%CI 0.67-0.74).^{90, 99} Forest plots are shown in Figure 3. Unfortunately, many studies could not be included, and the reported estimators are not adjusted for relevant covariates, in particular age and sex for some, and overweight, insulin resistance, and smoking for all.

Other, non-enlisted biomarkers

In Supplementary Table 7, 179 articles for all other biomarkers for diagnosis or prognosis of MetS are enlisted and the main data is summarized. These included ratios of our studied biomarkers. All studies investigating leptin/adiponectin ratio as prognostic^{62, 68, 73, 102-105} or diagnostic study^{68, 102, 104-107} showed a possible relevance. Apolipoprotein A1 (apoA1) and apoB/apoA1 ratio seem valuable in predicting the MetS (6 studies with a protective effect of apoA1^{82, 86, 87, 108, 109}, and 8 studies with an effect of increasing risk of increasing apoB/apoA1 ratio.^{108, 111} Other recurrently reported, potentially useful biomarkers were Gamma GT, (non-high sensitivity) CRP, ferritin, leukocyte count, hemoglobin and urine pH and sodium excretion.

DISCUSSION

This is the first systematic literature review investigating newer biomarkers for metabolic syndrome (MetS) in CCS, with the aim to obtain the highest level of evidence by including validated tools for risk of bias assessment and summary of evidence, and by performing a meta-analysis.

For five biomarkers, numerous studies with moderate to high quality of evidence were found for diagnosing and predicting MetS: uric acid, adiponectin, leptin, hsCRP, and apoB. The evidence was not sufficient to confirm the value of candidate biomarkers lp(a), IL-1, IL-6, and TNF-alpha.

Meta-analysis of eligible studies showed a predictive value of uric acid for MetS, with a positive association, and a diagnostic value for hsCRP.

These findings suggest that uric acid, adiponectin, leptin, hsCRP, and apoB may be used in a screening setting for CCS, in addition to standard MetS criteria, in order to provide better diagnosis and prediction of MetS (risk). Systematic reviews in other populations have identified not only elevated leptin,¹¹⁵ uric acid,¹¹⁵⁻¹¹⁸ and low

(HWM) adiponectin, $^{\rm 115,\ 119,\ 120}$ but also Il-6 $^{\rm 115}$ and TNF-alpha $^{\rm 115}$ as potential MetS biomarkers.

As anticipated, the number of publications for survivors on this topic was rather limited: we identified only five studies in CCS specifically, which found a possible predictive value for hsCRP and uric acid, and conflicting or no evidence for the value of adiponectin, leptin and TNF-alpha. Disadvantages of these survivor studies were low patient numbers and moderate to high (detection and confounding) bias risk. No studies investigated the diagnostic value of newer biomarkers. Survivor studies with information on altered biomarker values but no direct comparison between biomarker and MetS occurrence, were excluded.^{60, 121-130} We expected to miss many relevant studies when designing the study, if we based our conclusions only on survivor studies. Therefore, evidence in the younger general adult population without childhood cancer history was included in our search as well, leading to 175 general population studies with relevant data which were generalizable to young adult survivors.

CCS can have an increased risk to develop MetS, in particular after treatment with cranial and/or abdominal radiotherapy, intensive chemotherapy, nephrectomy, adrenalectomy, or stem cell transplantation.^{43, 131-139} These therapies can lead to several underlying conditions that can increase the risk for (components of) MetS, such as hypothalamic damage, growth hormone deficiency, pancreatic beta cell dysfunction, hypogonadism, hypothyroidism, and altered body composition with increased abdominal fat.^{43, 131-139}

Furthermore, it is well acknowledged, that in CCS the biological age progresses faster than their true age, as can be derived from their high level of frailty.⁴⁻⁹ Previous studies have shown, that the physiologic reserve of CCS with a median age of 33 is similar to that of adults in the general population who are aged 65 years.⁶ For this reason, we included studies investigating biomarkers for MetS in the general population, with >75% of participants aged below 65 years, as may be very well applicable to CCS. We excluded studies investigating MetS biomarkers among elderly people on purpose, since they have an even higher level of frailty than CCS, comorbidities and aging factors, which may be confounders in the association between the newer biomarker and metabolic syndrome. We considered that extrapolating conclusions from a general elderly population to CCS could draw invalid conclusions . Based on this approach, all available literature applicable to survivors is now discussed in this review, as it includes both survivor studies as well as all generalizable data from a reasoned selection of the general population studies.

On the other hand, several studies excluded people with certain chronic illnesses.^{73, 101, 140-150} This may limit applicability of results to the population of CCS, in which the prevalence of comorbidities is high.^{3, 25, 126, 151} This was taken into account when scoring the risk of bias. Additionally, childhood cancer (treatment) related long-term side effects, such as altered fat distribution, sarcopenic obesity, and hormonal disbalances, may play a survivor specific role in the pathogenesis of MetS;¹¹ development of future studies that apply the use of biomarkers in large cohorts of CCS is therefore important.

Due to differences in study designs and statistical analyses, a wide variety of outcome measures was used. There was also substantial diversity in follow-up time in longitudinal studies. By employing the GRADE tool for summarizing evidence, we were able to draw conclusions for each biomarker from this heterogeneity of results. The meta-analysis was based on few studies, as many studies could not be included. Also, heterogeneity was high in the meta-analysis on uric acid per unit increase, as the study of Liu et al. had a remarkably higher OR than the other two studies.

Furthermore, although the ability of different MetS definitions to predict diabetes and CVD appears to be similar,^{152, 153} the use of different definitions (Table 1) can lead to differences in occurrence of MetS. There are subtle differences between the definitions that were mostly used in the included studies (Table 1). The potential consequence of choice of definition is illustrated by studies that tested the biomarker use in diagnosing or predicting MetS according to multiple definitions, and sometimes found different results depending on the definition used.^{67, 143, 154, 155} Therefore, comparing different studies and interpreting results of the meta-analysis requires some caution, as a full comparison of the studies is often not possible.

Adiposity, and hence the MetS, can be underdiagnosed in survivors, due to altered body composition after radiotherapy, stem cell transplantation, or amputations. For clinical applicability to survivors, it is important that newer biomarkers play an independent role in MetS, and measurement of newer biomarkers is only useful when their effect is not yet captured by established MetS components. Therefore, we did not investigate routine dyslipidemia and insulin resistance markers in our search (e.g., LDL, HOMA-IR). Although apoB and lp(a) are also lipid markers, they are of interest because they are better predictors of atherogenicity than triglycerides, HDL and LDL – particularly apoB, because it gives an estimate of the total number of circulating atherogenic particles.¹⁵⁶⁻¹⁵⁸

In this light, it is also favorable that studies adjust for MetS components, such as adiposity and insulin resistance, in order to adjust for potentially major correlations

and interactions,¹⁵⁹⁻¹⁶² and to yield the independent/additional diagnostic and predictive value of the biomarker. Furthermore, it remains important to evaluate other traditional risk factors, including smoking, physical activity, socio-economic status, and family history⁷⁸.¹⁶³ In addition, genetic profile may still be relevant for MetS risk, although so far this is not included in standard screening.¹⁶⁴⁻¹⁶⁶ Risk of detection and confounding bias remains high, especially in the diagnostic studies, as many studies did not adjust for MetS components and traditional risk factors. In particular for the diagnostic studies, a risk of (detection and) confounding bias remained.

The MetS is defined as a cluster of symptoms such as obesity, hypertension, impaired glucose tolerance and dyslipidemia.¹¹ These clustered symptoms are related to each other: an imbalance in energy intake and consumption causes a cascade of increased (visceral) adiposity, increased circulating free fatty acids and decreased adiponectin (which causes also an increase in insulin resistance), and high levels of pro-inflammatory and pro-thrombotic mediators, such as TNF-alpha, IL-1 and IL-6.^{11, 34} Insulin resistance is associated with a lowered excretion of uric acid by the kidneys, and higher uric acid production.^{167, 168} The adipokines leptin and adiponectin are produced by adipocytes.¹⁶⁹ Low leptin values trigger metabolic, behavioral and endocrine responses that aim at a preservation of the fuel reserves of the body.¹⁷⁰ Adiponectin enhances insulin sensitization and suppresses inflammation and cell death.^{170, 171, 172, 173} Another important molecule is apoB: all atherogenic lipoproteins carry one single apoB molecule as their structural protein, and therefore apoB represents the atherogenic burden.¹⁷⁴ Serum apoB is a strong predictor of cardiovascular risks^{156, 175, 176} and comes in as an important player for the MetS in this review as well. One of the low density lipoproteins carrying an apoB molecule, is Lp(a).¹⁷⁷ The interpretation of Lp(a) values in an individual can be difficult due to a high heterogeneity and wide distribution of Lp(a) concentrations.¹⁷⁸ Although evidence for relevance of Lp(a) for MetS evaluation in survivors was unavailable, it remains a marker of interest, since elevated Lp(a) levels were an independent predictor for cardio- and cerebrovascular outcomes¹⁷⁹⁻¹⁸⁷ and were inversely associated with T2DM.¹⁸⁸

An important inflammatory marker is (hs) CRP, which is synthesized by hepatocytes^{189,190} in response to infection, inflammation, tissue damage and malignant neoplasia.^{189,190} CRP binds to LDL^{189,191} and may have a causal role in atherogenesis,¹⁸⁹ as it is present in atherosclerotic plaques.^{189,192} Inflammatory markers may reflect a transient state instead of chronic state of inflammation.¹⁹³ Still, in the study of Oda et al., the diagnostic value of hsCRP was reproducible when the measurement was repeated after one year.¹⁹⁴ Many studies had a high CRP,^{70,101,195,196} or infection^{100,197-200} as exclusion criterion. Regarding inflammation; smooth muscle cells, endothelial cells and macrophages produce cytokines such as IL-1 and IL-6²⁰¹⁻²⁰³ in reaction to metabolic stress,^{203, 204} by other inflammatory mediators such as interferon-gamma and TNF, and cholesterol itself.²⁰³ Still, the evidence for the usefulness as marker for the MetS is rather limited.

Due to the systemic nature of MetS, our secondary objective to reveal other interesting biomarkers yielded many markers. Interesting markers for further research include Gamma GT, ferritin, leukocytes and hemoglobin. In several studies biomarkers were related to each other, as MetS components are related as well.²⁰⁵ In one study, leptin was inversely associated with uric acid excretion;²⁰⁵ in another, a synergistic effect between hsCRP and high molecular weight adiponectin was found.²⁰⁶ Also, ratios of biomarkers (e.g. leptin/adiponectin, apoB/apoA1) include extra information and may be better diagnostic or prognostic agents than single biomarkers. Future studies may investigate the value of combining biomarkers.

Some limitations are present in this systematic literature.

Many of the included studies had a cross-sectional design, which is suboptimal to investigate causality; this was taken into account for the GRADE and level of evidence. Some authors conducted prospective longitudinal studies^{81, 193, 207, 208} and associated MetS risk at end of follow-up with baseline and/or change in biomarker level. Study designs even more suitable for determining prediction and causality include prediction models and Mendelian randomization.²⁰⁹⁻²¹² These study designs require more time and financial resources, and large cohorts. These types of studies where either not performed or unsuitable for our research question.

Many studies were performed among Asian cohorts. Asian people are more susceptible to insulin resistance,^{213, 214} which is accounted for in lower waist circumference thresholds. Additionally, there may be an ethnicity specific component in the relationship between biomarker and MetS.²¹⁵⁻²²² This may limit the applicability to a Caucasian population.

For this literature study, we focused on diagnosis and prediction of the full MetS; other outcomes such as resolution of the MetS,²²³ components of the MetS, CVD or T2DM were out of scope.^{186, 224-235} Therefore, our findings do not provide a complete overview of the use of the newer biomarkers in diagnosing and predicting cardiovascular risk factors in CCS.

We have two suggestions for future research that are relevant for the implementations of our findings in the follow-up of CCS. The newer biomarkers could be added as a sixth criterion for MetS. This application can be especially of value in cases of doubt of MetS diagnosis for individuals who had abdominal irradiation: it may be valuable to replace waist circumference with the adipokines leptin or adiponectin. This may identify MetS in more survivors, and can potentially improve the predictive ability for T2DM and CVD.²³¹

An important requirement for the applicability of these newer biomarkers in such a screening setting for MetS (risk) in CCS, is the determination of a threshold. For uric acid, this is relatively well-established (Supplementary Table 5); for other biomarkers, this is less clear, as is illustrated by the range of applied thresholds (Supplementary Table 5). This is partly because of the use of different assays and testing of subfractions of a biomarker, such as high molecular weight adiponectin. Also, a tradeoff between sensitivity and specificity may influence the determination of an optimal threshold.

In conclusion, based on this systematic literature search, we suggest to consider the additional use of uric acid, adiponectin, hsCRP, leptin, and apoB in the screening setting for metabolic syndrome in CCS. As our conclusions are largely based on general population studies, studies in CCS are needed. Furthermore, future studies may specifically test the use of newer biomarkers as additional MetS components, and define optimal thresholds. The addition of one or more of these newer biomarkers as a criterion for MetS may lead to a newer and better classification and enhanced identification of risk of developing T2DM and CVD, especially in CCS in whom components are difficult to evaluate in the currently applied definitions. Early intervention can delay or prevent complications, and hence improve very long-term survival outcomes and quality of life.

Funding:

This work was supported by the Erasmus MC, Rotterdam, the Netherlands, and Princess Máxima Center for Pediatric Oncology, Utrecht, the Netherlands.

Disclosure Statement:

Mr. Pluimakers has nothing to disclose.Mrs. van Santen has nothing to disclose.Prof. dr. Fiocco has nothing to disclose.Mrs. Bakker has nothing to disclose.Prof. dr. van der Lelij has nothing to disclose.Prof. dr. van den Heuvel-Ebrink has nothing to disclose.Associate prof. dr. Neggers has nothing to disclose.

Table 1. Commonly	used metabolic syndron	ne definitions in selected	l studies				
	NCEP ATP III	IDF 2006	Joint interim statement/ Harmonized definition	Japanese Obesity Society	Chinese Diabetes Society	Children and adolescents	Modified with BMI instead of waist circumference
Required for MetS diagnosis	3 or more criteria	Obesity plus 2 or more criteria	3 or more criteria	Obesity plus 2 or more criteria	3 or more criteria	3 or more criteria	
Obesity	Waist circumference >102cm (men) or >88cm (women)	Waist circumference >90cm (men) or >80cm (women)	Waist circumference with ethnic-specific thresholds	Waist circumference >85cm (men) or >90cm (women)	Body mass index ≥25kg/m²	Waist circumference ≥90th percentile	Body mass index ≥30 (Caucasians) or ≥25 (Asians) kg/m²
Insulin resistance	Fasting plasma glucose ≥5.6 mmol/L or treatment	Fasting plasma glucose ≥5.6 mmol/L or treatment	Fasting plasma glucose ≥5.6 mmol/L or treatment	Fasting plasma glucose ≥6.1 mmol/L or treatment	Fasting plasma glucose ≥6.1 mmol/L or treatment	Fasting plasma glucose ≥5.6 mmol/L or treatment	D
Dyslipidemia	Triglycerides ≥1.7mmol/L or treatment	Triglycerides ≥1.7mmol/L or treatment	Triglycerides ≥1.7mmol/L or treatment	Triglycerides ≥1.7mmol/L, or HDL cholesterol <1mmol/L (men) or <1.3mmol/L (women), or treatment	Triglycerides ≥1.7mmol/L, or HDL cholesterol <0.9mmol/L (men) or <1.0mmol/L (women), or treatment	Triglycerides ≥1.7mmol/L or treatment	
	HDL cholesterol <1 mmol/L (men) or <1.3 mmol/L (women) or treatment	HDL cholesterol <1 mmol/L (men) or <1.3mmol/L (women) or treatment	HDL cholesterol <1mmol/L (men) or <1.3mmol/L (women) or treatment			HDL cholesterol <1mmol/L or treatment	
Hypertension	≥130/85mmHg or treatment	≥130/85mmHg or treatment	≥130/85mmHg or treatment	≥130/85mmHg or treatment	≥140/90mmHg or treatment	≥130/85mmHg or treatment	

TABLES AND FIGURES

Summary of outcomes in diagnostic studies					
Biomarker	Total number of studies and	Outcome	Number of	Range	
Leptin, in general population	participants 6 studies, 8,209 participants ^{68, 102, 106, 154, 236, 237}	AUC	studies 5 ^{68, 102, 106, 236, 237}	0.68-0.93	
Uric acid, in general population	9 studies, 73,190 participants ^{66,101,150,237-242}	Sensitivity Specificity AUC	$\begin{array}{c} 3^{102,\ 154,\ 237}\\ 3^{102,\ 154,\ 237}\\ 7^{66,\ 101,\ 150,\ 237,}\\ 239,\ 240,\ 242\end{array}$	48.0-92.6% 56.3-72.0% 0.56-0.85	
Adiponectin, in general population	12 studies, 21,888 participants ^{63, 65, 67-69, 102, 106, 140, 143, 243-245}	Sensitivity Specificity AUC	$3^{237-239} \\ 3^{237-239} \\ 12^{63, 65, 67-69, } \\ {}^{102, 106, 140, 143, } \\ 243-245$	38.0-76.0% 56.0-85.0% 0.55-0.92	
hsCRP, in general population	7 studies, 18,211 participants ^{64, 91, 100, 101, 208, 246,}	Sensitivity Specificity AUC	$2^{102, 243} \\ 2^{102, 243} \\ 6^{64, 91, 100,} \\ {}_{101, 208, 247}$	64.7-69.3% 56.0-66.0% 0.55-0.74	
	247	Sensitivity Specificity	$3^{208, 246, 247} \\ 3^{208, 246, 247}$	51.0-69.0% 56.6-72.0%	
ApoB, in general population TNF-alpha, in general population	1 study, 8,120 participants ¹¹¹ 1 study, 976 participants ⁶⁴	AUC AUC	1^{111} 1^{64}	0.68 0.54	
IL-6, in general population IL-1 and lp(a), in general	1 study, 976 participants ⁶⁴ No studies	AUC n.a.	1 ⁶⁴ n.a.	0.56 n.a.	
All biomarkers, in survivors	No studies	n.a.	n.a.	n.a.	

Table 2.	Summary	of outcomes
----------	---------	-------------

Summary of outcomes in prognostic studies Number of studies Range Biomarker Total number of studies Outcome and participants $\underset{254,\,257,\,258,\,264,\,266,\,267,\,274,}{21}$ Uric acid, 78 studies, 447,559 participants ^{71, 72, 75, 77,} ^{82, 88, 90, 92-96, 98,} 1.00-5.17 OR dichotomous in general 280, 287, 296, 297, 304 population 99, 101, 150, 207, 210, $19^{90, \, 92, \, 98, \, 99, \, 101, \, 238, \, 249, }_{250, \, 255, \, 260, \, 262, \, 263, \, 265, }$ OR per unit 1.001-2.14 238, 239, 241, 242, 248-303 283, 285, 288, 289, 298 $2^{239, 302}$ OR per unit log-1.16, 2.08 transformed 24^{72, 95, 251, 253, 256-259, 261, 268, 271, 272, 275, 277-279,} OR highest quantile 1.00-8.04 281, 282, 290-292, 296, 301, 303 5^{207, 239, 242, 270, 300} HR dichotomous 1.06-2.99 4241, 285, 286, 294 HR per unit 1.10-2.35 3^{207, 210, 295} HR per SD 0.86-1.36 8207, 241, 252, 269, 286, HR highest quantile 0.74-3.47 293-295 $2^{207, 285}$ 1.05, 1.31 HR per unit longitudinal increase 1^{276} RR per unit log-7.25 for men, 13.26 transformed for women 1^{71} RR per SD 1.10 1^{88} RR per 1.4mg/dl 1.54 for men, 1.82 for women $2^{71, 299}$ RR highest quantile 1.69, 1.76 275, 284 PR 1.47, 2.10

IRR

 1^{75}

1.73

Table 2. Contin	ued			
Summary of ou	tcomes in prognostic studi	ies		
Biomarker	Total number of studies	Outcome	Number of studies	Range
	and participants		. 50	
Uric acid, in	2 studies, 390	MetS prevalence in uric	155	28.5% vs 12.5%
survivors	survivors	acid Q4 vs Q1-3	. 50	(p=0.0044)
		MetS component(s)	150	60% vs 24%
		prevalence high vs low		(p=0.04)
		uric acid	278 103	0.00.0.0
Adiponectin,	38 studies, 56,656	OR dichotomous (low	2/0,105	0.90, 2.68
in general	participants 65, 67, 69, 70, 73, 74,	adiponectin)	067. 73. 76. 141.	0.66.1.00
population	76, 78, 81, 102, 103,	OR per unit	144, 148, 159, 309, 314	0.66-1.08
	105, 140-148, 159, 245, 305-319		1 69	0.02.0
		OR per 5 units	1	0.82 for men, 0.90
		0.0	74, 146, 245, 306	for women
		OR per unit log-	4	0.10-0.6/
		OR non SD	3 315, 317	0.50.0.01
		OR per SD	1 ¹⁰²	0.30-0.91
		transformed 7 second	1	0.76 for boys, 0.09
		OD bishest suggit	1 269, 105, 140, 142, 143, 145,	
		OR nignest quantile	15 147, 305, 308, 315, 317-319	0.10-0.6/
		OR lowest quantile	665, 81, 307, 312, 313, 316	1.82-18.6
		HR high baseline and	1 ³¹¹	0.33
		increase during follow-		
		up vs low baseline and		
		decrease		
		HR decreased at	1 ⁸¹	4.37
		follow-up		
		Time ratio of	1 ³¹⁰	0.15 (=85% shorter
		developing MetS Q1		time to develop
		vs Q4		MetS)
		Baseline ratio (value in	1 ⁷⁰	1.27
		MetS subjects divided		
		by value in non-MetS,		
		adjusted for covariates)		
Adiponectin, in	3 studies, 139	OR highest quantile at	162	0.5 (n.s.) for
survivors	survivors ^{58, 60, 62}	baseline and follow-up		baseline, 0.9 (n.s.) for follow-up
		HR dichotomous (low	158	6.7
		adiponectin)		
		P-value of Kruskal-	160	n.s.
		Wallis test median		
		adiponectin in 0, 1,		
		2-4 MetS components		

Summary of a	nucu	iec		
Biomarker	Total number of studies	Outcome	Number of studies	Range
Dioiliai Kei	and participants	Outcome	Trumber of studies	Range
heCRP	32 studies 119 138	OR dichotomous	2 155, 196	1 20 2 74
in general	participants ^{70, 74, 83-85,}	OR activition	2 790, 91, 198, 199, 249, 328, 330	1.20, 2.74
population	88, 90, 91, 147, 155, 193, 195-199,	OR per unit log	/ /74, 246, 265, 324	1.00/-2.9/
population	208, 246, 249, 265, 269, 320-330	transformed	7	1.1)-J.2
		OP per SD	185	1.21
		OR per SD loc	1 2208, 325	0.06 1.07
		transformed	2	0.90, 1.07
		OR highest quantile	11 ^{84, 147, 193, 197, 321-323, 325-327, 329}	1.07-7.11
		OR highest of three	3 ¹⁹⁵ , 320, 324	1 65-18 86
		groups (<1.0, 1.0-3.0	5	1.09-10.00
		LID por unit log	1 269	1 15
		rik per unit log-	1	1.1)
		DD 1 C11	1 88	1 1 2
		increase	1	1.15
		Baseline ratio	1^{70}	0.80 (n.s.)
		P-value of likelihood	1 ⁸³	n.s.
		test in multivariable		
		model		
hsCRP, in survivors	1 study, 87 survivors and 87 controls ⁶¹	OR dichotomous	161	7.26
ApoB, in	10 studies, 66,924	OR dichotomous	182	2.55
general	participants ^{74, 79, 82, 86,}	OR per unit	1 ⁷⁴	2.99
population	87, 108, 109, 331-333	OR per 30mg/dl	1 ⁸⁷	1.76 for men, 2.10
		0		for women
		OR per SD	1 ³³¹	1.56
		OR highest quantile	6 ^{79, 108, 109, 331-333}	0.96-6.03
		OR highest of three	1 ⁸⁶	2.69 for men, 1.69
		groups (<90, 90-119		for women
		and $\geq 120 \text{ mg/dl}$)		
		RR per SD	1 ³³¹	1.17 (n.s.)
		RR highest quantile	1 ³³¹	1.79
ApoB, in	No studies	n.a.	n.a.	n.a.
Leptin.	17 studies, 28.797	OR dichotomous	1^{103}	2.39
in general	participants 68, 73, 74,	OR per unit	4 ^{73, 148, 314, 336}	0.96-1.91
population	102, 103, 147-149, 236, 306,	OR per 10ng/ml	1 149	1.06 (adjusted
1 -1	314, 315, 319, 334-337		-	for WC), 1.22
				(adjusted for BMI)
		OR per unit log-	2 ^{74, 306}	1.47, 2.76
		transformed	-	
		OR per SD	3 ^{68, 315, 335}	1.01-1.31
		OR per unit log-	1 ¹⁰²	1.81 for boys 1.32
		transformed 7-score	-	for girls
		OR highest quantile	6 ^{147, 236, 315, 319, 334,}	1.16-3.02
		Or ingliest qualitie	337	1.10-3.02

Chapter 3

Table 2. Continued
Table 2. Continued					
Summary of ou	tcomes in prognostic stud	ies			
Biomarker	Total number of studies	Outcome	Number of studies	Range	
	and participants				
Leptin, in	3 studies, 139 survivors	OR highest quantile at	1 ⁶²	4.8 for baseline, 5.7	
survivors	58, 60, 62	baseline and follow-up		for follow-up	
		MetS component(s)	1 ⁵⁸	54% vs 17%	
		prevalence high vs low		(p=0.03)	
		leptin			
		P-value of Kruskal-	1^{60}	n.s.	
		Wallis test median			
		adiponectin in 0, 1,			
		2-4 MetS components	(7.100		
IL-6, in general	5 studies, 3,370	OR per unit	2 ^{67, 199}	0.98-1.47	
population	participants ^{07, 80,}	OR highest quantile	2 ^{145, 200}	0.98 (n.s.), 4.10	
	145, 199, 200	P-value in	1**	n.s.	
		multivariable model	(1		
IL-6, in	1 study, 87 survivors and	OR dichotomous	161	1.53 (n.s.)	
survivors	87 controls ⁶¹		- 320		
Lp(a), in	5 studies, 15,162	OR dichotomous	1 338	8.27	
general	participants	OR highest of three	1556	0.82 (n.s.)	
population		groups (<18.40, 18.40-			
		$33.84 \text{ and } \ge 33.85 \mu\text{g}/$			
		MI) OB non unit	1 339	1.0(n.s)	
		OR per unit	1 1 ³⁴⁰	0.45	
		UD highest quantile	1 1 ⁸⁹	1.01(n-1)	
$\mathbf{I}_{\mathbf{n}}(\mathbf{a})$ in	No studios	rik nignest quantile	1	1.01 (n.s.)	
Lp(a), in	ino studies	n.a.	n.a.	n.a.	
II 1 in conoral	1 studies 1 59/	OP per unit	2 199, 341	2.28 (II label)	
nopulation	participants ^{70, 199, 200, 341}	OK per unit	Z	2.20 (11-1 aprila),	
population				1.00), 2.01 (IL- 1beta)	
		OR highest quartile	1 200	0.98 (n s)	
		Baseline ratio	1 ⁷⁰	1 17 (suggests effect	
		Dasenne ratio	1	in other direction)	
TNF-alpha.	3 studies, 1,458	OR per unit	1 199	1.45	
in general	participants ^{80, 199, 200}	OR highest quartile	1 ²⁰⁰	0.78 (n.s.)	
population	1	P-value in	1 80	n.s.	
		multivariable model		······	
TNF-alpha, in	1 study, 87 survivors and	OR dichotomous	1 ⁶¹	0.52 (n.s.)	
survivors	87 controls ⁶¹				

Can biomarkers be used to improve diagnosis and prediction of metabolic syndrome? (...)



Figure 1 Flow chart of in- and excluded articles from the systematic literature search

	1 Leptin	Survivors	No diagnostic studies available	e Conflicting evidence in predictive studies (very low level of evidence)	General population	Diagnostic value (high level of evidence)	Independent predictor (moderate level of evidence)
	ل المعالم المحمد ال	Survivors	No diagnostic studies available	Conflicting evidence in predictive studies (very low level of evidence)	น้ำน้ำน้ำน้ำนี้ General population	Diagnostic value (low level of evidence)	Independent predictor (high level of evidence)
	Uric acid	Survivors	No diagnostic studies available	Possibly independent predictor (very low level of evidence)	AAAAAA General population	Diagnostic value (moderate level of evidence)	Independent predictor (high level of evidence)
	h sCRP	Survivors	No diagnostic studies available	Possibly independent predictor (very low level of evidence)	ትትትትትት General population	Diagnostic value (low level of evidence)	Independent predictor (high level of evidence)
	τ. TNF-α	Survivors	tudies available	No apparent predictive value (very low level of evidence)	Ceneral population	No diagnostic value (low level of evidence)	Conflicting evidence in predictive studies (very low level of evidence)
rkers	IL-6	Survivors	vo uagnostic studies available	No apparent predictive value (very low level of evidence)	AAAAAAA General population	No diagnostic value (low level of evidence)	Conflicting evidence in predictive studies (very low evidence)
me bioma	·•:•=	Survivors	No ulaynostic studies available	No predictive studies available	ด้ห้านี้ที่ที่ที่ General population	No diagnostic studies available	Conflicting evidence in predictive studies (very low level of evidence)
lic syndro	L _P (a)	Survivors	No diagnostic studies available	No predictive studies available	ค้ที่ที่ที่ที่ที่ General population	No diagnostic studies available	Conflicting evidence in predictive studies (very low level of evidence)
Metabo	ApoB	Survivors	No diagnostic studies available	No predictive studies available	슈슈슈슈슈 General population	Possible diagnostic value (moderate level of evidence)	Independent predictor (high level of evidence)

Figure 2 Summary of conclusions: predictive and diagnostic value of novel biomarkers for the MetS

Study	Beta	SE		Odd	ls Ra	tio		OR	95%-CI	Weight
Cheserek 2018	1.45	0.206					-	4.26	[2.84; 6.38]	25.4%
Moulin 2017	0.51	0.224				-:		1.66	[1.07; 2.57]	24.0%
Porchia 2018	1.28	0.414			12	18		3.60	[1.60; 8.10]	12.3%
Wei 2015	1.12	0.045				+		3.08	[2.82; 3.37]	38.3%
Random effects mod	del		_		_	-	•	2.94	[2.08; 4.15]	100.0%
Heterogeneity: I ² = 71%, 1	$r^{-} = 0.0784, p$	= 0.02	10							
			0.2	0.5	1	2	5			

A. Pooled results for odds ratio of hyperuricemia

B. Pooled results for odds ratio per unit increase in uric acid

Study	Beta	SE	Odds	Ratio	OR	95%-CI	Weight
Liu 2019	0.489	0.0188		+	1.630	[1.571; 1.691]	15.8%
Petrikova 2018	0.005	0.0008			1.005	[1.004; 1.007]	42.0%
Zhang 2017	0.007	0.0003			1.007	[1.006; 1.008]	42.1%
Random effects model					1.086	[1.066; 1.106]	100.0%
Heterogeneity: $I^2 = 100\%$, $\tau^2 =$	= 0.0002, p	< 0.01		1			
		0.75	1	1.5	2		

C. Pooled results for AUC of hsCRP

Study	AUC	SE	AUC	AUC	95%-CI	Weight
Chen 2019	0.68	0.018		0.68	[0.65; 0.71]	33.8%
Kawada 2012	0.73	0.008	+	0.73	[0.72; 0.74]	45.0%
Stefanska 2011	0.70	0.031		0.70	[0.64; 0.76]	21.2%
Random effects model Heterogeneity: $I^2 = 72\%$, $\tau^2 = 100$	0.0007, p	= 0.03 F		0.71	[0.67; 0.74]	100.0%
		0.5	5 0.6 0.7 0.8 0.9	1		

Figure 3 Forest plots for different study-specific outcomes.

A. Odds ratio (OR) for hyperuricemia. B. OR for per unit increase in uric acid. C. Area under the curve (AUC) of hsCRP. The sizes of the square boxes on the forest plots are proportional to the total number of patients in the selected trials

SUPPLEMENTARY MATERIAL

Part 1 Enlisted biomarkers

Textual desc	ription results enlisted biomarkers	Page 3-24
Table 1	Search terms	Page 25-27
Selection crit	teria title/abstract screening	Page 28
Risk of bias	assessment criteria	Page 29-30
Table 2	Risk of bias predictor studies (QUIPS tool)	Page 31-62
	Table 2A In the general population, Table 2B In survivors	
Table 3	Diagnostic studies in general population (QUADAS-2 tool)	Page 63-69
Table 4	Data-extraction and GRADE enlisted biomarkers	Page 70-304

Summary of evidence diagnostic and predictive value of biomarkers for MetS

Table 4A	Diagnostic value adiponectin general population	Page 72-82
Table 4B	Predictive value adiponectin general population	Page 83-120
Table 4C	Predictive value adiponectin survivors	Page 121-123
Table 4D	Diagnostic value leptin general population	Page 124-129
Table 4E	Predictive value leptin general population	Page 129-143
Table 4F	Predictive value leptin in survivors	Page 144-146
Table 4G	Diagnostic value uric acid general population	Page 147-155
Table 4H	Predictive value uric acid general population	Page 156-220
Table 4I	Predictive value uric acid survivors	Page 221-222
Table 4J	Diagnostic value hsCRP general population	Page 223-230
Table 4K	Predictive value hsCRP general population	Page 231-257
Table 4L	Predictive value hsCRP survivors	Page 258-260
Table 4M	Predictive value TNF-alpha general population	Page 261-266
Table 4N	Diagnostic value TNF-alpha general population	Page 267-268
Table 4O	Predictive value TNF-alpha survivors	Page 269-271
Table 4P	Predictive value IL-1 general population	Page 272-278
Table 4Q	Predictive value IL-6 general population	Page 279-285
Table 4R	Predictive value IL-6 survivors	Page 286-288
Table 4S	Diagnostic value IL-6 general population	Page 289-290
Table 4T	Diagnostic value apoB general population	Page 291-292
Table 4U	Predictive value apoB general population	Page 293-299
Table 4V	Predictive value Lp(a) general population	Page 300-304
Table 5	Applied thresholds for enlisted biomarkers	Page 305-307
Table 6	Overview applicability of studies for meta-analysis	Page 308-218

Part 2 Other biomarkers (not enlisted in advance)

List of other biomarkers	Page 319-327
Summary and brief discussion or results	Page 328-329
Table 7 Data-extraction	Page 330-661
Shortlist outcomes other biomarkers with 4 or more publications	Page 662-701
Abbreviations	Page 702
References	Page 703-723

The supplementary material is available online: https://onlinelibrary.wiley.com/doi/full/10.1111/obr.13312

REFERENCES

- Howlader N, Noone, A.M., Krapcho, M., Miller, D., Bishop, K., Kosary, C.L., Yu, M., Ruhl,, J. T, Z., Mariotto, A., Lewis, D.R., Chen, H.S., Feuer, E.J., Cronin, K.A. SEER Cancer Statistics Review (CSR) 1975-2014, . National Cancer Institute.: Bethesda, MD 2017.
- 2 Gibson TM, Mostoufi-Moab S, Stratton KL, Leisenring WM, Barnea D, Chow EJ, *et al.* Temporal patterns in the risk of chronic health conditions in survivors of childhood cancer diagnosed 1970-99: a report from the Childhood Cancer Survivor Study cohort. *The Lancet Oncology*. 2018; 19: 1590-601.
- 3 Hudson MM, Ness KK, Gurney JG, Mulrooney DA, Chemaitilly W, Krull KR, *et al.* Clinical ascertainment of health outcomes among adults treated for childhood cancer. *Jama.* 2013; 309: 2371-81.
- 4 Ness KK, Howell CR, Bjornard KL. Frailty and quality of life in adult survivors of childhood cancer. *Expert review of quality of life in cancer care*. 2017; 2: 79-85.
- 5 Ness KK, Armstrong GT, Kundu M, Wilson CL, Tchkonia T, Kirkland JL. Frailty in childhood cancer survivors. *Cancer*. 2015; 121: 1540-7.
- 6 Ness KK, Kirkland JL, Gramatges MM, Wang Z, Kundu M, McCastlain K, *et al.* Premature Physiologic Aging as a Paradigm for Understanding Increased Risk of Adverse Health Across the Lifespan of Survivors of Childhood Cancer. *Journal of clinical oncology : official journal of the American Society of Clinical Oncology.* 2018; 36: 2206-15.
- 7 Ness KK, Wogksch MD. Frailty and aging in cancer survivors. *Translational research : the journal of laboratory and clinical medicine*. 2020; 221: 65-82.
- 8 Armenian SH, Gibson CJ, Rockne RC, Ness KK. Premature Aging in Young Cancer Survivors. *Journal of the National Cancer Institute*. 2019; 111: 226-32.
- 9 Vatanen A, Hou M, Huang T, Söder O, Jahnukainen T, Kurimo M, *et al.* Clinical and biological markers of premature aging after autologous SCT in childhood cancer. *Bone marrow transplantation.* 2017; 52: 600-05.
- 10 Bhakta N, Liu Q, Ness KK, Baassiri M, Eissa H, Yeo F, *et al.* The cumulative burden of surviving childhood cancer: an initial report from the St Jude Lifetime Cohort Study (SJLIFE). *Lancet* (*London, England*). 2017; 390: 2569-82.
- Pluimakers VG, van Waas M, Neggers S, van den Heuvel-Eibrink MM. Metabolic syndrome as cardiovascular risk factor in childhood cancer survivors. *Critical reviews in oncology/hematology*. 2019; 133: 129-41.
- 12 Reulen RC, Winter DL, Frobisher C, Lancashire ER, Stiller CA, Jenney ME, *et al.* Long-term cause-specific mortality among survivors of childhood cancer. *Jama*. 2010; 304: 172-9.
- 13 Tukenova M, Guibout C, Oberlin O, Doyon F, Mousannif A, Haddy N, *et al.* Role of cancer treatment in long-term overall and cardiovascular mortality after childhood cancer. *Journal of clinical oncology : official journal of the American Society of Clinical Oncology.* 2010; 28: 1308-15.
- 14 Castellino SM, Geiger AM, Mertens AC, Leisenring WM, Tooze JA, Goodman P, *et al.* Morbidity and mortality in long-term survivors of Hodgkin lymphoma: a report from the Childhood Cancer Survivor Study. *Blood.* 2011; 117: 1806-16.
- 15 Prasad PK, Signorello LB, Friedman DL, Boice JD, Jr., Pukkala E. Long-term non-cancer mortality in pediatric and young adult cancer survivors in Finland. *Pediatric blood & cancer*. 2012; 58: 421-7.
- 16 Armstrong GT, Oeffinger KC, Chen Y, Kawashima T, Yasui Y, Leisenring W, *et al.* Modifiable risk factors and major cardiac events among adult survivors of childhood cancer. *Journal of clinical oncology : official journal of the American Society of Clinical Oncology.* 2013; 31: 3673-80.

- 17 Perkins SM, Fei W, Mitra N, Shinohara ET. Late causes of death in children treated for CNS malignancies. *Journal of neuro-oncology*. 2013; 115: 79-85.
- 18 Kero AE, Järvelä LS, Arola M, Malila N, Madanat-Harjuoja LM, Matomäki J, et al. Cardiovascular morbidity in long-term survivors of early-onset cancer: a population-based study. *International journal of cancer*. 2014; 134: 664-73.
- 19 van Laar M, Feltbower RG, Gale CP, Bowen DT, Oliver SE, Glaser A. Cardiovascular sequelae in long-term survivors of young peoples' cancer: a linked cohort study. *British journal of cancer*. 2014; 110: 1338-41.
- 20 Olsen M, Schmidt M, Lash TL, Sorensen K, Pedersen L, Sorensen HT. Cardiovascular disease risk in childhood cancer survivors. *American journal of epidemiology*. 2014; 180: 120-3.
- 21 Kero AE, Järvelä LS, Arola M, Malila N, Madanat-Harjuoja LM, Matomäki J, *et al.* Late mortality among 5-year survivors of early onset cancer: a population-based register study. *International journal of cancer.* 2015; 136: 1655-64.
- 22 Gudmundsdottir T, Winther JF, de Fine Licht S, Bonnesen TG, Asdahl PH, Tryggvadottir L, *et al.* Cardiovascular disease in Adult Life after Childhood Cancer in Scandinavia: A populationbased cohort study of 32,308 one-year survivors. *International journal of cancer.* 2015; 137: 1176-86.
- 23 Bhakta N, Liu Q, Yeo F, Baassiri M, Ehrhardt MJ, Srivastava DK, *et al.* Cumulative burden of cardiovascular morbidity in paediatric, adolescent, and young adult survivors of Hodgkin's lymphoma: an analysis from the St Jude Lifetime Cohort Study. *The Lancet Oncology*. 2016; 17: 1325-34.
- 24 Schindler M, Spycher BD, Ammann RA, Ansari M, Michel G, Kuehni CE. Cause-specific long-term mortality in survivors of childhood cancer in Switzerland: A population-based study. *International journal of cancer*. 2016; 139: 322-33.
- 25 Kero AE, Madanat-Harjuoja LM, Järvelä LS, Malila N, Matomäki J, Lähteenmäki PM. Cardiovascular medication after cancer at a young age in Finland: A nationwide registry linkage study. *International journal of cancer*. 2016; 139: 683-90.
- 26 Armstrong GT, Liu Q, Yasui Y, Neglia JP, Leisenring W, Robison LL, et al. Late mortality among 5-year survivors of childhood cancer: a summary from the Childhood Cancer Survivor Study. Journal of clinical oncology : official journal of the American Society of Clinical Oncology. 2009; 27: 2328-38.
- Reaven GM. Banting lecture 1988. Role of insulin resistance in human disease. *Diabetes*. 1988;
 37: 1595-607.
- 28 Wittcopp C, Conroy R. Metabolic Syndrome in Children and Adolescents. *Pediatrics in review*. 2016; 37: 193-202.
- 29 Kaplan NM. The deadly quartet. Upper-body obesity, glucose intolerance, hypertriglyceridemia, and hypertension. *Archives of internal medicine*. 1989; 149: 1514-20.
- 30 Wijnen M, Olsson DS, van den Heuvel-Eibrink MM, Hammarstrand C, Janssen J, Van der Lely AJ, *et al.* The metabolic syndrome and its components in 178 patients treated for craniopharyngioma after 16 years of follow-up. *European journal of endocrinology*. 2017; 178: 11-22.
- 31 Alberti KG, Zimmet PZ. Definition, diagnosis and classification of diabetes mellitus and its complications. Part 1: diagnosis and classification of diabetes mellitus provisional report of a WHO consultation. *Diabetic medicine : a journal of the British Diabetic Association*. 1998; 15: 539-53.

- 32 Balkau B, Charles MA. Comment on the provisional report from the WHO consultation. European Group for the Study of Insulin Resistance (EGIR). *Diabetic medicine : a journal of the British Diabetic Association*. 1999; 16: 442-3.
- 33 Executive Summary of The Third Report of The National Cholesterol Education Program (NCEP) Expert Panel on Detection, Evaluation, And Treatment of High Blood Cholesterol In Adults (Adult Treatment Panel III). *Jama*. 2001; 285: 2486-97.
- 34 Grundy SM, Cleeman JI, Daniels SR, Donato KA, Eckel RH, Franklin BA, et al. Diagnosis and management of the metabolic syndrome: an American Heart Association/National Heart, Lung, and Blood Institute Scientific Statement. *Circulation*. 2005; 112: 2735-52.
- 35 Alberti KG, Zimmet P, Shaw J. Metabolic syndrome--a new world-wide definition. A Consensus Statement from the International Diabetes Federation. *Diabetic medicine : a journal of the British Diabetic Association*. 2006; 23: 469-80.
- 36 Alberti KG, Eckel RH, Grundy SM, Zimmet PZ, Cleeman JI, Donato KA, *et al.* Harmonizing the metabolic syndrome: a joint interim statement of the International Diabetes Federation Task Force on Epidemiology and Prevention; National Heart, Lung, and Blood Institute; American Heart Association; World Heart Federation; International Atherosclerosis Society; and International Association for the Study of Obesity. *Circulation*. 2009; 120: 1640-5.
- Oda E. Metabolic syndrome: its history, mechanisms, and limitations. *Acta diabetologica*. 2012;
 49: 89-95.
- Ju SY, Lee JY, Kim DH. Association of metabolic syndrome and its components with all-cause and cardiovascular mortality in the elderly: A meta-analysis of prospective cohort studies. *Medicine*. 2017; 96: e8491.
- 39 Alberti G; Zimmet P; Shaw J GS. The IDF consensus worldwide definition of the Metabolic Syndrome. International Diabetes Federation: 2006.
- 40 Wilson PW, D'Agostino RB, Parise H, Sullivan L, Meigs JB. Metabolic syndrome as a precursor of cardiovascular disease and type 2 diabetes mellitus. *Circulation*. 2005; 112: 3066-72.
- 41 Ford ES. The metabolic syndrome and mortality from cardiovascular disease and all-causes: findings from the National Health and Nutrition Examination Survey II Mortality Study. *Atherosclerosis.* 2004; 173: 309-14.
- 42 Blijdorp K, van den Heuvel-Eibrink MM, Pieters R, Boot AM, Delhanty PJ, van der Lely AJ, *et al.* Obesity is underestimated using body mass index and waist-hip ratio in long-term adult survivors of childhood cancer. *PLoS One.* 2012; 7: e43269.
- 43 van Waas M, Neggers SJ, Raat H, van Rij CM, Pieters R, van den Heuvel-Eibrink MM. Abdominal radiotherapy: a major determinant of metabolic syndrome in nephroblastoma and neuroblastoma survivors. *PLoS One*. 2012; 7: e52237.
- 44 van Waas M, Neggers SJ, Uitterlinden AG, Blijdorp K, van der Geest IM, Pieters R, *et al.* Treatment factors rather than genetic variation determine metabolic syndrome in childhood cancer survivors. *European journal of cancer (Oxford, England : 1990).* 2013; 49: 668-75.
- 45 Ward LC. Bioelectrical impedance analysis for body composition assessment: reflections on accuracy, clinical utility, and standardisation. *European journal of clinical nutrition*. 2019; 73: 194-99.
- 46 Horber FF, Thomi F, Casez JP, Fonteille J, Jaeger P. Impact of hydration status on body composition as measured by dual energy X-ray absorptiometry in normal volunteers and patients on haemodialysis. *The British journal of radiology*. 1992; 65: 895-900.
- 47 O'Brien C, Young AJ, Sawka MN. Bioelectrical impedance to estimate changes in hydration status. *International journal of sports medicine*. 2002; 23: 361-6.

- 48 van Santen SS, Olsson DS, Hammarstrand C, Wijnen M, van den Heuvel-Eibrink MM, Van der Lely AJ, *et al.* Diagnosing Metabolic Syndrome in Craniopharyngioma Patients: Body Composition versus BMI. *European journal of endocrinology*. 2019.
- 49 Perreault L. Determining body composition in adults. In: Pi-Sunyer F, Sullivan D (eds.). UpToDate: Waltham, MA 2017.
- 50 Sommer I, Teufer B, Szelag M, Nussbaumer-Streit B, Titscher V, Klerings I, *et al.* The performance of anthropometric tools to determine obesity: a systematic review and meta-analysis. *Scientific reports*. 2020; 10: 12699.
- 51 Coffman E, Richmond-Bryant J. Multiple biomarker models for improved risk estimation of specific cardiovascular diseases related to metabolic syndrome: a cross-sectional study. *Population health metrics.* 2015; 13: 7.
- 52 Shea BJ, Reeves BC, Wells G, Thuku M, Hamel C, Moran J, *et al.* AMSTAR 2: a critical appraisal tool for systematic reviews that include randomised or non-randomised studies of healthcare interventions, or both. *BMJ (Clinical research ed).* 2017; 358: j4008.
- 53 Whiting PF, Rutjes AW, Westwood ME, Mallett S, Deeks JJ, Reitsma JB, *et al.* QUADAS-2: a revised tool for the quality assessment of diagnostic accuracy studies. *Annals of internal medicine*. 2011; 155: 529-36.
- 54 Hayden JA, van der Windt DA, Cartwright JL, Côté P, Bombardier C. Assessing bias in studies of prognostic factors. *Annals of internal medicine*. 2013; 158: 280-6.
- 55 Guyatt G, Oxman AD, Akl EA, Kunz R, Vist G, Brozek J, *et al.* GRADE guidelines: 1. Introduction-GRADE evidence profiles and summary of findings tables. *Journal of clinical epidemiology*. 2011; 64: 383-94.
- 56 DerSimonian R, Laird N. Meta-analysis in clinical trials. *Controlled clinical trials*. 1986; 7: 177-88.
- 57 R Core Team (2020). R: A language and environment for statistical computing. R Foundation for Statistical Computing, Vienna, Austria. URL https://www.R-project.org/.
- 58 Bielorai B, Weintraub Y, Hutt D, Hemi R, Kanety H, Modan-Moses D, *et al.* The metabolic syndrome and its components in pediatric survivors of allogeneic hematopoietic stem cell transplantation. *Clin Transplant.* 2017; 31.
- 59 Cheung YT, Edelmann MN, Mulrooney DA, Green DM, Chemaitilly W, John N, *et al.* Uric Acid and Neurocognitive Function in Survivors of Childhood Acute Lymphoblastic Leukemia Treated with Chemotherapy Only. *Cancer epidemiology, biomarkers & prevention : a publication* of the American Association for Cancer Research, cosponsored by the American Society of Preventive Oncology. 2016; 25: 1259-67.
- 60 Kojima C, Kubota M, Nagai A, Adachi S, Watanabe K, Nakahata T. Adipocytokines in childhood cancer survivors and correlation with metabolic syndrome components. *Pediatr Int.* 2013; 55: 438-42.
- 61 Ariffin H, Azanan MS, Abd Ghafar SS, Oh L, Lau KH, Thirunavakarasu T, *et al.* Young adult survivors of childhood acute lymphoblastic leukemia show evidence of chronic inflammation and cellular aging. *Cancer.* 2017; 123: 4207-14.
- 62 Barbosa-Cortes L, Lopez-Alarcon M, Mejia-Arangure JM, Klunder-Klunder M, Del Carmen Rodriguez-Zepeda M, Rivera-Marquez H, *et al.* Adipokines, insulin resistance, and adiposity as a predictors of metabolic syndrome in child survivors of lymphoma and acute lymphoblastic leukemia of a developing country. *BMC cancer.* 2017; 17: 125.
- 63 Eglit T, Lember M, Ringmets I, Rajasalu T. Gender differences in serum high-molecular-weight adiponectin levels in metabolic syndrome. *European journal of endocrinology*. 2013; 168: 385-91.

- 64 González-Jiménez E, Schmidt-Riovalle J, Sinausía L, Carmen Valenza M, Perona JS. Predictive value of ceruloplasmin for metabolic syndrome in adolescents. *BioFactors (Oxford, England)*. 2016; 42: 163-70.
- 65 Henneman P, Janssens AC, Zillikens MC, Frolich M, Frants RR, Oostra BA, *et al.* Menopause impacts the relation of plasma adiponectin levels with the metabolic syndrome. *Journal of internal medicine*. 2010; 267: 402-9.
- 66 Krzystek-Korpacka M, Patryn E, Kustrzeba-Wojcicka I, Chrzanowska J, Gamian A, Noczynska A. Gender-specific association of serum uric acid with metabolic syndrome and its components in juvenile obesity. *Clinical chemistry and laboratory medicine*. 2011; 49: 129-36.
- 67 Ntzouvani A, Fragopoulou E, Panagiotakos D, Pitsavos C, Antonopoulou S. Reduced circulating adiponectin levels are associated with the metabolic syndrome independently of obesity, lipid indices and serum insulin levels: a cross-sectional study. *Lipids in health and disease*. 2016; 15: 140.
- 68 Stefanska A, Ponikowska I, Cwiklinska-Jurkowska M, Sypniewska G. Association of FSH with metabolic syndrome in postmenopausal women: a comparison with CRP, adiponectin and leptin. *Biomarkers in medicine*. 2014; 8: 921-30.
- 69 Kim JY, Ahn SV, Yoon JH, Koh SB, Yoon J, Yoo BS, *et al.* Prospective study of serum adiponectin and incident metabolic syndrome: the ARIRANG study. *Diabetes care*. 2013; 36: 1547-53.
- 70 Ahonen TM, Saltevo JT, Kautiainen HJ, Kumpusalo EA, Vanhala MJ. The association of adiponectin and low-grade inflammation with the course of metabolic syndrome. *Nutrition, metabolism, and cardiovascular diseases : NMCD.* 2012; 22: 285-91.
- 71 Bombelli M, Quarti-Trevano F, Tadic M, Facchetti R, Cuspidi C, Mancia G, *et al.* Uric acid and risk of new-onset metabolic syndrome, impaired fasting glucose and diabetes mellitus in a general Italian population: data from the Pressioni Arteriose Monitorate E Loro Associazioni study. *J Hypertens.* 2018; 36: 1492-98.
- 72 Cohen E, Krause I, Fraser A, Goldberg E, Garty M. Hyperuricemia and metabolic syndrome: lessons from a large cohort from Israel. *Isr Med Assoc J.* 2012; 14: 676-80.
- 73 Delitala AP, Scuteri A, Fiorillo E, Lakatta EG, Schlessinger D, Cucca F. Role of Adipokines in the Association between Thyroid Hormone and Components of the Metabolic Syndrome. *Journal of Clinical Medicine*. 2019; 8.
- 74 Fernandez-Berges D, Consuegra-Sanchez L, Penafiel J, Cabrera de Leon A, Vila J, Felix-Redondo FJ, et al. Metabolic and inflammatory profiles of biomarkers in obesity, metabolic syndrome, and diabetes in a Mediterranean population. DARIOS Inflammatory study. Rev Esp Cardiol (Engl Ed). 2014; 67: 624-31.
- 75 Goncalves JP, Oliveira A, Severo M, Santos AC, Lopes C. Cross-sectional and longitudinal associations between serum uric acid and metabolic syndrome. *Endocrine*. 2012; 41: 450-7.
- 76 Juonala M, Saarikoski LA, Viikari JS, Oikonen M, Lehtimaki T, Lyytikainen LP, et al. A longitudinal analysis on associations of adiponectin levels with metabolic syndrome and carotid artery intima-media thickness. The Cardiovascular Risk in Young Finns Study. Atherosclerosis. 2011; 217: 234-9.
- 77 Kanagasabai T, Alkhalaqi K, Churilla JR, Ardern CI. The Association Between Metabolic Syndrome and Serum Concentrations of Micronutrients, Inflammation, and Oxidative Stress Outside of the Clinical Reference Ranges: A Cross-Sectional Study. *Metabolic syndrome and related disorders*. 2019; 17: 29-36.
- 78 Khan RJ, Gebreab SY, Sims M, Riestra P, Xu R, Davis SK. Prevalence, associated factors and heritabilities of metabolic syndrome and its individual components in African Americans: the Jackson Heart Study. *BMJ open.* 2015; 5: e008675.

- 79 Koskinen J, Magnussen CG, Wurtz P, Soininen P, Kangas AJ, Viikari JS, *et al.* Apolipoprotein B, oxidized low-density lipoprotein, and LDL particle size in predicting the incidence of metabolic syndrome: the Cardiovascular Risk in Young Finns study. *European journal of preventive cardiology*. 2012; 19: 1296-303.
- 80 Kosola J, Vaara JP, Ahotupa M, Kyrolainen H, Santtila M, Oksala N, *et al.* Elevated concentration of oxidized LDL together with poor cardiorespiratory and abdominal muscle fitness predicts metabolic syndrome in young men. *Metabolism: clinical and experimental.* 2013; 62: 992-9.
- 81 Lindberg S, Jensen JS, Bjerre M, Frystyk J, Flyvbjerg A, Jeppesen J, *et al.* Low adiponectin levels at baseline and decreasing adiponectin levels over 10 years of follow-up predict risk of the metabolic syndrome. *Diabetes & metabolism.* 2017; 43: 134-39.
- 82 Luptáková L, Siváková D, Cvíčelová M, Wsólová L, Danková Z, Michnová A, *et al.* Power of biomarkers and their relative contributions to metabolic syndrome in Slovak adult women. *Annals of human biology.* 2013; 40: 132-8.
- 83 Mani P, Ren HY, Neeland IJ, McGuire DK, Ayers CR, Khera A, *et al.* The association between HDL particle concentration and incident metabolic syndrome in the multi-ethnic Dallas Heart Study. *Diabetes & metabolic syndrome.* 2017; 11 Suppl 1: S175-s79.
- 84 Mazidi M, Toth PP, Banach M. C-reactive Protein Is Associated With Prevalence of the Metabolic Syndrome, Hypertension, and Diabetes Mellitus in US Adults. *Angiology*. 2018; 69: 438-42.
- 85 Musani SK, Vasan RS, Bidulescu A, Liu J, Xanthakis V, Sims M, *et al.* Aldosterone, C-reactive protein, and plasma B-type natriuretic peptide are associated with the development of metabolic syndrome and longitudinal changes in metabolic syndrome components: findings from the Jackson Heart Study. *Diabetes care.* 2013; 36: 3084-92.
- 86 Onat A, Komurcu-Bayrak E, Can G, Kucukdurmaz Z, Hergenc G, Erginel-Unaltuna N. Apolipoprotein A-I positively associated with diabetes in women independently of apolipoprotein E genotype and apolipoprotein B levels. *Nutrition (Burbank, Los Angeles County, Calif)*. 2010; 26: 975-80.
- 87 Onat A, Can G, Ornek E, Ayhan E, Erginel-Ünaltuna N, Murat SN. High serum apolipoprotein E determines hypertriglyceridemic dyslipidemias, coronary disease and apoA-I dysfunctionality. *Lipids*. 2013; 48: 51-61.
- 88 Onat A, Can G, Cakr H, Ozpamuk-Karadeniz F, Karadeniz Y, Yuksel H, *et al.* Sex-Specific Predictors of Metabolic Syndrome Independent of Its Components. *Journal of investigative medicine : the official publication of the American Federation for Clinical Research.* 2015; 63: 796-801.
- 89 Onat A, Can G, Çoban N, Dönmez İ, Çakır H, Ademoğlu E, *et al.* Lipoprotein(a) level and MIF gene variant predict incident metabolic syndrome and mortality. *Journal of investigative medicine : the official publication of the American Federation for Clinical Research.* 2016; 64: 392-9.
- 90 Petrikova J, Janicko M, Fedacko J, Drazilova S, Madarasova Geckova A, Marekova M, *et al.* Serum Uric Acid in Roma and Non-Roma-Its Correlation with Metabolic Syndrome and Other Variables. *International journal of environmental research and public health*. 2018; 15.
- 91 Stefanska A, Sypniewska G, Blaszkiewicz B, Ponikowska I, Cwiklinska-Jurkowska M. Comparison between C-reactive protein and adipocyte fatty acid-binding protein as a component of metabolic syndrome in middle-aged women. *Clinical biochemistry*. 2011; 44: 304-6.
- 92 Sun D, Li S, Zhang X, Fernandez C, Chen W, Srinivasan SR, *et al.* Uric acid is associated with metabolic syndrome in children and adults in a community: the Bogalusa Heart Study. *PLoS One.* 2014; 9: e89696.
- 93 Cheserek MJ, Shi Y, Le G. Association of hyperuricemia with metabolic syndrome among university workers: sex and occupational differences. *Afr Health Sci.* 2018; 18: 842-51.

- 94 Moulin SR, Baldo MP, Souza JB, Luchi WM, Capingana DP, Magalhaes P, et al. Distribution of Serum Uric Acid in Black Africans and Its Association With Cardiovascular Risk Factors. J Clin Hypertens (Greenwich). 2017; 19: 45-50.
- 95 Wei CY, Sun CC, Wei JC, Tai HC, Sun CA, Chung CF, *et al.* Association between Hyperuricemia and Metabolic Syndrome: An Epidemiological Study of a Labor Force Population in Taiwan. *BioMed research international.* 2015; 2015: 369179.
- 96 Porchia LM, Gonzalez-Mejia ME, Torres-Rasgado E, Ruiz-Vivanco G, Perez-Fuentes R. Low serum uric acid concentration augments insulin effects on the prevalence of metabolic syndrome. *Diabetes & metabolic syndrome*. 2018; 12: 325-31.
- 97 Yu KH, Luo SF, Tsai WP, Huang YY. Intermittent elevation of serum urate and 24-hour urinary uric acid excretion. *Rheumatology (Oxford, England)*. 2004; 43: 1541-5.
- 98 Liu CW, Chen KH, Tseng CK, Chang WC, Wu YW, Hwang JJ. The dose-response effects of uric acid on the prevalence of metabolic syndrome and electrocardiographic left ventricular hypertrophy in healthy individuals. *Nutrition, metabolism, and cardiovascular diseases : NMCD*. 2019; 29: 30-38.
- 99 Zhang Y, Fu J, Yang S, Yang M, Liu A, Wang L, *et al.* Prevalence of metabolically obese but normal weight (MONW) and metabolically healthy but obese (MHO) in Chinese Beijing urban subjects. *Bioscience trends.* 2017; 11: 418-26.
- 100 Chen H, Xiong C, Shao X, Ning J, Gao P, Xiao H, *et al.* Lymphocyte To High-Density Lipoprotein Ratio As A New Indicator Of Inflammation And Metabolic Syndrome. *Diabetes, metabolic syndrome and obesity : targets and therapy.* 2019; 12: 2117-23.
- 101 Kawada T. Relationship between several markers and presence of metabolic syndrome or components of the metabolic syndrome in Japanese workers. *Journal of occupational and environmental medicine*. 2012; 54: 984-8.
- 102 Li G, Xu L, Zhao Y, Li L, Fu J, Zhang Q, et al. Leptin-adiponectin imbalance as a marker of metabolic syndrome among Chinese children and adolescents: The BCAMS study. PLoS One. 2017; 12: e0186222.
- 103 Chearskul S, Homsanit M, Chearskul S, Homsanit M, Udol K, Kooptiwut S, *et al.* Certain hormonal markers in urban Thai adults with metabolic syndrome. *Journal of the Medical Association of Thailand = Chotmaihet thangphaet.* 2014; 97: 77-84.
- 104 Yoon JH, Park JK, Oh SS, Lee KH, Kim SK, Cho IJ, *et al.* The ratio of serum leptin to adiponectin provides adjunctive information to the risk of metabolic syndrome beyond the homeostasis model assessment insulin resistance: the Korean Genomic Rural Cohort Study. *Clinica chimica acta; international journal of clinical chemistry.* 2011; 412: 2199-205.
- 105 Yun JE, Won S, Mok Y, Cui W, Kimm H, Jee SH. Association of the leptin to high-molecularweight adiponectin ratio with metabolic syndrome. *Endocrine journal*. 2011; 58: 807-15.
- 106 Adejumo EN, Adejumo OA, Azenabor A, Ekun AO, Enitan SS, Adebola OK, et al. Leptin: Adiponectin ratio discriminated the risk of metabolic syndrome better than adiponectin and leptin in Southwest Nigeria. Diabetes & metabolic syndrome. 2019; 13: 1845-49.
- 107 Mirza S, Qu HQ, Li Q, Martinez PJ, Rentfro AR, McCormick JB, et al. Adiponectin/leptin ratio and metabolic syndrome in a Mexican American population. *Clinical and investigative medicine Medecine clinique et experimentale*. 2011; 34: E290.
- 108 Chou YC, Kuan JC, Bai CH, Yang T, Chou WY, Hsieh PC, *et al.* Predictive value of serum apolipoprotein B/apolipoprotein A-I ratio in metabolic syndrome risk: a Chinese cohort study. *Endocrine*. 2015; 49: 404-14.

- 109 Yang MH, Sung J, Gwak GY. The associations between apolipoprotein B, A1, and the B/A1 ratio and nonalcoholic fatty liver disease in both normal-weight and overweight Korean population. *Journal of clinical lipidology*. 2016; 10: 289-98.
- 110 Belfki H, Ben Ali S, Bougatef S, Ben Ahmed D, Haddad N, Jmal A, et al. The Apolipoprotein B/Apolipoprotein A 1 ratio in relation to metabolic syndrome and its components in a sample of the Tunisian population. Experimental and molecular pathology. 2011; 91: 622-5.
- 111 Jing F, Mao Y, Guo J, Zhang Z, Li Y, Ye Z, et al. The value of Apolipoprotein B/Apolipoprotein A1 ratio for metabolic syndrome diagnosis in a Chinese population: a cross-sectional study. *Lipids in health and disease*. 2014; 13: 81.
- 112 Jung CH, Hwang JY, Yu JH, Shin MS, Bae SJ, Park JY, *et al.* The value of apolipoprotein B/ A1 ratio in the diagnosis of metabolic syndrome in a Korean population. *Clinical endocrinology*. 2012; 77: 699-706.
- 113 Kim SW, Jee JH, Kim HJ, Jin SM, Suh S, Bae JC, et al. Non-HDL-cholesterol/HDL-cholesterol is a better predictor of metabolic syndrome and insulin resistance than apolipoprotein B/ apolipoprotein A1. International journal of cardiology. 2013; 168: 2678-83.
- 114 Zhong L, Li Q, Jiang Y, Cheng D, Liu Z, Wang B, *et al.* The ApoB/ApoA1 ratio is associated with metabolic syndrome and its components in a Chinese population. *Inflammation*. 2010; 33: 353-8.
- 115 Srikanthan K, Feyh A, Visweshwar H, Shapiro JI, Sodhi K. Systematic Review of Metabolic Syndrome Biomarkers: A Panel for Early Detection, Management, and Risk Stratification in the West Virginian Population. *International journal of medical sciences*. 2016; 13: 25-38.
- 116 Goli P, Riahi R, Daniali SS, Pourmirzaei M, Kelishadi R. Association of serum uric acid concentration with components of pediatric metabolic syndrome: A systematic review and metaanalysis. *J Res Med Sci.* 2020; 25: 43.
- 117 Yuan H, Yu C, Li X, Sun L, Zhu X, Zhao C, et al. Serum Uric Acid Levels and Risk of Metabolic Syndrome: A Dose-Response Meta-Analysis of Prospective Studies. The Journal of clinical endocrinology and metabolism. 2015; 100: 4198-207.
- 118 Liu Z, Que S, Zhou L, Zheng S. Dose-response Relationship of Serum Uric Acid with Metabolic Syndrome and Non-alcoholic Fatty Liver Disease Incidence: A Meta-analysis of Prospective Studies. *Scientific reports.* 2015; 5: 14325.
- 119 Liu Z, Liang S, Que S, Zhou L, Zheng S, Mardinoglu A. Meta-Analysis of Adiponectin as a Biomarker for the Detection of Metabolic Syndrome. *Frontiers in physiology*. 2018; 9: 1238.
- 120 Falahi E, Khalkhali Rad AH, Roosta S. What is the best biomarker for metabolic syndrome diagnosis? *Diabetes & metabolic syndrome*. 2015; 9: 366-72.
- 121 Ketterl TG, Chow EJ, Leisenring WM, Goodman P, Koves IH, Petryk A, *et al.* Adipokines, Inflammation, and Adiposity in Hematopoietic Cell Transplantation Survivors. *Biology of blood and marrow transplantation : journal of the American Society for Blood and Marrow Transplantation*. 2018; 24: 622-26.
- 122 Tonorezos ES, Vega GL, Sklar CA, Chou JF, Moskowitz CS, Mo Q, et al. Adipokines, body fatness, and insulin resistance among survivors of childhood leukemia. *Pediatric blood & cancer*. 2012; 58: 31-6.
- 123 Argyrou C, Hatziagapiou K, Theodorakidou M, Nikola OA, Vlahopoulos S, Lambrou GI. The role of adiponectin, LEPTIN, and ghrelin in the progress and prognosis of childhood acute lymphoblastic leukemia. *Leukemia & lymphoma*. 2019; 60: 2158-69.
- 124 Annaloro C, Airaghi L, Giannarelli D, Mometto G, Orsatti A, Saporiti G, *et al.* Prospective evaluation of metabolic syndrome and its features in a single-center series of hematopoietic stem cell transplantation recipients. *Annals of hematology.* 2018; 97: 2471-78.

- 125 Moschovi M, Trimis G, Vounatsou M, Katsibardi K, Margeli A, Damianos A, *et al.* Serial plasma concentrations of adiponectin, leptin, and resistin during therapy in children with acute lymphoblastic leukemia. *Journal of pediatric hematology/oncology*. 2010; 32: e8-13.
- 126 Cepelova M, Kruseova J, Luks A, Capek V, Cepela P, Potockova J, *et al.* Accelerated atherosclerosis, hyperlipoproteinemia and insulin resistance in long-term survivors of Hodgkin lymphoma during childhood and adolescence. *Neoplasma.* 2019; 66: 978-87.
- 127 Chow EJ, Simmons JH, Roth CL, Baker KS, Hoffmeister PA, Sanders JE, *et al.* Increased cardiometabolic traits in pediatric survivors of acute lymphoblastic leukemia treated with total body irradiation. *Biology of blood and marrow transplantation : journal of the American Society for Blood and Marrow Transplantation.* 2010; 16: 1674-81.
- 128 Cooksey R, Wu SY, Klesse L, Oden JD, Bland RE, Hodges JC, *et al.* Metabolic syndrome is a sequela of radiation exposure in hypothalamic obesity among survivors of childhood brain tumors. *Journal of investigative medicine : the official publication of the American Federation for Clinical Research.* 2019; 67: 295-302.
- 129 Frisk P, Arvidson J, Larsson M, Naessén T. Risk factors for cardiovascular disease are increased in young adults treated with stem cell transplantation during childhood. *Pediatric transplantation*. 2012; 16: 385-91.
- 130 Grote S, Almstedt HC, Tarleton HP. Cardiometabolic Health Among Cancer Survivors: A 13-Week Pilot Study of a Combined Aerobic and Resistance Training Program. *Oncology nursing forum.* 2016; 43: 306-15.
- 131 Wei C, Crowne E. The impact of childhood cancer and its treatment on puberty and subsequent hypothalamic pituitary and gonadal function, in both boys and girls. *Best practice & research Clinical endocrinology & metabolism.* 2019; 33: 101291.
- 132 Pluimakers VG, van Waas M, Looman CWN, de Maat MP, de Jonge R, Delhanty P, *et al.* Metabolic syndrome detection with biomarkers in childhood cancer survivors. *Endocrine connections*. 2020; 9: 676-86.
- 133 Marouli E, Graff M, Medina-Gomez C, Lo KS, Wood AR, Kjaer TR, *et al.* Rare and low-frequency coding variants alter human adult height. *Nature*. 2017; 542: 186-90.
- 134 Chemaitilly W, Cohen LE. DIAGNOSIS OF ENDOCRINE DISEASE: Endocrine late-effects of childhood cancer and its treatments. *European journal of endocrinology*. 2017; 176: R183-r203.
- 135 de Vathaire F, El-Fayech C, Ben Ayed FF, Haddy N, Guibout C, Winter D, *et al.* Radiation dose to the pancreas and risk of diabetes mellitus in childhood cancer survivors: a retrospective cohort study. *The Lancet Oncology*. 2012; 13: 1002-10.
- 136 Darzy KH, Shalet SM. Hypopituitarism following Radiotherapy Revisited. *Endocrine development.* 2009; 15: 1-24.
- 137 Oudin C, Simeoni MC, Sirvent N, Contet A, Begu-Le Coroller A, Bordigoni P, *et al.* Prevalence and risk factors of the metabolic syndrome in adult survivors of childhood leukemia. *Blood.* 2011; 117: 4442-8.
- 138 Rose SR, Horne VE, Howell J, Lawson SA, Rutter MM, Trotman GE, *et al.* Late endocrine effects of childhood cancer. *Nature reviews Endocrinology*. 2016; 12: 319-36.
- 139 de Haas EC, Oosting SF, Lefrandt JD, Wolffenbuttel BH, Sleijfer DT, Gietema JA. The metabolic syndrome in cancer survivors. *The Lancet Oncology*. 2010; 11: 193-203.
- 140 de Abreu VG, Martins CJM, de Oliveira PAC, Francischetti EA. High-molecular weight adiponectin/HOMA-IR ratio as a biomarker of metabolic syndrome in urban multiethnic Brazilian subjects. *PLoS One.* 2017; 12: e0180947.

- 141 Baudrand R, Campino C, Carvajal CA, Olivieri O, Guidi G, Faccini G, *et al.* High sodium intake is associated with increased glucocorticoid production, insulin resistance and metabolic syndrome. *Clinical endocrinology*. 2014; 80: 677-84.
- 142 de Oliveira A, Hermsdorff HH, Cocate PG, Santos EC, Bressan J, Natali AJ. Accuracy of plasma interleukin-18 and adiponectin concentrations in predicting metabolic syndrome and cardiometabolic disease risk in middle-age Brazilian men. *Applied physiology, nutrition, and metabolism = Physiologie appliquee, nutrition et metabolisme.* 2015; 40: 1048-55.
- 143 Ding Y, Li S, Ma RL, Guo H, Zhang J, Zhang M, *et al.* Association of homeostasis model assessment of insulin resistance, adiponectin, and low-grade inflammation with the course of the metabolic syndrome. *Clinical biochemistry*. 2015; 48: 503-7.
- 144 Hirose H, Takayama T, Hozawa S, Hibi T, Saito I. Prediction of metabolic syndrome using artificial neural network system based on clinical data including insulin resistance index and serum adiponectin. *Computers in biology and medicine*. 2011; 41: 1051-6.
- 145 Ju H, Zhou Z, Sun M, Chen H. Association of fetuin-A to adiponectin ratio with metabolic syndrome: a cross-sectional study. *Endocrine*. 2017; 58: 190-93.
- 146 Kawada T, Hasegawa M. Predictive ability of serum high-molecular-weight adiponectin in combination with serum insulin and serum C-reactive protein for the presence of metabolic syndrome. *Annals of human biology*. 2012; 39: 108-12.
- 147 Abu-Farha M, Behbehani K, Elkum N. Comprehensive analysis of circulating adipokines and hsCRP association with cardiovascular disease risk factors and metabolic syndrome in Arabs. *Cardiovascular diabetology*. 2014; 13: 76.
- 148 Cheng KH, Huang SP, Huang CN, Lee YC, Chu CS, Chang CF, et al. The impact of estradiol and 1,25(OH)2D3 on metabolic syndrome in middle-aged Taiwanese males. PLoS One. 2013; 8: e60295.
- 149 Esteghamati A, Khalilzadeh O, Anvari M, Rashidi A, Mokhtari M, Nakhjavani M. Association of serum leptin levels with homeostasis model assessment-estimated insulin resistance and metabolic syndrome: the key role of central obesity. *Metabolic syndrome and related disorders*. 2009; 7: 447-52.
- 150 Jeong J, Suh YJ. Association between Serum Uric Acid and Metabolic Syndrome in Koreans. *Journal of Korean medical science*. 2019; 34: e307.
- 151 Lipshultz SE, Landy DC, Lopez-Mitnik G, Lipsitz SR, Hinkle AS, Constine LS, *et al.* Cardiovascular status of childhood cancer survivors exposed and unexposed to cardiotoxic therapy. *Journal of clinical oncology : official journal of the American Society of Clinical Oncology.* 2012; 30: 1050-7.
- 152 Meigs JB, Rutter MK, Sullivan LM, Fox CS, D'Agostino RB, Sr., Wilson PW. Impact of insulin resistance on risk of type 2 diabetes and cardiovascular disease in people with metabolic syndrome. *Diabetes care*. 2007; 30: 1219-25.
- 153 Mottillo S, Filion KB, Genest J, Joseph L, Pilote L, Poirier P, *et al.* The metabolic syndrome and cardiovascular risk a systematic review and meta-analysis. *Journal of the American College of Cardiology*. 2010; 56: 1113-32.
- 154 Esteghamati A, Zandieh A, Zandieh B, Khalilzadeh O, Meysamie A, Nakhjavani M, *et al.* Leptin cut-off values for determination of metabolic syndrome: third national surveillance of risk factors of non-communicable diseases in Iran (SuRFNCD-2007). *Endocrine*. 2011; 40: 117-23.
- 155 Bermúdez V, Rojas J, Salazar J, Calvo MJ, Morillo J, Torres W, *et al.* The Maracaibo city metabolic syndrome prevalence study: Primary results and agreement level of 3 diagnostic criteria. *Revista Latinoamericana de Hipertension.* 2014; 9: 20-32.

- 156 Lamarche B, Moorjani S, Lupien PJ, Cantin B, Bernard PM, Dagenais GR, *et al.* Apolipoprotein A-I and B levels and the risk of ischemic heart disease during a five-year follow-up of men in the Québec cardiovascular study. *Circulation.* 1996; 94: 273-8.
- 157 Wilson DP, Jacobson TA, Jones PH, Koschinsky ML, McNeal CJ, Nordestgaard BG, *et al.* Use of Lipoprotein(a) in clinical practice: A biomarker whose time has come. A scientific statement from the National Lipid Association. *Journal of clinical lipidology*. 2019; 13: 374-92.
- 158 Barter PJ, Ballantyne CM, Carmena R, Castro Cabezas M, Chapman MJ, Couture P, *et al.* Apo B versus cholesterol in estimating cardiovascular risk and in guiding therapy: report of the thirty-person/ten-country panel. *Journal of internal medicine*. 2006; 259: 247-58.
- 159 Huffman FG, Knight-Sepulveda K, McLean M, Vaccaro JA, Zarini GG. Serum Adiponectin and Ghrelin, Metabolic Syndrome and Diabetes Status in Cuban Americans *Int J Health Res.* 2010; 3: 93-103.
- 160 Norvik JV, Storhaug HM, Ytrehus K, Jenssen TG, Zykova SN, Eriksen BO, *et al.* Overweight modifies the longitudinal association between uric acid and some components of the metabolic syndrome: The Tromsø Study. *BMC cardiovascular disorders.* 2016; 16: 85.
- 161 Dallmeier D, Larson MG, Vasan RS, Keaney JF, Jr., Fontes JD, Meigs JB, *et al.* Metabolic syndrome and inflammatory biomarkers: a community-based cross-sectional study at the Framingham Heart Study. *Diabetology & metabolic syndrome.* 2012; 4: 28.
- 162 Côté M, Cartier A, Reuwer AQ, Arsenault BJ, Lemieux I, Després JP, *et al.* Adiponectin and risk of coronary heart disease in apparently healthy men and women (from the EPIC-Norfolk Prospective Population Study). *The American journal of cardiology*. 2011; 108: 367-73.
- 163 Goldberg JF, Ness KK, Chi X, Santucci AK, Plana JC, Joshi VM, *et al.* Cardiovascular Family History Increases Risk for Late-Onset Adverse Cardiovascular Outcomes in Childhood Cancer Survivors: A St. Jude Lifetime Cohort Report. *Cancer epidemiology, biomarkers & prevention : a publication of the American Association for Cancer Research, cosponsored by the American Society of Preventive Oncology*. 2020.
- 164 Povel CM, Boer JM, Reiling E, Feskens EJ. Genetic variants and the metabolic syndrome: a systematic review. *Obesity reviews : an official journal of the International Association for the Study of Obesity.* 2011; 12: 952-67.
- 165 Monda KL, North KE, Hunt SC, Rao DC, Province MA, Kraja AT. The genetics of obesity and the metabolic syndrome. *Endocrine, metabolic & immune disorders drug targets.* 2010; 10: 86-108.
- 166 Fenwick PH, Jeejeebhoy K, Dhaliwal R, Royall D, Brauer P, Tremblay A, *et al.* Lifestyle genomics and the metabolic syndrome: A review of genetic variants that influence response to diet and exercise interventions. *Critical reviews in food science and nutrition*. 2019; 59: 2028-39.
- 167 Al-Meshaweh AF, Jafar Y, Asem M, Akanji AO. Determinants of Blood Uric Acid Levels in a Dyslipidemic Arab Population. *Medical Principles and Practice*. 2012; 21: 209-16.
- 168 Tsouli SG, Liberopoulos EN, Mikhailidis DP, Athyros VG, Elisaf MS. Elevated serum uric acid levels in metabolic syndrome: an active component or an innocent bystander? *Metabolism: clinical and experimental.* 2006; 55: 1293-301.
- 169 Zhang F, Basinski MB, Beals JM, Briggs SL, Churgay LM, Clawson DK, *et al.* Crystal structure of the obese protein leptin-E100. *Nature*. 1997; 387: 206-9.
- 170 Funcke JB, Scherer PE. Beyond adiponectin and leptin: adipose tissue-derived mediators of inter-organ communication. *Journal of lipid research*. 2019; 60: 1648-84.
- 171 Scherer PE, Williams S, Fogliano M, Baldini G, Lodish HF. A novel serum protein similar to C1q, produced exclusively in adipocytes. *The Journal of biological chemistry*. 1995; 270: 26746-9.
- 172 van Andel M, Heijboer AC, Drent ML. Adiponectin and Its Isoforms in Pathophysiology. *Advances in clinical chemistry.* 2018; 85: 115-47.

- 173 Ronsley R, Rassekh SR, Fleming A, Empringham B, Jennings W, Portwine C, *et al.* High molecular weight adiponectin levels are inversely associated with adiposity in pediatric brain tumor survivors. *Scientific reports.* 2020; 10: 18606.
- Young SG. Recent progress in understanding apolipoprotein B. *Circulation*. 1990; 82: 1574-94.
- 175 Kastelein JJ, van der Steeg WA, Holme I, Gaffney M, Cater NB, Barter P, *et al.* Lipids, apolipoproteins, and their ratios in relation to cardiovascular events with statin treatment. *Circulation.* 2008; 117: 3002-9.
- 176 Sniderman AD, Williams K, Contois JH, Monroe HM, McQueen MJ, de Graaf J, *et al.* A meta-analysis of low-density lipoprotein cholesterol, non-high-density lipoprotein cholesterol, and apolipoprotein B as markers of cardiovascular risk. *Circulation Cardiovascular quality and outcomes.* 2011; 4: 337-45.
- 177 Devaraj S, Semaan JR, Jialal I. Biochemistry, Apolipoprotein B. *StatPearls*. StatPearls Publishing
- Copyright © 2020, StatPearls Publishing LLC.: Treasure Island (FL) 2020.
- 178 Clouet-Foraison N, Marcovina SM, Guerra E, Aarsand AK, Coşkun A, Díaz-Garzón J, *et al.* Analytical Performance Specifications for Lipoprotein(a), Apolipoprotein B-100, and Apolipoprotein A-I Using the Biological Variation Model in the EuBIVAS Population. *Clinical chemistry.* 2020; 66: 727-36.
- 179 Danesh J, Collins R, Peto R. Lipoprotein(a) and coronary heart disease. Meta-analysis of prospective studies. *Circulation*. 2000; 102: 1082-5.
- 180 Kar S. Elevated Lipoprotein A in South Asians and the Associated Risk of Cardiovascular Disease: A Systematic Review. *Current problems in cardiology*. 2020: 100581.
- 181 Craig WY, Neveux LM, Palomaki GE, Cleveland MM, Haddow JE. Lipoprotein(a) as a risk factor for ischemic heart disease: metaanalysis of prospective studies. *Clinical chemistry*. 1998; 44: 2301-6.
- 182 Forbes CA, Quek RG, Deshpande S, Worthy G, Wolff R, Stirk L, *et al.* The relationship between Lp(a) and CVD outcomes: a systematic review. *Lipids in health and disease*. 2016; 15: 95.
- 183 Smolders B, Lemmens R, Thijs V. Lipoprotein (a) and stroke: a meta-analysis of observational studies. *Stroke*. 2007; 38: 1959-66.
- Nave AH, Lange KS, Leonards CO, Siegerink B, Doehner W, Landmesser U, *et al.* Lipoprotein(a) as a risk factor for ischemic stroke: a meta-analysis. *Atherosclerosis*. 2015; 242: 496-503.
- 185 Wang Z, Zhai X, Xue M, Cheng W, Hu H. Prognostic value of lipoprotein (a) level in patients with coronary artery disease: a meta-analysis. *Lipids in health and disease*. 2019; 18: 150.
- 186 Clarke R, Peden JF, Hopewell JC, Kyriakou T, Goel A, Heath SC, *et al.* Genetic variants associated with Lp(a) lipoprotein level and coronary disease. *The New England journal of medicine*. 2009; 361: 2518-28.
- 187 van Iperen EP, Sivapalaratnam S, Holmes MV, Hovingh GK, Zwinderman AH, Asselbergs FW. Genetic analysis of emerging risk factors in coronary artery disease. *Atherosclerosis*. 2016; 254: 35-41.
- 188 Paige E, Masconi KL, Tsimikas S, Kronenberg F, Santer P, Weger S, *et al.* Lipoprotein(a) and incident type-2 diabetes: results from the prospective Bruneck study and a meta-analysis of published literature. *Cardiovascular diabetology*. 2017; 16: 38.
- 189 Kaptoge S, Di Angelantonio E, Lowe G, Pepys MB, Thompson SG, Collins R, *et al.* C-reactive protein concentration and risk of coronary heart disease, stroke, and mortality: an individual participant meta-analysis. *Lancet (London, England).* 2010; 375: 132-40.
- 190 Pepys MB, Hirschfield GM. C-reactive protein: a critical update. *The Journal of clinical investigation*. 2003; 111: 1805-12.

- 191 Pepys MB, Rowe IF, Baltz ML. C-reactive protein: binding to lipids and lipoproteins. *International review of experimental pathology*. 1985; 27: 83-111.
- 192 Zhang YX, Cliff WJ, Schoefl GI, Higgins G. Coronary C-reactive protein distribution: its relation to development of atherosclerosis. *Atherosclerosis*. 1999; 145: 375-9.
- 193 Yoon K, Ryu S, Lee J, Park JD. Higher and increased concentration of hs-CRP within normal range can predict the incidence of metabolic syndrome in healthy men. *Diabetes & metabolic* syndrome. 2018; 12: 977-83.
- 194 Oda E, Kawai R. Reproducibility of high-sensitivity C-reactive protein as an inflammatory component of metabolic syndrome in Japanese. *Circulation journal : official journal of the Japanese Circulation Society*. 2010; 74: 1488-93.
- 195 Hillman AJ, Lohsoonthorn V, Hanvivatvong O, Jiamjarasrangsi W, Lertmaharit S, Williams MA. Association of High Sensitivity C-Reactive Protein Concentrations and Metabolic Syndrome among Thai Adults. *Asian Biomed (Res Rev News)*. 2010; 4: 385-93.
- 196 Kim J, Pyo S, Yoon DW, Lee S, Lim JY, Heo JS, *et al.* The co-existence of elevated high sensitivity C-reactive protein and homocysteine levels is associated with increased risk of metabolic syndrome: A 6-year follow-up study. *PLoS One.* 2018; 13: e0206157.
- 197 Abu-Farha M, Abubaker J, Al-Khairi I, Cherian P, Noronha F, Kavalakatt S, *et al.* Circulating angiopoietin-like protein 8 (betatrophin) association with HsCRP and metabolic syndrome. *Cardiovascular diabetology.* 2016; 15: 25.
- 198 Chen T, Chen H, Xiao H, Tang H, Xiang Z, Wang X, et al. Comparison of the Value of Neutrophil to High-Density Lipoprotein Cholesterol Ratio and Lymphocyte to High-Density Lipoprotein Cholesterol Ratio for Predicting Metabolic Syndrome Among a Population in the Southern Coast of China. Diabetes, metabolic syndrome and obesity : targets and therapy. 2020; 13: 597-605.
- 199 Mirhafez SR, Pasdar A, Avan A, Esmaily H, Moezzi A, Mohebati M, *et al.* Cytokine and growth factor profiling in patients with the metabolic syndrome. *The British journal of nutrition.* 2015; 113: 1911-9.
- 200 Orsatti CL, Petri Nahas EA, Nahas-Neto J, Orsatti FL, Giorgi VI, Witkin SS. Evaluation of Toll-Like receptor 2 and 4 RNA expression and the cytokine profile in postmenopausal women with metabolic syndrome. *PLoS One.* 2014; 9: e109259.
- 201 Fearon WF, Fearon DT. Inflammation and cardiovascular disease: role of the interleukin-1 receptor antagonist. *Circulation*. 2008; 117: 2577-9.
- 202 Dinarello CA. Immunological and inflammatory functions of the interleukin-1 family. *Annual review of immunology*. 2009; 27: 519-50.
- 203 Ridker PM, Howard CP, Walter V, Everett B, Libby P, Hensen J, *et al.* Effects of interleukin-1β inhibition with canakinumab on hemoglobin A1c, lipids, C-reactive protein, interleukin-6, and fibrinogen: a phase IIb randomized, placebo-controlled trial. *Circulation.* 2012; 126: 2739-48.
- 204 Popovic M, Ebrahimi F, Urwyler SA, Donath MY, Christ-Crain M. The role of IL-1 in the regulation of copeptin in patients with metabolic syndrome. *Endocrine connections*. 2020; 9: 715-23.
- 205 D'Elia L, Giaquinto A, Cappuccio FP, Iacone R, Russo O, Strazzullo P, *et al.* Circulating leptin is associated with serum uric acid level and its tubular reabsorption in a sample of adult middleaged men. *Journal of endocrinological investigation*. 2020; 43: 587-93.
- 206 Tabara Y, Osawa H, Kawamoto R, Tachibana-Iimori R, Yamamoto M, Nakura J, *et al.* Reduced high-molecular-weight adiponectin and elevated high-sensitivity C-reactive protein are synergistic risk factors for metabolic syndrome in a large-scale middle-aged to elderly population:

the Shimanami Health Promoting Program Study. *The Journal of clinical endocrinology and metabolism*. 2008; 93: 715-22.

- 207 Oda E. Serum uric acid is an independent predictor of metabolic syndrome in a Japanese health screening population. *Heart Vessels*. 2014; 29: 496-503.
- 208 Oda E. High-sensitivity C-reactive protein and white blood cell count equally predict development of the metabolic syndrome in a Japanese health screening population. *Acta diabetologica*. 2013; 50: 633-8.
- 209 Dai X, Yuan J, Yao P, Yang B, Gui L, Zhang X, *et al.* Association between serum uric acid and the metabolic syndrome among a middle- and old-age Chinese population. *European journal of epidemiology*. 2013; 28: 669-76.
- 210 Wang L, Zhang T, Liu Y, Tang F, Xue F. Association of Serum Uric Acid with Metabolic Syndrome and Its Components: A Mendelian Randomization Analysis. *BioMed research international*. 2020; 2020: 6238693.
- 211 White J, Sofat R, Hemani G, Shah T, Engmann J, Dale C, *et al.* Plasma urate concentration and risk of coronary heart disease: a Mendelian randomisation analysis. *The lancet Diabetes & endocrinology*. 2016; 4: 327-36.
- 212 Mente A, Meyre D, Lanktree MB, Heydarpour M, Davis AD, Miller R, *et al.* Causal relationship between adiponectin and metabolic traits: a Mendelian randomization study in a multiethnic population. *PLoS One.* 2013; 8: e66808.
- 213 Kodama K, Tojjar D, Yamada S, Toda K, Patel CJ, Butte AJ. Ethnic Differences in the Relationship Between Insulin Sensitivity and Insulin Response. A systematic review and meta-analysis. 2013; 36: 1789-96.
- 214 WHO. Appropriate body-mass index for Asian populations and its implications for policy and intervention strategies. *Lancet (London, England)*. 2004; 363: 157-63.
- 215 DeBoer MD, Dong L, Gurka MJ. Racial/ethnic and sex differences in the relationship between uric acid and metabolic syndrome in adolescents: an analysis of National Health and Nutrition Survey 1999-2006. *Metabolism: clinical and experimental.* 2012; 61: 554-61.
- 216 Tsimikas S, Clopton P, Brilakis ES, Marcovina SM, Khera A, Miller ER, *et al.* Relationship of oxidized phospholipids on apolipoprotein B-100 particles to race/ethnicity, apolipoprotein(a) isoform size, and cardiovascular risk factors: results from the Dallas Heart Study. *Circulation*. 2009; 119: 1711-9.
- 217 Chen W, Srinivasan SR, Bao W, Wattigney WA, Berenson GS. Sibling aggregation of low- and high-density lipoprotein cholesterol and apolipoproteins B and A-I levels in black and white children: the Bogalusa Heart Study. *Ethnicity & disease*. 1997; 7: 241-9.
- 218 Iso H, Harada S, Shimamoto T, Folsom AR, Koike K, Sato S, *et al.* Polymorphism of the apolipoprotein B gene and blood lipid concentrations in Japanese and Caucasian population samples. *Atherosclerosis.* 1996; 126: 233-41.
- 219 Vermaak WJ, Ubbink JB, Delport R, Becker PJ, Bissbort SH, Ungerer JP. Ethnic immunity to coronary heart disease? *Atherosclerosis*. 1991; 89: 155-62.
- 220 Hackler E, 3rd, Lew J, Gore MO, Ayers CR, Atzler D, Khera A, et al. Racial Differences in Cardiovascular Biomarkers in the General Population. *Journal of the American Heart Association*. 2019; 8: e012729.
- 221 Osei K, Gaillard T. Disparities in Cardiovascular Disease and Type 2 Diabetes Risk Factors in Blacks and Whites: Dissecting Racial Paradox of Metabolic Syndrome. *Frontiers in endocrinology*. 2017; 8: 204.

- 222 Jacobs S, Kroeger J, Schulze MB, Frank LK, Franke AA, Cheng I, *et al.* Dietary Patterns Derived by Reduced Rank Regression Are Inversely Associated with Type 2 Diabetes Risk across 5 Ethnic Groups in the Multiethnic Cohort. *Current developments in nutrition*. 2017; 1: e000620.
- 223 Kim JY, Yadav D, Ahn SV, Koh SB. A prospective study of serum adiponectin and regression of metabolic syndrome: The ARIRANG study. *Biochemical and biophysical research communications*. 2015; 466: 201-5.
- 224 Kramer CK, von Mühlen D, Jassal SK, Barrett-Connor E. Serum uric acid levels improve prediction of incident type 2 diabetes in individuals with impaired fasting glucose: the Rancho Bernardo Study. *Diabetes care*. 2009; 32: 1272-3.
- 225 Feig DI, Kang DH, Johnson RJ. Uric acid and cardiovascular risk. *The New England journal of medicine*. 2008; 359: 1811-21.
- 226 Chuang SY, Chen JH, Yeh WT, Wu CC, Pan WH. Hyperuricemia and increased risk of ischemic heart disease in a large Chinese cohort. *International journal of cardiology*. 2012; 154: 316-21.
- 227 Sattar N, Wannamethee G, Sarwar N, Tchernova J, Cherry L, Wallace AM, *et al.* Adiponectin and coronary heart disease: a prospective study and meta-analysis. *Circulation*. 2006; 114: 623-9.
- Li S, Shin HJ, Ding EL, van Dam RM. Adiponectin levels and risk of type 2 diabetes: a systematic review and meta-analysis. *Jama*. 2009; 302: 179-88.
- 229 Meier U, Gressner AM. Endocrine regulation of energy metabolism: review of pathobiochemical and clinical chemical aspects of leptin, ghrelin, adiponectin, and resistin. *Clinical chemistry*. 2004; 50: 1511-25.
- 230 Sabatine MS, Morrow DA, Jablonski KA, Rice MM, Warnica JW, Domanski MJ, *et al.* Prognostic significance of the Centers for Disease Control/American Heart Association highsensitivity C-reactive protein cut points for cardiovascular and other outcomes in patients with stable coronary artery disease. *Circulation*. 2007; 115: 1528-36.
- 231 Povel CM, Beulens JW, van der Schouw YT, Dolle ME, Spijkerman AM, Verschuren WM, *et al.* Metabolic syndrome model definitions predicting type 2 diabetes and cardiovascular disease. *Diabetes care.* 2013; 36: 362-8.
- 232 Yusuf S, Hawken S, Ounpuu S, Dans T, Avezum A, Lanas F, *et al.* Effect of potentially modifiable risk factors associated with myocardial infarction in 52 countries (the INTERHEART study): case-control study. *Lancet (London, England).* 2004; 364: 937-52.
- 233 Gianfagna F, Veronesi G, Guasti L, Chambless LE, Brambilla P, Corrao G, *et al.* Do apolipoproteins improve coronary risk prediction in subjects with metabolic syndrome? Insights from the North Italian Brianza cohort study. *Atherosclerosis.* 2014; 236: 175-81.
- 234 Swerdlow DI, Holmes MV, Kuchenbaecker KB, Engmann JE, Shah T, Sofat R, *et al.* The interleukin-6 receptor as a target for prevention of coronary heart disease: a mendelian randomisation analysis. *Lancet (London, England).* 2012; 379: 1214-24.
- 235 Zinman B, Hanley AJ, Harris SB, Kwan J, Fantus IG. Circulating tumor necrosis factor-alpha concentrations in a native Canadian population with high rates of type 2 diabetes mellitus. *The Journal of clinical endocrinology and metabolism.* 1999; 84: 272-8.
- 236 Li WC, Hsiao KY, Chen IC, Chang YC, Wang SH, Wu KH. Serum leptin is associated with cardiometabolic risk and predicts metabolic syndrome in Taiwanese adults. *Cardiovascular diabetology*. 2011; 10: 36.
- 237 Obeidat AA, Ahmad MN, Haddad FH, Azzeh FS. Leptin and uric acid as predictors of metabolic syndrome in jordanian adults. *Nutr Res Pract.* 2016; 10: 411-7.
- 238 Chiou WK, Wang MH, Huang DH, Chiu HT, Lee YJ, Lin JD. The relationship between serum uric acid level and metabolic syndrome: differences by sex and age in Taiwanese. *J Epidemiol.* 2010; 20: 219-24.

- 239 Chen YY, Kao TW, Yang HF, Chou CW, Wu CJ, Lai CH, *et al.* The association of uric acid with the risk of metabolic syndrome, arterial hypertension or diabetes in young subjects- An observational study. *Clinica chimica acta; international journal of clinical chemistry.* 2018; 478: 68-73.
- 240 Lee JG, Park SW, Cho BM, Lee S, Kim YJ, Jeong DW, *et al.* Serum amylase and risk of the metabolic syndrome in Korean adults. *Clinica chimica acta; international journal of clinical chemistry.* 2011; 412: 1848-53.
- 241 Lee YJ, Cho S, Kim SR. A possible role of serum uric acid as a marker of metabolic syndrome. *Intern Med J.* 2014; 44: 1210-6.
- 242 Sun HL, Pei D, Lue KH, Chen YL. Uric Acid Levels Can Predict Metabolic Syndrome and Hypertension in Adolescents: A 10-Year Longitudinal Study. *PLoS One*. 2015; 10: e0143786.
- 243 Fujikawa R, Ito C, Tsuboi A. Is the screening of metabolic syndrome using adiponectin possible? *Diabetol Int.* 2015; 6: 313-20.
- 244 Oh J, Kim JY, Park S, Youn JC, Son NH, Shin DJ, *et al.* The relationship between insulin-like growth factor-1 and metabolic syndrome, independent of adiponectin. *Clinica chimica acta; international journal of clinical chemistry.* 2012; 413: 506-10.
- 245 Saisho Y, Hirose H, Roberts R, Abe T, Kawabe H, Itoh H. C-reactive protein, high-molecularweight adiponectin and development of metabolic syndrome in the Japanese general population: a longitudinal cohort study. *PLoS One*. 2013; 8: e73430.
- 246 Jeong H, Baek SY, Kim SW, Park EJ, Lee J, Kim H, et al. C reactive protein level as a marker for dyslipidaemia, diabetes and metabolic syndrome: results from the Korea National Health and Nutrition Examination Survey. BMJ open. 2019; 9: e029861.
- 247 Oda E, Kawai R. Comparison between high-sensitivity C-reactive protein (hs-CRP) and white blood cell count (WBC) as an inflammatory component of metabolic syndrome in Japanese. *Internal medicine (Tokyo, Japan).* 2010; 49: 117-24.
- 248 Abbasian M, Ebrahimi H, Delvarianzadeh M, Norouzi P, Fazli M. Association between serum uric acid (SUA) levels and metabolic syndrome (MetS) components in personnel of Shahroud University of Medical Sciences. *Diabetes & metabolic syndrome*. 2016; 10: 132-6.
- 249 Ahmadnezhad M, Arefhosseini SR, Parizadeh MR, Tavallaie S, Tayefi M, Darroudi S, *et al.* Association between serum uric acid, high sensitive C-reactive protein and pro-oxidantantioxidant balance in patients with metabolic syndrome. *BioFactors (Oxford, England).* 2018; 44: 263-71.
- Ali N, Miah R, Hasan M, Barman Z, Mou AD, Hafsa JM, et al. Association between serum uric acid and metabolic syndrome: a cross-sectional study in Bangladeshi adults. Scientific reports. 2020; 10: 7841.
- 251 Baygi F, Herttua K, Sheidaei A, Ahmadvand A, Jensen OC. Association of Serum Uric Acid with cardio-metabolic risk factors and metabolic syndrome in seafarers working on tankers. *BMC public health.* 2020; 20: 442.
- 252 Chen D, Zhang H, Gao Y, Lu Z, Yao Z, Jiang Y, *et al.* Cross-sectional and longitudinal associations between serum uric acid and metabolic syndrome: Results from Fangchenggang Area Male Health and Examination Survey in China. *Clinica chimica acta; international journal of clinical chemistry.* 2015; 446: 226-30.
- 253 Cho MH, Kim YM, Yoon JH, Kim DH, Lim JS. Serum uric acid in Korean children and adolescents: reference percentiles and association with metabolic syndrome. *Annals of pediatric endocrinology & metabolism.* 2020; 25: 104-11.

- 254 de Oliveira A, Hermsdorff HHM, Guedes Cocate P, Bressan J, Azevedo Novello A, Cardoso dos Santos E, *et al.* The impact of serum uric acid on the diagnostic of metabolic syndrome in apparently healthy brazilian middle-aged men. *Nutricion hospitalaria.* 2014; 30: 562-9.
- 255 Ferrara LA, Wang H, Umans JG, Franceschini N, Jolly S, Lee ET, et al. Serum uric acid does not predict incident metabolic syndrome in a population with high prevalence of obesity. Nutrition, metabolism, and cardiovascular diseases : NMCD. 2014; 24: 1360-4.
- 256 Fu YQ, Yang H, Zheng JS, Zeng XY, Zeng W, Fan ZF, et al. Positive association between metabolic syndrome and serum uric acid in Wuhan. Asia Pacific journal of clinical nutrition. 2017; 26: 343-50.
- 257 Hara S, Tsuji H, Ohmoto Y, Amakawa K, Hsieh SD, Arase Y, *et al.* High serum uric acid level and low urine pH as predictors of metabolic syndrome: a retrospective cohort study in a Japanese urban population. *Metabolism: clinical and experimental.* 2012; 61: 281-8.
- 258 He SJ, Chan C, Xie ZD, Shi D, Hu XB, Li HY, *et al.* The relationship between serum uric acid and metabolic syndrome in premenopausal and postmenopausal women in the Jinchang Cohort. *Gynecological endocrinology : the official journal of the International Society of Gynecological Endocrinology.* 2017; 33: 141-44.
- 259 Huang S, Liu X, Li H, Xu W, Jia H. Sex difference in the association of serum uric acid with metabolic syndrome and its components: a cross-sectional study in a Chinese Yi population. *Postgrad Med.* 2017; 129: 828-33.
- 260 Huang LL, Dou DM, Liu N, Wang XX, Fu LY, Wu X, *et al.* Association of erythrocyte parameters with metabolic syndrome in the Pearl River Delta region of China: a cross sectional study. *BMJ* open. 2018; 8: e019792.
- 261 Jaipakdee J, Jiamjarasrangsri W, Lohsoonthorn V, Lertmaharit S. Prevalence of metabolic syndrome and its association with serum uric acid levels in Bangkok Thailand. *The Southeast Asian journal of tropical medicine and public health*. 2013; 44: 512-22.
- 262 Kawada T, Otsuka T, Inagaki H, Wakayama Y, Li Q, Li YJ, *et al.* Association of smoking status, insulin resistance, body mass index, and metabolic syndrome in workers: A 1-year follow-up study. *Obesity research & clinical practice.* 2010; 4: e163-246.
- 263 Kawada T, Otsuka T. Factor structure of indices of the second derivative of the finger photoplethysmogram with metabolic components and other cardiovascular risk indicators. *Diabetes & metabolism journal.* 2013; 37: 40-5.
- 264 Kawada T. Insulin-related biomarkers to predict the risk of metabolic syndrome. *International journal of endocrinology and metabolism.* 2013; 11: e10418.
- 265 Kawada T, Otsuka T, Inagaki H, Wakayama Y, Katsumata M. Biological markers, lifestyles and metabolic syndrome in workers. *Diabetes & metabolic syndrome*. 2015; 9: 71-3.
- 266 Lan Y, Mai Z, Zhou S, Liu Y, Li S, Zhao Z, *et al.* Prevalence of metabolic syndrome in China: An up-dated cross-sectional study. *PLoS One.* 2018; 13: e0196012.
- 267 Lee HJ, Park HT, Cho GJ, Yi KW, Ahn KH, Shin JH, *et al.* Relationship between uric acid and metabolic syndrome according to menopausal status. *Gynecological endocrinology : the official journal of the International Society of Gynecological Endocrinology*. 2011; 27: 406-11.
- 268 Lee JM, Kim HC, Cho HM, Oh SM, Choi DP, Suh I. Association between serum uric acid level and metabolic syndrome. *J Prev Med Public Health*. 2012; 45: 181-7.
- 269 Lee JK, Ryoo JH, Choi JM, Park SK. Serum uric acid level and the incidence of metabolic syndrome in middle-aged Korean men: a 5-year follow-up study. *J Prev Med Public Health*. 2014; 47: 317-26.

- 270 Lee YB, Jun JE, Lee SE, Ahn J, Kim G, Jee JH, *et al.* Utility of Serum Albumin for Predicting Incident Metabolic Syndrome according to Hyperuricemia. *Diabetes & metabolism journal*. 2018; 42: 529-37.
- 271 Li Y, Chen S, Shao X, Guo J, Liu X, Liu A, *et al.* Association of uric acid with metabolic syndrome in men, premenopausal women and postmenopausal women. *International journal of environmental research and public health.* 2014; 11: 2899-910.
- 272 Lim JH, Kim YK, Kim YS, Na SH, Rhee MY, Lee MM. Relationship between serum uric Acid levels, metabolic syndrome, and arterial stiffness in korean. *Korean Circ J*. 2010; 40: 314-20.
- 273 Lin CC, Liu CS, Li CI, Lin WY, Lai MM, Lin T, *et al.* The relation of metabolic syndrome according to five definitions to cardiovascular risk factors--a population-based study. *BMC public health*. 2009; 9: 484.
- 274 Lin YC, Chen JD, Lo SH, Chen PC. Worksite health screening programs for predicting the development of Metabolic Syndrome in middle-aged employees: a five-year follow-up study. BMC public health. 2010; 10: 747.
- 275 Liu PW, Chang TY, Chen JD. Serum uric acid and metabolic syndrome in Taiwanese adults. *Metabolism: clinical and experimental.* 2010; 59: 802-7.
- 276 Liu Y, Fan Y, Liu Q, Liu K, Chen F, Tang X, *et al.* Sex-specific association of serum uric acid dynamics with the incidence of metabolic syndrome in a health check-up Chinese population: a prospective cohort study. *BMJ open.* 2020; 10: e035289.
- 277 Meshkani R, Zargari M, Larijani B. The relationship between uric acid and metabolic syndrome in normal glucose tolerance and normal fasting glucose subjects. *Acta diabetologica*. 2011; 48: 79-88.
- 278 Nagahama K, Inoue T, Kohagura K, Ishihara A, Kinjo K, Ohya Y. Hyperuricemia predicts future metabolic syndrome: a 4-year follow-up study of a large screened cohort in Okinawa, Japan. *Hypertens Res.* 2014; 37: 232-8.
- 279 Ni W, Wang R, Liu Z, Yuan X, Chi H, Lv D, et al. Association of Serum Uric Acid with Metabolic Syndrome and Its Components: A Cross-Sectional Study in Chinese Coastal Population. Metabolic syndrome and related disorders. 2020; 18: 103-09.
- 280 Obokata M, Negishi K, Ohyama Y, Okada H, Imai K, Kurabayashi M. A risk score with additional four independent factors to predict the incidence and recovery from metabolic syndrome: Development and validation in large Japanese cohorts. *PLoS ONE*. 2015; 10.
- 281 Rhee C, Kim J, Kim JY, Chang E, Park SY, Lee W, *et al.* Clinical Markers Associated with Metabolic Syndrome Among Military Aviators. *Aerosp Med Hum Perform.* 2015; 86: 970-5.
- 282 Safiri S, Qorbani M, Heshmat R, Tajbakhsh R, Eslami Shahr Babaki A, Djalalinia S, et al. Association of Serum Uric Acid With Cardiometabolic Risk Factors and Metabolic Syndrome in Iranian Adolescents: the CASPIAN-III Study. *Iranian journal of kidney diseases*. 2016; 10: 126-34.
- 283 Salehidoost R, Aminorroaya A, Zare M, Amini M. Is uric acid an indicator of metabolic syndrome in the first-degree relatives of patients with type 2 diabetes? *J Res Med Sci.* 2012; 17: 1005-10.
- 284 Schmitt AC, Cardoso MR, Lopes H, Pereira WM, Pereira EC, de Rezende DA, *et al.* Prevalence of metabolic syndrome and associated factors in women aged 35 to 65 years who were enrolled in a family health program in Brazil. *Menopause (New York, NY).* 2013; 20: 470-6.
- 285 Song YM, Lee K. Genetic and Environmental Influences on the Associations Between Uric Acid Levels and Metabolic Syndrome Over Time. *Metabolic syndrome and related disorders*. 2018; 16: 299-304.

- 286 Sumiyoshi H, Ohyama Y, Imai K, Kurabayashi M, Saito Y, Nakamura T. Association of Uric Acid with Incident Metabolic Syndrome in a Japanese General Population. *Int Heart J.* 2019; 60: 830-35.
- 287 Tani S, Matsuo R, Imatake K, Suzuki Y, Takahashi A, Yagi T, *et al.* The serum uric acid level in females may be a better indicator of metabolic syndrome and its components than in males in a Japanese population. *J Cardiol.* 2020; 76: 100-08.
- 288 Tao LX, Li X, Zhu HP, Huo D, Zhou T, Pan L, et al. Association of hematological parameters with metabolic syndrome in Beijing adult population: a longitudinal study. *Endocrine*. 2014; 46: 485-95.
- 289 Tsai TY, Cheng JF, Lai YM. Prevalence of metabolic syndrome and related factors in Taiwanese high-tech industry workers. *Clinics (Sao Paulo, Brazil)*. 2011; 66: 1531-5.
- 290 Wakabayashi I. Inverse associations between serum urate and glycemic status in a general population and in persons with diabetes mellitus. *Diabetology & metabolic syndrome*. 2020; 12: 21.
- 291 Wang JY, Chen YL, Hsu CH, Tang SH, Wu CZ, Pei D. Predictive value of serum uric acid levels for the diagnosis of metabolic syndrome in adolescents. *The Journal of pediatrics*. 2012; 161: 753-6.e2.
- 292 Yadav D, Lee ES, Kim HM, Choi E, Lee EY, Lim JS, et al. Prospective study of serum uric acid levels and incident metabolic syndrome in a Korean rural cohort. Atherosclerosis. 2015; 241: 271-7.
- 293 Yang T, Chu CH, Bai CH, You SL, Chou YC, Chou WY, *et al.* Uric acid level as a risk marker for metabolic syndrome: a Chinese cohort study. *Atherosclerosis*. 2012; 220: 525-31.
- 294 Yu TY, Jee JH, Bae JC, Jin SM, Baek JH, Lee MK, *et al.* Serum uric acid: A strong and independent predictor of metabolic syndrome after adjusting for body composition. *Metabolism: clinical and experimental.* 2016; 65: 432-40.
- 295 Yu TY, Jin SM, Jee JH, Bae JC, Lee MK, Kim JH. The Protective Effects of Increasing Serum Uric Acid Level on Development of Metabolic Syndrome. *Diabetes & metabolism journal*. 2019; 43: 504-20.
- 296 You L, Liu A, Wuyun G, Wu H, Wang P. Prevalence of hyperuricemia and the relationship between serum uric acid and metabolic syndrome in the Asian Mongolian area. J Atheroscler Thromb. 2014; 21: 355-65.
- 297 Zhang Q, Lou S, Meng Z, Ren X. Gender and age impacts on the correlations between hyperuricemia and metabolic syndrome in Chinese. *Clinical rheumatology*. 2011; 30: 777-87.
- 298 Zhang Z, Bian L, Choi Y. Serum uric acid: a marker of metabolic syndrome and subclinical atherosclerosis in Korean men. *Angiology*. 2012; 63: 420-8.
- 299 Zhang Q, Zhang C, Song X, Lin H, Zhang D, Meng W, et al. A longitudinal cohort based association study between uric acid level and metabolic syndrome in Chinese Han urban male population. BMC public health. 2012; 12: 419.
- 300 Zhang ML, Gao YX, Wang X, Chang H, Huang GW. Serum uric acid and appropriate cutoff value for prediction of metabolic syndrome among Chinese adults. J Clin Biochem Nutr. 2013; 52: 38-42.
- 301 Zhang H, Li Y, Mao Z, Liu X, Zhang X, Yang K, et al. Sex-specific associations of serum uric acid with metabolic syndrome in Chinese rural population: The RuralDiab study. *Clinica chimica* acta; international journal of clinical chemistry. 2018; 480: 119-25.
- 302 Zhao Y, Yu Y, Li H, Li M, Zhang D, Guo D, *et al.* The Association between Metabolic Syndrome and Biochemical Markers in Beijing Adolescents. *International journal of environmental research and public health.* 2019; 16.

- 303 Ziaee A, Esmailzadehha N, Ghorbani A, Asefzadeh S. Association between Uric Acid and Metabolic Syndrome in Qazvin Metabolic Diseases Study (QMDS), Iran. *Glob J Health Sci.* 2012; 5: 155-65.
- 304 Lin KP. The relationship between serum uric acid concentration and metabolic syndrome in university freshmen. *J Nurs Res.* 2009; 17: 286-92.
- 305 Ahn SV, Jung DH, Yadav D, Kim JY, Koh SB. Relative contribution of obesity and menopause to the association between serum adiponectin and incident metabolic syndrome. *Menopause (New York, NY).* 2018; 25: 154-59.
- 306 Ayina CNA, Endomba FTA, Mandengue SH, Noubiap JJN, Ngoa LSE, Boudou P, et al. Association of the leptin-to-adiponectin ratio with metabolic syndrome in a sub-Saharan African population. Diabetology & metabolic syndrome. 2017; 9: 66.
- 307 Ben Ali S, Jemaa R, Ftouhi B, Kallel A, Feki M, Slimene H, *et al.* Adiponectin and metabolic syndrome in a Tunisian population. *Inflammation*. 2012; 35: 828-33.
- 308 Cho SA, Joo HJ, Cho JY, Lee SH, Park JH, Hong SJ, *et al.* Visceral Fat Area and Serum Adiponectin Level Predict the Development of Metabolic Syndrome in a Community-Based Asymptomatic Population. *PLoS One.* 2017; 12: e0169289.
- 309 Jean-Luc Gradidge P, Norris SA, Jaff NG, Crowther NJ. Metabolic and Body Composition Risk Factors Associated with Metabolic Syndrome in a Cohort of Women with a High Prevalence of Cardiometabolic Disease. *PLoS One.* 2016; 11: e0162247.
- 310 Hata A, Yonemoto K, Shikama Y, Aki N, Kosugi C, Tamura A, *et al.* Cut-off value of total adiponectin for managing risk of developing metabolic syndrome in male Japanese workers. *PLoS One.* 2015; 10: e0118373.
- 311 Huh JH, Yoon TW, Kang DR, Kim JY. Prospective Study of Sex-Specific Adiponectin Changes and Incident Metabolic Syndrome: The ARIRANG Study. *Journal of Clinical Medicine*. 2019; 8.
- 312 Klünder-Klünder M, Flores-Huerta S, García-Macedo R, Peralta-Romero J, Cruz M. Adiponectin in eutrophic and obese children as a biomarker to predict metabolic syndrome and each of its components. *BMC public health*. 2013; 13: 88.
- 313 Koh SB, Park JK, Yoon JH, Chang SJ, Oh SS, Kim JY, *et al.* Preliminary report: a serious link between adiponectin levels and metabolic syndrome in a Korean nondiabetic population. *Metabolism: clinical and experimental.* 2010; 59: 333-7.
- 314 Lee SK, Yoon DW, Kim J, Lee SW, Kim JY, Jee SH, *et al.* Association of adiponectin, ghrelin, and leptin with metabolic syndrome and its metabolic components in Sasang constitutional type. *Eur J Integr Med.* 2018; 22: 16-21.
- 315 Lee KW, Shin D. Prospective Associations of Serum Adiponectin, Leptin, and Leptin-Adiponectin Ratio with Incidence of Metabolic Syndrome: The Korean Genome and Epidemiology Study. *International journal of environmental research and public health.* 2020; 17.
- 316 Li P, Jiang R, Li L, Liu C, Yang F, Qiu Y. Correlation of serum adiponectin and adiponectin gene polymorphism with metabolic syndrome in Chinese adolescents. *European journal of clinical nutrition.* 2015; 69: 62-7.
- 317 Song YM, Lee K, Sung J. Adiponectin Levels and Longitudinal Changes in Metabolic Syndrome: The Healthy Twin Study. *Metabolic syndrome and related disorders*. 2015; 13: 312-8.
- 318 Tanihara S, Imatoh T, Momose Y, Miyazaki M, Une H. Inverse correlation between adiponectin and the risk of metabolic syndrome in middle-aged Japanese male workers. *Acta Med Okayama*. 2009; 63: 325-30.
- 319 Yu D, Yu Z, Sun Q, Sun L, Li H, Song J, *et al.* Effects of body fat on the associations of highmolecular-weight adiponectin, leptin and soluble leptin receptor with metabolic syndrome in Chinese. *PLoS One.* 2011; 6: e16818.

- 320 Chatterjee B, Mehta M, Shah T, Mahant H, Katwa V, Gosai K. Association of Lipoprotein(a) and hsCRP Levels with Metabolic Syndrome and its Components. *Indian Journal of Clinical Biochemistry*. 2015; 30: 394-402.
- 321 Darroudi S, Fereydouni N, Tayefi M, Ahmadnezhad M, Zamani P, Tayefi B, *et al.* Oxidative stress and inflammation, two features associated with a high percentage body fat, and that may lead to diabetes mellitus and metabolic syndrome. *BioFactors (Oxford, England).* 2019; 45: 35-42.
- 322 Hong GB, Gao PC, Chen YY, Xia Y, Ke XS, Shao XF, *et al.* High-Sensitivity C-Reactive Protein Leads to Increased Incident Metabolic Syndrome in Women but Not in Men: A Five-Year Follow-Up Study in a Chinese Population. *Diabetes, metabolic syndrome and obesity : targets and therapy.* 2020; 13: 581-90.
- 323 Lai MM, Li CI, Kardia SL, Liu CS, Lin WY, Lee YD, *et al.* Sex difference in the association of metabolic syndrome with high sensitivity C-reactive protein in a Taiwanese population. *BMC public health.* 2010; 10: 429.
- 324 Mahajan A, Jaiswal A, Tabassum R, Podder A, Ghosh S, Madhu SV, *et al.* Elevated levels of C-reactive protein as a risk factor for metabolic syndrome in Indians. *Atherosclerosis*. 2012; 220: 275-81.
- 325 Meng G, Zhu Q, Shao J, Zhang Q, Liu L, Wu H, *et al.* Comparing the diagnostic ability of inflammatory markers in metabolic syndrome. *Clinica chimica acta; international journal of clinical chemistry.* 2017; 475: 1-6.
- 326 Perez CM, Ortiz AP, Guzman M, Suarez E. Distribution and correlates of the metabolic syndrome in adults living in the San Juan Metropolitan Area of Puerto Rico. *Puerto Rico health sciences journal*. 2012; 31: 114-22.
- 327 Song Y, Yang SK, Kim J, Lee DC. Association between C-Reactive Protein and Metabolic Syndrome in Korean Adults. *Korean J Fam Med.* 2019; 40: 116-23.
- 328 Thompson AM, Zhang Y, Tong W, Xu T, Chen J, Zhao L, *et al.* Association of inflammation and endothelial dysfunction with metabolic syndrome, prediabetes and diabetes in adults from Inner Mongolia, China. *BMC endocrine disorders*. 2011; 11: 16.
- 329 Yang T, Chu CH, Hsieh PC, Hsu CH, Chou YC, Yang SH, *et al.* C-reactive protein concentration as a significant correlate for metabolic syndrome: a Chinese population-based study. *Endocrine*. 2013; 43: 351-9.
- 330 Ying X, Jiang Y, Qin G, Qian Y, Shen X, Jiang Z, et al. Association of body mass index, waist circumference, and metabolic syndrome with serum cystatin C in a Chinese population. *Medicine*. 2017; 96: e6289.
- 331 Du R, Wu X, Peng K, Lin L, Li M, Xu Y, *et al.* Serum apolipoprotein B is associated with increased risk of metabolic syndrome among middle-aged and elderly Chinese: A cross-sectional and prospective cohort study. *J Diabetes.* 2019; 11: 752-60.
- 332 Ryoo JH, Park SK. Association of apolipoprotein B and incidence of metabolic syndrome in Korean men: a 5-years' follow-up study. *Atherosclerosis*. 2013; 226: 496-501.
- 333 Yi DW, Jeong DW, Lee SY, Son SM, Kang YH. The Association between Apolipoprotein A-II and Metabolic Syndrome in Korean Adults: A Comparison Study of Apolipoprotein A-I and Apolipoprotein B. *Diabetes & metabolism journal*. 2012; 36: 56-63.
- 334 Choi JR, Kim JY, Huh JH, Kim SH, Koh SB. Contribution of obesity as an effect regulator to an association between serum leptin and incident metabolic syndrome. *Clinica chimica acta; international journal of clinical chemistry.* 2018; 487: 275-80.
- 335 Esteghamati A, Noshad S, Khalilzadeh O, Morteza A, Nazeri A, Meysamie A, *et al.* Contribution of serum leptin to metabolic syndrome in obese and nonobese subjects. *Archives of medical research.* 2011; 42: 244-51.

- 336 Suriyaprom K, Tungtrongchitr R, Thawnasom K. Measurement of the levels of leptin, BDNF associated with polymorphisms LEP G2548A, LEPR Gln223Arg and BDNF Val66Met in Thai with metabolic syndrome. *Diabetology & metabolic syndrome*. 2014; 6: 6.
- 337 Yun JE, Kimm H, Jo J, Jee SH. Serum leptin is associated with metabolic syndrome in obese and nonobese Korean populations. *Metabolism: clinical and experimental.* 2010; 59: 424-9.
- 338 Bermudez V, Rojas J, Salazar J, Bello L, Anez R, Toledo A, *et al.* Variations of lipoprotein(a) levels in the metabolic syndrome: a report from the Maracaibo City Metabolic Syndrome Prevalence Study. *Journal of diabetes research.* 2013; 2013: 416451.
- 339 Jain SR, Shah KH, Acharya HN, Barot K, Sharma KH. Prevalence and Predictors of Metabolic Syndrome in Young Asymptomatic Gujarati Population. Int J Chronic Dis. 2015; 2015: 365217.
- 340 Wu XY, Lin L, Qi HY, Du R, Hu CY, Ma LN, *et al.* Association between Lipoprotein (a) Levels and Metabolic Syndrome in a Middle-aged and Elderly Chinese Cohort. *Biomed Environ Sci.* 2019; 32: 477-85.
- 341 Abdel-Moneim A, Mahmoud B, Sultan E, Mahmoud R. Association of erythrocytes indices and interleukin-1 beta with metabolic syndrome components. *University of Toronto Medical Journal*. 2020 97: 6-13.



Metabolic syndrome parameters, determinants, and biomarkers in adult survivors of childhood cancer: protocol of the Dutch Childhood Cancer Survivor Study Metabolic Syndrome (Dutch LATER METS study)

V.G. Pluimakers, M. Fiocco, J.E. van Atteveld, M. Hobbelink, D. Bresters, E. van Dulmen-den Broeder, M. van der Heiden-van der Loo, G.O. Janssens, L.C.M. Kremer, J.J. Loonen, M. Louwerens, H.J. van der Pal, C.M. Ronckers, H. van Santen, A.B. Versluijs, A.C.H. de Vries, M.M. van den Heuvel-Eibrink and S.J.C.M.M. Neggers

JMIR Research Protocols, 2021 Jan;10(1):e21256

ABSTRACT

Background

Potential late effects of treatment for childhood cancer include adiposity, insulin resistance, dyslipidemia and hypertension. These risk factors cluster together as metabolic syndrome (MetS) and increase the risk for development of diabetes mellitus and cardio- and cerebrovascular disease. Knowledge on risk factors, timely diagnosis and preventive strategies is of importance to prevent cardio- and cerebrovascular complications and improve quality of life. Currently, no studies in national cohorts on prevalence and determinants of MetS in childhood cancer survivors including biomarkers and genetic predisposition are available.

Objectives

The objectives of the Dutch LATER METS study are to assess 1) the prevalence and risk factors of MetS and its separate components, and 2) the potential value of additional biomarkers, in the national cohort of adult long-term survivors of childhood cancer.

Methods

This is a cross-sectional study, based on recruitment of all survivors treated in the Netherlands between 1963 and 2002. MetS will be classified according to the definitions of the National Cholesterol Education Program (NCEP-ATP III) as well as the Joint Interim Statement (JIS), and compared to reference data. Dual-energy X-ray absorptiometry (DXA) scans were performed to assess body composition in more detail. The effect of patient characteristics, previous treatment, and genetic variation on the risk of MetS will be assessed. The diagnostic and predictive value of novel biomarkers will be tested.

Results

Patient accrual started in 2016 and lasted until April 2020. A total of 2380 survivors has participated, in seven pediatric oncology hospitals. From July 2020, biomarker testing, SNP analysis and data analysis will be performed.

Conclusions

The Dutch LATER METS study will provide knowledge on clinical and genetic determinants of MetS, and the diagnostic value of biomarkers, in childhood cancer survivors. The results of this study will be used to optimize surveillance guidelines for MetS in survivors, based on enhanced risk stratification and screening strategies. This will improve diagnosis of MetS, and prevent complications.

INTRODUCTION

Because of increasing survival of patients with childhood cancer, late side effects have become more prominent. Potential late effects include adiposity, insulin resistance, dyslipidemia and hypertension, which cluster together as metabolic syndrome (MetS). MetS is associated with a higher risk of diabetes mellitus, as well as cardio- and cerebrovascular morbidity and mortality later in life [1-3]. The separate components are in itself risk factors for diabetes and cardiovascular disease, but when coexisting, the components can aggravate each other, leading to an even higher risk [4, 5].

Studies in childhood cancer survivors have reported a prevalence of MetS of over 30% after 25 years follow-up, substantially higher compared to age- and sex-matched controls (odds ratio 1.76) [6, 7]. This apparent risk difference for MetS further increases the elevated risks for cardiovascular outcomes and endothelial damage from anthracyclines, alkylating agents and irradiation [8, 9]. Consequently, the mortality due to coronary and cerebrovascular disease in long-term survivors is up to 12.7 times higher than the general population [10-13]. The fact that MetS can be subclinical for many years, emphasizes the need for timely identification of MetS in survivors and early intervention strategies. Lifestyle and diet advices, exercise and medication may prevent the development of diabetes and cardio- and cerebrovascular disease, improving survival rates and quality of life.

Several underlying conditions have been reported to increase the risk for (components of) MetS in survivors: growth hormone deficiency, pancreatic beta cell dysfunction, hypogonadism, hypothyroidism, and altered body composition with increased intraabdominal fat [14-19]. Hence, an increased risk of MetS might be associated with treatment for a brain tumor, treatment with radiotherapy, intensive chemotherapy, nephrectomy, adrenalectomy, or stem cell transplantation (SCT) [7, 16, 20-32]. The effects of other potentially harmful treatments, for example corticosteroids, and patient-related factors such as sex, age, BMI at diagnosis, and lifestyle, are still not clear [3]. Also, heterogeneity in incidence of MetS among homogeneously treated survivors, suggests a role of genetic susceptibility [33, 34]. A few studies, using candidate gene approach as well as one genome-wide association study, have identified genetic variants that might be associated with development of MetS and its components in survivors [24, 35, 36]. Results based on these studies have not yet been replicated nor functionally validated.

Multiple definitions of MetS have been developed over the past years. The two most commonly used are those of the National Cholesterol Education Program (Adult Treatment Panel, third report: NCEP-ATPIII) and the Joint Interim Statement (JIS) of the International Diabetes Federation (IDF), National Heart, Lung, and Blood Institute (NHLBI), and the American Heart Association (AHA). Both definitions overlap largely but they differ in waist circumference cut-off point (Table 1) [37, 38]. Apart from the four components, pro-inflammatory and pro-thrombotic markers have been reported to be relevant biomarkers of MetS, as well as hyperuricemia [39, 40].

Adequate assessment of MetS in survivors using the NCEP-ATP III and JIS definitions has specific challenges, particularly after abdominal radiotherapy. It has been shown that BMI and waist circumference underestimate adiposity, due to deformation of spine, muscles and fat, particularly in past treatment eras when higher radiotherapy doses and larger fields were used [21, 41, 42]. Similarly, adiposity can be disguised due to sarcopenic obesity after SCT [43, 44]. Body composition can be more reliably measured by Dual-energy X-ray absorptiometry (DXA scan), but this is time consuming and expensive to be implemented for standard follow-up of all survivors. Serum biomarkers may be more cost-effective surrogate markers for MetS. In smaller survivor cohorts, and in the general population, biomarkers – other than triglycerides and high-density lipoprotein (HDL) cholesterol – that have been proposed as predictors of MetS include low-density lipoprotein (LDL), apolipoprotein-B (apo-B), leptin, adiponectin, uric acid, and C-reactive protein (CRP) [39, 45-50].

So far, large studies on clinically diagnosed MetS in survivors are scarce. Two large multi-center cohort studies with clinically diagnosed MetS are the American St. Jude Lifetime (SJLIFE, all types of childhood cancer) [6, 7] and the French Leucémies de l'Enfant et l'Adolescent (leukemia) [31, 51, 52] studies. Other studies have yielded heterogeneous and sometimes conflicting results, and can be difficult to compare. This may be due to MetS components being analyzed only separately, or due to small patient cohorts, a questionnaire based or retrospective design, insufficient treatment data (e.g., only childhood cancer diagnosis is known, not treatment), and short follow-up (MetS risk increases continuously with age, so a follow-up of 10 to 20 years likely underestimates this) [22, 32, 53-55]. In addition, comparison of study outcomes can be difficult due to the use of different classifications. Currently, no studies in national cohorts on prevalence and determinants of MetS in childhood cancer survivors, including biomarkers and genetic predisposition to MetS, are available.

Here we describe the methodology of the Dutch LATER METS study in the adult cohort of survivors treated between 1963 and 2002. This nationwide study assesses MetS prevalence, clinical and genetic risk factors, and the diagnostic and predictive value of additional biomarkers. The results of this study will be used to identify survivors at risk and to optimize surveillance guidelines.

		NCEP-ATP III [37]	JIS [38]	Alternative with DXA scan
Required for diagno	osis		≥3	
Adiposity	Waist circumference (cm)	>102 ^ª /88 ^b	≥94 ^a /80 ^{b c}	Body fat Z-score >2
Insulin resistance	Fasting glucose (mmol/L)	≥	5.5 or treatme	nt
Dyslipidemia	Triglycerides (mmol/L)	≥	1.7 or treatme	nt
	HDL cholesterol (mmol/L)	<1.0	*/1.3** or treat	tment
Hypertension	Blood pressure (mmHg)	≥1	30/85 or treatn	nent

Table 1. NCEP-ATP III and JIS classifications of metabolic syndrome, and alternative classification with adiposity measured by DXA scan.

^a men; ^b women; ^c cut-off for Caucasian population

NCEP-ATP III = National Cholesterol Education Program Adult Treatment Panel III; JIS = Joint Interim Statement of IDF, NHLBI and AHA.

METHODS

Objectives

The objectives of this study are to assess 1) the prevalence and risk factors (patient characteristics, previous treatment, and genetic variation) of MetS and its separate components, compared to reference data, and 2) the potential diagnostic and predictive value of novel biomarkers for surveillance for MetS, in the national cohort of adult long-term survivors of childhood cancer.

Study population and design

The current study is part of the nationwide Dutch LATER study (Figure 1). This study started accrual in all seven pediatric oncology centers in The Netherlands in 2016, thereby inviting the national cohort of all survivors treated in these hospitals between 1963 and 2002 to participate. Survivors were identified from registries of children with newly diagnosed cancer, that are maintained in each of the seven pediatric oncology centers in The Netherlands. For the current study, this information was merged to a specific childhood cancer survivors registry, containing all registered survivors. Dependent on completeness of these sources in the centers, the starting year varied from 1963 to 1977. The LATER METS study was approved by the Medical Research Ethics Committee of the Amsterdam UMC, The Netherlands (registered at toetsingonline.nl, NL32117.018.10)

In the Dutch LATER study, data for fifteen sub-studies of late effects were collected, including cardiotoxicity, bone density, frailty, growth hormone deficiency, renal toxicity, fatigue, and psychological late effects. Individuals who survived at least five years after diagnosis of histologically confirmed malignancies defined in the

International Classification of Childhood Cancer, edition 3 [56], or Langerhans cell histiocytosis treated with chemotherapy and/or radiotherapy, who were between 0 and 17 years of age at diagnosis, were invited. Exclusion criteria were treatment for a malignancy in the past year, and living abroad.

For all eligible survivors, prior to the visit of the late-effects clinic, sex, date of birth, date of cancer diagnosis, and detailed data on cancer type and treatment, including chemotherapy regimens and doses, radiotherapy fields and (fractionated) dose, SCT and corticosteroids, have been collected in a pseudonymized, web-based, central database. This includes primary diagnosis as well as, if present, recurrences and subsequent malignancies.

Subsequently, data collection for all studies was combined with the survivors' regular care visit to the late-effects clinic for the majority of survivors. Before the visit, survivors received information about the study, sent by mail by the study personnel. If they agreed to participate, study data was collected by the treating physician and/ or the study personnel.

The entire cohort, at formation in 2008, contained 6165 eligible survivors. By mail, survivors were provided the option to opt-out of future study participation. For the Dutch LATER study, the cleaned cohort was frozen in 2016, leaving 5160 subjects eligible. For the LATER METS study, only adults (n=4741) were invited. Inclusion took place until April 2020. Written informed consent was obtained from all study participants.

Reference population

Normative data from the Dutch Lifelines study cohort will serve as reference population [57]. This is a three-generational cohort of 167,000 inhabitants (10%) of the north of the Netherlands, of whom, among other data, the following parameters relevant to our study were collected between 2006 and 2013: age, sex, height, weight, waist and hip circumference, blood pressure, co-morbidities, medication use, smoking, physical activity, HDL, triglycerides, glucose, apo-B, LDL, total cholesterol, uric acid, and high sensitivity CRP (hsCRP). We aim to use a subset of this reference cohort of controls that have the same age and sex distribution as our study cohort.

Data collection

Data collected before visit of late-effects clinic

An overview of collected variables is presented in Table 2. In addition to the previously mentioned data, the following variables relevant for the METS study were extracted
from the medical records: height and weight at cancer diagnosis, and relevant comorbidities.

Data collected at visit of late-effects clinic

Weight was measured without shoes and with light clothing, on an electronic scale to the nearest 0.1kg. Height was measured without shoes to the nearest centimeter. Body mass index (BMI) was calculated from weight and height. Waist circumference was measured in the middle between the lower rib and iliac crest to the nearest centimeter. Hip circumference was measured at the greater trochanter to the nearest centimeter. Waist/hip ratio was calculated. Blood pressure was measured after at least five minutes rest with an electronic oscillometric meter (the mean of two measurements).

Survivors completed a general health questionnaire, containing questions about comorbidities, current medication use, smoking and alcohol habits, education level, and family history of diabetes mellitus and cardiovascular disease. They also completed the SQUASH questionnaire on physical activity [58].

Total body DXA scans (Hologic and Lunar types) were used to assess body composition [41]. These measurements include fat percentage and lean body mass.

The 6-minutes walking test was performed in a subset of the survivors (those treated in the Sophia children's hospital/Erasmus MC, Rotterdam), as a measure of functional exercise capacity [59, 60].

Data determined from stored samples

Venous blood samples were drawn after overnight fasting, and stored at -80°C in the biobank. To assess dyslipidemia, a lipid spectrum will be measured, consisting of triglycerides, HDL, LDL, total cholesterol, and apo-B. Insulin resistance will be assessed by measuring glucose and insulin. Additionally, adiponectin, leptin and uric acid will be measured. Inflammatory markers include hsCRP, interleukin-6 (IL-6), high sensitivity tumor necrosis factor alpha (hsTNF α) and IL-1. The following possible confounders will be measured: IGF-1, kidney function (creatinine, urea), sex hormones (luteinizing hormone (LH), follicle stimulating hormone (FSH), anti-Müllerian hormone (AMH) (women), estradiol (women), testosterone (men)), thyroid function (thyroid-stimulating hormone (TSH), free T4), cortisol.

DNA for analysis of single nucleotide polymorphisms (SNPs) will be isolated from blood or, in survivors who received allogeneic SCT, saliva. Saliva was obtained by spitting into a collection tube (Oragene kit), after not drinking or eating for 30 minutes.

Metabolic syndrome definition

MetS will be classified according to the NCEP-ATPIII and JIS definitions [37, 38] (Table 1). Should these criteria be updated during our analysis, we will strive to take these adjustments into account.

Risk of bias

Sex, date of birth, date of cancer diagnosis, and disease and treatment data are also available for non-participating survivors. Hence, comparing participating and nonparticipating survivors is feasible, in order to determine the risk of selection bias. We will also compare these data between survivors with complete and incomplete data, to judge the risk of attrition bias. Neither physician nor study personnel were blinded to the exposures of the survivors. Objectively measurable outcomes will reduce the risk of bias in this setting.

Statistical analysis

Prevalence of MetS

The percentage of subjects with MetS and the separate components will be assessed in survivors and in the Lifelines reference cohort according to both aforementioned MetS definitions. Both cohorts will be compared by chi-square (or Fisher's exact) test. The relative risk for survivors to develop MetS, compared to Lifelines reference data, will be calculated by employing a log-binomial regression model. The agreement between both MetS definitions will be investigated with *kappa* statistic, in the whole cohort and stratified by sex.

A total body fat percentage of more than two standard deviations above the mean, as assessed by DXA, will be used as most reliable marker for adiposity. We will estimate the correlation between waist circumference and fat percentage measured by DXA scan, and we will compare overweight classification with both definitions.

Risk factors

Treatment-related risk factors for occurrence of MetS and the separate components will be assessed using multiple uni- and multivariable logistic regression models. Based on literature, an initial model will be built with cranial radiotherapy, abdominal radiotherapy, and alkylating agents (total alkylating dose calculated using cyclophosphamide equivalent dose [61]) as treatment-related independent variables, and age, sex, follow-up time, and smoking as patient-related independent variables. The effect of potential additional risk factors will be assessed by adding them to the initial model, and variables with a p-value <0.20 will be kept in the final model. These potential risk factors include all other chemotherapy agents (type and total cumulative

dose), other radiotherapy fields (body location and dose), corticosteroids, education level, family history, physical activity, functional exercise capacity, and comorbidities.

We will also investigate different abdominal radiotherapy fields involved (pancreas, liver), and the influence of SCT conditioning regimens. We will also study patientand treatment-related risk factors for the outcome underdiagnosis of overweight measured by waist circumference.

Biomarkers

Biomarker values will be reported, with reference values from the local laboratory where the samples are measured. This will be compared to Lifelines reference data by chi-square (or Fisher's exact) test. A risk factor analysis of altered biomarker values will be performed similar to the abovementioned strategy for risk factor analysis of MetS occurrence.

The diagnostic and predictive value of the biomarkers to detect MetS will be investigated in multiple steps. We will stratify the survivors by MetS presence or absence, and compare mean or median values with the *t*-test or Mann-Whitney U test. We will evaluate sensitivity and specificity, and positive and negative predictive value, based on the reference values of the local laboratory where the samples are measured. We will compare the area under the curve for a model with MetS components and for a model with each biomarker added, in order to investigate the additional diagnostic value of the novel biomarkers. We will build multivariable logistic regression models with MetS as dependent variable, and the biomarker as independent variable. In these models, we will also include MetS components as covariates, in order to investigate the MetS risk by including the biomarker as categorical as well as continuous variable.

Correlation (Pearson or Spearman) between biomarkers and fat percentage by DXA scan will be used to measure the potential use as surrogate markers for adiposity.

Genetic susceptibility analysis

Genotyping will be performed with the Infinium Global Screening Array (Illumina, San Diego, California, USA [62]), on DNA isolated from blood or, in post-SCT survivors, saliva. Quality control of the genotype data will be performed following a standardized protocol [63] including filtering based on call rate (excluded when <0.975 for either SNP or individual call rate), Hardy-Weinberg equilibrium, excess heterozygosity, gender mismatches, and familial relationships. Genetic ancestry will be assessed based on principal component analysis. Imputation will be performed

with the Michigan Imputation Server using standard settings [64] with reference panel Haplotype Reference Consortium version r1.1 [65].

The SNP analysis will be performed with the RVtests software package [66], using multiple logistic regression models with MetS and its separate components as outcomes. The initial analysis will be adjusted for age at follow-up, sex, and genetic ancestry. Then, potentially relevant covariates will be added to the model using forward selection, to study whether they influence the SNP analysis; if so, they will be kept in the model. These covariates include: BMI at follow-up, comorbidities (growth hormone deficiency, hypogonadism, diabetes mellitus, and hypothyroidism), cranial and abdominal radiotherapy, and alkylating agents (cyclophosphamide equivalent dose). We will also perform a time-to-event analysis (with left-censoring) on identified hits in order to get clinically relevant effect estimates.

Quality control of the SNP analysis will be performed with the EasyQC package using standard settings [67]. This includes filtering based on minor allele frequency (MAF, excluded when <0.05) and imputation quality (excluded when <0.3).

Visualization of the genetic associations, and annotation of biological function for the top SNPs, will be performed with the FUMA platform [68]. Findings will be replicated in available independent international cohorts.

RESULTS

Patient accrual

Patient accrual started in 2016 and lasted until April 2020. A total of 2380 survivors has participated (participation rate 50.2%). From July 2020, biomarker testing, SNP analysis and data analysis will be performed.

Power calculation

We performed a power calculation with an expected prevalence of MetS in our study cohort of 30%. This percentage is based on results from the SJLIFE cohort, in which the prevalence of clinically diagnosed MetS in 1598 survivors, after a mean of 25.6 years since diagnosis, was 31.8% [6]. This is the only large cohort study so far with clinically diagnosed MetS in survivors of heterogeneous malignancies, with a follow-up time comparable to our cohort.

Based on the sample size of 2380 survivors, expected MetS prevalence of 30%, power of 80%, and type I error of 0.05, we will have sufficient power to detect a -3% difference in MetS prevalence with the reference cohort. For risk factor analysis among survivors, depending on in how many survivors the risk factor (for example, a treatment regimen) is present, e.g. in 10, 25 and 50% of survivors, a minimum difference of -9%, -7%, and -6%, respectively, can be detected.

A genetic power calculator was used to estimate the relative risk that can be found in the genetic susceptibility analysis for an assumed MAF of 0.25 [69]. Based on the sample size of 2380, MetS population prevalence of 15% [70], a power of 80%, a type I error of $5*10^{-8}$, and a case-control ratio of 1:2, the relative risk per high risk allele that can be found is 1.5.

DISCUSSION

In the current study, we will assess the prevalence and patient and treatment-related risk factors for MetS and its separate components, in adult survivors of childhood cancer, as well as the additional diagnostic value of novel biomarkers for surveillance, and the genetic susceptibility to (treatment-related) MetS by SNP analysis.

A total of 2380 survivors has participated in the study. This corresponds to 38.6% of all survivors in the Dutch LATER cohort, and a participation rate of 50.2% of invited adult survivors. The definitive numbers of refusals, non-responders, deaths or otherwise excluded subjects are not available yet. We will report these in the paper with the results of our study.

Strengths of this study include the availability of a national cohort of survivors, the availability of comprehensive disease and treatment data, and the clinical assessment of late effects, in addition to questionnaire based endpoints. So far, the role of biomarkers and genetic susceptibility to MetS has not been well defined in survivors. We specifically intend to use DXA scans and relevant biomarkers (those with a high independent diagnostic or prognostic value, and a high correlation with fat percentage on DXA scan), to enable identification of survivors at risk for MetS, in whom waist circumference measurement is not feasible due to abdominal radiotherapy.

In conclusion, our study will provide knowledge on clinical and genetic determinants of MetS, and the diagnostic value of biomarkers, in adult childhood cancer survivors. The results of this study will be used to optimize surveillance guidelines for MetS among survivors, based on enhanced risk stratification and screening strategies. This will improve the diagnosis of MetS, and prevent complications, thereby improving quality of life.

Category	Variable	Unit(s) or categories
Collected before visit of late-eff	fects clinic	
Childhood cancer type and	Primary childhood cancer diagnosis	ICCC-3 classification
treatment	Treatment protocol	Name and arm
	Chemotherapy, per regimen	TCD
	Radiotherapy fields	TCD, fractions (if applicable)
	Cranial/craniospinal	
	Total body	
	Abdominal	
	Pancreas involvement	
	Surgery procedure	
	Autologous SCT	Yes / No, conditioning regimen
	Allogeneic SCT	Yes / No, conditioning regimen
	Relapse	Yes / No
Patient characteristics	Sex	Male / Female
	Date of birth	Date
	Date of childhood cancer diagnosis	Date
	Date of study measurements (follow-up date)	Date
Medical history	Height at cancer diagnosis	Centimeter
	Weight at cancer diagnosis	Kilogram
	Growth hormone deficiency	Yes / No
	Growth hormone replacement	Yes / No
	Hypothyroidism	Yes / No
	Hypogonadism	Yes / No
	Hypocortisolism with steroid replacement	Yes / No
Collected at visit of late-effects	clinic	
Physical examination	Height	Centimeter
	Weight	Kilogram
	Waist circumference	Centimeter
	Hip circumference	Centimeter
	Blood pressure	mmHg
General health questionnaire	Does the survivor have or has the survivor	Yes / No, age at diagnosis
	experienced	
	High cholesterol	
	Hypertension	
	Diabetes mellitus	
	Myocardial infarction	
	Stroke	
	Medication use	Type, dose, age at start
	Smoking status	Yes / Former / No
	Cardiovascular disease in family	Relative, type of disease, age at diagnosis

Table 2. Overview of collected variables

SQUASH questionnaire on physical activity

Category	Variable	Unit(s) or categories
DXA scan	Total body fat	Percentage
	Z-score total body fat	Z-score
	Lean body mass	Kilogram per m ²
	Appendicular lean body mass	Kilogram per m ²
6-minutes walking test		Meter
Data determined from st	ored samples	
Serum biomarkers	HDL	mmol/L
	LDL	mmol/L
	Total cholesterol	mmol/L
	Аро-В	g/L
	Glucose	mmol/L
	Insulin	pmol/L
	Adiponectin	ug/mL
	Leptin	ng/mL
	Uric acid	mmol/L
	hsCRP	mg/L
	IL-6	pg/mL
	hsTNFα	pg/mL
	IL-1	pg/mL
	IGF-1	ug/L
	Creatinine	mg/mmol
	Urea	mmol/L
	LH	U/L
	FSH	U/L
	AMH	ug/L
	Estradiol	pmol/L
	Testosterone	nmol/L
DNTA C 11 1/ 1		

DNA from blood/saliva

ICCC-3= International Classification of Childhood Cancer, edition 3; TCD=total cumulative dose; SCT=stem cell transplantation; DXA=Dual-energy X-ray absorptiometry; HDL=high-density lipoprotein; LDL=low-density lipoprotein; Apo-B=apolipoprotein B; hsCRP=high-sensitivity C-reactive protein; IL-6=interleukin-6; hsTNFα=highsensitivity tumor necrosis factor alpha; IL-1=interleukin-1; IGF-1=insulin-like growth factor 1; LH=luteinizing hormone; FSH=follicle stimulating hormone; AMH=anti-Müllerian hormone.



Figure 1. Overview of the Dutch LATER cohort and embeddedness of the METS study cohort within the underlying cohort. Percentages indicate proportion of Dutch LATER cohort (n=6165).

REFERENCES

- 1. de Haas EC, Oosting SF, Lefrandt JD, Wolffenbuttel BHR, Sleijfer DT, Gietema JA. The metabolic syndrome in cancer survivors. Lancet Oncol. 2010;11:193-203. PMID:20152771.
- 2. Rosen GP, Nguyen HT, Shaibi GQ. Metabolic syndrome in pediatric cancer survivors: a mechanistic review. Pediatr Blood Cancer. 2013;60(12):1922-8. PMID:23913590.
- Pluimakers VG, van Waas M, Neggers S, van den Heuvel-Eibrink MM. Metabolic syndrome as cardiovascular risk factor in childhood cancer survivors. Crit Rev Oncol Hematol. 2019;133:129-41. PMID:30661649.
- 4. Reaven GM. Banting lecture 1988. Role of insulin resistance in human disease. Diabetes. 1988;37(12):1595-607. PMID:3056758.
- DeFronzo RA, Ferrannini E. Insulin resistance. A multifaceted syndrome responsible for NIDDM, obesity, hypertension, dyslipidemia, and atherosclerotic cardiovascular disease. Diabetes Care. 1991;14(3):173-94. PMID:2044434.
- Smith WA, Li C, Nottage KA, Mulrooney DA, Armstrong GT, Lanctot JQ, Chemaitilly W, Laver JH, Srivastava DK, Robison LL, et al. Lifestyle and metabolic syndrome in adult survivors of childhood cancer: a report from the St. Jude Lifetime Cohort Study. Cancer. 2014;120(17):2742-50. PMID:25070001.
- Nottage KA, Ness KK, Li C, Srivastava D, Robison LL, Hudson MM. Metabolic syndrome and cardiovascular risk among long-term survivors of acute lymphoblastic leukaemia - From the St. Jude Lifetime Cohort. Br J Haematol. 2014;165(3):364-74. PMID:24467690.
- van der Pal HJ, van Dalen EC, van Delden E, van Dijk IW, Kok WE, Geskus RB, Sieswerda E, Oldenburger F, Koning CC, van Leeuwen FE, et al. High risk of symptomatic cardiac events in childhood cancer survivors. J Clin Oncol. 2012;30(13):1429-37. PMID:22473161.
- Brouwer CA, Postma A, Hooimeijer HL, Smit AJ, Vonk JM, van Roon AM, van den Berg MP, Dolsma WV, Lefrandt JD, Bink-Boelkens MT, et al. Endothelial damage in long-term survivors of childhood cancer. J Clin Oncol. 2013;31(31):3906-13. PMID:24062395.
- Reulen RC, Winter DL, Frobisher C, Lancashire ER, Stiller CA, Jenney ME, Skinner R, Stevens MC, Hawkins MM. Long-term Cause-Specific Mortality Among Survivors of Childhood Cancer. JAMA. 2010;304(2):172-9. PMID:20628130.
- Schindler M, Spycher BD, Ammann RA, Ansari M, Michel G, Kuehni CE, Swiss Paediatric Oncology G. Cause-specific long-term mortality in survivors of childhood cancer in Switzerland: A population-based study. Int J Cancer. 2016;139(2):322-33. PMID:26950898.
- 12. Armstrong GT, Liu Q, Yasui Y, Neglia JP, Leisenring W, Robison LL, Mertens AC. Late mortality among 5-year survivors of childhood cancer: a summary from the Childhood Cancer Survivor Study. J Clin Oncol. 2009;27(14):2328-38. PMID:19332714.
- Bagnasco F, Caruso S, Andreano A, Valsecchi MG, Jankovic M, Biondi A, Miligi L, Casella C, Terenziani M, Massimino M, et al. Late mortality and causes of death among 5-year survivors of childhood cancer diagnosed in the period 1960-1999 and registered in the Italian Off-Therapy Registry. Eur J Cancer. 2019;110:86-97. PMID:30772657.
- 14. Chemaitilly W, Cohen LE. Diagnosis of Endocrine Disease: Endocrine late-effects of childhood cancer and its treatments. Eur J Endocrinol. 2017;176(4):R183-R203. PMID:28153840.
- Friedman DN, Hilden P, Moskowitz CS, Suzuki M, Boulad F, Kernan NA, Wolden SL, Oeffinger KC, Sklar CA. Cardiovascular Risk Factors in Survivors of Childhood Hematopoietic Cell Transplantation Treated with Total Body Irradiation: A Longitudinal Analysis. Biol Blood Marrow Transplant. 2017;23(3):475-82. PMID:28040534.

- de Vathaire F, El-Fayech C, Ben Ayed FF, Haddy N, Guibout C, Winter D, Thomas-Teinturier C, Veres C, Jackson A, Pacquement H, et al. Radiation dose to the pancreas and risk of diabetes mellitus in childhood cancer survivors: a retrospective cohort study. The Lancet Oncology. 2012;13(10):1002-10. PMID:22921663.
- Wei C, Thyagiarajan M, Hunt L, Cox R, Bradley K, Elson R, Hamilton-Shield J, Stevens M, Crowne E. Reduced beta-cell reserve and pancreatic volume in survivors of childhood acute lymphoblastic leukaemia treated with bone marrow transplantation and total body irradiation. Clin Endocrinol (Oxf). 2015;82(1):59-67. PMID:25132503.
- van Dorp W, Blijdorp K, Laven JS, Pieters R, Visser JA, van der Lely AJ, Neggers SJ, van den Heuvel-Eibrink MM. Decreased ovarian function is associated with obesity in very long-term female survivors of childhood cancer. Eur J Endocrinol. 2013;168(6):905-12. PMID:23557987.
- de Haas EC, Altena R, Boezen HM, Zwart N, Smit AJ, Bakker SJ, van Roon AM, Postma A, Wolffenbuttel BH, Hoekstra HJ, et al. Early development of the metabolic syndrome after chemotherapy for testicular cancer. Ann Oncol. 2013;24(3):749-55. PMID:23131388.
- 20. Darzy KH, Shalet SM. Hypopituitarism following Radiotherapy Revisited. Endocr Dev. 2009;15:1-24. PMID:19293601.
- 21. van Waas M, Neggers SJ, Raat H, van Rij CM, Pieters R, van den Heuvel-Eibrink MM. Abdominal radiotherapy: a major determinant of metabolic syndrome in nephroblastoma and neuroblastoma survivors. PLoS One. 2012;7(12):e52237. PMID:23251703.
- Mostoufi-Moab S, Seidel K, Leisenring WM, Armstrong GT, Oeffinger KC, Stovall M, Meacham LR, Green DM, Weathers R, Ginsberg JP, et al. Endocrine Abnormalities in Aging Survivors of Childhood Cancer: A Report From the Childhood Cancer Survivor Study. J Clin Oncol. 2016;34(27):3240-7. PMID:27382091.
- Belle FN, Weiss A, Schindler M, Goutaki M, Bochud M, Zimmermann K, Von der Weid N, Ammann RA, Kuehni CE. Overweight in childhood cancer survivors: the Swiss Childhood Cancer Survivor Study. Am J Clin Nutr. 2018;107:3-11. PMID:29381792.
- Wilson CL, Liu W, Yang JJ, Kang G, Ojha RP, Neale GA, Srivastava DK, Gurney JG, Hudson MM, Robison LL, et al. Genetic and clinical factors associated with obesity among adult survivors of childhood cancer: A report from the St. Jude Lifetime Cohort. Cancer. 2015;121(13):2262-70. PMID:25963547.
- 25. van Santen HM, Geskus RB, Raemaekers S, van Trotsenburg AS, Vulsma T, van der Pal HJ, Caron HN, Kremer LC. Changes in body mass index in long-term childhood cancer survivors. Cancer. 2015;121(23):4197-204. PMID:26287726.
- Meacham LR, Sklar CA, Li S, Liu Q, Gimpel N, Yasui Y, Whitton JA, Stovall M, Robison LL, Oeffinger KC. Diabetes mellitus in long-term survivors of childhood cancer. Increased risk associated with radiation therapy: a report for the childhood cancer survivor study. Arch Intern Med. 2009;169(15):1381-8. PMID:19667301.
- Felicetti F, D'Ascenzo F, Moretti C, Corrias A, Omede P, Marra WG, Arvat E, Fagioli F, Brignardello E, Gaita F. Prevalence of cardiovascular risk factors in long-term survivors of childhood cancer: 16 years follow up from a prospective registry. Eur J Prev Cardiol. 2015;22(6):762-70. PMID:24691151.
- Kooijmans EC, Bokenkamp A, Tjahjadi NS, Tettero JM, van Dulmen-den Broeder E, van der Pal HJ, Veening MA. Early and late adverse renal effects after potentially nephrotoxic treatment for childhood cancer. Cochrane Database Syst Rev. 2019;3:CD008944. PMID:30855726.
- 29. Gibson TM, Li Z, Green DM, Armstrong GT, Mulrooney DA, Srivastava D, Bhakta N, Ness KK, Hudson MM, Robison LL. Blood Pressure Status in Adult Survivors of Childhood

Cancer: A Report from the St. Jude Lifetime Cohort Study. Cancer Epidemiol Biomarkers Prev. 2017;26(12):1705-13. PMID:29167278.

- 30. van Waas M, Neggers SJ, van Eck JP, van Noesel MM, van der Lely AJ, de Jong FH, Pieters R, van den Heuvel-Eibrink MM. Adrenal function in adult long-term survivors of nephroblastoma and neuroblastoma. Eur J Cancer. 2012;48(8):1159-66. PMID:22513228.
- Oudin C, Simeoni M-C, Sirvent N, Contet A, Begu-Le Coroller A, Bordignoni P, Curtillet C, Poiree M, Thuret I, Play B, et al. Prevalence and risk factors of the metabolic syndrome in adult survivors of childhood leukemia. Blood. 2011;117(17):4442-8. PMID:21278355.
- 32. Chow EJ, Simmons JH, Roth CL, Baker KS, Hoffmeister PA, Sanders JE, Friedman DL. Increased cardiometabolic traits in pediatric survivors of acute lymphoblastic leukemia treated with total body irradiation. Biol Blood Marrow Transplant. 2010;16(12):1674-81. PMID:20685399.
- 33. Clemens E, van der Kooi ALF, Broer L, van Dulmen-den Broeder E, Visscher H, Kremer L, Tissing W, Loonen J, Ronckers CM, Pluijm SMF, et al. The influence of genetic variation on late toxicities in childhood cancer survivors: A review. Critical Reviews in Oncology/Hematology. 2018;126:154-67. PMID:29759558.
- Gramatges MM, Bhatia S. Evidence for Genetic Risk Contributing to Long-Term Adverse Treatment Effects in Childhood Cancer Survivors. Annu Rev Med. 2018;69:247-62. PMID:29029584.
- 35. van Waas M, Neggers SJ, Uitterlinden AG, Blijdorp K, van der Geest IM, Pieters R, van den Heuvel-Eibrink MM. Treatment factors rather than genetic variation determine metabolic syndrome in childhood cancer survivors. Eur J Cancer. 2013;49(3):668-75. PMID:23036851.
- England J, Drouin S, Beaulieu P, St-Onge P, Krajinovic M, Laverdiere C, Levy E, Marcil V, Sinnett D. Genomic determinants of long-term cardiometabolic complications in childhood acute lymphoblastic leukemia survivors. BMC Cancer. 2017;17(1):751. PMID:29126409.
- 37. National Cholesterol Education Program Expert Panel on Detection E, and Treatment of High Blood Cholesterol in Adults (Adult Treatment Panel III). Third Report of the National Cholesterol Education Program (NCEP) Expert Panel on Detection, Evaluation, and Treatment of High Blood Cholesterol in Adults (Adult Treatment Panel III). 2002.
- 38. Alberti KG, Eckel RH, Grundy SM, Zimmet PZ, Cleeman JI, Donato KA, Fruchart JC, James WP, Loria CM, Smith SC, Jr., et al. Harmonizing the metabolic syndrome: a joint interim statement of the International Diabetes Federation Task Force on Epidemiology and Prevention; National Heart, Lung, and Blood Institute; American Heart Association; World Heart Federation; International Atherosclerosis Society; and International Association for the Study of Obesity. Circulation. 2009;120(16):1640-5. PMID:19805654.
- 39. Koh KK, Han SH, Quon MJ. Inflammatory markers and the metabolic syndrome: insights from therapeutic interventions. J Am Coll Cardiol. 2005;46(11):1978-85. PMID:16325028.
- Kanbay M, Jensen T, Solak Y, Le M, Roncal-Jimenez C, Rivard C, Lanaspa MA, Nakagawa T, Johnson RJ. Uric acid in metabolic syndrome: From an innocent bystander to a central player. Eur J Intern Med. 2016;29:3-8. PMID:26703429.
- 41. Blijdorp K, van den Heuvel-Eibrink MM, Pieters R, Boot AM, Delhanty PJ, van der Lely AJ, Neggers SJ. Obesity is underestimated using body mass index and waist-hip ratio in long-term adult survivors of childhood cancer. PLoS One. 2012;7(8):e43269. PMID:22905245.
- 42. Karlage RE, Wilson CL, Zhang N, Kaste S, Green DM, Armstrong GT, Robison LL, Chemaitilly W, Srivastava DK, Hudson MM, et al. Validity of anthropometric measurements for characterizing obesity among adult survivors of childhood cancer: A report from the St. Jude Lifetime Cohort Study. Cancer. 2015;121(12):2036-43. PMID:25728221.

- 43. Inaba H, Yang J, Kaste SC, Hartford CM, Motosue MS, Chemaitilly W, Triplett BM, Shook DR, Pui CH, Leung W. Longitudinal changes in body mass and composition in survivors of childhood hematologic malignancies after allogeneic hematopoietic stem-cell transplantation. J Clin Oncol. 2012;30(32):3991-7. PMID:23032628.
- 44. Wei C, Thyagiarajan MS, Hunt LP, Shield JP, Stevens MC, Crowne EC. Reduced insulin sensitivity in childhood survivors of haematopoietic stem cell transplantation is associated with lipodystropic and sarcopenic phenotypes. Pediatr Blood Cancer. 2015;62(11):1992-9. PMID:25989749.
- 45. Parish S, Offer A, Clarke R, Hopewell JC, Hill MR, Otvos JD, Armitage J, Collins R, Heart Protection Study Collaborative G. Lipids and lipoproteins and risk of different vascular events in the MRC/BHF Heart Protection Study. Circulation. 2012;125(20):2469-78. PMID:22539783.
- 46. Mach F, Baigent C, Catapano AL, Koskinas KC, Casula M, Badimon L, Chapman MJ, De Backer GG, Delgado V, Ference BA, et al. 2019 ESC/EAS Guidelines for the management of dyslipidaemias: lipid modification to reduce cardiovascular risk. Eur Heart J. 2019. PMID:31504418.
- Ketterl TG, Chow EJ, Leisenring WM, Goodman P, Koves IH, Petryk A, Steinberger J, Baker KS. Adipokines, Inflammation, and Adiposity in Hematopoietic Cell Transplantation Survivors. Biol Blood Marrow Transplant. 2018;24(3):622-6. PMID:29197678.
- 48. Skoczen S, Tomasik PJ, Bik-Multanowski M, Surmiak M, Balwierz W, Pietrzyk JJ, Sztefko K, Gozdzik J, Galicka-Latala D, Strojny W. Plasma levels of leptin and soluble leptin receptor and polymorphisms of leptin gene -18G > A and leptin receptor genes K109R and Q223R, in survivors of childhood acute lymphoblastic leukemia. J Exp Clin Cancer Res. 2011;30:64. PMID:21631924.
- 49. Im JA, Kim SH, Lee JW, Shim JY, Lee HR, Lee DC. Association between hypoadiponectinemia and cardiovascular risk factors in nonobese healthy adults. Metabolism. 2006;55(11):1546-50. PMID:17046559.
- Cicero AFG, Fogacci F, Giovannini M, Grandi E, Rosticci M, D'Addato S, Borghi C. Serum uric acid predicts incident metabolic syndrome in the elderly in an analysis of the Brisighella Heart Study. Sci Rep. 2018;8(1):11529. PMID:30068918.
- 51. Saultier P, Auquier P, Bertrand Y, Vercasson C, Oudin C, Contet A, Plantaz D, Poiree M, Ducassou S, Kanold J, et al. Metabolic syndrome in long-term survivors of childhood acute leukemia treated without hematopoietic stem cell transplantation: an L.E.A. study. Haematologica. 2016;101(12):1603-10. PMID:27515247.
- 52. Oudin C, Auquier P, Bertrand Y, Contet A, Kanold J, Sirvent N, Thouvenin S, Tabone MD, Lutz P, Ducassou S, et al. Metabolic syndrome in adults who received hematopoietic stem cell transplantation for acute childhood leukemia: an LEA study. Bone Marrow Transplant. 2015;50(11):1438-44. PMID:26191949.
- 53. Holmqvist AS, Olsen JH, Andersen KK, de Fine Licht S, Hjorth L, Garwicz S, Moell C, Anderson H, Wesenberg F, Tryggvadottir L, et al. Adult life after childhood cancer in Scandinavia: diabetes mellitus following treatment for cancer in childhood. Eur J Cancer. 2014;50(6):1169-75. PMID:24507548.
- Bhakta N, Liu Q, Ness KK, Baassiri M, Eissa H, Yeo F, Chemaitilly W, Ehrhardt MJ, Bass J, Bishop MW, et al. The cumulative burden of surviving childhood cancer: an initial report from the St Jude Lifetime Cohort Study (SJLIFE). Lancet. 2017;390(10112):2569-82. PMID:28890157.
- Prasad M, Arora B, Chinnaswamy G, Vora T, Narula G, Banavali S, Kurkure P. Nutritional status in survivors of childhood cancer: Experience from Tata Memorial Hospital, Mumbai. Indian J Cancer. 2015;52(2):219-23. PMID:26853412.

- 56. Steliarova-Foucher E, Stiller C, Lacour B, Kaatsch P. International Classification of Childhood Cancer, third edition. Cancer. 2005;103(7):1457-67. PMID:15712273.
- 57. Lifelines. Lifelines cohort and biobank: https://www.lifelines.nl/. 2017.
- 58. Wendel-Vos GC, Schuit AJ, Saris WH, Kromhout D. Reproducibility and relative validity of the short questionnaire to assess health-enhancing physical activity. J Clin Epidemiol. 2003;56(12):1163-9. PMID:14680666.
- A. T. S. Committee on Proficiency Standards for Clinical Pulmonary Function Laboratories statement: guidelines for the six-minute walk test. Am J Respir Crit Care Med. 2002;166(1):111-7. PMID:12091180.
- 60. Casanova C, Celli BR, Barria P, Casas A, Cote C, de Torres JP, Jardim J, Lopez MV, Marin JM, Montes de Oca M, et al. The 6-min walk distance in healthy subjects: reference standards from seven countries. Eur Respir J. 2011;37(1):150-6. PMID:20525717.
- 61. Green DM, Nolan VG, Goodman PJ, Whitton JA, Srivastava D, Leisenring WM, Neglia JP, Sklar CA, Kaste SC, Hudson MM, et al. The cyclophosphamide equivalent dose as an approach for quantifying alkylating agent exposure: a report from the Childhood Cancer Survivor Study. Pediatr Blood Cancer. 2014;61(1):53-67. PMID:23940101.
- 62. Illumina. Infinium Global Screening Array-24 v3.0 BeadChip.
- 63. Anderson CA, Pettersson FH, Clarke GM, Cardon LR, Morris AP, Zondervan KT. Data quality control in genetic case-control association studies. Nat Protoc. 2010;5(9):1564-73. PMID:21085122.
- 64. Das S, Forer L, Schonherr S, Sidore C, Locke AE, Kwong A, Vrieze SI, Chew EY, Levy S, McGue M, et al. Next-generation genotype imputation service and methods. Nat Genet. 2016;48(10):1284-7. PMID:27571263.
- 65. McCarthy S, Das S, Kretzschmar W, Delaneau O, Wood AR, Teumer A, Kang HM, Fuchsberger C, Danecek P, Sharp K, et al. A reference panel of 64,976 haplotypes for genotype imputation. Nat Genet. 2016;48(10):1279-83. PMID:27548312.
- Zhan X, Hu Y, Li B, Abecasis GR, Liu DJ. RVTESTS: an efficient and comprehensive tool for rare variant association analysis using sequence data. Bioinformatics. 2016;32(9):1423-6. PMID:27153000.
- 67. Heid I, Winkler T, Gorski M, Olden M. Easy QC: https://www.uni-regensburg.de/medizin/ epidemiologie-praeventivmedizin/genetische-epidemiologie/software/.
- 68. Watanabe K, Taskesen E, van Bochoven A, Posthuma D. Functional mapping and annotation of genetic associations with FUMA. Nat Commun. 2017;8(1):1826. PMID:29184056.
- 69. Purcell S, Cherny S, Sham P. Genetic Power Calculator: design of linkage and association genetic mapping studies of complex traits. Bioinformatics. 2003;19(1):149-50. PMID:12499305.
- Blokstra A, Vissink P, Venmans LMAJ, Holleman P, van der Schouw YT, Smit HA, Verschuren WM. Measuring The Netherlands. A monitoring study of risk factors in hte general population, 2009-2010. RIVM Rapport 260152001/2011. 2011.

Chapter 5

Prevalence, risk factors and optimal way to determine overweight, obesity and morbid obesity, in the first Dutch cohort of 2,338 very long-term survivors of childhood cancer: a DCCSS-LATER study

V.G. Pluimakers, J.E. van Atteveld, D.T.C. de Winter, M. Fiocco, M. Bolier, R.A.J. Nievelstein, G.O.R. Janssens, D. Bresters, M. van der Heiden-van der Loo, A.C.H. de Vries, M. Louwerens, H.J. van der Pal, S.M.F. Pluijm, C.M. Ronckers, A.B. Versluijs, L.C.M. Kremer, J.J. Loonen, E. van Dulmen-den Broeder, W.J.E. Tissing, H.M. van Santen, M.M. van den Heuvel-Eibrink and S.J.C.M.M. Neggers

Submitted

ABSTRACT

Objective

Overweight and obesity are common challenges among childhood cancer survivors. Overweight may be disguised, as survivors can have normal weight but high fat percentage (fat%) on dualenergy X-ray absorptiometry (DXA). We aimed to assess prevalence, identify determinants and assess which method captures overweight best, in a nationwide cohort.

Research design and methods

Prevalence of overweight and obesity, primarily defined by body mass index, was assessed in the DCCSS-LATER cohort of adult survivors treated 1963-2002, with the LifeLines cohort as reference. Associations between risk factors and outcomes were investigated using logistic regression. Additional overweight metrics included DXA-fat%, waist circumference (WC), waist/hip-ratio (WHR), waist/height-ratio (WHtR), and high-molecular-weight-(HMW)-adiponectin.

Results

2,338 (mean age 35.5y, follow-up 28.3y) survivors participated. Overweight prevalence was 46.3% in men and 44.3% in women (obesity 11.2% and 15.9%, morbid obesity 2.4% and 5.4%), with highest rates among brain tumor survivors. Compared to controls, overweight rate was higher in women >50y, morbid obesity in men >50y. Overweight at cancer diagnosis (adjusted odds ratio (aOR)=3.83, 95%CI=2.19-6.69), cranial radiotherapy (aOR=3.21, 95%CI=1.99-5.18) and growth hormone deficiency (separate model, aOR=1.61, 95%CI=1.00-2.59) were associated with overweight. Using BMI, WC, WHR and WHtR, overweight prevalence was similar. Low HMW-adiponectin, present in only 4.5% of survivors, was an insensitive overweight marker. DXA-based classification identified overweight in an additional 30%, particularly after abdominal radiotherapy, total body irradiation, anthracyclines and platinum.

Conclusions

Overweight occurs in almost half of long-term survivors. We identified factors associated with overweight, as well as subgroups in whom DXA can more reliably assess overweight.

INTRODUCTION

Although childhood cancer survival rates have increased impressively, excess treatmentrelated morbidity and mortality among childhood cancer survivors are observed(1). Overweight and obesity are examples of long-term morbidity after treatment. These are components of metabolic syndrome and risk factors for diabetes mellitus, atherosclerotic disease and consequent mortality.

While in the general population overweight is a common problem, among survivors it is even more frequent. The reported prevalence of overweight (8.5-40.7%) and obesity (1.4-42%) in survivors varies due to different follow-up times, size and selection of cohorts(2-6). Except for Switzerland(4), overweight prevalence has so far not been assessed in a nationwide, unselected cohort of survivors.

Risk factors for overweight in the general population are related to lifestyle (unhealthy diet, lack of exercise and smoking) and genetic susceptibility(7, 8). Additionally, in survivors, increased overweight risk can be of endocrine origin: cranial and abdominal radiotherapy and alkylating chemotherapy can cause growth hormone deficiency (GHD), hypogonadism and hypothyroidism(2-6). Also, abdominal radiotherapy and stem cell transplantation (SCT) have been shown to be associated with an altered body composition, i.e. increased abdominal fat(9, 10). For other potential risk factors, e.g. corticosteroid use, study results are conflicting(5, 6). Lastly, survivors may experience visual, neurologic or orthopedic problems causing decreased ability to exercise.

Overweight and obesity are mostly reported as high body mass index (BMI)(7). BMI can however underestimate the true adiposity status, or overestimate overweight in muscular people. Other overweight measurements include waist circumference (WC) (11-13), the ratio of waist to hip circumference (WHR)(13) and the ratio of waist to height (WHtR)(14). These methods involve easily performed assessments that are specifically directed at measuring abdominal fat, which in the general population correlates better with fat% than BMI(11-13). Still, these methods are not ideal, and can be more challenging in survivors treated with abdominal radiotherapy, in whom due to tissue damage waist circumference does not reflect the total fat%(9, 15, 16). Fat percentage (fat%) on dual-energy X-ray absorptiometry (DXA) is regarded as a more accurate method for overweight assessment(17), but performing it as standard of care can be a logistic and financial challenge. Therefore, BMI is still the primary method to measure overweight in the current Dutch surveillance guideline for survivors. Serum adiponectin might serve as another overweight diagnostic. Low adiponectin is associated with overweight and higher intra-abdominal fat in the general population(18), but studies on its diagnostic value for assessing overweight in survivors are lacking(19). The underestimation of overweight in survivors who in fact may have an increased risk of developing subsequent health problems is a major challenge in surveillance of survivors and prohibits adequate counseling.

We studied overweight in the first treated Dutch national cohort of long-term survivors of childhood cancer, aiming to assess prevalence based on a nationwide survivor cohort and compare this to the general population, to further clarify risk factors for developing overweight, and to assess optimal overweight measurement methods for future survivor surveillance.

METHODS

Study cohort

This study is part of the nationwide Dutch Childhood Cancer Survivor Study – Long Term Effects (DCCSS-LATER) study(20). The first national cohort of all adult survivors treated in a pediatric oncology center in The Netherlands between 1963 and 2002 was invited (N=4,671, Figure 1). This study was approved by the Amsterdam UMC Medical Research Ethics Committee, The Netherlands (toetsingonline.nl, NL32117.018.10). Written informed consent was obtained from all participants.

Reference cohort

Data on BMI from the Dutch LifeLines study cohort served as reference population(21). This is a large three-generational cohort of which we included all members aged between 18 and 65 years, without a history of cancer.

Data collection and definitions of overweight

An overview of definitions of outcomes and covariates is provided in Supplemental Table 1. During a late effects clinic visit (2016-2020), height, weight (adjusted for amputation when applicable) and waist and hip circumference were measured. From this data BMI, WHR and WHtR were calculated. Total body DXA scans were performed in survivors <40 years to measure fat% (converted to Hologic values when applicable). Overweight, obesity and morbid obesity, as primarily defined by BMI, were defined as ≥ 25 , ≥ 30 and ≥ 35 kg/m², respectively(7). Thresholds for overweight for the other modalities were (in men/women) WC $\geq 94/\geq 80$ cm(11) and $\geq 102/88$ cm(12), WHR $\geq 0.90/0.85(13)$, WHtR $\geq 0.50/0.50(14)$ and DXA fat% $\geq 25/30\%(22)$. During outpatient clinic visit, venous samples were drawn after overnight fasting, for assessment of high-molecular-weight-(HMW)-adiponectin and insulin-like growth factor 1 (IGF-1) levels. Smoking habits were collected in

a questionnaire and physical activity information was acquired with the SQUASH questionnaire. From the medical records we extracted height and weight at cancer diagnosis to calculate BMI at diagnosis. If present, we also extracted data on GHD tests and treatment. Childhood cancer treatment data were collected on a national level in our central database.

Statistical analysis

Analyses were conducted in R version 3.6.3 (R Foundation, Vienna, Austria)(23). Demographic and treatment characteristics were compared between participants and non-participants. Sex-specific prevalence of overweight, obesity and morbid obesity in men and women was compared to that in the Lifelines reference population using chi-squared or Fisher's exact test. Risk factors for these outcomes were assessed using logistic regression. Significant variables (p<0.05) in univariable analysis and patient factors known to be relevant from literature (age, sex, smoking, physical activity) were included in the multivariable model. Multicollinearity was inspected with the variance inflation factor. As GHD is a potential mechanism for overweight caused by therapies, it was analyzed in a separate model. Sensitivity analyses were performed to inspect the role of missing data. The influence of overweight measurement methods was evaluated by assessing overweight prevalence according to each, and discrepancies were calculated with BMI <25kg/m² and high fat% as references. We studied what factors were associated with disguised overweight, in logistic regression models with all treatment groups as predictors.

RESULTS

Study cohort description

In total, 2,338 very long-term survivors (52.1% male) participated (50.1% participation rate) (Figure 1, Table 1). Mean age was 35.5 (standard deviation \pm 9.3) years, and mean follow-up time was 28.3 (\pm 8.4) years. The most common childhood cancer diagnosis groups had been leukemias (35.5%), lymphomas (19.2%), renal tumors (11.5%) and central nervous system (CNS) tumors (9.1%). Participants had more often received cranial and abdominal radiotherapy, alkylating agents, anthracyclines, platinum derivatives and vinca alkaloids (compared to non-responders (n=1,599) only, this data was unavailable for survivors who declined participation), but did not differ regarding age at diagnosis, treatment period, age at invitation and follow-up time (Supplemental Table 2).

The Lifelines reference cohort consisted of 132,150 subjects (58.6% female), with mean age of 42.0, SD \pm 11.0 years (Supplemental Table 3).

Prevalence of overweight, obesity and morbid obesity

Based on BMI values, overweight prevalence was 46.3% (males) and 44.3% (females). For obesity this was 11.2% and 15.9%, and for morbid obesity 2.4% and 5.4%, respectively. Compared to LifeLines, there was a higher overweight rate among women aged 50+ (68.7 vs 57.0%, p=0.032) and a higher morbid obesity rate among men aged 50+ (6.7 vs 2.5%, p=0.040) (Figure 2). Lower rates were observed for overweight in male survivors aged 30-40 (44.0 vs 56.3%, p<0.001) and 40-50 (59.7 vs 66.9%, p=0.012) and for obesity in male survivors aged 30-40 (8.5 vs 12.2%, p=0.019).

Overweight was most common in survivors of a CNS tumor (52.3%), retinoblastoma (50.0%) and lymphomas (49.4%). Obesity and morbid obesity rates were particularly high after CNS tumor (22.1% and 10.8%, respectively) and retinoblastoma (30.0% and 10.0%) (Supplemental figure 1).

Risk factors for overweight, obesity and morbid obesity

The strongest risk factor for overweight in univariable analysis was overweight at cancer diagnosis (odds ratio (OR) 3.16, 95%CI 2.08-4.79) (Supplemental table 4). CrRT was another risk factor, with stronger association for lower dose CrRT (OR for 1-25 Gy 3.58 (95%CI 2.60-4.94), for >25 Gy 1.75 (95%CI 1.31-2.34)) as was GHD (OR 2.28, 95%CI 1.59-3.27). Corticosteroids use was not associated with overweight. In a sex-stratified analysis we observed that treatment-related risk estimates were similar in men and women (data not shown).

In multivariable analysis, risk factors for overweight were overweight at diagnosis (OR 3.83, 95%CI 2.19-6.69), CrRT 1-25 Gy (OR 3.21, 95%CI 1.99-5.18) and older age at clinic visit (OR 1.03, 95%CI 1.01-1.05 per year) (Table 2). In the separate model GHD was also associated with overweight (OR 1.61, 95%CI 1.00-2.59).

Female sex, CrRT and GHD were identified as independent risk factors for obesity and morbid obesity (Supplemental table 5). Inspection of variance inflation in all multivariable models suggested that multicollinearity was not present. Sensitivity analyses showed similar results (data not shown).

Assessment of overweight using different methods

In the overall cohort, BMI, WC (\geq 94/80cm), WHR and WHtR revealed an overweight prevalence in the range 45.3%-49.0% (Supplemental table 6). When stratified by sex, high WC was observed less in men (33.6%) and more in women (60.8%). High fat%

on DXA identified overweight in 83.7% of women and 58.4% of men as well as in 77.7% of abdominally irradiated survivors. When using Gallagher's threshold for high fat%, higher prevalence was observed particularly in men (82.3%, and 66.1% in women). When using Heo's threshold, prevalence remained similar to BMI, WHR and WHtR.

There were differences in the classification of overweight according to different methods. High WC was observed in 11.2% of survivors with normal BMI, high WHR in 18.8%, and high WHtR in 9.3%. DXA identified an additional 31.6% survivors with normal BMI as overweight, and this was 39.3% in the abdominally irradiated group.

When compared to fat% on DXA scan, underestimation of overweight was comparable with BMI, WC, WHR and WHtR for the whole cohort (range 29.9-32.9%) (Figure 2). BMI, WHR and WHtR underestimated overweight more often in women (up to 47.4%). After abdominal irradiation, the percentage of underestimation of overweight was highest for the methods that use waist circumference (range 45.9-63.1%).

Consequently, in the regression model with outcome underestimation of overweight with BMI, female sex was a strong risk factor (OR 3.02, 95%CI 2.27-4.03) (Table 3, Supplemental table 7). Therapies significantly associated with underestimation were total body irradiation (TBI, OR 9.06, 95%CI 2.41-34.04) and anthracyclines (second tertile OR 1.57 (95%CI 1.02-2.24), a trend for the highest tertile 1.48 (95%CI 0.99-2.23)). Abdominal radiotherapy was a major risk factor for misclassification with the methods that use WC (for WC OR up to 3.06 (95%CI 1.64-5.72, data for WHR and WHtR were similar (data not shown)). In this model anthracyclines (OR highest tertile 1.58, 95%CI 1.12-2.23) and platinum (OR 1.66, 95%CI 1.21-2.29) also emerged as independent risk factors.

Adiponectin as marker for overweight

Low HMW-adiponectin was present in only 4.5% of survivors. Consequently, only 1.9% of survivors with normal BMI were additionally diagnosed as overweight. Also, when compared to fat%, low adiponectin underestimated overweight in 66.1% of survivors, with higher rates in women (78.9%) and abdominally irradiated survivors (72.3%). Sensitivity and specificity for low adiponectin compared to high BMI were 6.2% and 96.5%, respectively. When compared to high fat% on DXA, sensitivity was 6.1% and specificity 97.1%.

DISCUSSION

This study shows that overweight occurs in almost half of all adult long-term childhood cancer survivors, and that associated factors include overweight at diagnosis, CrRT and GHD. We also show that DXA scans identified overweight in an additional 30% of survivors not identified with conventional methods such as BMI and WC.

There was a significantly higher prevalence in our cohort for overweight among women aged 50+ and for morbid obesity among men aged 50+. Our findings may suggest that the increase in prevalence per age category is more pronounced in survivors than in the general population. This was particularly the case in women, which may be partly attributed to the effect of menopause. So, while aging, prevalence of overweight, obesity and morbid obesity may be higher in survivors, increasing their risk of overweight-associated comorbidity and mortality. It could also be that this increase slows down later on, as was observed for cardiac disease in survivors(24). Another potential reason for increased prevalence in the oldest age groups, is that younger participants were treated more recently, and may therefore suffer from less treatment-related side effects. ALL is the most prevalent cancer type in this cohort. Use of prophylactic CrRT was reduced in the 1980-1990s with the introduction of the ALL-6 and ALL-9 protocols. Accordingly, ALL survivors who underwent CrRT are overrepresented among the oldest survivors in this study. Longitudinal follow-up is required to elucidate this.

The only other study so far on overweight in a heterogeneous nationwide survivor cohort, the Swiss Childhood Cancer Survivor Study (N=2,365), found a prevalence of 26% after fifteen years, which was not different from sibling and general population controls(4). The lower prevalence compared to our study might be due to the shorter follow-up (15 v 28 years), since overweight prevalence increases with aging. The Childhood Cancer Survivor Study (CCSS)(3) and the St Jude Lifetime Cohort Study (SJLIFE)(5), had a comparable follow-up time (~24 years). Higher overweight rates in these studies seem in part to be due to an already higher general population overweight prevalence. The CCSS found no difference with siblings. In SJLIFE, the general population obseivy prevalence was already twice as high as in our control group. Still, the authors observed more obesity among survivors (standardized morbidity ratio 1.14). It is clear that follow-up time and general population risk impede a full comparison between studies.

We observed that overweight prevalence was highest among survivors of CNS tumors, but differences between diagnosis groups were small. Obesity and morbid obesity prevalence was clearly higher after CNS tumors. This is in line with previous

findings(4, 25, 26) and likely related to damage to the hypothalamus and pituitary gland due to tumor and treatment. Whereas other studies also observed a higher overweight prevalence after ALL(6), we did not, as there was no excess overweight in ALL survivors unexposed to CrRT.

We further explored the role of CrRT and confirmed the association with overweight, obesity and morbid obesity, as has been reported multiple times(3-6). In multivariable analysis, the effect of dosages <25Gy on overweight was stronger. This group most likely consists of survivors who received craniospinal radiotherapy. The higher dose group more likely received local radiotherapy to a brain tumor, with the exception of medulloblastoma and a few ALL survivors who received higher dose craniospinal radiation. In the low dose group the hypothalamus and pituitary may therefore have been in the radiation field more often. Radiotherapy affects the somatotropic axis first, with GHD occurring from 15-20 Gy(27). In our regression models GHD was independently associated with overweight, obesity and morbid obesity. Another potential mechanism is hypogonadism, which can occur after hypothalamic and gonadal radiation and alkylating chemotherapy(28). Alkylating agents were no independent risk factor in our analysis. Unfortunately no data were available yet on presence of hypogonadism. This and other potential mechanisms will be further explored in additional studies in this cohort.

In our cohort, corticosteroids use during cancer treatment did not impact overweight or obesity. Apparently, the short-term metabolic side effects of these compounds, have not led to overweight on the long-term. Two previous studies among ALL survivors, 10 and 12.7 years after treatment, still observed an association with corticosteroids and overweight(29, 30). This may suggest that steroids exposure does influence the middle-long-term, but that this resolves later on. In the SJLIFE cohort, the association between glucocorticoid treatment and obesity was significant even after 24.6 years(5). This difference with our cohort might be explained by environmental and lifestyle factors in the United States that make it harder to lose weight after weight gain during adolescence. Our study is the first to investigate the effect of total cumulative dose steroids on development of overweight, obesity and morbid obesity in a large, unselected, national cohort of very long-term survivors of all types of childhood cancer. The absence of an association therefore appears to be convincing.

Overweight at cancer diagnosis emerged as a strong patient factor associated with overweight and obesity. Other studies have observed similar results(5, 6). This may reflect a genetic susceptibility to weight gain, socio-economic status and associated lifestyle, and for some brain tumors a hypothalamus/pituitary damaging effect prior to diagnosis. Smoking was also an independent risk factor for overweight, as is in the general population. Our findings show that these patient factors have an additional effect on top of historical treatment and emphasize that they need to be acknowledged in surveillance.

Our third aim was to explore how adequate currently used methods assess overweight in survivors. After abdominal irradiation, WC and WHR do not provide optimal overweight assessment, and that fat% assessment with DXA may be more valuable(9, 15, 16). Furthermore, short stature due to GHD, reduced bone mineral density, sarcopenia, and amputations can hinder the estimation of overweight with BMI, WC, WHR and WHtR. We show that waist-circumference-based methods classified a substantial number of survivors with normal BMI as overweight. Moreover, we show that DXA scan measured overweight in an additional 30% of survivors, and even 40% in the subgroup of abdominally irradiated survivors. Underestimation occurred more often in females, which is also observed in the general population(31). The underestimation rate was similar for all types of childhood cancer except hepatic tumors (Supplemental figure 2). Previous studies observed similar underestimation rates when comparing anthropometric measurements to DXA measured fat(9, 15, 16). Karlage used the obesity threshold for BMI, hence, the observed discrepancy was higher(16). Subsequently, we identified subgroups of survivors that may benefit from assessing overweight with DXA. These include survivors treated with abdominal irradiation and TBI. Altered fat distribution has been described after SCT preconditioned with TBI(10). Furthermore, for the first time, we identified anthracyclines and platinum chemotherapy to be associated with disguised overweight. How these therapies might lead to an altered body composition is yet unknown. Anthracyclines were associated with low BMI in a previous study in Dutch survivors, but the mechanism, e.g., sarcopenia, was not clear(32). In another study in the DCCSS-LATER cohort, an association between platinum and meningioma appeared to be confounded by medulloblastoma survivors also receiving high dose CrRT(33), but in a sensitivity analysis excluding these survivors the effect remained. Hence, survivors with BMI or WC near the upper limit of normal and who received abdominal radiation or TBI, and possibly anthracyclines and platinum chemotherapy, may benefit from a DXA scan as most reliable diagnostic method.

It was remarkable how our analysis of the three previously reported thresholds for high fat% influenced overweight prevalence. The common use of 25% for men and 30% for women may in part be caused by a misinterpretation of a World Health Organization statement on body fat(34). In the two other studies we compared, the authors attempted to calibrate BMI values of 25 and 30kg/m^2 to corresponding DXA fat% values(35, 36). This yields a grey area until 28% in men and 40% in women with somewhat unclear overweight diagnosis.

Low HMW-adiponectin could serve as alternative marker for overweight. HMWadiponectin is the most biologically active isoform of adiponectin, an adipokine that enhances insulin sensitization and suppresses inflammation and cell death(37). In the general population low adiponectin is associated with overweight, increased intra-abdominal fat, as well as metabolic syndrome, diabetes and atherosclerotic disease(18). In our recent systematic literature review we proposed that it may be used to replace the overweight component of metabolic syndrome in survivors with unreliable WC after abdominal radiotherapy(19). However, less than 5% of the cohort had low adiponectin, so it was not a sensitive marker for overweight, and particularly compared to DXA many overweight survivors are missed. An explanation of this finding could be that the study cohort was still relatively young and that as the cohort ages, low adiponectin levels may develop. Alternatively, some underlying mechanisms for overweight development may be different in survivors and less correlated with adiponectin than in the general population.

A few limitations of this study require consideration. First, prescribed radiotherapy dose is not the same as dose received by organs involved in metabolic side effects. For full CrRT the prescribed dose can be assumed to reflect dose received by the hypothalamus and pituitary, but for other malignancies such dosimetric data were not available yet. Second, DXA scans were intentionally only performed in survivors <40 years to avoid bias caused by menopause, but it may limit full generalizability of our DXA related findings. Third, we did not have data on hypogonadism. Fourth, due to historic changes in treatment protocols and the cross-sectional design of this study, treatment exposure is correlated with attained age at study participation, and interpretation of findings regarding the impact of attained age require caution.

To further deepen our understanding of late effects of childhood cancer, future perspectives may include longitudinal designs shedding more light on potential causative mechanisms, dosimetry for specific organs, and further elucidated pathophysiological mechanisms. Also, obesity is often not a sole side effect, but related to other cardiovascular risk factors – insulin resistance, dyslipidemia and hypertension – as metabolic syndrome, further increasing the risk of diabetes and cardiovascular disease(11, 12). This will be further explored in our study. Lastly, it is important not only to identify risk factors for overweight and other metabolic sequelae, but also to invest in lifestyle interventions.

In conclusion, in this study in our nationwide Dutch cohort of the first treated (1963-2002) childhood cancer survivors, we show that overweight occurs in almost half of all long-term survivors, and that overweight at diagnosis, CrRT and GHD, but not corticosteroids, are associated with long-term overweight. Of several assessment

methods, DXA was most sensitive, as it identified overweight in an additional 30% of survivors, particularly those treated with abdominal irradiation, TBI, anthracyclines and platinum chemotherapy. HMW-adiponectin did not have added diagnostic value.

ACKNOWLEDGMENTS

Personal Thanks

We thank all participating survivors, physicians, research nurses and data managers for their contribution. In addition, the authors wish to acknowledge services of the Lifelines Cohort Study, the contributing research centers delivering data to Lifelines, and all the study participants.

Funding and Assistance

This study was supported by the Children Cancer-free Foundation (KiKa), KiKaRoW Association, and the ODAS Foundation.

Conflict of Interest

All authors have no conflicts of interest to declare.

Author Contributions and Guarantor Statement

VGP, DTCdW, MMvdH-E and SJCMMN contributed to the study funding, concept, and design. VGP, JEcA, DTCdW, RAJN, DB, MvdH-vdL, ACHdV, ML, HJvdP, LCMK, JJL, EcD-dB, WJET, MMvdH-E and SJCMMN contributed to data acquisition. VGP, SJCMMN and MMvdH-E contributed to data analysis and interpretation. VGP, MMvdH-E and SJCMMN drafted the manuscript. VGP, JEvA, DTCdW, MF, MB, RAJN, GORJ, DB, MvdH-vdL, ACHdV, ML, HJvdP, SMFP, CMR, ABV, LCMK, JJL, EvDdB, WJET, HMvS, MMvdH-E and SJCMMN contributed to manuscript revision and approval. VGP, JEvA, DTCdW, MB, MF, MvdH-vdL, SJCMMN, and MMvdH-E had full access to all data in the study and take responsibility for data integrity, verification, and analysis. All authors had access to all the data reported in the study and accept responsibility to submit for publication.

Data Sharing Statement

The data underlying this Article were provided by the DCCSS-LATER consortium under license. Data will be shared on reasonable request to the corresponding author with permission of the DCCSS-LATER consortium. The data are not publicly available due to privacy or ethical restrictions.

TABLES AND FIGURES

Table 1. Baseline characteristics of the study cohort

	Entire cohort	Male	Female	Comparison male vs female
Number of participants	2338	1198 (51.2%)	1140 (48.8%)	
Patient, cancer and treatment characteristics Age and follow-up time				
Age at clinic visit (y) (mean (SD))	35.5 (9.3)	35.5 (9.0)	35.5 (9.5)	0.87
Age at clinic visit categorized (y) (%)				0.77
18/30	728 (31.3)	371 (31.0)	357 (31.6)	
30/39	883 (38.0)	462 (38.7)	421 (37.2)	
40+	715 (30.7)	362 (30.3)	353 (31.2)	
Follow-up time (y) (mean (SD))	28.3 (8.4)	28.0 (8.2)	28.5 (8.7)	0.17
Follow-up time categorized (y) (%)				0.63
10/19	472 (20.2)	241 (20.1)	231 (20.3)	
20/29	928 (39.7)	489 (40.8)	439 (38.5)	
30/39	693 (29.6)	352 (29.4)	341 (29.9)	
40/49	220 (9.4)	103 (8.6)	117 (10.3)	
50/59	25 (1.1)	13 (1.1)	12 (1.1)	
Childhood cancer characteristics				
Childhood cancer diagnosis per ICCC-3 site group (%)				<0.001
1 Leukemias, myeloproliferative diseases and myelodysplastic diseases	831 (35.5)	437 (36.5)	394 (34.6)	
2 Lymphomas and reticuloendothelial neoplasms	448 (19.2)	288 (24.0)	160 (14.0)	
3 CNS and miscellaneous intracranial and intraspinal neoplasms	213 (9.1)	107 (8.9)	106 (9.3)	
4 Neuroblastoma and other peripheral nervous cell tumors	135 (5.8)	44 (3.7)	91 (8.0)	
5 Retinoblastoma	11 (0.5)	5 (0.4)	6 (0.5)	
6 Renal tumors	269 (11.5)	111 (9.3)	158 (13.9)	
7 Hepatic tumors	18 (0.8)	9 (0.8)	9 (0.8)	
8 Bone tumors	136 (5.8)	63 (5.3)	73 (6.4)	
9 Soft tissue and other extraosseous sarcomas	167 (7.1)	93 (7.8)	74 (6.5)	
10 Germ cell tumors, trophoblastic tumors, and neoplasms of gonads	76 (3.3)	28 (2.3)	48 (4.2)	
11 Other malignant epithelial neoplasms and malignant melanomas	31 (1.3)	12 (1.0)	19 (1.7)	
12 Other and unspecified malignant neoplasms	3 (0.1)	1 (0.1)	2 (0.2)	
Age at diagnosis (y) (mean (SD))	6.71 (4.69)	6.90 (4.63)	6.51 (4.74)	0.045
Age at diagnosis categorized (y) (%)				0.14

Chapter 5

Table 1. Continued

	Entire cohort	Male	Female	Comparison male vs female
0 /5	1071 (45.8)	525 (43.8)	546 (47.9)	
5/10	651 (27.8)	347 (29.0)	304 (26.7)	
10/15	481 (20.6)	261 (21.8)	220 (19.3)	
15/18	135 (5.8)	65 (5.4)	70 (6.1)	
Treatment period (%)				0.15
1960/69	33 (1.4)	15 (1.3)	18 (1.6)	
1970/79	317 (13.6)	142 (11.9)	175 (15.4)	
1980/89	732 (31.3)	386 (32.2)	346 (30.4)	
1990/99	1013 (43.3)	528 (44.1)	485 (42.5)	
2000/09	243 (10.4)	127 (10.6)	116 (10.2)	
Height at cancer diagnosis (cm) (mean (SD))	120.4 (30.6)	122.0 (30.5)	118.6 (30.7)	0.016
Weight at cancer diagnosis (kg) (mean (SD))	25.6 (15.4)	26.1 (15.2)	25.1 (15.7)	0.16
BMI at cancer diagnosis (kg/m2) (mean (SD))	16.22 (2.88)	16.14 (2.90)	16.30 (2.85)	0.26
Overweight at cancer diagnosis	116 (6.1)	51 (5.1)	65 (6.8)	0.11
Obesity at cancer diagnosis	22 (1.1)	12 (1.2)	10 (1.1)	0.75
Cancer treatment characteristics				
Cranial radiotherapy (%)	432 (18.5)	239 (20.1)	193 (16.9)	0.053
Cranial radiotherapy categorized (%)				0.008
No	1898 (81.6)	952 (80.1)	946 (83.2)	
1-25 Gy	204 (8.8)	100 (8.4)	104 (9.1)	
25+ Gy	223 (9.6)	136 (11.4)	87 (7.7)	
Total body irradiation	88 (3.8)	63 (5.3)	25 (2.2)	<0.001
Abdominal/pelvic radiotherapy (%)	201 (8.6)	84 (7.1)	117 (10.3)	0.006
Abdominal/pelvic radiotherapy categorized (%)				0.006
No	2126 (91.4)	1104 (93.0)	1022 (89.8)	
1-29 Gy	112 (4.8)	41 (3.5)	71 (6.2)	
30+ Gy	87 (3.7)	42 (3.5)	45 (4.0)	
Alkylating agents (CED) (%)	1175 (55.3)	648 (59.0)	527 (51.3)	<0.001
Cyclophosphamide equivalent dose categorized (%)				0.001
No	951 (44.7)	451 (41.0)	500 (48.7)	
1-4000mg/m2	445 (20.9)	229 (20.8)	216 (21.0)	
4000-8000mg/m2	340 (16.0)	201 (18.3)	139 (13.5)	
8000+ mg/m2	390 (18.3)	218 (19.8)	172 (16.7)	
Anthracyclines (DED) (%)	1172 (53.2)	635 (55.7)	537 (50.5)	0.014
Doxorubicin equivalent dose categorized (%)				0.041
No	1032 (46.8)	505 (44.3)	527 (49.5)	
tertile 1 (range 9-138mg/m2)	391 (17.7)	200 (17.5)	191 (18.0)	
tertile 2 (range 139-273mg/m2)	391 (17.7)	216 (18.9)	175 (16.4)	

Table 1. Continued

	Entire cohort	Male	Female	Comparison male vs female
tertile 3 (range 275-1764mg/m2)	390 (17.7)	219 (19.2)	171 (16.1)	
Corticosteroids (SED) (%)	1190 (50.9)	675 (56.3)	515 (45.2)	<0.001
Steroid equivalent dose categorized (%)				<0.001
No	1148 (49.1)	523 (43.7)	625 (54.8)	
1-10g/m2	1083 (46.3)	598 (49.9)	485 (42.5)	
10+ g/m2	107 (4.6)	77 (6.4)	30 (2.6)	
Asparaginase (%)	584 (25.0)	316 (26.4)	268 (23.6)	0.11
Platinum derivatives (%)	311 (13.3)	142 (11.9)	169 (14.9)	0.03
Vinca alkaloids (%)	1829 (78.3)	966 (80.7)	863 (75.8)	0.004
Amputation (%)	67 (2.9)	24 (2.0)	43 (3.8)	0.010
Amputation type				0.084
Elbow/upper arm	3 (0.1)	1 (0.1)	2 (0.2)	
Shoulder/scapula	4 (0.2)	3 (0.3)	1 (0.1)	
Ankle/lower leg	9 (0.4)	3 (0.3)	6 (0.5)	
Knee/upper leg	42 (1.8)	13 (1.1)	29 (2.5)	
Hip/pelvis	9 (0.4)	4 (0.3)	5 (0.4)	
Allogeneic SCT (%)	99 (4.3)	65 (5.4)	34 (3.0)	0.004
Measurements assessed at clinic visit Physcial examination				
Height (cm) (mean (SD))	173.6 (10.1)	179.5 (8.8)	167.2 (7.2)	<0.001
Weight (kg) (mean (SD))	76.1 (16.0)	81.1 (15.3)	70.8 (15.0)	<0.001
BMI (kg/m2) (mean (SD))	25.20 (4.65)	25.10 (4.15)	25.30 (5.12)	0.33
BMI >25kg/m2	1020 (45.3)	535 (46.3)	485 (44.3)	0.33
BMI >30kg/m2	303 (13.5)	129 (11.2)	174 (15.9)	0.001
BMI >35kg/m2	87 (3.9)	28 (2.4)	59 (5.4)	<0.001
Waist circumference (cm) (mean (SD))	87.2 (12.7)	89.8 (11.6)	84.5 (13.3)	<0.001
Hip circumference (cm) (mean (SD))	99.2 (10.3)	98.5 (8.8)	99.9 (11.7)	0.001
High waist circumference JIS	1040 (46.8)	382 (33.6)	658 (60.8)	<0.001
High waist circumference NCEP	543 (24.5)	159 (14.0)	384 (35.5)	<0.001
Waist hip ratio (mean (SD))	0.88 (0.09)	0.91 (0.07)	0.85 (0.09)	<0.001
High waist hip ratio	1082 (49.0)	608 (53.8)	474 (43.9)	<0.001
Waist height ratio (mean (SD))	0.50 (0.07)	0.50 (0.06)	0.51 (0.08)	0.087
High waist height ratio	1042 (47.0)	524 (46.1)	518 (47.9)	0.41
Mean systolic blood pressure (mmHg) (mean (SD))	124 (16)	126 (15)	121 (16)	<0.001
Mean diastolic blood pressure (mmHg) (mean (SD))	76 (10)	77 (11)	75 (10)	<0.001
Hypertension (%)	806 (36.1)	470 (41.2)	336 (30.8)	<0.001

DXA scan, laboratory and questionnaire data

	Entire cohort	Male	Female	Comparison male vs female
Fat% on DXA scan (mean (SD))	31.1 (7.7)	26.3 (5.3)	36.4 (6.2)	<0.001
High fat% on DXA 25/30% (%)	1150 (70.4)	503 (58.4)	647 (83.7)	<0.001
High fat% on DXA Gallagher (%)	649 (39.7)	395 (45.9)	254 (32.9)	<0.001
High fat% on DXA Heo	346 (21.2)	206 (23.9)	140 (18.1)	0.004
Adiponectin (µg/ml) (mean (SD))	3.96 (2.31)	3.17 (1.78)	4.79 (2.50)	<0.001
Low adiponectin (%)	106 (4.5)	57 (4.8)	49 (4.3)	0.59
IGF1 (nmol/l) (mean (SD))	24.40 (8.04)	24.86 (7.66)	23.91 (8.41)	0.006
Low IGF1 (%)	27 (1.2)	15 (1.3)	12 (1.1)	0.67
Growth hormone deficiency (%)	116 (5.0)	55 (4.6)	61 (5.4)	0.40
Smoking (current or former) (%)	665 (32.5)	372 (35.4)	293 (29.4)	0.003
Minutes per week of moderate activity (median [IQR])	390 [135, 960]	450 [180, 1132.5]	352.50 [120, 845]	<0.001
Low physical activity (%)	436 (25.5)	191 (21.6)	245 (29.8)	<0.001

Table 1. Continued

BMI was available for 2,252 (96.3%) survivors, WC for 2,220 (95.0%), WHR for 2,210 (94.5%), WHtR for 2,218 (94.9%), a DXA scan was performed in 1,652 (70.7%), and adiponectin was measured in 2,219 (94.9%) survivors. Abbreviations: BMI = body mass index; IGF-1 = insulin-like growth factor 1; ICCC-3 = International Classification of Childhood Cancer, Third edition; CNS = central nervous system; SCT = stem cell transplantation

	Frequency of high BMI (n (%))	М	ultivariable m	odel 1	Mu	ltivariable m	odel 2
		OR	95% CI	P-value	OR	95% CI	P-value
Patient characteristics							
Age at clinic visit		1.03	1.01 - 1.05	0.006	1.04	1.02 - 1.06	<0.001
First tertile	258 (34.1%)						
Second tertile	326 (43.8%)						
Third tertile	435 (58.1%)						
Sex		0.88	0.69 - 1.13	0.32	0.96	0.76 - 1.20	0.70
Female	485 (44.3%)						
Male	535 (46.3%)						
Age at diagnosis		1.02	0.99 - 1.05	0.27	1.00	0.98 - 1.03	0.77
First tertile	299 (40.0%)						
Second tertile	340 (45.0%)						
Third tertile	381 (50.9%)						
Overweight at cancer diagnosis		3.83	2.19 - 6.69	<0.001	3.44	2.04 - 5.78	<0.001
Yes	80 (70.8%)						
No	782 (43.4%)						
Smoking (current or former)		1.26	0.97 - 1.63	0.079	1.24	0.97 - 1.58	0.086
Yes	338 (52.7%)						
No	590 (44.2%)						
Low physical activity		1.12	0.85 - 1.47	0.42	1.14	0.88 - 1.48	0.31
Yes	188 (44.8%)						
No	500 (40.3%)						
Treatment characteristics							
Cranial radiotherapy							
Yes	260 (63.1%)						
No	755 (41.2%)						
Cranial radiotherapy categorized				< 0.001			
No	755 (41.2%)	Ref					
1-25 Gy	143 (71.5%)	3.21	1.99 - 5.18	< 0.001			
25+ Gy	114 (55.1%)	1.67	0.98 - 2.87	0.061			
Total body irradiation		0.39	0.12 - 1.31	0.13			
Yes	12 (13.8%)						
No	1001 (46.5%)						
Abdominal/pelvic radiotherapy							
Yes	89 (44.5%)						
No	924 (45.2%)						
Abdominal/pelvic radiotherapy							
categorized							
No	924 (45.2%)						
1-29 Gy	52 (46.4%)						
30+ Gy	35 (41.4%)						
Alkylating agents (CED)							
Yes	477 (41.7%)						
No	431 (47.3%)						

Table 2. Multivariable logistic regression analysis of variables associated with overweight (BMI ≥25kg/m²)

	Frequency of high BMI (n (%))	М	ultivariable m	odel 1	Mu	ltivariable m	odel 2
		OR	95% CI	P-value	OR	95% CI	P-value
Cyclophosphamide equivalent dose categorized				0.050			
No	431 (47.3%)	Ref					
$1-4000 \text{mg/m}^2$	183 (42.3%)	0.99	0.73 - 1.35	0.95			
$4000-8000 \text{mg/m}^2$	140 (42.9%)	0.92	0.65 - 1.32	0.67			
$8000 + mg/m^2$	154 (40.1%)	0.62	0.44 - 0.88	0.008			
Corticosteroids							
Yes	536 (46.7%)						
No	484 (43.8%)						
Corticosteroids categorized							
No	484 (43.8%)						
$0-10g/m^2$	490 (46.9%)						
$10 + g/m^2$	46 (44.2%)						
Allogeneic SCT		0.56	0.16 - 1.91	0.35			
Yes	17 (17.7%)						
No	996 (46.5%)						
Comorbidity							
Growth hormone deficiency					1.61	1.00 - 2.59	0.048
Yes	88 (64.2%)						
No	932 (44.1%)						

Table 2. Continued

Model 1: patient and cancer treatment characteristics. Model 2: patient characteristics and growth hormone deficiency. For variables with more than two categories the overall p-value was calculated with the Wald test. Significant variables in univariable analysis and patient factors known to be relevant from literature (age, sex, smoking, physical activity) were included in the multivariable model. If for a treatment factor both the dichotomous and categorized variable were significant, only the latter was included.

Abbreviations: CI = confidence interval; OR = odds ratio; CI = confidence interval; Ref = reference



Figure 1. Comparison of overweight, obesity and morbid obesity between our survivor cohort and the LifeLines control cohort.

Prevalence of overweight (BMI >25kg/m2), obesity (>30kg/m2) and morbid obesity (>35kg/m2) in the LATER study cohort and LifeLines control cohort in men and women. Significantly different proportions are indicated by an asterisk.



Underdiagnosed

yesno

Figure 2. Comparison of overweight classification using BMI and waist circumference versus DXA scan

Underdiagnosis of overweight, when compared to fat percentage on DXA scan, with BMI in men (figure A) and in women (figure B), and with waist circumference in men (figure C) and in women (figure D). The red square and percentage in each figure indicate the survivors with high fat percentage and normal waist circumference or BMI.

SUPPLEMENTAL MATERIAL:SUPPLEMENTAL TABLE 1. DEFINITIONS OF OUTCOMES AND COVARIATES

Supplemental table 2.	Childhood cancer and treatment characteristics of study participants, non-participants and underlying cohort
Supplemental table 3.	Baseline characteristics of Lifelines reference cohort
Supplemental table 4.	Univariable logistic regression analysis of variables associated with overweight $(BMI \ge 25 \text{kg/m}^2)$
Supplemental table 5.	Multivariable logistic regression models for outcomes obesity (BMI $\geq 30 \text{kg/m}^2$) and morbid obesity (BMI $\geq 35 \text{kg/m}^2$)
Supplemental table 6.	Prevalence of overweight according to different definitions, and percentage of false-negative when compared to BMI or fat% on DXA
Supplemental table 7.	Univariable logistic regression of factors associated with outcome false-negative classification of overweight based on BMI and waist circumference, when compared to overweight based on fat% on DXA scan.
Supplemental table 8.	Multivariable logistic regression of factors associated with outcome false-negative classification of overweight based on BMI and waist circumference, when compared to overweight based on fat% on DXA scan.
Supplemental figure 1.	Flow chart of study participants
Supplemental figure 2.	Overweight, obesity and morbid obesity prevalence per childhood cancer diagnosis group
Supplemental figure 3.	Overweight outcomes per childhood cancer diagnosis group

Supplemental table 1. Definitions of a	utcomes and covariates		
Outcome or covariate	Measurement, calculation or derivation	Definition of aberrant value	Source
Body mass index (kg/m²)	$Weight(kg)/Height(m)^2$	Overweight ≥25, obesity ≥30 and morbid obesity ≥35kg/m ²	World Health Organization ⁽¹⁾
Weight adjusted for amputation (kg)	Percentage added to body weight based on amputation level	п.а.	The Amputee Coalition ⁽²⁾
Waist circumference (cm)	n.a.	≥94cm in European men, ≥80cm in European women ≥102cm in men, ≥88cm in women	Joint Interim Statement ⁽³⁾ National Cholesterol Education Program ATP III ⁽⁴⁾
Waist hip ratio	Waist circumference(cm)/Hip circumference(cm)	≥0.90 in men, ≥0.85 in women	World Health Organization ⁽⁵⁾
Waist height ratio	Waist circumference(cm)/Height(cm)	≥0.50 in both sexes	Ashwell et al. ⁽⁶⁾
DXA fat% Lunar conversion to Hologic (%)	Conversion factor to account for difference in beam techniques	п.а.	Shepherd et al. $^{(7)}$
DXA far% (%)	n.a.	Primary threshold used in main analyses: 225% in men, 230% in women Age-dependent, 221.0-23.0% in white men, 233.0-35.0% in white women Age-dependent, 224.9-28.0% in white men, 237.0-39.9% in white women	Sommer et al. ⁽⁸⁾ , Shah et al. ⁽⁹⁾ Gallagher et al. ⁽¹⁰⁾ Heo et al. ⁽¹¹⁾
HMW adiponectin (µg/ml)	Lumipulse G immunoassay (Fujirebio, Tokyo, Japan)	≤1.46µg/ml in men, 0.90µg/ml in women	Reference values provided by the manufacturer
IGF-1	IDS-iSYS immunoassay (Immunodiagnostic Systems, Tyne & Wear, United Kingdom)	Age and sex-dependent reference values	Bidlingmaier et al. ⁽¹²⁾
GHD tests or treatment	Insulin tolerance, arginine, clonidine or levodopa-propranolol test, or treatment with growth hormone	n.a.	Medical records
Growth hormone deficiency		 Low IGF-1 and having had a brain tumor, CrRT or TBI, in order to include low IGF-1 due to hypothalamic and pituitary damage only, or Diagnosed with GHD or receiving treatment according to medical records 	Consensus between authors

Chapter 5

Supplemental table 1. Continued			
Outcome or covariate	Measurement, calculation or derivation	Definition of aberrant value	Source
Smoking status (yes or no)	General health questionnaire	Any (current or former) smoking	Consensus between authors
Physical activity	SQUASH questionnaire data, converted to average weekly exercise capacity following the corresponding protocol ^(13,13)	Not fulfilling norm of 150 minutes of medium exercise per week	World Health Organization ⁽¹⁶⁾
BMI at diagnosis (kg/m²)	Weight at diagnosis(kg)/Height at diagnosis(m) ²	Age and sex-dependent reference values for overweight and obesity	Obesity Taskforce Classification for childhood obesity ⁽¹⁷⁾
Cyclophosphamide equivalent dose (mg/m ²)	Equivalence ratios for alkylating agents	Categorization based on literature	Green et al. ⁽¹⁸⁾
Doxorubicin equivalent dose (mg/m ²)	Equivalence ratios for anthracyclines	Categorization based on tertiles	Feijen et al. ⁽¹⁹⁾
Steroid equivalent dose (mg/m²) Cranial radiotherapy dose (Gy)	Equivalence ratios for steroids n.a.	Categorization based on literature Categorized as lower or higher than 25Gy, in order to distinguish between irradiation to the whole cranium and local, higher dose radiotherapy	Parente et al. ⁽²⁰⁾ , Christoffersen et al. ⁽²¹⁾ Sherlock et al. ⁽²²⁾ , Crowne et al. ⁽²³⁾ , Blijdorp et al. ⁽²⁴⁾
 World Health Organization. Obes overweight. 	ity and Overweight fact sheet 2021 [updated	2021-06-09. Available from: https://www.wh	10.int/news-room/fact-sheets/detail/obesity-and-
2. The Amputee Coalition. [Available 1	from: https://www.amputee-coalition.org/limb-lo	ss-resource-center/resources-filtered/resources-	by-topic/healthy-living/about-bmi/.
3. Alberti KG, Eckel RH, Grundy SM, Deconsion: National Hoose, 1, 100	et al. Harmonizing the metabolic syndrome: a jo	int interim statement of the International Diab	etes Federation Task Force on Epidemiology and
for the Study of Obesity. Circulation	and 2009;120(16):1640-5.		osciciosis oociety, and michinarional association
4. National Cholesterol Education Pro	gram Expert Panel on Detection E, and Treatmen	t of High Blood Cholesterol in Adults (Adult Tr	eatment Panel III). Third Report of the National
5. World Health Organization. Waist c	JEV) Expert Fanel on Detection, Evaluation, and circumference and waist-hip ratio: report of a WF	Lireatment of High blood Cholesterol in Adul 40 expert consultation, Geneva, 8-11 Decemb	ts (Adult Treatment Fanel III). 2002. ber 2008. 2011.
6. Ashwell M, Gibson S. Waist to heigh	t ratio is a simple and effective obesity screening t	ool for cardiovascular risk factors: Analysis of da	tta from the British National Diet And Nutrition
Survey of adults aged 19-64 years. C	Dbes Facts. 2009;2(2):97-103.		
 Shepherd JA, Fan B, Lu Y, et al. A r DXA systems. J Bone Miner Res. 20 	nultinational study to develop universal standard 312;27(10):2208-16.	ization of whole-body bone density and comp	ssition using GE Healthcare Lunar and Hologic
8. Sommer I, Teufer B, Szelag M, et al	. The performance of anthropometric tools to det	ermine obesity: a systematic review and meta-a	malysis. Sci Rep. 2020;10(1):12699.
9. Shah NR, Braverman ER. Measurin	ig adiposity in patients: the utility of body mass i	ndex (BMI), percent body fat, and leptin. PLoS	5 One. 2012;7(4):e33308.
10. Gallagher D, Heymsfield SB, Heo N	A, Jebb SA, Murgatroyd PR, Sakamoto Y. Health	y percentage body fat ranges: an approach for d	leveloping guidelines based on body mass index.
Am J Clin Nutr. 2000;72(3):694-7(01.		
11	. Heo M, Faith MS, Pietrobelli A, Heymsfield SB. Percentage of body fat cutoffs by sex, age, and race-ethnicity in the US adult population from NHANES 1999-2004. Am J Clin		
----	---		
	Nutr. 2012;95(3):594-602.		
12	. Bidlingmaier M, Friedrich N, Emeny RT, et al. Reference intervals for insulin-like growth factor-1 (igf-i) from birth to senescence: results from a multicenter study using a new		
	automated chemiluminescence IGF-I immunoassay conforming to recent international recommendations. J Clin Endocrinol Metab. 2014;99(5):1712-21.		
13	. Wendel-Vos GC, Schuit AJ, Saris WH, Kromhout D. Reproducibility and relative validity of the short questionnaire to assess health-enhancing physical activity. J Clin Epidemiol.		
	2003;56(12):1163-9.		
14	. Wendel-Vos W, Schuit J. SQUASH: Short QUestionnaire to ASses Health enhancing physical activity. Bilthoven: Centrum voor Preventie en Zorgonderzoek Rijksinstituut voor		
	Volksgezondheid en Milieu; 2004.		
15	. Ainsworth BE, Haskell WL, Leon AS, et al. Compendium of physical activities: classification of energy costs of human physical activities. Med Sci Sports Exerc. 1993;25(1):71-80.		
16	. World Health Organization. Physical activity fact sheet 2020 [updated 2020-11-26. Available from: https://www.who.int/news-room/fact-sheets/detail/physical-activity.		
17	. Cole TJ, Bellizzi MC, Flegal KM, Dietz WH. Establishing a standard definition for child overweight and obesity worldwide: international survey. BMJ. 2000;320(7244):1240-3.		
18	. Green DM, Nolan VG, Goodman PJ, et al. The cyclophosphamide equivalent dose as an approach for quantifying alkylating agent exposure: a report from the Childhood Cancer		
	Survivor Study. Pediatr Blood Cancer. 2014;61(1):53-67.		
19	: Feijen EAM, Leisenring WM, Stratton KL, et al. Derivation of Anthracycline and Anthraquinone Equivalence Ratios to Doxorubicin for Late-Onset Cardiotoxicity. JAMA Oncol.		
	2019;5(6):864-71.		
20	· Parente L. Deflazacort: therapeutic index, relative potency and equivalent doses versus other corticosteroids. BMC Pharmacol Toxicol. 2017;18(1):1.		
21	. Christoffersen L, Gibson TM, Pui CH, et al. Cardiac autonomic dysfunction in survivors of childhood acute lymphoblastic leukemia: The St. Jude Lifetime Cohort Study. Pediatr		
	Blood Cancer. 2020;67(7):e28388.		
22	. Sherlock M, Toogood AA. Sensitivity of anterior pituitary hormones to irradiation. Expert Rev Endocrinol Metab. 2006;1(5):633-49.		
23	. Crowne E, Gleeson H, Benghiat H, Sanghera P, Toogood A. Effect of cancer treatment on hypothalamic-pituitary function. The Lancet Diabetes & Endocrinology. 2015;3(7):568-		
	76.		
24	. Blijdorp K, van den Heuvel-Eibrink M, Pieters R, et al. The limited screening value of insulin-like growth factor-I as a marker for alterations in body composition in very long-term		
	adult survivors of childhood cancer. Pediatr Blood Cancer. 2012;59(4):711-6.		

Supplemental table 2. Childhood cancer and treatment characteristics	of study participant	ts, non-participants ar	nd underlying col	Jort	
Characteristics	Participants (n=2,338)	Non-participants (n=2,333)	Underlying cohort (N=6,165)	Comparison participants vs non- participants	Comparison participants vs cohort
Sex				a	
Male	1198 (51.2)	1418 (60.8)	3433 (55.7)	P<0.001	P<0.001
Female	1140 (48.8)	915 (39.2)	2731 (44.3)		
Transgender	0	0	1(0.0)		
Primary childhood cancer (ICCC)					
Leukemias, myeloproliferative diseases and myelodysplastic diseases	831 (35.5)	779 (33.4)	2094 (34.0)	P<0.001	P<0.001
Lymphomas and reticulo endothelial neoplasms	448 (19.2)	416 (17.8)	1062 (17.2)		
CNS and miscellaneous intracranial and intraspinal neoplasms	213 (9.1)	341 (14.6)	844 (13.7)		
Neuroblastoma and other peripheral nervous cell tumors	135 (5.8)	103(4.4)	324 (5.3)		
Retinoblastoma	11 (0.5)	15 (0.6)	3 (0.5)		
Renal tumors	269 (11.5)	222 (9.5)	596 (9.7)		
Hepatic tumors	18 (0.8)	28 (1.2)	52 (0.8)		
Bone tumors	136 (5.8)	126 (5.4)	370 (6.0)		
Soft tissue and other extraosseous sarcomas	167 (7.1)	170 (7.3)	450 (7.3)		
Germ cell tumors, trophoblastic tumors, and neoplasms of gonads	76 (3.3)	97 (4.2)	232 (3.8)		
Other malignant epithelial neoplasms and malignant melanomas	31 (1.3)	35 (1.5)	102 (1.7)		
Other and unspecified malignant neoplasms	3 (0.1)	1(0.0)	6 (0.1)		
Age at diagnosis (yr)					
0-4	1071 (45.8)	1073 (46.1)	2727 (45.3)	P=0.94	P=0.66
5-9	651 (27.8)	630 (27.1)	1628 (27.1)		
10-14	481 (20.6)	489 (21.0)	1285 (21.4)		
15-17	135 (5.8)	136 (5.8)	376 (6.2)		
Transformet married					
10/3-10/0	33 (1 1)	34 (1 5)	(0) (1) 0) 1	D-070	D-0.03
1970-1979	317(13.6)	311(13.3)	978 (15.9)	7.00	
1980-1989	732 (31.3)	722 (30.9)	1931 (31.3)		

Supplemental table 2. Continued					
Characteristics	Participants (n=2,338)	Non-participants (n=2,333)	Underlying cohort (N=6,165)	Comparison participants vs non- participants	Comparison participants vs cohort
1990-1999	1013 (43.3)	1048(44.9)	2541 (41.2)		
2000-2001	243 (10.4)	218 (9.3)	596 (9.7)		
Age at invitation					
<18	NA	NA	49 (0.8)	P=0.43	P < 0.001
18-30	784 (33.5)	526 (22.5)	1313 (21.3)		
30-40	878 (37.6)	632 (27.1)	1511 (24.5)		
>40	676 (28.9)	441 (18.9)	1118 (18.1)		
Follow-up time since childhood cancer diagnosis					
10-20	472 (20.2)	433 (18.6)	981 (15.9)	P=0.48	P=0.96
20-30	928 (39.7)	968 (41.5)	1931 (31.3)		
30-40	693 (29.6)	680 (29.1)	1393 (22.6)		
40-50	220 (9.4)	232 (9.9)	460 (7.5)		
50-60	25 (1.1)	20 (0.9)	46 (0.7)		
Yes	1145 (49.1)	1212 (52.0)	3185 (51.7)	P=0.03	P=0.01
Amputation	67 (2.9)	41 (2.6)	١	P=0.57	
${f Radiotherapy}^a$					
Yes	905 (38.7)	86 (33.7)	2527 (41.0)	P<0.001	P=0.04
Cranial	432 (18.5)	248 (15.6)	ı	P=0.016	
Total body	88 (3.8)	30 (1.9)	١	P=0.001	
Abdominal/pelvic	201 (8.6)	93 (5.8)	ı	P=0.001	
Chemotherapy ⁴					
Yes	2057 (88.0)	1824 (78.2)	5005 (81.2)	P<0.001	P<0.001
Alkylating agents	1176 (55.3)	657 (45.6)	ı	P<0.001	
Anthracyclines	1172 (53.2)	676 (45.7)	١	P<0.001	
Corticosteroids	1190 (50.9)	757 (47.3)	١	P=0.088	
Asparaginase	584 (25.0)	357 (22.4)	١	P=0.057	
Platinum	311 (13.3)	175 (11.0)	ı	P=0.028	

Supplemental table 2. Continued					
Characteristics	Participants (n=2,338)	Non-participants (n=2,333)	Underlying cohort (N=6,165)	Comparison participants vs non- participants	Comparison participants vs cohort
Vinca alkaloids	1829 (78.3)	1128 (70.7)	1	P<0.001	
Bone marrow transplant a					
Allogeneic	99 (4.3)	59 (2.5)	231 (3.7)	P<0.001	P=0.70
Autologous	57 (2.5)	34 (1.5)	155 (2.5)		
Therapy					
No treatment	10(0.4)	28 (1.2)	61 (1.0)	P<0.001	P<0.001
Surgery only	150(6.4)	295 (12.6)	575 (9.3)		
Chemotherapy only (± surgery)	1272 (54.4)	1216 (52.1)	2967 (48.1)		
Radiotherapy only (± surgery)	121 (5.2)	178 (7.6)	484 (7.9)		
Chemotherapy and radiotherapy (± surgery)	784 (33.5)	607 (26.0)	2030 (32.9)		
^a For primary cancer and recurrences.					

Not all numbers and percentages add up to group total because data for refusers is not available.

NA Not applicable, no inclusion <18y

We compared participants with non-participants and with the underlying cohort, which contains all invited survivors as well as survivors who refused participation in an earlier phase of the study and survivors who died after the formation of the cohort.

	Total	Men	Women
Number of participants	132,150	54,733 (41.4%)	77,417 (58.6%)
Age (mean/SD)	42.0 (11.0)	42.5 (10.9)	41.7 (11.1)
Age (median/IQR)	43.0 (34.0-49.0)	43.0 (35.0-50.0)	43.0 (34.0-49.0)
Age range	18.0-65.0	18.0-65.0	18.0-65.0
Age 18-30y	20,812 (15.7%)	7,793 (14.2%)	13,019 (16.8%)
Age 30-40y	30,028 (22.7%)	12,488 (22.8%)	17,540 (22.7%)
Age 40-50y	49,006 (37.1%)	20,469 (37.4%)	28,537 (36.9%)
Age 50+y	32,304 (24.4%)	13,983 (25.5%)	18,321 (23.7%)
BMI (mean/SD)	25.93 (4.35)	26.29 (3.72)	25.68 (4.73)
BMI (median/IQR)	25.28 (22.95-28.09)	25.88 (23.81-28.28)	24.74 (22.39-27.97)
BMI range	13.38-73.57	14.43-73.57	13.38-66.26
Overweight	69,989 (53.0%)	33,274 (60.8%)	36,715 (47.4%)
Obesity	19,941 (15.1%)	7,656 (14.0%)	12,285 (15.9%)
Morbid obesity	4,966 (3.8%)	1,289 (2.4%)	3,677 (4.7%)

Supplemental table 3. Baseline characteristics of Lifelines reference cohort

Supplemental table 4. Univariable log	istic regression analysis of variables asso	ciated with overweight (BMI ≥25kg/m ²)

Dependent variable	Frequency of high BMI (n (%))		Univariable mo	dels
		OR	95% CI	P-value
Patient characteristics				
Age at clinic visit		1.05	1.04 - 1.06	<0.001
First tertile	258 (34.1%)			
Second tertile	326 (43.8%)			
Third tertile	435 (58.1%)			
Sex		0.92	0.78 - 1.09	0.33
Female	485 (44.3%)			
Male	535 (46.3%)			
Age at diagnosis		1.04	1.02 - 1.06	<0.001
First tertile	299 (40.0%)			
Second tertile	340 (45.0%)			
Third tertile	381 (50.9%)			
Overweight at cancer diagnosis		3.16	2.08 - 4.79	<0.001
Yes	80 (70.8%)			
No	782 (43.4%)			
Smoking (current or former)		1.41	1.17 - 1.70	<0.001
Yes	338 (52.7%)			
No	590 (44.2%)			
Low physical activity		1.20	0.96 - 1.50	0.11
Yes	188 (44.8%)			
No	500 (40.3%)			
Treatment characteristics				
Cranial radiotherapy		2.44	1.96 - 3.05	<0.001
Yes	260 (63.1%)			
No	755 (41.2%)			
Cranial radiotherapy categorized				<0.001
No	755 (41.2%)	Ref		
1-25 Gy	143 (71.5%)	3.58	2.60 - 4.94	

Supplemental table 4. Continued

Dependent variable	Frequency of high BMI (n (%))		Univariable mo	dels
		OR	95% CI	P-value
25+ Gy	114 (55.1%)	1.75	1.31 - 2.34	
Total body irradiation		0.18	0.10 - 0.34	<0.001
Yes	12 (13.8%)			
No	1001 (46.5%)			
Abdominal/pelvic radiotherapy		0.97	0.72 - 1.30	0.84
Yes	89 (44.5%)			
No	924 (45.2%)			
Abdominal/pelvic radiotherapy categorized				0.70
No	924 (45.2%)	Ref		
1-29 Gy	52 (46.4%)	1.05	0.72 - 1.54	
30+ Gy	35 (41.4%)	0.83	0.54 - 1.29	
Alkylating agents (CED)		0.80	0.67 - 0.95	0.012
Yes	477 (41.7%)			
No	431 (47.3%)			
Cyclophosphamide equivalent dose categorized				0.081
No	431 (47.3%)	Ref		
1-4000mg/m ²	183 (42.3%)	0.82	0.65 - 1.03	
4000-8000mg/m ²	140 (42.9%)	0.84	0.65 - 1.08	
8000+ mg/m ²	154 (40.1%)	0.75	0.59 - 0.95	
Corticosteroids		1.12	0.95 - 1.32	0.18
Yes	536 (46.7%)			
No	484 (43.8%)			
Corticosteroids categorized				0.35
No	484 (43.8%)	Ref		
$0-10g/m^2$	490 (46.9%)	1.13	0.96 - 1.34	
$10 + g/m^2$	46 (44.2%)	1.02	0.68 - 1.52	
Allogeneic SCT		0.25	0.15 - 0.42	<0.001
Yes	17 (17.7%)			
No	996 (46.5%)			
Comorbidity				
Growth hormone deficiency		2.28	1.59 - 3.27	<0.001
Yes	73 (64.0%)			
No	947 (44.3%)			

Dependent variable	0	utcome BMI >30)kg/m ²	0	utcome BMI >35	kg/m ²
	OR	95% CI	P-value	OR	95% CI	P-value
Age at clinic visit	1.02	0.99 - 1.05	0.12	1.04	0.98 - 1.09	0.18
Female sex	2.05	1.44 - 2.94	<0.001	3.33	1.58 - 7.02	0.002
Overweight at cancer diagnosis	2.85	1.60 - 5.08	<0.001	5.83	2.53 - 13.45	<0.001
Smoking (current or former)	1.29	0.90 - 1.86	0.17	1.32	0.66 - 2.63	0.43
Cranial radiotherapy categorized			<0.001			0.036
No	Ref			Ref		
1-25 Gy	2.48	1.45 - 4.23		3.06	1.24 - 7.50	
25+ Gy	1.94	1.11 - 3.39		1.97	0.70 - 5.55	
Total body irradiation	0.11	0.01 - 1.50	0.097			
Allogeneic SCT	1.52	0.22 - 10.61	0.67			
Growth hormone deficiency*	2.40	1.37 – 4.21	0.002	1.11	1.23 – 7.50	0.016

 $Supplemental \ table \ 5. \ Multivariable \ logistic \ regression \ models \ for \ outcomes \ obesity \ (BMI \ge 30 kg/m2) \ and \ morbid \ obesity \ (BMI \ge 35 kg/m2)$

*tested in separate model

All Men	valence of rerweight		4	Additional as ove	lly diagnose rweight	q		Undero	liagnosis of rweight	
	Women	Abd. irr.	All	Men	Women	Abd. irr.	IIV	Men	Women	Abd. irr.
High BMI (≥25kg/m ²) 45.3% 46.3%	6 44.3%	44.5%	Ref	Ref	Ref	Ref	31.6%	21.9%	42.4%	39.3%
High waist JIS (>80/94cm) 46.8% 33.6%	60.8%	42.6%	11.2%	3.5%	19.3%	10.2%	30.1%	30.0%	30.2%	46.8%
High waist NCEP ATP III (≥88/102cm) 24.5% 14.0%	6 35.5%	18.8%	2.3%	0.1%	4.6%	1.5%	48.7%	46.5%	51.1%	63.1%
High waist hip ratio (≥0.85/0.90) 49.0% 53.8%	6 43.9%	35.5%	18.8%	20.4%	17.0%	13.7%	32.9%	19.9%	47.4%	53.2%
High waist height ratio (≥0.50) 47.0% 46.1%	6 47.9%	40.1%	9.3%	8.5%	10.2%	6.6%	29.9%	20.2%	40.7%	45.9%
High fat% on DXA (≥30/25%) 70.4% 58.4%	6 83.7%	77.7%	31.6%	21.9%	42.4%	39.3%	Ref	Ref	Ref	Ref
High fat% on DXA (≥33-35/21-23%) 74.7% 82.3%	66.1%	74.1%	33.6%	39.5%	27.1%	35.7%	8.3%	0.0%	17.6%	10.7%
High fat% on DXA (≥37.0-39.9%/24.9- 51.4% 56.7% 28.0%)	ó 45.5%	58.0%	16.2%	20.8%	11.0%	22.3%	19.0%	1.7%	38.2%	19.6%
Low adiponectin (≤1.46/0.90µg/ml) 4.5% 4.8%	4.3%	4.5%	1.9%	2.2%	1.6%	2.0%	66.1%	54.6%	78.9%	72.3%

Dependent variable	Outcome false-nega	tive classif	fication with BMI	[
	Frequency of false-negative classification with BMI (n (%))		Univariable moo	lels
		OR	95% CI	P-value
Patient characteristics				
Age and sex				
Age at clinic visit		1.00	0.98 - 1.01	0.52
First tertile	203 (32.2%)			
Second tertile	199 (31.9%)			
Third tertile	111 (29.9%)			
Sex		2.62	2.11 - 3.26	<0.001
Female	327 (42.4%)			
Male	187 (21.9%)			
Age and weight at diagnosis				
Age at diagnosis		1.00	0.98 - 1.03	0.90
First tertile	183 (30.9%)			
Second tertile	177 (31.4%)			
Third tertile	154 (32.7%)			
Overweight at cancer diagnosis		0.56	0.33 - 0.96	0.035
Yes	18 (21.4%)			
No	437 (32.6%)			
Lifestyle				
Smoking (current or former)		0.73	0.57 - 0.93	0.011
Yes	128 (26.8%)			
No	321 (33.4%)			
Low physical activity		1.54	1.20 - 1.97	0.001
Yes	143 (39.7%)			
No	323 (30.0%)			
Treatment characteristics				
Cranial radiotherapy		0.76	0.56 - 1.02	0.068
Yes	68 (26.8%)			
No	445 (32.6%)			
Cranial radiotherapy categorized				0.12
No	445 (32.6%)	Ref		
1-25 Gy	28 (23.5%)	0.64	0.41 - 0.99	
25+ Gy	39 (30.0%)	0.89	0.60 - 1.31	
Total body irradiation		3.85	2.32 - 6.39	<0.001
Yes	42 (62.7%)			
No	471 (30.4%)			
Abdominal/pelvic radiotherapy		1.43	0.96 - 2.12	0.076
Yes	44 (39.3%)			
No	469 (31.2%)			
Abdominal/pelvic radiotherapy				0.19
categorized				
No	469 (31.2%)	Ref		
1-29 Gy	25 (37.9%)	1.35	0.81 - 2.24	
30+ Gy	19 (41.3%)	1.55	0.86 - 2.82	

Supplemental table 7. Univariable logistic regression of factors associated with outcome false-negative classification of overweight based on BMI and waist circumference, when compared to overweight based on fat% on DXA scan.

Supplemental table 7. Continued

Dependent variable	Outcome false-nega	tive classi	fication with BM	[
	Frequency of false-negative classification with BMI (n (%))		Univariable moo	lels
		OR	95% CI	P-value
Alkylating agents (CED)		1.36	1.09 - 1.70	0.006
Yes	293 (34.4%)			
No	188 (27.8%)			
Cyclophosphamide equivalent dose				0.021
categorized				
No	188 (27.8%)	Ref		
$1-4000 \text{mg/m}^2$	114 (32.7%)	1.26	0.95 - 1.67	
4000-8000mg/m ²	76 (33.0%)	1.28	0.93 - 1.77	
$8000 + mg/m^2$	103 (37.7%)	1.58	1.17 - 2.12	
Anthracyclines (DED)		1.35	1.09 - 1.68	0.007
Yes	296 (34.3%)			
No	195 (27.9%)			
Doxorubicin equivalent dose categorized				0.0022
No	195 (27.9%)	Ref		
First tertile	86 (28.7%)	1.04	0.77 - 1.40	
Second tertile	110 (35.8%)	1.45	1.09 - 1.92	
Third tertile	100 (38.9%)	1.65	1.22 - 2.23	
Corticosteroids		0.86	0.70 - 1.06	0.16
Yes	260 (30.1%)			
No	254 (33.3%)			
Corticosteroids categorized				0.17
No	254 (33.3%)	Ref		
$0-10g/m^2$	234 (29.5%)	0.84	0.67 - 1.04	
$10 + g/m^2$	26 (36.7%)	1.16	0.70 - 1.92	
Asparaginase		0.85	0.67 - 1.08	0.19
Yes	129 (29.1%)			
No	384 (32.5%)			
Platinum derivatives		1.37	1.03 - 1.82	0.029
Yes	91 (37.6%)			
No	422 (30.5%)			
Vinca alkaloids		1.02	0.79 - 1.33	0.88
Yes	410 (31.7%)			
No	103 (31.2%)			
Allogeneic SCT		2.92	1.83 - 4.66	<0.001
Yes	42 (56.0%)			
No	469 (30.4%)			
Comorbidity	· · · ·			
Growth hormone deficiency		0.94	0.59 - 1.50	0.79
Yes	27 (30.3%)			
No	487 (31.7%)			

Dependent variable	Outcome false-nega	tive classif	ication with WC	
	Frequency of false-negative classification with WC (n (%))		Univariable mo	dels
		OR	95% CI	P-value
Patient characteristics				
Age and sex				
Age at clinic visit		0.98	0.97 - 1.00	0.014
First tertile	203 (32.2%)			
Second tertile	190 (30.5%)			
Third tertile	94 (25.6%)			
Sex		1.01	0.82 - 1.25	0.92
Female	231 (30.2%)			
Male	256 (30.0%)			
Age and weight at diagnosis				
Age at diagnosis		0.97	0.95 - 1.00	0.035
First tertile	184 (31.1%)			
Second tertile	179 (31.9%)			
Third tertile	124 (26.6%)			
Overweight at cancer diagnosis		0.70	0.42 - 1.17	0.17
Yes	20 (23.8%)			
No	413 (31.0%)			
Lifestyle		0.83	0.65 - 1.06	0.13
Smoking (current or former)				
Yes	128 (26.9%)			
No	294 (30.8%)			
Low physical activity		1.10	0.85 - 1.43	0.45
Yes	117 (32.9%)			
No	330 (30.7%)			
Treatment characteristics				
Cranial radiotherapy		0.79	0.59 - 1.08	0.14
Yes	66 (26.2%)			
No	420 (30.9%)			
Cranial radiotherapy categorized				0.15
No	420 (30.9%)	Ref		
1-25 Gy	26 (22.2%)	0.64	0.41 - 1.00	
25+ Gy	40 (30.8%)	0.99	0.67 - 1.47	
Total body irradiation		3.20	1.95 - 5.25	<0.001
Yes	38 (56.7%)			
No	448 (29.1%)			
Abdominal/pelvic radiotherapy		2.16	1.46 - 3.19	<0.001
Yes	52 (46.8%)			
No	434 (29.0%)			
Abdominal/pelvic radiotherapy				<0.001
categorized				
No	434 (29.0%)	Ref		
1-29 Gy	28 (43.1%)	1.86	1.12 - 3.07	
30+ Gy	24 (52.2%)	2.67	1.48 - 4.82	
Alkylating agents (CED)		1.47	1.18 - 1.85	0.001
Yes	285 (33.5%)			

Dependent variable	Outcome false-negat	tive classif	ication with WC	
	Frequency of false-negative classification with WC (n (%))		Univariable mo	dels
		OR	95% CI	P-value
No	171 (25.5%)			
Cyclophosphamide equivalent dose				0.0027
categorized				
No	171 (25.5%)	Ref		
$1-4000 \text{mg/m}^2$	106 (30.6%)	1.29	0.97 - 1.72	
4000-8000mg/m ²	85 (36.8%)	1.70	1.24 - 2.34	
$8000 + mg/m^2$	94 (34.4%)	1.54	1.13 - 2.08	
Anthracyclines (DED)		1.49	1.19 - 1.86	<0.001
Yes	290 (33.7%)			
No	177 (25.5%)			
Doxorubicin equivalent dose categorized				<0.001
No	177 (25.5%)	Ref		
First tertile	89 (29.7%)	1.23	0.91 - 1.67	
Second tertile	102 (33.7%)	1.49	1.11 - 1.99	
Third tertile	99 (38.5%)	1.83	1.35 - 2.48	
Corticosteroids		0.89	0.72 - 1.10	0.29
Yes	248 (28.9%)			
No	239 (31.4%)			
Corticosteroids categorized				0.064
No	239 (31.4%)	Ref		
0-10g/m ²	220 (28.0%)	0.85	0.68 - 1.06	0.14
$10 + g/m^2$	28 (40.0%)	1.46	0.88 - 2.41	0.14
Asparaginase		0.89	0.70 - 1.13	0.33
Yes	124 (28.3%)			
No	363 (30.8%)			
Platinum derivatives		1.63	1.23 - 2.17	0.001
Yes	95 (39.4%)			
No	392 (28.5%)			
Vinca alkaloids		1.06	0.82 - 1.39	0.65
Yes	391 (30.4%)			
No	96 (29.1%)			
Allogeneic SCT		2.37	1.49 - 3.78	<0.001
Yes	37 (49.3%)			
No	447 (29.1%)			
Comorbidity	. ,			
Growth hormone deficiency		1.07	0.68 - 1.70	0.77
Yes	28 (31.5%)			
No	459 (30.0%)			

Supplemental table 7. Continued

For variables with more than two categories the overall p-value was calculated with the Wald test. Significant variables in univariable analysis were included in the multivariable model. If for a treatment factor both the dichotomous and categorized variable were significant, only the latter was included.

Abbreviations: CI = confidence interval; OR = odds ratio; CI = confidence interval; Ref = reference

Supplemental table 8. Multivariable logistic regression of factors associated with outcome false-negative classification of overweight based on BMI and waist circumference, when compared to overweight based on fat% on DXA scan.

Dependent variable	Outcome false-negation	ive classif	ication with BMI	
	Frequency of false-negative classification with BMI (n (%))		Multivariable mo	odel
		OR	95% CI	P-value
Patient characteristics Age and sex				
Age at clinic visit		1	0.98 - 1.03	0.74
First tertile	203 (32.2%)			
Second tertile	199 (31.9%)			
Third tertile	111 (29.9%)			
Sex		3.02	2.27 - 4.03	<0.001
Female	327 (42.4%)			
Male	187 (21.9%)			
Weight at diagnosis		0.57	0.20 1.11	0.008
Ves	19 (21 (0/)	0.37	0.29 - 1.11	0.098
No	18 (21.4%)			
Lifestyle	457 (52.6%)			
Smoking (current or former)		0.95	0.70 - 1.29	0.76
Yes	128 (26.8%)			
No	321 (33.4%)			
Low physical activity		1.36	1.00 - 1.85	0.052
Yes	143 (39.7%)			
No	323 (30.0%)			
Treatment characteristics				
Cranial radiotherapy				
Yes	68 (26.8%)			
No	445 (32.6%)			
Cranial radiotherapy categorized				0.036
No	445 (32.6%)	Ref		
1-25 Gy	28 (23.5%)	0.47	0.26 - 0.83	
25+ Gy	39 (30.0%)	0.89	0.45 - 1.76	
Total body irradiation		9.06	2.41 - 34.04	0.001
Yes	42 (62.7%)			
No	471 (30.4%)			
Cyclophosphamide equivalent dose categorized				0.55
No	188 (27.8%)	Ref		
1-4000mg/m ²	114 (32.7%)	1.13	0.72 - 1.77	
4000-8000mg/m ²	76 (33.0%)	1.15	0.72 - 1.84	

Supplemental table 8. Continued

Dependent variable	Outcome false-negat	ive classifi	cation with BMI	
	Frequency of false-negative classification with BMI (n (%))		Multivariable mo	odel
		OR	95% CI	P-value
$8000 + mg/m^2$	103 (37.7%)	1.36	0.90 - 2.05	
Doxorubicin equivalent dose categorized				0.13
No	195 (27.9%)	Ref		
First tertile	86 (28.7%)	1.29	0.80 - 2.09	
Second tertile	110 (35.8%)	1.57	1.02 - 2.42	
Third tertile	100 (38.9%)	1.48	0.99 - 2.23	
Platinum derivatives		1.35	0.89 - 2.04	0.16
Yes	91 (37.6%)			
No	422 (30.5%)			
Allogeneic SCT		0.63	0.16 - 2.44	0.5
Yes	42 (56.0%)			
No	469 (30.4%)			
Dependent variable	Outcome false-negat	ive classif	ication with WC	

1	Frequency of false-negative			
	classification with WC (n (%))		Multivariable mo	odel
		OR	95% CI	P-value
Patient characteristics				
Age at clinic visit		0.99	0.97 - 1.01	0.28
First tertile	203 (32.2%)			
Second tertile	190 (30.5%)			
Third tertile	94 (25.6%)			
Age at diagnosis		0.97	0.94 - 1.00	0.023
First tertile	184 (31.1%)			
Second tertile	179 (31.9%)			
Third tertile	124 (26.6%)			
Treatment characteristics				
Total body irradiation		4.35	1.78 - 10.67	<0.001
Yes	38 (56.7%)			
No	448 (29.1%)			
Abdominal/pelvic radiotherapy categorized				<0.001
No	434 (29.0%)	Ref		
1-29 Gy	28 (43.1%)	2.05	1.17 - 3.56	
30+ Gy	24 (52.2%)	3.06	1.64 - 5.72	
Cyclophosphamide equivalent dose categorized				0.71

Dependent variable	Outcome false-negat	tive classif	ication with WC	
	Frequency of false-negative classification with WC (n (%))		Multivariable m	odel
		OR	95% CI	P-value
No	171 (25.5%)	Ref		
$1-4000 \text{mg/m}^2$	106 (30.6%)	1.07	0.74 - 1.56	
4000-8000mg/m ²	85 (36.8%)	1.25	0.85 - 1.83	
8000+ mg/m ²	94 (34.4%)	1.11	0.79 - 1.57	
Doxorubicin equivalent dose categorized				0.049
No	177 (25.5%)	Ref		
First tertile	89 (29.7%)	1.33	0.90 - 1.97	
Second tertile	102 (33.7%)	1.43	0.99 - 2.05	
Third tertile	99 (38.5%)	1.58	1.12 - 2.23	
Platinum derivatives		1.66	1.21 - 2.29	0.002
Yes	95 (39.4%)			
No	392 (28.5%)			
Allogeneic SCT		1.08	0.46 - 2.51	0.86
Yes	37 (49.3%)			
No	447 (29.1%)			

Supplemental table 8. Continued

For variables with more than two categories the overall p-value was calculated with the Wald test. Significant variables in univariable analysis were included in the multivariable model. If for a treatment factor both the dichotomous and categorized variable were significant, only the latter was included.

Abbreviations: CI = confidence interval; OR = odds ratio; CI = confidence interval; Ref = reference



Supplemental figure 1. Flow chart of study participants

The current study used the cohort of the Metabolic Syndrome sub-study of the DCCSS-LATER study. This flow chart shows the number of participants in the current study and how they are embedded in the overall cohort. Percentages indicate the proportion of the overall cohort (N=6,165). In the lower two blocks, there are two percentages: the first indicates the proportion of the overall cohort, the second is the proportion of the invited survivors, indicating the (non-) participation rate.

At formation in 2008, the entire survivors cohort contained 6,165 eligible survivors. By mail, survivors were provided the option to opt-out of future study participation. At the start of the DCCSS-LATER study, the cleaned cohort was frozen in 2016, leaving 5,160 subjects eligible. For the Metabolic Syndrome sub-study, only adults (n=4,671) were invited.



Supplemental figure 2. Overweight, obesity and morbid obesity prevalence per childhood cancer diagnosis group ICCC-3 group "Other and unspecified" is not included. Dashed lines indicate the overall mean for each outcome. Abbreviations: ALL = acute lymphoid leukemia; CNS = central nervous system



Supplemental figure 3. Overweight outcomes per childhood cancer diagnosis group

Only survivors with all three measurements available are shown in this figure. ICCC-3 group "Other and unspecified" is not included. Dashed lines indicate the overall mean for each outcome.

Abbreviations: ALL = acute lymphoid leukemia; CNS = central nervous system

REFERENCES

- Bhakta N, Liu Q, Ness KK, Baassiri M, Eissa H, Yeo F, et al. The cumulative burden of surviving childhood cancer: an initial report from the St Jude Lifetime Cohort Study (SJLIFE). Lancet. 2017;390(10112):2569-82.
- 2. van Waas M, Neggers SJ, Pieters R, van den Heuvel-Eibrink MM. Components of the metabolic syndrome in 500 adult long-term survivors of childhood cancer. Ann Oncol. 2010;21(5):1121-6.
- Mostoufi-Moab S, Seidel K, Leisenring WM, Armstrong GT, Oeffinger KC, Stovall M, et al. Endocrine Abnormalities in Aging Survivors of Childhood Cancer: A Report From the Childhood Cancer Survivor Study. J Clin Oncol. 2016;34(27):3240-7.
- Belle FN, Weiss A, Schindler M, Goutaki M, Bochud M, Zimmermann K, et al. Overweight in childhood cancer survivors: the Swiss Childhood Cancer Survivor Study. Am J Clin Nutr. 2018;107:3-11.
- Wilson CL, Liu W, Yang JJ, Kang G, Ojha RP, Neale GA, et al. Genetic and clinical factors associated with obesity among adult survivors of childhood cancer: A report from the St. Jude Lifetime Cohort. Cancer. 2015;121(13):2262-70.
- van Santen HM, Geskus RB, Raemaekers S, van Trotsenburg AS, Vulsma T, van der Pal HJ, et al. Changes in body mass index in long-term childhood cancer survivors. Cancer. 2015;121(23):4197-204.
- 7. World Health Organization. Obesity and Overweight fact sheet 2021 [updated 2021-06-09. Available from: https://www.who.int/news-room/fact-sheets/detail/obesity-and-overweight.
- 8. Locke AE, Kahali B, Berndt SI, Justice AE, Pers TH, Day FR, et al. Genetic studies of body mass index yield new insights for obesity biology. Nature. 2015;518(7538):197-206.
- 9. van Waas M, Neggers SJ, Raat H, van Rij CM, Pieters R, van den Heuvel-Eibrink MM. Abdominal radiotherapy: a major determinant of metabolic syndrome in nephroblastoma and neuroblastoma survivors. PLoS One. 2012;7(12):e52237.
- 10. Inaba H, Yang J, Kaste SC, Hartford CM, Motosue MS, Chemaitilly W, et al. Longitudinal changes in body mass and composition in survivors of childhood hematologic malignancies after allogeneic hematopoietic stem-cell transplantation. J Clin Oncol. 2012;30(32):3991-7.
- 11. Alberti KG, Eckel RH, Grundy SM, Zimmet PZ, Cleeman JI, Donato KA, et al. Harmonizing the metabolic syndrome: a joint interim statement of the International Diabetes Federation Task Force on Epidemiology and Prevention; National Heart, Lung, and Blood Institute; American Heart Association; World Heart Federation; International Atherosclerosis Society; and International Association for the Study of Obesity. Circulation. 2009;120(16):1640-5.
- 12. National Cholesterol Education Program Expert Panel on Detection E, and Treatment of High Blood Cholesterol in Adults (Adult Treatment Panel III). Third Report of the National Cholesterol Education Program (NCEP) Expert Panel on Detection, Evaluation, and Treatment of High Blood Cholesterol in Adults (Adult Treatment Panel III). 2002.
- 13. World Health Organization. Waist circumference and waist-hip ratio: report of a WHO expert consultation, Geneva, 8-11 December 2008. 2011.
- Ashwell M, Gibson S. Waist to height ratio is a simple and effective obesity screening tool for cardiovascular risk factors: Analysis of data from the British National Diet And Nutrition Survey of adults aged 19-64 years. Obes Facts. 2009;2(2):97-103.
- 15. Blijdorp K, van den Heuvel-Eibrink MM, Pieters R, Boot AM, Delhanty PJ, van der Lely AJ, et al. Obesity is underestimated using body mass index and waist-hip ratio in long-term adult survivors of childhood cancer. PLoS One. 2012;7(8):e43269.

- Karlage RE, Wilson CL, Zhang N, Kaste S, Green DM, Armstrong GT, et al. Validity of anthropometric measurements for characterizing obesity among adult survivors of childhood cancer: A report from the St. Jude Lifetime Cohort Study. Cancer. 2015;121(12):2036-43.
- 17. LaForgia J, Dollman J, Dale MJ, Withers RT, Hill AM. Validation of DXA body composition estimates in obese men and women. Obesity (Silver Spring). 2009;17(4):821-6.
- 18. Frankenberg ADV, Reis AF, Gerchman F. Relationships between adiponectin levels, the metabolic syndrome, and type 2 diabetes: a literature review. Arch Endocrinol Metab. 2017;61(6):614-22.
- Pluimakers VG, van Santen SS, Fiocco M, Bakker ME, van der Lelij AJ, van den Heuvel-Eibrink MM, et al. Can biomarkers be used to improve diagnosis and prediction of metabolic syndrome in childhood cancer survivors? A systematic review. Obes Rev. 2021;22(11):e13312.
- 20. Pluimakers V, Fiocco M, van Atteveld J, Hobbelink M, Bresters D, Van Dulmen-den Broeder E, et al. Metabolic Syndrome Parameters, Determinants, and Biomarkers in Adult Survivors of Childhood Cancer: Protocol for the Dutch Childhood Cancer Survivor Study on Metabolic Syndrome (Dutch LATER METS). JMIR Res Protoc. 2021;10(1):e21256.
- 21. Scholtens S, Smidt N, Swertz MA, Bakker SJ, Dotinga A, Vonk JM, et al. Cohort Profile: LifeLines, a three-generation cohort study and biobank. Int J Epidemiol. 2015;44(4):1172-80.
- 22. Sommer I, Teufer B, Szelag M, Nussbaumer-Streit B, Titscher V, Klerings I, et al. The performance of anthropometric tools to determine obesity: a systematic review and meta-analysis. Sci Rep. 2020;10(1):12699.
- 23. R Core Team (2020). R: A language and environment for statistical computing. R Foundation for Statistical Computing, Vienna, Austria. URL https://www.R-project.org/.
- Henson KE, Reulen RC, Winter DL, Bright CJ, Fidler MM, Frobisher C, et al. Cardiac Mortality Among 200 000 Five-Year Survivors of Cancer Diagnosed at 15 to 39 Years of Age: The Teenage and Young Adult Cancer Survivor Study. Circulation. 2016;134(20):1519-31.
- 25. Prasad M, Arora B, Chinnaswamy G, Vora T, Narula G, Banavali S, et al. Nutritional status in survivors of childhood cancer: Experience from Tata Memorial Hospital, Mumbai. Indian J Cancer. 2015;52(2):219-23.
- 26. van Schaik J, van Roessel I, Schouten-van Meeteren N, van Iersel L, Clement SC, Boot AM, et al. High Prevalence of Weight Gain in Childhood Brain Tumor Survivors and Its Association With Hypothalamic-Pituitary Dysfunction. J Clin Oncol. 2021;39(11):1264-73.
- Merchant TE, Rose SR, Bosley C, Wu S, Xiong X, Lustig RH. Growth hormone secretion after conformal radiation therapy in pediatric patients with localized brain tumors. J Clin Oncol. 2011;29(36):4776-80.
- 28. Chemaitilly W, Mertens AC, Mitby P, Whitton J, Stovall M, Yasui Y, et al. Acute ovarian failure in the childhood cancer survivor study. J Clin Endocrinol Metab. 2006;91(5):1723-8.
- van Dongen-Melman JEWM, Hokken-Koelega ACS, Hählen K, de Groot A, Tromp CG, Egeler RM. Obesity after Successful Treatment of Acute Lymphoblastic Leukemia in Childhood. Pediatr Res. 1995;38(1):86-90.
- 30. van Beek RD, de Muinck Keizer-Schrama SM, Hakvoort-Cammel FG, van der Sluis IM, Krenning EP, Pieters R, et al. No difference between prednisolone and dexamethasone treatment in bone mineral density and growth in long term survivors of childhood acute lymphoblastic leukemia. Pediatr Blood Cancer. 2006;46(1):88-93.
- 31. Shah NR, Braverman ER. Measuring adiposity in patients: the utility of body mass index (BMI), percent body fat, and leptin. PLoS One. 2012;7(4):e33308.

- 32. Brouwer CA, Gietema JA, Vonk JM, Tissing WJ, Boezen HM, Zwart N, et al. Body mass index and annual increase of body mass index in long-term childhood cancer survivors; relationship to treatment. Support Care Cancer. 2012;20(2):311-8.
- 33. Kok JL, Teepen JC, van Leeuwen FE, Tissing WJE, Neggers S, van der Pal HJ, et al. Risk of benign meningioma after childhood cancer in the DCOG-LATER cohort: contributions of radiation dose, exposed cranial volume, and age. Neuro Oncol. 2019;21(3):392-403.
- Ho-Pham LT, Campbell LV, Nguyen TV. More on body fat cutoff points. Mayo Clin Proc. 2011;86(6):584; author reply -5.
- Gallagher D, Heymsfield SB, Heo M, Jebb SA, Murgatroyd PR, Sakamoto Y. Healthy percentage body fat ranges: an approach for developing guidelines based on body mass index. Am J Clin Nutr. 2000;72(3):694-701.
- 36. Heo M, Faith MS, Pietrobelli A, Heymsfield SB. Percentage of body fat cutoffs by sex, age, and race-ethnicity in the US adult population from NHANES 1999-2004. Am J Clin Nutr. 2012;95(3):594-602.
- 37. Funcke JB, Scherer PE. Beyond adiponectin and leptin: adipose tissue-derived mediators of inter-organ communication. J Lipid Res. 2019;60(10):1648-84.

Chapter 6

Genetic susceptibility to treatment related dyslipidemia in adult long-term survivors of childhood cancer: a report from the Childhood Cancer Survivor Study and the St Jude Lifetime cohorts

V.G. Pluimakers, L. Broer*, S.J.C.M.M. Neggers*, D.T.C. de Winter, J.L. Baedke, A.G. Uitterlinden, K.C. Oeffinger, L.L. Robison, R.M. Howell, E. Chow, J.N. Sampson, W.M. Leisenring, M.M. Hudson, CCSS Chronic Disease Working Group, G.T. Armstrong**, S. Bhatia**, Y. Yasui**, M.M. van den Heuvel-Eibrink**

In preparation

ABSTRACT

Background

Dyslipidemia can occur as a long-term side effect of childhood cancer treatment. Difference in occurrence in children receiving comparable treatment may suggest a role for genetic variation. We performed the first genome-wide association study on dyslipidemia in three large childhood cancer survivor cohorts.

Methods

Discovery analysis was performed in the Original (diagnosis 1970-1986) Cohort of the Childhood Cancer Survivor Study (CCSS, N=4,332). Replication analyses were performed in the St Jude Lifetime Cohort (SJLIFE, N=2,274) and the CCSS Expansion (diagnosis 1986-1999) Cohort (N=2,212). In the CCSS cohorts, dyslipidemia was defined as CTCAE grade 2 self-reported high cholesterol or high triglycerides, in the SJLIFE this was based on serum lipid assessment. Association was studied in the entire cohort as well as by treatment stratification. Additionally, we performed a candidate SNP approach of three variants identified in the only available genetic study on dyslipidemia in acute lymphoblastic leukemia (ALL) survivors.

Results

The initial discovery analysis yielded one significant ($p<5x10^{-8}$) and 16 suggestive ($p<5x10^{-6}$) loci. Nine with biological plausibility were included for replication analysis but none replicated. Using a treatment-stratified analysis of nine additional significant loci, variant rs114017774 was statistically significant among cranial radiotherapy (CRT) exposed survivors when pooling CCSS and SJLIFE (OR=11.30, 95%CI=5.03-25.40, p=4.5x10⁻⁹), but did not individually replicate in the CCSS Expansion or SJLIFE cohorts. This variant had an interaction with CRT (genotypexCRT OR=14.61, 95%CI=4.59-56.55) independent of BMI. This variant could potentially alter *LRRTM4* or *CTNNA2* gene expression or function, making survivors more susceptible to hypothalamic damage caused by CRT. From the candidate SNP study we replicated rs676210 in the *APOB* gene in the CCSS cohort (p=0.0061 in ALL survivors, p=0.0049 in the total cohort).

Discussion

We identified rs114017774 on chromosome 2 as a potential genetic variant for dyslipidemia, specifically in cranially irradiated survivors of childhood cancer. We replicated for the first time variant rs676210 in a heterogeneous survivor cohort.

INTRODUCTION

The survival rates of childhood cancer have increased tremendously over the past decades. Currently, more than 80% of children with cancer will attain 5-year survival⁽¹⁾. This number, however, fails to reflect cancer treatment related morbidity and consequent mortality occur⁽²⁾. Dyslipidemia, an established risk factor for atherosclerotic disease and mortality, is among the chronic health conditions experienced by long-term childhood cancer survivors ⁽³⁾. Dyslipidemia is also a component of the metabolic syndrome, together with adiposity, insulin resistance and hypertension, a synergistic cluster strongly associated with subsequent diabetes mellitus and cardiovascular disease⁽⁴⁾.

Previous studies have revealed that childhood cancer survivors have an early increased risk for developing dyslipidemia and metabolic syndrome⁽⁵⁻¹¹⁾. Several factors have been associated with dyslipidemia, either directly or consequent to obesity. Endocrinopathies such as growth hormone deficiency (GHD), hypothyroidism and hypogonadism can lead to dyslipidemia and obesity. Disease and treatment related risk factors for dyslipidemia include diagnosis of brain tumor, brain surgery, cranial radiotherapy (CRT), and radiation to other relevant organs involved in metabolism, e.g., pancreas, gonads, thyroid and abdominal fat tissue^(6, 11-16).

Variation in the occurrence and severity of dyslipidemia that is observed in comparably treated survivors suggests a potential role for genetic susceptibility^(17, 18). This genetic susceptibility could be explained by carrier status of genetic variants known to be associated with dyslipidemia in the general population, unique variants interacting with cancer treatment (e.g., drug metabolism), or a combination of the two. Although genome-wide associated with different lipid traits⁽¹⁹⁻²³⁾, evidence for relevance in survivors and for unique variants in survivors is scarce. So far, only one candidate SNP study in a cohort of 209 acute lymphoblastic leukemia (ALL) survivors has been performed, identifying dyslipidemia associated single nucleotide polymorphisms (SNP) in three genes⁽²⁴⁾. These findings have, however, not been replicated in other survivors.

In order to identify unique genetic variants, we performed the first GWAS on dyslipidemia in childhood cancer survivors, aiming to identify SNPs uniquely associated with dyslipidemia in this group, with discovery and replication analyses in three large survivor cohorts: the Childhood Cancer Survivor Study (CCSS) Original and Expansion cohort and the St. Jude Lifetime Cohort (SJLIFE). We also performed a candidate SNP analysis of the three variants found in ALL survivors.

METHODS

Discovery cohort

The discovery analysis was performed on the original cohort of the CCSS. The CCSS original cohort consists of 14,361 5-year survivors, diagnosed between 1970 and 1986⁽²⁵⁾. Participants in the study completed a baseline and follow-up questionnaire to capture self-reported outcomes. Of these, 5,739 survivors were genotyped (Supplemental Figure 1). This was done with DNA from blood, saliva or buccal cells, using the HumanOmni5Exome array (Illumina, San Diego, CA, USA) and imputed with the 1000Genomes phase 3 version 5 reference panel⁽²⁶⁾. We restricted our analysis to participants with European genetic ancestry. Survivors participating in SJLIFE were excluded, to avoid double inclusion. Survivors who received stem cell transplantation were also excluded. Hence, the cohort for the discovery analysis consisted of 4,332 survivors, of whom 759 (17.5%) had developed dyslipidemia.

Replication cohorts

The SJLIFE and CCSS Expansion Cohorts both served as replication cohorts. SJLIFE is a cohort of 4,713 survivors, diagnosed between 1962 and 2012, who completed health questionnaires and underwent a comprehensive clinical assessment at a late effects clinic⁽²⁷⁾. The CCSS Expansion Cohort is questionnaire-based and consists of 11,304 5-year survivors diagnosed between 1987 and 1999. For both replication cohorts genotyping was performed with whole genome sequencing using HiSeq X10 and NovaSeq sequencers (Illumina, San Diego, USA). Replication analyses were also restricted to European genetic ancestry, and survivors who received stem cell transplantation were excluded. Participants in both SJLIFE and CCSS Expansion Cohort were excluded from the latter. This resulted in 2,274 SJLIFE survivors and 2,212 CCSS Expansion Cohort survivors included as our replication cohorts (Supplemental Figure 1). Participants in all study cohorts provided informed consent for the use of genetic data for future analyses.

Phenotype definition

In all three cohorts, dyslipidemia was defined as high total cholesterol or hypertriglyceridemia grade 2 (=moderate) or higher, according to the modified CTCAE criteria⁽²⁸⁾ (Supplemental Table 1). In the CCSS and CCSS Expansion Cohorts, this is self-reported dyslipidemia, requiring medication, based on health questionnaires⁽²⁹⁾. In the SJLIFE Cohort, grade 2 or higher dyslipidemia was defined as total cholesterol >300 mg/dL (7.8 mmol/L) or triglycerides >300 mg/dL (3.4 mmol/L) or treatment with one or more lipid lowering agents.

GWAS analysis

Discovery and replication analysis

A genetic power calculator was used to estimate the minimally detectable odds ratio (OR) that could be achieved in the discovery analysis for variants to reach the genomewide significance threshold of $p<5x10^{-8}$ for several minor allele frequencies (MAF) ⁽³⁰⁾. For a survivor population dyslipidemia prevalence of $15\%^{(16)}$, under an allelic model with additive risk, with the required power of 80%, the minimal detectable ORs per high risk allele with MAF = 0.01, 0.05, 0.1 and 0.25 were 2.7, 1.8, 1.5 and 1.2, respectively. In addition to the genome-wide significance threshold, we set a suggestive threshold of $p<5x10^{-6}$. To reach the suggestive threshold, the minimum ORs per high risk allele with the four MAFs were 2.4, 1.6, 1.4 and 1.1, respectively.

GWAS was performed in *Rvtests*⁽³¹⁾, using the score test option. This first analysis used a logistic regression model adjusting for age at follow-up, sex and the first four genetic principal components. Additionally, treatment-stratified GWAS were performed, in order to identify SNP effects in survivors with a specific cancer treatment, as well as to find SNPs whose effects are otherwise outweighed/masked by cancer treatment and relevant only in non-exposed survivors^(13, 32-36). To identify treatment factors to stratify by, hypothesized relevant factors (i.e., CRT, abdominal radiotherapy, asparaginase, prednisone and dexamethasone)^(6, 11-15) were examined by a logistic regression model and those with a p-value less than 0.10 were used as stratification variables.

Quality control of the GWAS analysis was performed in R (R Core Foundation, Vienna, Austria)⁽³⁷⁾, with the package *EasyQC*⁽³⁸⁾, using an imputation quality filter of 0.4 and otherwise default settings. Genetic associations were visualized in a Manhattan plot using the R package *EasySTRATA*⁽³⁹⁾.

For replication analysis we included initial discovery analysis SNPs that either reached the genome-wide significance threshold, or that reached the suggestive threshold and had biological plausibility to be associated with dyslipidemia (see section Functional evidence below). For treatment-stratified models we were more stringent and only included genome-wide significant SNPs for replication analysis. For the replication analysis the same logistic regression model was used. For a significant replication, the SNP effect on the outcome needed to be in the same direction, with meeting the statistical significance threshold at p-value of 0.05 divided by the number of SNPs tested (Bonferroni correction).

The results of the discovery and replication analyses were pooled using the R package $rmeta^{(40)}$, with a fixed effects model or, in case of significant heterogeneity (p<0.05), a

random effects model. Due to its relatively young age, lower dyslipidemia prevalence and differences in treatment characteristics reflecting newer protocols, the CCSS Expansion cohort was judged differently from the CCSS and SJLIFE cohorts. It was therefore decided to first pool the CCSS original and SJLIFE cohorts in meta-analysis, and as the next step add the CCSS Expansion results. The results of the meta-analysis were regarded as significant if the p-value became lower.

Functional evidence

We used the UCSC Genome Browser⁽⁴¹⁾, Locus Zoom⁽⁴²⁾, Functional Mapping and Annotation of GWAS (FUMA-GWAS)⁽⁴³⁾, ROADMAP⁽⁴⁴⁾, GWAS catalog⁽⁴⁵⁾, GTEx⁽⁴⁶⁾, Mouse Genome Informatics⁽⁴⁷⁾, International Mouse Phenotyping Consortium⁽⁴⁸⁾ and published literature to explore supportive information that links the discovered loci to potential genes, and these genes to dyslipidemia. This was performed for the variant as well as variants in high LD (R^2 >0.80).

Further analyses of replicated SNP

For the SNP replicated in treatment-stratified analysis, additional analyses were performed by adding terms to the regression model, to further explore a potential gene-treatment exposure interaction. First, the interaction term (i.e., SNPxTherapy) was included. Furthermore, a dose-effect relation was evaluated by adding the dose variable to the model instead of the dichotomous variable. For cranial radiotherapy, an additional variable was available specifying the dose received by several brain regions (Supplemental figure 2). This variable was added to the model to find out whether the gene-radiotherapy interaction was associated with radiation to the hypothalamic-pituitary region specifically^(49, 50). In order to evaluate whether the effect was restricted to this treatment exposure only, the association for this locus in the other treatment-stratified analyses and the full cohort analysis was looked up in the GWAS results. Finally, we evaluated whether the effect of the SNP on dyslipidemia was a direct effect or mediated by overweight, by adding body mass index (BMI) to the model, as well as by building a model with BMI, overweight or obesity as outcomes. The additional models were built in R and tested with the Wald test.

Comparison with general population

To verify that the genetic signal is specific to survivors, we performed a look-up of the identified variant and variants in high linkage disequilibrium with it (LD, according to FUMA-GWAS results and LD $\text{proxy}^{(51)}$) in two of the largest GWASs in the general population published to date, that had the complete summary data publicly available^(22, 23). The genome-wide significance threshold (p<5x10⁻⁸) was set for association in the general population.

Candidate SNP analysis

We looked up associations for the three variants (rs676210, rs2286615, rs62079523) previously identified in the candidate SNP study by England et al. among ALL survivors⁽²⁴⁾ in the largest of our three cohorts, i.e., the CCSS Original cohort (overall cohort and ALL subgroup). The significance threshold after Bonferroni correction of 0.05/3=0.017 was used for this analysis.

RESULTS

Cohort description

In the CCSS discovery cohort (mean age 43.3 years old, mean follow-up time 35.3 years), the prevalence of dyslipidemia was 17.5% (Table 1). The most prevalent childhood cancer diagnoses included ALL (26.9%), Hodgkin lymphoma (13.3%), and kidney tumors (9.8%). Survivors with dyslipidemia were on average 6.5 years older and were more often overweight or obese. Growth hormone deficiency, hypogonadism, diabetes and hypothyroidism were more prevalent among dyslipidemia cases. Administration of abdominal radiotherapy and alkylating agents was more frequent among cases, whereas controls had more often received asparaginase.

In SJLIFE (mean age 36.2 years old, mean follow-up 28.1 years) the dyslipidemia prevalence was 19.7% and in CCSS Expansion (mean age 30.0 years old, mean follow-up 21.3 years) this was 5.4%. Participants in CCSS Expansion had significantly less often received cranial and abdominal radiotherapy.

GWAS analysis

Discovery and replication analysis

In the discovery analysis of the full CCSS Original cohort, one locus with a genomewide significant association and 16 suggestive loci were identified (Table 2, Figure 1, Supplemental Figure 3). For nine of these 17 loci, we were able to find supportive biological information for an association with dyslipidemia. However, in both replication cohorts none of these variants replicated with statistical significance. In the meta-analysis with CCSS Original and SJLIFE, and with all three cohorts pooled, none of the variants were significant either.

Next, in the model without genetic factors (Supplemental Table 2), in order to determine the stratification variables, an independent positive association with dyslipidemia was found for administration of CRT (OR=1.57, 95%CI=1.26-1.94) and

abdominal radiotherapy (1.86, 1.53-2.25) and a negative association for asparaginase (0.68, 0.50-0.92). Prednisone (1.23, 0.98-1.55) was also included in the treatmentstratified analyses. By this treatment-stratified approach, nine additional genomewide significant loci were identified (Table 3). In the abdominal radiotherapy exposed group, one locus on chromosome 6 showed a statistically significant association in SJLIFE but the effect was in the opposite direction so the meta-analysis did not reveal a significant association. In the CRT exposed group (Table 3, Figure 2, Supplemental table 3, Supplemental figure 3 and 4), one locus on chromosome 2, rs114017774, revealed a trend for a positive association with dyslipidemia in the SJLIFE cohort (OR=4.27, 95%CI=0.97-22.11, p=0.055) and was significant in the meta-analysis of the two cohorts (OR=11.30, 95%CI=5.03-25.40, p=4.46x10⁻⁹). This locus was, however, not associated in the CCSS Expansion cohort and not significant in the meta-analysis of all three cohorts. The other seven loci were not replicated in either cohort and not significant in the meta-analysis.

Functional evidence

Additional analyses were performed for our top locus, rs114017774 on chromosome 2. We could not identify any expression quantitative trait locus between our top locus and genes in the region. Chromatin interactions have been observed with the genes Leucine Rich Repeat Transmembrane Neuronal 4 (*LRRTM4*), Catenin Alpha 2 (*CTNNA2*) and Regenerating Islet-derived 1 Alpha, 3 Alpha, 1 Beta and 3 Gamma (*REG1A, 3A, 1B* and *3G*) (Supplemental table 4). Based on protein expression in the brain, protein function including nervous system development and functioning, and associated phenotypes for genetic variants, including high-density-lipoprotein (HDL) and very-low-density-lipoprotein (VLDL) cholesterol levels, *LRRTM4* and *CTNNA2* are the most promising to be the causal genes. However, genetic variants studied were not in LD with our identified variant, and knockout models for these genes leading to a dyslipidemia phenotype are not available yet.

Further analyses of replicated SNP

When the interaction term of the top locus genotypexCRT was added to the model, it was significantly associated with dyslipidemia (OR=14.61, 95%CI=4.59-46.55, $p=5.75 \times 10^{-6}$, Table 4). When CRT dose was added to the model, a dose-response effect was observed: as continuous variable the OR for the interaction term, per Gy increase, was 1.06 (95%CI=1.03-1.09, $p=1.88 \times 10^{-5}$, Supplemental table 5). When categorized, only the interaction term with high dose radiotherapy was significant (OR=12.51, 95%CI=2.24-69.86, p=0.004). In dosimetry analysis of different brain regions, only radiation dose to the frontal region was continuously associated with dyslipidemia (OR per Gy 1.01, 95%CI=1.00-1.03, p=0.043). High dose to the hypothalamic/ pituitary region was associated with dyslipidemia (OR=1.50, 95%CI=1.10-2.04,

p=0.0096), as was high dose to the other three regions. The association was identified in the CRT exposed group only (Supplemental table 6). Overweight and obesity, when added as covariates, were significantly associated with dyslipidemia, but the genotype-dyslipidemia association remained (Supplemental table 5). Also, when the model was built with overweight or obesity as outcomes, the genetic variant was not associated with these.

Comparison with general population

rs114017774 on chromosome 2 has not been reported as a susceptibility variant for dyslipidemia in prior general population GWASs. We were also unable to find this locus in the entire GWAS datasets that were made available by Teslovich et al.⁽²²⁾ and Liu et al.⁽²³⁾.

Candidate SNP analysis

The protective SNP rs676210 in the *APOB* gene identified by England et al.⁽²⁴⁾ was associated with a lower prevalence of dyslipidemia in our study as well, in both the overall cohort (p=0.0049) and the ALL subgroup (p=0.0061) (Table 5). The exact effect sizes are not comparable due to differences in included covariates that we did not have available including additional treatment and lifestyle variables. The two other variants did not replicate in the overall cohort nor the ALL subgroup.

DISCUSSION

We performed the first GWAS on dyslipidemia in three large survivor cohorts: the CCSS Original, SJLIFE and CCSS Expansion cohorts. In the analysis of the entire cohort, none of nine SNPs replicated. Next, in survivors exposed to cranial radiotherapy, we found one locus, rs114017774 on chromosome 2, which was genome-wide significant after pooling results of the CCSS and SJLIFE cohorts. Our analyses show that survivors carrying the genetic polymorphism and treated with CRT have a significantly higher risk of developing dyslipidemia, with a pooled OR of 11.30 (95%CI=5.03-25.40, p=4.46x10⁻⁹). Sequential inclusion of the CCSS Expansion Cohort in meta-analysis was not significant, which may be due to younger age, lower dyslipidemia prevalence and cancer treatment differences. This genetic variant was not reported in large general population GWASs on dyslipidemia, potentially suggesting a survivor-specific effect. Survivors carrying this variant may be more susceptible to damage caused by CRT and therefore have an increased dyslipidemia risk.

Potential candidate genes for rs114017774 to cause dyslipidemia are *LRRTM4* and *CTNNA2*. *LRRTM4* is expressed in the hypothalamus and other brain parts

and involved in development and maintenance of the nervous system and synaptic functioning⁽⁵²⁾. In the Framingham Heart Study, a variant in this gene was associated with high VLDL cholesterol levels, but this was not genome-wide significant and this variant is not in LD with our variant⁽⁵³⁾. Knock-out mice were found to have decreased system synapse formation and abnormal excitatory postsynaptic potential, but development of dyslipidemia was not studied⁽⁵⁴⁾. CTNNA2 is also expressed in the hypothalamus and other brain parts, and involved in neuronal growth and migration⁽⁵⁵⁾. A non-genome-wide significant association was found between a variant in CTNNA2 and HDL cholesterol in HIV patients⁽⁵⁶⁾ and genome-wide significant associations were observed for coronary artery disease⁽⁵⁷⁾, body fat distribution⁽⁵⁸⁾ and BMI⁽⁵⁹⁾, but all these variants were not in LD with our variant. An *in vitro* model showed that CTNNA2 deficiency led to a cortical neuronal migration defect⁽⁵⁵⁾. Mice with CTNNA2 knock-out have been observed to have impaired cerebellar and body growth, but dyslipidemia was not described⁽⁶⁰⁾. Therefore, for both LRRTM4 and CTNNA2, more evidence is required that the variant affects gene expression in relevant (i.e., brain) tissue, and that this leads to dyslipidemia. Mouse models with mutations in the REG genes, for which chromatin interactions with rs114017774 were also observed, showed interesting phenotypes: mice with a mutation in REG3 showed increased levels of circulating total, LDL and HDL cholesterol⁽⁶¹⁾ and were at increased risk of developing fatty liver⁽⁶²⁾, and mice with overexpression of *REG1A* showed delayed development of autoimmune diabetes⁽⁶³⁾. However, these genes are mostly expressed in the pancreas, so a relationship with CRT is hard to argue.

We performed treatment-stratified GWASs because we hypothesized that the genetic effect may be different (present/absent) depending on exposure to certain treatments. Therefore, adding them as a covariate to a regression model would not capture this effect. Adding several interaction terms to a regression model would be another strategy, but this could lead to a complex and difficult to understand model. Also, *Rvtests* was unable to add interaction terms to the model. Therefore, we decided to first test whether potentially important variables (from literature) were associated with dyslipidemia in our study cohort to a certain significance (p<0.10) and as next step included these variables for stratified analyses.

The regression models that we additionally built suggest a significant interaction effect between genotype and CRT, with a stronger effect for higher dose. Previous studies observed that CRT was associated with dyslipidemia^(6, 64, 65) and this is likely caused by disruption of the hypothalamic-pituitary axis leading to hormonal deficiencies, particularly growth and sex hormone^(49, 66, 67). CRT was also a risk factor for obesity in previous studies^(12, 13, 68-70), which can in turn lead to dyslipidemia⁽⁴⁾, but in our analysis the genetic effect on dyslipidemia was direct, not through overweight. Conceivably, high dose radiotherapy (>25Gy), with the exception of medulloblastoma survivors, is most often local radiotherapy for a brain tumor, whereas low dose is likely radiotherapy to the whole cranium for leukemia treatment. We could take this difference into account in our dosimetry analysis of four brain regions. We observed that the frontal region, rather than hypothalamic pituitary region, appeared to be mostly associated with dyslipidemia. We did observe an association with higher dose radiation to the hypothalamic-pituitary region and dyslipidemia, as in fact for all four brain regions. An association between frontal lobe radiation damage and dyslipidemia has not been previously reported in literature.

Our results suggest that survivors carrying the variant rs114017774 are more vulnerable to CRT, supported by a high pooled odds ratio of 11.30 for this effect. It may be noted that the true risk can be lower, as genetic discovery analyses tend to overestimate the true effect size as a consequence of the "winner's curse"⁽⁷¹⁾. The fact that this variant is not reported in general population GWASs may suggest a survivor-specific effect, although its absence in the two available datasets may also suggest that it failed quality control. Replication in another cohort of cranially irradiated survivors is required to further argue its relevance. Preferably, as our cohorts only had survivors with dyslipidemia that were heterozygous for the variant, this additional cohort would contain homozygous survivors as well. If this locus turns out to be relevant, assessment of a survivor's genotype for this variant, in addition to clinical risk factors, may facilitate personalized risk stratification.

The protective variant rs676210 in the *APOB* gene, which encodes apolipoprotein B, that we replicated from the ALL survivors study, is also known in the general population⁽⁷²⁾. We could not replicate the variants in *BAD* and *OGFOD3*. This is the first time that a genetic variant related to dyslipidemia, in addition to ALL survivors, was replicated in an independent cohort of survivors of heterogeneous malignancies. In addition to treatment-related dyslipidemia, accelerated aging phenotypes are observed in survivors⁽⁷³⁾, suggesting that general population SNPs may also be relevant, and potentially earlier in life.

Three aspects of our study merit consideration. First, for suggestive loci to be included for replication analysis, supportive information for relatedness to dyslipidemia was required. This requirement lowers the risk of false-positive findings, but it excludes the possibility of identifying new pathways involved in the phenotype. In a next replication step, it may therefore still be interesting to also replicate the other seven SNPs identified in the initial discovery analysis. Second, the CCSS was a questionnaire-based study. In larger genetic studies it has been shown that self-reported phenotypes, including high cholesterol, perform better than one may expect^(74, 75), but in smaller samples as

ours self-report might lead to bias as survivors may be unaware of having dyslipidemia or be using lipid lowering medication as secondary prevention. At this age the first is more likely, leading to a bias towards the null hypothesis, hence, there may be other variants that we did not find due to the self-reported nature of the data. Third, we do not know whether non-replication in the CCSS Expansion cohort resulted from a power issue due to low dyslipidemia prevalence in CRT exposed survivors, or valid non-replication.

In conclusion, in our GWAS we identified variant rs114017774 on chromosome 2 as a potential genetic predisposing factor for dyslipidemia specifically in cranially irradiated survivors. We also, for the first time, replicated variant rs676210 in *APOB* in a heterogeneous survivor cohort, indicating that this general population variant is relevant in survivors as well.

Acknowledgments

We thank all survivors that participated in the CCSS, SJLIFE and CCSS Expansion cohort studies.

Conflicts of interest

The authors have no conflicts of interest to declare.

	- do
	ouo
	cati
	epli
	u pu
	ry a
	OVE
	disc
	the
	s of
	stic
	teri
Ē	arac
2	t ch
P.	nen
SI	eatr
T	dtr
	c an
5	phi
)gra
Ē	eme
SL	. D
AF	le 1
Ê	Tab

ion cohorts	
and replicat	
le discovery	
eristics of th	
nent charact	
c and treatn	
Jemographi	Í
Table 1. I	

		CCSS Ori	ginal				SJLIFE		
Characteristics	Entire cohort	Cases	Controls	Cases v controls	Entire cohort	Cases	Controls	Cases v controls	CCSS Original vs SJLIFE
Z	4,332	759 (17.5%)	3,573 (82.5%)	p<0.001	2,274	448 (19.7%)	1826 (80.3%)	p<0.001	p=0.03°
Male sex	2,081 (48.0%)	444 (58.5%)	1,637 (45.8%)	p<0.001	1,194 (52.5%)	278 (62.1%)	916 (50.2%)	p<0.001	p<0.001
Age at dyslipidemia liomosie*	37.3 (8.8)	37.3 (8.8)	n.a.	n.a.	37.2 (10.5)	37.2 (10.5)	n.a.	n.a.	p=0.70
Are at follow-un*	(1) (1)	48 7 (7 4)	(0 6) 6 7	00 0/u	36.7 (9.8)	(7) 2 (6) 4)	34 8 (0 3)	n~0.001	100 0/u
ablow-up time*	35.3 (6.8)	38.1 (5.6)	34.7 (6.8)	n<0.001	28.1 (9.3)	32.8 (8.8)	27.0 (9.1)	n<0.001	n<0.001
3MI at follow-up*	27.2 (6.8)	28.5 (6.0)	26.9 (6.9)	p<0.001	29.4 (7.7)	32.0 (7.1)	28.8 (7.7)	p<0.001	p<0.001
Overweight or obesity	2,521 (58.3%)	559 (73.6%)	1,962 (55.1%)	p<0.001	1,573 (69.2%)	383 (85.5%)	1,190 (65.2%)	p<0.001	p<0.001
Dhesity	1,109 (25.7%)	251 (33.1%)	858 (24.1%)	p<0.001	914 (40.2%)	261 (58.3%)	653 (35.8%)	p<0.001	p<0.001
Comorbidities (CTCAE gr	ade)								ı
GHD gr3	0	0	0	n.a.	2	0	2	n.a.	n.a.
GHD gr2	196 (4.5%)	50 (6.6%)	146(4.1%)	p=0.003	190 (8.4%)	47~(10.5%)	143 (7.8%)	p=0.06	p<0.001
GHD gr1	119 (2.7%)	27 (3.6%)	92 (2.6%)	_p=0.13	288 (12.7%)	88 (19.6%)	200 (11.0%)	p<0.001	p<0.001
Hypogon gr3	232 (5.4%)	58 (7.6%)	174(4.9%)	p=0.002	134 (5.9%)	45 (10.0%)	(4.9%)	p<0.001	p=0.36
Hypogon gr2	1	0	1	n.a	224 (9.9%)	103 (23.0%)	121 (6.6%)	p<0.001	p<0.001
Hypogon gr1	12(0.3%)	2(0.3%)	10(0.3%)	p=0.94	250 (11.0%)	57 (12.7%)	193(10.6%)	p=0.19	p<0.001
Diabetes gr3	119 (2.7%)	56 (7.4%)	63~(1.8%)	p<0.001	89 (3.9%)	34 (7.6%)	55 (3.0%)	p<0.001	p=0.01
Diabetes gr2	125 (2.9%)	83 (10.9%)	42 (1.2%)	p<0.001	104(4.6%)	49 (10.9%)	55 (3.0%)	p<0.001	p<0.001
Diabetes gr1	55 (1.3%)	19 (2.5%)	36~(1.0%)	p=0.0008	464 (20.4%)	130 (29.0%)	334 (18.3%)	p<0.001	p<0.001
Hypothyr gr3	0	0	0	n.a.	3	1	2	p=0.48	n.a.
Hypothyr gr2	678 (15.7%)	209 (27.5%)	469 (13.1%)	p<0.001	513 (22.6%)	151 (33.7%)	362 (19.8%)	p<0.001	p<0.001
Hypothyr grl	100(2.3%)	20 (2.6%)	80 (2.2%)	p=0.51	16(0.7%)	2	14(0.8%)	p=0.75	p<0.001
Cardiomyo gr5	3	0	3(0.1%)	n.a.	0	0	0	n.a.	n.a.
Cardiomyo gr4	2	1	1	n.a.	2	2	0	n.a.	p=0.61
Cardiomyo gr3	6(0.1%)	1	5(0.1%)	p=0.94	87 (3.8%)	39 (8.7%)	48 (2.6%)	p<0.001	p<0.001
Cardiomyo gr2	0	0	0	n.a.	225 (9.9%)	61 (13.6%)	164 (9.0%)	p=0.003	n.a.
Cardiomyo gr1	17~(0.4%)	7 (0.9%)	10(0.3%)	p=0.01	0	0	0	n.a.	n.a.

Table 1. Continued									
		CCSS Ori	iginal				SJLIFE		
Characteristics	Entire cohort	Cases	Controls	Cases v controls	Entire cohort	Cases	Controls	Cases v controls	CCSS Original vs SJLIFE
Smoking	881 (21.0%)	164 (21.8%)	717 (20.9%)	p=0.34	826 (36.3%)	174 (38.8%)	652 (35.7%)	p=0.22	p<0.001
Cancer diagnosis									
ALL	1165 (26.9%)	148 (19.5%)	1017 (28.5%)	p<0.001	792 (34.8%)	162 (36.2%)	630 (34.5%)	p=0.51	p<0.001
AML	83 (1.9%)	11(1.4%)	72 (2.0%)	p=0.30	45 (2.0%)	7 (1.6%)	38 (2.1%)	p=0.48	p=0.86
Astrocytoma	386 (8.9%)	49 (6.5%)	337 (9.4%)	p=0.009	0	0	0	n.a.	n.a.
Ewing sarcoma	124 (2.9%)	17 (2.2%)	107 (3.0%)	_p=0.26	81 (3.6%)	13 (2.9%)	68 (3.7%)	p=0.40	p=0.12
Hodgkin	575 (13.3%)	189 (24.9%)	386~(10.8%)	p<0.001	285 (12.5%)	79 (17.6%)	206 (11.3%)	p<0.001	p=0.40
Kidney tumor	426 (9.8%)	51 (6.7%)	375 (10.5%)	p=0.002	150 (6.6%)	24 (5.4%)	126 (6.9%)		p<0.001
Medullo/PNET	134(3.1%)	41 (5.4%)	93 (2.6%)	p<0.001	0	0	0	n.a.	n.a.
Neuroblast	323 (7.5%)	29 (3.8%)	294 (8.2%)	p<0.001	108(4.7%)	10 (2.2%)	98 (5.4%)	p=0.005	p<0.001
Non-Hodgkin	346 (8.0%)	78 (10.3%)	268 (7.5%)	p=0.01	171 (7.5%)	34 (7.6%)	137 (7.5%)	p=0.95	p=0.50
Osteosarc	237 (5.5%)	54 (7.1%)	183 (5.1%)	p=0.03	80 (3.5%)	27 (6.0%)	53 (2.9%)	p=0.001	p<0.001
Other bone	15(0.3%)	5 (0.7%)	10(0.3%)	p=0.11	0	0	0	n.a.	n.a.
Other CNS	91 (2.1%)	19 (2.5%)	72 (2.0%)	p=0.39	227 (10.0%)	46(10.3%)	181 (9.9%)	p=0.82	p<0.001
Other leukem	15 (3.5%)	5 (0.7%)	10(0.3%)	p=0.11	4 (0.2%)	1(0.2%)	3 (0.2%)	p=0.58	p=0.33
Soft tissue sarc	412 (9.5%)	63 (8.3%)	349~(9.8%)	p=0.21	61 (2.7%)	6(1.3%)	55 (3.0%)	p=0.050	p<0.001
Rhabdomyosarc	0	0	0	n.a.	74 (3.3%)	14(3.1%)	60(3.3%)	p=0.86	n.a.
Germ cell	0	0	0	n.a.	45 (2.0%)	3 (0.7%)	42 (2.3%)	p=0.02	n.a.
Histiocytosis	0	0	0	n.a.	17~(0.7%)	1(0.2%)	16(0.9%)	p=0.22	n.a.
Liver	0	0	0	n.a.	16(0.7%)	2(0.4%)	14(0.8%)	p=0.75	n.a.
Melanoma	0	0	0	n.a.	14(0.6%)	3 (0.7%)	11 (0.6%)	p=0.75	n.a.
Retinoblastoma	0	0	0	n.a.	62 (2.7%)	9 (2.0%)	53 (2.9%)	p=0.30	n.a.
Other	0	0	0	n.a.	42 (1.8%)	7 (1.6%)	35 (1.9%)	p=0.62	n.a.

.
Table 1. Continued									
		CCSS Ori	ginal				SJLIFE		
Characteristics	Entire cohort	Cases	Controls	Cases v controls	Entire cohort	Cases	Controls	Cases v controls	CCSS Original vs SJLIFE
Radiotherapy									
Cranial	1242 (30.5%)	236 (33.0%)	1006(30.0%)	p=0.10	669 (29.4%)	193(43.1%)	476 (26.1%)	p<0.001	p=0.52
Dose (cGy)**	200 (20-2400)	20 (20-2400)	200 (20-2400)	p=0.066	1800 (20-	1800 (20-2400)	1250 (20-2400)	***	***
Region 1	1050 (25.8%)	270 (37.7%)	780 (23.3%)	p<0.001	2400)	135 (30.1%)	283 (15.5%)	p<0.01	p<0.001
Region 2	200 (20-2400)	200 (20-3100)	200 (20-2100)	p<0.001	n.a.	200 (20-2100)	20 (20-1500)	***	***
Region 3	15(0.4%)	6 (0.8%)	9(0.3)	p=0.02	n.a.	0	0	n.a.	n.a.
Region 4					n.a.				
Abdominal					n.a.				
Dose (cGy)**					418 (18.4%)				
Total body					20 (20-1800)				
					0				
Chemotherapy									
Anthrac (DED)	1444 (34.9%)	239 (32.7%)	1205 (35.4%)	p=0.24	1320 (58.0%)	240 (53.6%)	1080 (59.1%)	p=0.03	p<0.001
Dose (mg/m ²)**	258 (153-370)	277 (170-397)	253 (152-363)	p=0.037	154 (52-241)	174 (75-306)	152 (52-213)	***	***
Alkyl (CED)	1959 (47.4%)	378 (51.9%)	1581 (46.5%)	p=0.005	1302 (57.3%)	288 (64.3%)	1014 (44.5%)	p<0.001	p<0.001
Dose (g/m ²)**	8.7 (4.4-13.9)	10.5 (5.9-15.0)	8.2 (4.0-13.6)	p<0.001	8.6 (5.4-12.5)	8.6 (5.6-14.1)	8.5 (5.3-12.2)	* *	***
Asparagin	1016 (25.6%)	129 (18.5%)	887 (27.1%)	p<0.001	774 (34.0%)	135 (30.4%)	639 (35.0%)	p=0.052	p<0.001
Steroids									
Predniso(lo)ne, m-predn	1547 (39.3%)	270 (39.1%)	1277 (39.4%)	P=0.93	1129 (49.6%)	231 (51.6%)	898 (49.2%)	p=0.37	p<0.001
Dexameth	278 (6.8%)	45 (6.2%)	233 (6.9%)	P=0.55	182 (8.0%)	6 (1.3%)	176 (9.6%)	p<0.001	p=0.02

Table 1. Continued					
			CCSS Expansion		
Characteristics	Entire cohort	Cases	Controls	Cases v controls	CCSS Original v CCSS Expansion
N	2,212	119 (5.4%)	2093 (94.6%)	p<0.001	p<0.001°
Male sex	1051 (47.5%)	79 (66.4%)	972 (46.4%)	p<0.001	p=0.69
Age at dyslipidemia diagnosis*	27.5 (7.1)	27.5 (7.1)	n.a.	n.a.	p<0.001
Age at follow-up*	30.0(6.3)	34.0(6.4)	29.8 (6.3)	p<0.001	p<0.001
Follow-up time	21.3(3.9)	22.5 (3.8)	21.3(3.9)	p=0.001	p<0.001
BMI at follow-up*	26.4(6.0)	28.8(6.4)	26.3 (6.0)	p<0.001	p<0.001
Overweight or obesity	1144 (51.7%)	84 (70.6%)	1060 (50.6%)	p<0.001	p<0.001
Obesity	496 (22.4%)	41 (34.5%)	455 (21.7%)	p=0.0012	p=0.005
Comorbidities (CTCAE grade)					
GHD gr3	0	0	0	n.a.	n.a.
GHD gr2	159 (7.2%)	22 (18.5%)	137 (6.5%)	p<0.001	p<0.001
GHD gr1	47 (2.1%)	7 (5.9%)	40 (1.9%)	p=0.003	_p=0.13
Hypogon gr3	61 (2.8%)	9 (7.6%)	52 (2.5%)	p<0.001	p<0.001
Hypogon gr2	0	0	0	n.a.	n.a.
Hypogon gr1	5 (0.2%)	1 (0.8%)	4(0.2%)	p=0.24	p=0.70
Diabetes gr3	14 (0.6%)	2 (1.7%)	12 (0.6%)	p=0.17	p<0.001
Diabetes gr2	16 (0.7%)	5 (4.2%)	11 (0.5%)	p<0.001	p<0.001
Diabetes gr1	38(1.7%)	9 (7.6%)	29 (1.4%)	p<0.001	p=0.15
Hypothyr gr3	0	0	0	n.a.	n.a.
Hypothyr gr2	318 (14.4%)	50 (42.0%)	268 (12.8%)	p<0.001	p=0.17
Hypothyr gr1	52 (2.4%)	5 (4.2%)	47 (2.2%)	p=0.17	p=0.92
Cardiomyo gr5	0	0	0	n.a.	n.a.
Cardiomyo gr4	0	0	0	n.a.	n.a.
Cardiomyo gr3	0	0	0	n.a.	n.a.
Cardiomyo gr2	66(3.0%)	12 (10.1%)	54 (2.6%)	p<0.001	n.a.
Cardiomyo gr1	0	0	0	n.a.	n.a.

Table 1. Continued					
			CCSS Expansion		
Characteristics	Entire cohort	Cases	Controls	Cases v controls	CCSS Original v CCSS Expansion
Smoking	674 (30.5%)	42 (35.3%)	632 (30.2%)	p=0.24	p<0.001
Cancer diagnosis					
ALL	400(18.1%)	14(11.8%)	386~(18.4%)	p=0.066	p<0.001
AML	49 (2.2%)	1 (0.8%)	48 (2.3%)	p=0.52	p=0.42
Astrocytoma	352 (15.9%)	13(10.9%)	339 (16.2%)	p=0.13	p<0.001
Ewing sarcoma	74 (3.3%)	3 (2.5%)	71 (3.4%)	p=0.80	p=0.28
Hodgkin	274 (12.4%)	31 (26.1%)	243 (11.6%)	p<0.001	p=0.31
Kidney tumor	217(9.8%)	6(5.0%)	211 (10.1%)	p=0.072	p=0.98
Medullo/PNET	140(6.3%)	19(16.0%)	121 (5.8%)	p<0.001	p<0.001
Neuroblast	160(7.2%)	2 (1.7%)	158 (7.5%)	p=0.01	p=0.74
Non-Hodgkin	208(9.4%)	7 (5.9%)	201 (9.6%)	p=0.18	p=0.052
Osteosarc	101(4.6%)	10(8.4%)	91(4.3%)	p=0.04	p=0.12
Other bone	17(0.8%)	0	17~(0.8%)	n.a.	p=0.02
Other CNS	92 (4.2%)	4(3.4%)	88 (4.2%)	p=0.82	p<0.001
Other leukem	17(0.8%)	0	17~(0.8%)	n.a.	p=0.02
Soft tissue sarc	111(5.0%)	9 (7.6%)	102 (4.9%)	p=0.19	p<0.001
Rhabdomyosarc	0	0	0	n.a.	n.a.
Germ cell	0	0	0	n.a.	n.a.
Histiocytosis	0	0	0	n.a.	n.a.
Liver	0	0	0	n.a.	n.a.
Melanoma	0	0	0	n.a.	n.a.
Retinoblastoma	0	0	0	n.a.	n.a.
Other	0	0	0	n.a.	n.a.

Characteristics			CCSS Expansion		
	Entire cohort	Cases	Controls	Cases v controls	CCSS Original v CCSS Expansion
Kadiotnerapy					
Cranial	408 (18.4%)	41 (34.5%)	367 (17.5%)	p<0.001	p<0.001
dose**	1150 (20-5300)	1800 (20-5400)	830 (20-5200)	***	***
Region 1	n.a.	32 (26.9%)	306 (14.6%)	p<0.001	p<0.001
Region 2	n.a.	200 (20-2900)	200 (20-2100)	* * *	***
Region 3	n.a.	0	0	n.a.	P=0.006
Region 4	n.a.				
Abdominal	338 (15.3%)				
dose**	200 (20-2100)				
Total body	0				
Chemotherapy					
Anthrac (DED)	1158 (52.4%)	62 (52.1%)	1096 (52.4%)	p=0.96	p<0.001
dose**	165 (120-280)	200 (135-303)	163 (119-273)	***	***
Alkyl (CED)	1140 (51.5%)	72 (60.5%)	1068(51.0%)	p=0.04	p<0.001
dose**	(6.9 (3.8-11.5))	8.4(4.9-11.9)	6.9(3.6-11.5)	* * *	***
Asparagin	433 (19.6%)	17(14.3%)	71 (3.4%)	p<0.001	p<0.001
Steroids					
Prednisone and/or dexam	875 (39.6%)	47 (39.5%)	828 (39.6%)	p=0.99	n.a.°°

normally distributed continuous variables, Mann-Whitney U test for non-normally distributed continuous variables. *** Not possible to compare Mann-Whitney U test from summary Due to differences in childhood cancer classification some tumors have no occurrence. Smoking is defined as ever smoking, including current and former. Radiotherapy dose includes statistics without raw data. ^o Comparison of dyslipidemia prevalence between cohorts. ^{oo} in CCSS Expansion prednisone and dexamethasone are registered as one variable. ò registration of 20cGy for low stray exposure and 200cGy for high stray exposure. 1an (IUN). UIII2 IESI IOI LAIEBUI mean (JUC)

		, ,	3				CCSS	scovery				` [~]	eplicatio SJLIFE				[∞] S	eplicatio S Expan	n sion		Meta-ar	alysis three	cohorts
SNP	Interesting gene(s)	Biological function(s)	RA	EA	z	EAF	Imp.	g	95%CI	4	z	EAF	ß	95%CI	<u>~</u>	z	EAF	ä	95%CI	<u>م</u>	ß	95%CI	4
1:97869183	LPPR5 (c.i.)	Free fatty acid synthesis	Н	U	4,323	0.016	1	3.00	1.88-4.76	3.60E-6	2,274	0.013	1.26	0.64-2.36	0.49	2,210	0.014	2.60	0.90-7.54	0.079	2.26^{\dagger}	1.58-3.24	7.63E-6
2:49370715	,	· ,	U	A	4,323	0.014	0.922	3.55	2.13-5.90	1.10E-6	,	,	,	,	,	,	ï	,	,	,	,	,	,
2:222263469*	SGPP2 (c.i.)	Lipid metabolism	Α	G	4,323	0.020	1	3.62	2.40-5.48	1.02E-9	2,274	0.015	1.34	0.72-2.37	0.34	2,212	0.020	1.79	0.83-3.88	0.14	2.13^{\ddagger}	1.08-4.18	0.028
6:32472999	,		-	Ω	4,323	0.011	0.669	5.69	2.89-11.20	4.86E-7	,	,	,	,	,	,	,	,		,	,	,	,
6:52486401	,	,	A	U	4,323	0.157	0.999	0.69	0.59-0.81	4.94E-6	ï	ï	,	ï	,	,	ï	,	,	,	,	,	,
7:5187718	1	,	A	G	4,323	0.955	0.937	0.50	0.38-0.68	4.67E-6	,	,	,	,		,	,	,	,	,	,	,	,
8:29262306	LEPROTL1 (c.i.) MBOAT4 (c.i.)	Growth hormone signaling in liver Ghrelin and leptin function	G	O	4,323	0.204	0.948	0.71	0.61-0.82	3.33E-6	2,203	0.199	1.00	0.82-1.21	1.00	2,209	0.211	0.87	0.59-1.28	0.48	0.84^{4}	0.65-1.09	0.19
10:3302154	IDII (c.i.)	Cholesterol synthesis	O	Н	4,323	0.633	0.987	0.75	0.67-0.85	4.43E-6	2,271	0.638	1.09	0.93-1.28	0.29	2,212	0.644	0.84	0.62-1.15	0.28	0.89^{\ddagger}	0.68-1.16	0.39
			(1																	*		
10:01/>440	1FAM (n.g.) CISD1 (¢QTL) SLC16A9 (c.i.)	Mitochondrial transcription regulation Adipogenesis and energy metabolism Lipid transport	0	-	4,323	0.110	-	(<u>c</u> .1	1.29-1.8/	3.15E-6	7/7,7	0.098	 /1-1	06.1-16.0	0.22	2,212	0.102	10.1	0.61-1.66	0.0/	1.38	66.1-02.1).69E-6
10:74473228	,		U	A	4,323	0.016	0.981	3.13	1.92-5.09	4.46E-6	ï	ı	,	ï	ı	,	ı	,	,	,	,	,	,
15:70769440	UACA (n.g.) THSD4 (c.i.)	Apoptosis in pancreas and adipose tissue Free fatty acid synthesis	U	U	4,323	0.011	0.618	5.86	2.79-12.32	3.16E-6	2,273	0.013	0.87	0.43-1.63	0.67	2,211	0.012	0.63	0.17-2.38	0.50	1.55*	0.37-6.52	0.55
16:82711821	CDH13 (ng) S1P/MBTPS1 (c.i.) HSD17B2 (c.i.) SDR42E1 (c.i.) (c.i.)	Adiponectin receptor Lipid metabolism Androgen and estrogen metabolism Steroid hormone and lipid metabolism	U	<	4,323	0.093	-	0.60	0.49-0.73	6.25E-7	2,273	0.095	0.91	0.69-1.17	0.46	2,212	0.101	1.31	0.89-1.93	0.17	0.87*	0.56-1.33	0.52
17:34109023	CCLs (c.i.)	Pro-inflammatory	Υ	Н	4,323	0.020	0.883	2.93	1.90-4.54	1.32E-6	2,199	0.018	0.90	0.47-1.60	0.73	2,205	0.018	0.53	0.16-1.74	0.29	1.22^{\ddagger}	0.44-3.45	0.70
19:53876108	NLRP12 (c.i.)	pathway Pro-inflammatory pathwav	V	IJ	4,323	0.711	1	1.38	1.21-1.57	8.60E-7	2,271	0.708	0.94	0.80-1.12	0.51	2,212	0.718	1.09	0.74-1.61	0.66	1.13^{4}	0.85-1.51	0.39
19:54163026	,	-	U	F	4,323	0.031	0.613	2.92	1.88-4.53	1.78E-6	,	,		,	,	,	,	,	,	,	,	,	,
20:52550194	,	,	Υ	G	4,323	0.036	0.994	2.09	1.53-2.85	3.40E-6	,	,	,	,	,	,	,	,	,	,	'	,	,
22:16924826	ı	,	U	Н	4,323	0.017	0.858	3.25	1.99-5.29	2.20E-6	,	,	,	,	ī	,	,	,		,	,	,	,
SNP positio	n is accordin	ig to 1000Genoi	nes b	uild ŝ	37.					5													

*Hit that reached significant threshold; RA = reference allele; EA = effective allele; EAF = effective allele frequency; imp. = imputation quality; n.g. = nearest gene; eQTL = expression Logistic regression analysis adjusted for first four genetic ancestry principal components, age at last follow-up (continuous) and sex. quantitative trait locus; c.i. = chromatin interaction; † fixed effects model; ‡ random effects model

Genetic susceptibility to treatment related dyslipidemia in adult long-term survivors (...)

Chapter 6

						Disc CCSS (overy Driginal					Replicat SILIF	ion			1 00	teplicatio SS Expan	n sion		CCS	Meta-analysis S Original + SI	LIFE
Subgroup	SNP	RA	EA	z	EAF	Imp.	oR	95%CI	4	z	EAF	ß	95%CI	4	z	EAF	OR	95%CI	ď	OR	95%CI	4
Abd RT +	6:15911191	G	Υ	1,044	0.068	0.903	3.82	2.47-5.91	1.65E-9	418	0.084	0.53	0.27-0.96	0.036	338	0.089	1.02	0.42-2.45	0.97	1.44	0.21-10.00	0.71
CRT +	2:76547279	U	г	1,239	0.012	1	16.80	6.36-44.22	1.18E-8	699	0.006	4.27	0.97-22.11	0.055	407	0.014	0.98	0.13-7.55	0.99	11.30	5.03-25.40	4.46E-9
CRT -	8:49202507	D	Ι	3,084	0.707	0.946	0.62	0.53-0.73	6.39E-9	n.a.*	ï	ı	,	,	n.a.*	,	ï	,	ı	,	1	,
CRT -	16:54120330	Т	IJ	3,084	0.325	1	1.53	1.131-1.77	3.30E-8	1586	0.329	0.97	0.78-1.20	0.77	1,687	0.33	1.08	0.73-1.58	0.71	1.22	0.79-1.90	0.37
Asp +	5:73438634	G	¥	1,016	0.011	1	50.30	12.30- 205.26	4.88E-8	774	0.009	1.51	0.32-5.40	0.57	433	0.007	2.81	0.13-61.94	0.56	8.68	0.28-271.00	0.22
Asp +	8:138526638	Y	U	1,016	0.872	1	0.30	0.20-0.46	3.71E-8	773	0.878	1.24	0.82-1.93	0.32	433	0.857	1.29	0.37-4.45	0.68	0.61	0.16-2.42	0.48
Asp +	22:49419475	G	A	1,016	0.017	0.94	24.9	8.30-74.55	9.42E-9	774	0.01	0.45	0.02-2.40	0.40	433	0.009	n.a.**	,	,	3.69	0.07-187.00	0.51
Pred +	2:166087768	U	Τ	1,545	0.019	0.877	8.55	4.03-18.15	2.30E-8	1,122	0.014	0.81	0.28-2.01	0.66	875	0.016	0.55	0.05-5.93	0.59	2.70	0.27-27.20	0.40
Pred -	17:77873943	C***	Τ	2,778	0.04	0.903	3.17	2.11-4.76	2.66E-8	1,151	0.053	0.76	0.43-1.26	0.30	1,257	0.048	0.69	0.27-1.79	0.43	3.16	2.09-4.77	4.35E-8
SNP pos	tion is accord	ing to	10000	renomes	s build 3	7.																

RA = reference allele; EA = effective allele; EAF = effective allele frequency; imp. = imputation quality

* Failed quality control; ** No dyslipidemia cases with genetic variant; *** The reference allele C vs effect allele T in CCSS Original corresponded to an INDEL in SJLIFE and CCSS Expansion at this position with reference allele TTAAC vs effect allele T. The C vs T SNP also passed quality control but had a much lower EAF compared to the CCSS Original data.

Logistic regression analysis adjusted for first four genetic ancestry principal components, age at last follow-up (continuous) and æx.

Chapter 6

Table 3. Treatment-stratified GWASs

Table 4. Interaction analysis between genotype and cranial radiotherapy

Results of logistic regression analysis with outcome dyslipidemia and as predictors genotype, CRT, interaction between genotype and CRT, age, sex and genetic ancestry

Variable	OR	95% CI	P-value
Genotype dosage	0.59	0.26 - 1.36	0.22
CRT yes	1.31	1.08 - 1.58	0.005
Genotype*CRT	14.61	4.59 - 46.55	5.75*10 ⁻⁶
Age (per year)	1.1	1.09 - 1.11	<0.001
Female sex	0.54	0.45 - 0.64	<0.001
PC1	27.45	0.15 - 5051.4	0.21
PC2	5.6	0.02 - 1561.6	0.55
PC3	0.37	0 - 105.5	0.73
PC4	0.01	0 - 3.28	0.12

Table 5. Re _l	plication analys	sis of Sl	NPs id	entified in ca	andida	te SNI	P study	by England	l et al.										
						ALL	study E	ngland et al			S	SS Enti	e cohort			ő	S ALL	subgroup	
rsID	chr: pos37	REF	ALT	Gene	z	EAF	OR	95%CI	Ρ	z	EAF	OR	95%CI	Ρ	z	EAF	OR	95%CI	Ρ
rs676210	2:21231524	IJ	Α	APOB	209	0.23	0.43	0.22-0.88	0.020	4,323	0.223	0.82	0.71 - 0.94	0.0049	1,165	0.212	0.64	0.46 - 0.88	0.0061
rs2286615	11:64039175	ს	Α	BAD	209	0.10	4.02	1.44-11.23	0.008	4,323	0.163	1.08	0.92-1.26	0.37	1,165	0.164	1.29	0.91 - 1.83	0.16
rs62079523	17:80352303	IJ	Α	OGFOD3	209	0.33	2.71	1.35-5.44	0.005	4,323	0.355	0.97	0.86 - 1.10	0.65	1,165	0.369	1.01	0.77 - 1.31	0.96

England et al used a different regression model including more treatment and lifestyle covariables. So only direction of effect and significance are comparable.



Figure 1. Manhattan plot of the initial discovery analysis



Figure 2. Manhattan plot of the CRT+ discovery analysis

SUPPLEMENTAL DATA

Supplemental table 1. Definition of dyslipidemia in study cohorts

CCSS & CCSS Expansion	SJLIFE
CTCAE high total cholesterol or hypertriglyceridemia	CTCAE high total cholesterol or hypertriglyceridemia
grade 2: self-reported high cholesterol or triglycerides and	grade 2: total cholesterol >300 mg/dL or triglycerides
on medication	>300 mg/dL or treatment with >=1 lipid lowering agent

Supplemental table 2. Clinical model to determine treatment factors for stratified GWASs

Predictor	OR	95% CI	P-value
Male sex	1.90	1.59-2.28	<0.001
Age (per year)	1.09	1.09-1.11	<0.001
Cranial radiotherapy	1.57	1.26-1.94	<0.001
Abdominal radiotherapy	1.86	1.53-2.25	<0.001
Asparaginase	0.68	0.50-0.92	0.012
Prednisone	1.23	0.98-1.55	0.069
Dexamethasone	1.12	0.78-1.62	0.54

Supplemental table 3. Baseline characteristics of survivors carrying the alternative allele of the significant locus in CRT exposed analysis.

n	29
Heterozygous	29 (100%)
Dyslipidemia	17 (58.6%)
Female	15 (51.7%)
Age at dyslipidemia dx (mean (SD))	34.0 (9.8)
Age (mean (SD))	41.8 (7.0)
BMI at follow-up (mean (SD))	31.0 (7.2)
Overweight at follow-up = yes (%)	24 (82.8%)
Obese at follow-up = yes (%)	12 (41.4%)
Diagnosis (%)	
ALL	11 (37.9%)
AML	12 (41.4%)
Astrocytoma	3 (10.3%)
Hodgkin	2 (6.9%)
Non-Hodgkin	1 (3.4%)
CRT dose (median [IQR]) (Gy)	34 [18-52]
Abd RT	8 (27.6%)
TBI	1 (3.4%)
Anthracyclines	9 (31.0%)
Alkylating agents	16 (55.2%)
Asparaginase	9 (31.0%)
Prednisone	15 (51.7%)
Dexamethasone	2 (6.9%)
Growth hormone deficiency (%)	
No	17 (58.6%)
Grade 1	6 (20.7%)
Grade 2	6 (20.7%)
Hypogonadism (%)	

Supplemental table 3. Continued	
No	26 (89.7%)
Grade 3	3 (10.3%)
Diabetes (%)	
No	24 (82.8%)
Grade 1	2 (6.9%)
Grade 2	2 (6.9%)
Grade 3	1 (3.4%)
Hypothyroidism (%)	
No	18 (62.1%)
Grade 1	1 (3.4%)
Grade 2	10 (34.5%)
Cardiomyopathy (%)	
No	29 (100%)
Smoking (%)	4 (13.8%)

Supplemental table 3. Continued

(rem)	DMAP) in primary No ell, foreskin nosyrt, 14/15 n, foreskin n, foreskin optor	NP Chro n cell (ROA 9/15 i 9/15 i fibrob fibrob fibrob in live artiur melann melann	IP in the second

Supplement	tal table	4. Conti	nued										
SNP (rs/ CHR:POS_ REF/ALT)	R²	SNP in study	SNP in cell line	Chromatin state (ROADMAP)	eQTL	Chromatin interaction (dist)	Nearest protein coding gene (dist)	Protein function	Protein expression in tissue (GTeX)	Gene polymor- phisms in study on lipids	Gene polymor- phisms in study on other outcome	Gene knockout in mouse	Gene knockout in vitro
						REG1B (2.76Mbp)		Associated with pancreatitis and inflammation- associated pancreatic carcinogenesis	Pancreas				
						REG3A (2.84Mbp)		Associated with pancreatitis and inflammation- associated pancreatic carcinogenesis	Pancreas				
						REG3G (2.71Mbp)		Associated with pancreatitis and inflammation- associated pancreatic carcinogenesis	Pancreas, liver			Increased levels of circulating LDL and HDL cholesterol ⁽¹⁸⁾ Alcoholic steatohepatitis ⁽¹⁹⁾	
1. Schaffer migratio	AE, Breu n. Nat G	1ss MW, 1 enet. 201	Caglaya 18;50(8)	n AO, et al. Biall :1093-101.	lelic loss	of human (CTNNA2,	encoding alphaN	-catenin, leads	to ARP2/3 con	nplex overactivity	and disordered o	ortical neuronal
2. Cheng F	H, Sewda	A, Marq	uez-Lun	a C, et al. Genet	tic archit	ecture of car	diometab	olic risks in people	living with H	IV. BMC Med.	2020;18(1):288.	-	
 Koyama 2020;52(S, Ito K, (11):1169	, Tèrao C 9-77.	, et al. I	opulation-specif	hc and t	rans-ancestry	y genome-	wide analyses iden.	titly distinct a	nd shared genet	ic risk loci for co	ronary artery dise	ase. Nat Genet.
 Pulit SL, 2019;28(, Stonem; (1):166-7	an C, Mc 74.	orris AP,	et al. Meta-anal	ysis of g	enome-wide	associatio	n studies for body	fat distributio	n in 694 649 ir	ndividuals of Eur	opean ancestry. H	um Mol Genet.
 Hoffmar Cook SA 	nn TJ, Cł V, Bronso	hoquet H n RT, Dc	I, Yin J, 2014 J	et al. A Large Mı LR, Ben-Arie N,	ultiethni Davisso	ic Genome-V n MT. Cerel	Wide Asso bellar defic	ciation Study of Ac cient folia (cdf): a 1	Hult Body Mas new mutation	ss Index Identifi on mouse chroi	ies Novel Loci. G mosome 6. Mam	enetics. 2018;210 m Genome. 1997	(2):499-515. ;8(2):108-12.
7. Lauren J.	, Airaksin	nen MS, 5	Saarma l	M, Timmusk T. A	A novel g	ene family e	ncoding le	ucine-rich repeat ti	ransmembrane	e proteins differe	entially expressed	in the nervous sys	tem. Genomics.
8. Kathires	an S, Mai	nning Ak	ζ, Demi	ssie S, et al. A ger	nome-wi	ide associatic	on study fo	yr blood lipid phen	otypes in the l	Framingham He	eart Study. BMC	Med Genet. 2007	;8 Suppl 1:S17.
9. Meigs JE	3, Mannii	ng AK, F	ox CS, 6	et al. Genome-wi	ide assoc	iation with	diabetes-re	slated traits in the]	Framingham F	Heart Study. BN:	AC Med Genet. 2	007;8 Suppl 1:S1	6. 210 - 1 /2) 20 /
10. Jansen F 403.	K, watar	labe N, S	tringer :	o, et al. Genome-	-wide an.	alysis of insc	imnia in 1	, <i>22</i> 1,010 INGIVIAU	als identifies no	ew risk loci and	runctional patny	ays. Ivat Genet. 2	-+4C:(C)1C;410
11. Siddiqui	TJ, Tari	PK, Con	nor SA,	et al. An LRRTN	M4-HSF	PG complex	mediates (excitatory synapse o	development c	m dentate gyrus	s granule cells. No	euron. 2013;79(4)	:680-95.
12. Parikh A	., Stephar	n AF, Tza.	nakakis	ES. Regenerating	g protein	s and their	expression	, regulation and sig	3naling. Biom	ol Concepts. 20	12;3(1):57-70.		
13. Xing H,	Chen X,	Sun H,	Han Y,	Ding L, Chen X.	. Associa	ution of regen	nerating g	ene 1A single-nucl	eotide polymo	orphisms and na	usopharyngeal car	cinoma susceptibi	lity in southern

Chinese population. Eur Arch Otorhinolaryngol. 2020;277(1):221-6.

Xing H, Chen X, Sun H, Han Y, Ding L, Chen X. Association of regenerating gene 1A single-nucleotide polymorphisms and nasopharyngeal carcinoma susceptibility in southern

- Xing HJ, Chen XD, Sun HX, et al. The Relevance of Regenerating Gene 1a Polymorphisms to Radiation Sensitivity and Survival of Nasopharyngeal Carcinoma Receiving Radiotherapy in a Southern Chinese Population. Pharmgenomics Pers Med. 2021;14:1403-13.
- 15. Unno M, Nata K, Noguchi N, et al. Production and characterization of Reg knockout mice: reduced proliferation of pancreatic beta-cells in Reg knockout mice. Diabetes. 2002;51 Suppl 3:S478-83.
- Mao H, Jia J, Sheng J, et al. Protective and anti-inflammatory role of REG1A in inflammatory bowel disease induced by JAK/STAT3 signaling axis. Int Immunopharmacol. 2021;92:107304.
- 17. Hayashi K, Motoyama S, Koyota S, et al. REG I enhances chemo- and radiosensitivity in squamous cell esophageal cancer cells. Cancer Sci. 2008;99(12):2491-5.
- International Mouse Phenotyping Consortium. REG3G knockout mouse phenotype [Available from: https://www.mousephenotype.org/data/genes/MGI:109406.
- Wang L, Fouts DE, Starkel P, et al. Intestinal REG3 Lectins Protect against Alcoholic Steatohepatitis by Reducing Mucosa-Associated Microbiota and Preventing Bacterial Translocation. Cell Host Microbe. 2016;19(2):227-39.

react value OR 3790 114 114 3.14 0.014 CRT added (whole cohort) Genotype dosage 1.59 1.14 3.14 0.014 CRT yes 1.41 1.17 1.70 4.0014 Interaction with CRT added (whole cohort) Genotype dosage 0.59 0.26 -1.36 0.22 CRT yes 1.31 1.08 -1.58 0.0005 Genotype CRT 14.64 4.59 -46.55 -0.001 Interaction with CRT dose added (continuous) Genotype dosage 0.76 0.37 1.57 0.46 (whole cohort) CRT dose (cont.), per Gy 1.01 1.003 -1.013 0.002 Genotype dosage 1.82 1.09 -3.03 0.021 CRT dose added (categorized) (whole cohort) Genotype dosage 1.61 1.06 1.33 -2.46 0.001 Interaction with CRT dose added (categorized) Genotype dosage 1.01 1.02 -3.54 0.001	Model	Variable	OP	95% CI	Pavalua
CRT accet (whole conort) Centry per dosage 1.49 $1.17 - 1.70$ 0.001 Interaction with CRT added (whole cohort) Genotype dosage 0.59 $0.26 - 1.36$ 0.22 CRT yes 1.31 $1.08 - 1.58$ 0.005 Genotype CRT 14.61 $4.59 - 4.655$ -0.001 Interaction with CRT dose added (continuous) Genotype dosage 1.87 $1.13 - 3.11$ 0.015 CRT dose (cont.), per Gy 1.01 $1.005 - 1.015$ -0.001 Interaction with CRT dose added (continuous) Genotype dosage 0.76 $0.37 - 1.57$ 0.46 (whole cohort) Genotype dosage 1.06 $1.03 - 1.013$ 0.002 CRT dose (cont.), per GY 1.06 $1.03 - 1.013$ 0.021 CRT dose added (categorized) (whole cohort) Genotype dosage 1.82 $1.09 - 3.03$ 0.021 CRT dose added (categorized) Genotype dosage 1.01 $0.28 - 3.54$ 0.98 (whole cohort) Genotype dosage 1.01 $0.25 - 4.61$ 0.93 (whole cohort) Genotyp	CDT added (whole each = ===)	Conotino dorrer	1.00	1 1 / 2 1 /	0.014
$ \begin{array}{c crrrr} \mbox{CRT ges} & 1.41 & 1.1.7 - 1.70 & 0.0001 \\ \mbox{CRT added} (whole cohorr) & Genorype dosage & 0.59 & 0.26 - 1.36 & 0.22 \\ \mbox{CRT yes} & 1.31 & 1.08 - 1.58 & 0.005 \\ \mbox{Genorype*CRT} & 14.61 & 4.59 - 46.55 & 0.001 \\ \mbox{CRT dose added} (continuous) (whole cohorr) & Genorype dosage & 1.87 & 1.13 - 3.11 & 0.015 \\ \mbox{CRT dose added} (continuous) & Genorype dosage & 0.76 & 0.37 - 1.57 & 0.46 \\ \mbox{(whole cohorr)} & CRT dose (cont.), per Gy & 1.01 & 1.003 - 1.013 & 0.002 \\ \mbox{Genorype*CRT dose} & 1.06 & 1.03 - 1.09 & 0.001 \\ \mbox{(cont.)} & CRT dose (cont.), per Gy & 1.01 & 1.003 - 1.013 & 0.002 \\ \mbox{Genorype*CRT dose} & 1.06 & 1.03 - 1.09 & 0.001 \\ \mbox{(CRT dose added} (categorized) (whole cohorr) & Genorype dosage & 1.82 & 1.09 - 3.03 & 0.021 \\ \mbox{(CRT dose added} (categorized) & Genorype dosage & 1.82 & 1.09 - 3.03 & 0.021 \\ \mbox{(CRT dose added} (categorized) & Genorype dosage & 1.01 & 0.28 - 3.64 & 0.98 \\ \mbox{(whole cohort)} & CRT dose 0.2500cGy & 2.00 & 1.63 - 2.46 & 0.98 \\ \mbox{(whole cohort)} & CRT dose 0.2500cGy & 2.40 & 1.78 - 3.23 & 0.001 \\ \mbox{(CRT dose 2500cGy & 2.40 & 1.78 - 3.23 & 0.001 \\ \mbox{(CRT dose 2500cGy & 2.40 & 1.78 - 3.23 & 0.001 \\ \mbox{(CRT dose 2500cGy & 2.40 & 1.78 - 3.23 & 0.001 \\ \mbox{(CRT dose 2500cGy & 2.40 & 1.78 - 3.23 & 0.001 \\ \mbox{(CRT dose 2500cGy & 2.40 & 1.78 - 3.23 & 0.001 \\ \mbox{(CRT dose 2500cGy & 2.40 & 1.78 - 3.23 & 0.001 \\ \mbox{(CRT dose 2500cGy & 2.40 & 1.78 - 3.23 & 0.001 \\ \mbox{(CRT dose 2500cGy & 2.40 & 1.78 - 3.23 & 0.001 \\ \mbox{(CRT dose 2500cGy & 2.40 & 1.78 - 3.23 & 0.001 \\ \mbox{(CRT dose 2500cGy & 2.40 & 1.78 - 3.23 & 0.001 \\ \mbox{(CRT dose 2500cGy & 2.40 & 1.78 - 3.23 & 0.001 \\ \mbox{(CRT dose 2500cGy & 2.40 & 1.78 - 3.23 & 0.001 \\ \mbox{(categorized)} (CRT + only) & \mbox{(CRT dose 2500cGy & 1.50 & 1.10 - 0.40 & 0.23 \\ \mbox{(categorized)} (CRT + only) & \mbox{(CRT dose 2500cGy & 1.50 & 1.10 - 0.40 & 0.021 \\ \mbox{(categorized)} (CRT + only) & (CRT dose 2500cGy & 1.50 & 1.10 - 0.40 & 0.21 \\ \mbo$	CK1 added (whole conort)	Genotype dosage	1.89	1.14 - 3.14	0.014
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Internetic merick CDT all 1 (1111)	Critic yes	1.41	1.1/ - 1./0	<0.001
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	Interaction with CR1 added (whole cohort)	Genotype dosage	0.59	0.26 - 1.36	0.22
$ \begin{array}{c c} Genotype (XI & 14.01 & 4.99 - 40.59 & e0.001 \\ Genotype dosage & 1.87 & 1.13 - 3.11 & 0.015 \\ CRT dose added (continuous) (whole cohort) & Genotype dosage & 0.76 & 0.37 - 1.57 & 0.46 \\ (whole cohort) & CRT dose added (continuous) & Genotype dosage & 0.76 & 0.37 - 1.57 & 0.46 \\ (whole cohort) & CRT dose (cont.), per Gy & 1.01 & 1.003 - 1.013 & 0.002 \\ Genotype*CRT dose & 1.06 & 1.03 - 1.013 & 0.002 \\ Genotype*CRT dose & 1.06 & 1.03 - 1.013 & 0.002 \\ CRT dose added (categorized) (whole cohort) & Genotype dosage & 1.82 & 1.09 - 3.03 & 0.021 \\ CRT dose -2500CGY & 2.00 & 1.63 - 2.46 & 4.0001 \\ CRT dose -2500CGY & 2.00 & 1.63 - 2.46 & 4.0001 \\ CRT dose -2500CGY & 2.64 & 1.98 - 3.53 & <0.001 \\ CRT dose -2500CGY & 2.64 & 1.98 - 3.53 & <0.001 \\ CRT dose -2500CGY & 2.64 & 1.78 - 3.23 & <0.001 \\ CRT dose -2500CGY & 2.40 & 1.78 - 3.23 & <0.001 \\ CRT dose -2500CGY & 2.40 & 1.78 - 3.23 & <0.001 \\ CRT dose -2500CGY & 2.40 & 1.78 - 3.23 & <0.001 \\ CRT dose -2500CGY & 2.40 & 1.78 - 3.23 & <0.001 \\ CRT dose -2500CGY & 2.40 & 1.78 - 3.23 & <0.001 \\ CRT dose -2500CGY & 2.40 & 1.78 - 3.23 & <0.001 \\ CRT dose -2500CGY & 2.40 & 1.78 - 3.23 & <0.001 \\ CRT dose -2500CGY & 2.40 & 1.78 - 3.23 & <0.001 \\ CRT dose -2500CGY & 2.40 & 1.78 - 3.23 & <0.001 \\ CRT dose -2500CGY & 2.40 & 1.78 - 3.23 & <0.001 \\ Genotype*CRT dose & 1.07 & 0.25 - 4.61 & 0.93 \\ 0-2500CGY & 2.40 & 1.78 - 3.21 & 0.004 \\ dose -2500CGY & 2.40 & 1.00 & 0.99 - 1.02 & 0.48 \\ Region 1 & 1.01 & 1.00 - 1.02 & 0.23 \\ Region 1 & 1.01 & 1.00 - 1.03 & 0.043 \\ Region 1 & 1.01 & 1.00 - 1.03 & 0.043 \\ Region 2 & 1.00 & 0.99 - 1.01 & 0.40 \\ Brain radiotherapy region 2 dose added (CRT+ only) & Genotype dosage & 7.51 & 3.39 - 16.67 & 7.05*10^7 \\ BMI (cort.) & 1.01 & 0.99 - 1.04 & 0.21 \\ Overweight added (CRT+ only) & Genotype dosage & 7.66 & 3.45 - 16.38 & 5.41*10^7 \\ Overweight added (CRT+ only) & Genotype dosage & 7.64 & 3.43 - 1.630 & 9.68*10^7 \\ Overweight added (CRT+ only) & Genotype dosage & 7.64 & 3.45 - 16.38 & 5.41*10^7 \\ Overweight added (CRT+ only) & Genotype dosage$		CRI yes	1.31	1.08 - 1.58	0.005
$ \begin{array}{c} {\rm CRT} \ {\rm dose \ added \ (continuous) \ (whole \ cohorr)} & {\rm Genotype \ dosage} & 1.87 & 1.13 - 3.11 & 0.015 \\ {\rm CRT} \ {\rm dose \ (cont.), \ per \ Gy} & 1.01 & 1.005 - 1.015 & <0.001 \\ {\rm Interaction \ with \ CRT \ dose \ added \ (continuous) \ Genotype \ dosage & 0.76 & 0.37 - 1.57 & 0.46 \\ {\rm (whole \ cohort)} & {\rm CRT \ dose \ (cont.), \ per \ Gy} & 1.01 & 1.003 - 1.013 & 0.002 \\ {\rm Genotype \ dosage \ 1.82 & 1.09 - 3.03 & 0.021 \\ {\rm (cont.)} & {\rm CRT \ dose \ 0.2500 \ Gy} & 2.00 & 1.63 - 2.46 & <0.001 \\ {\rm (CRT \ dose \ 0.2500 \ Gy} & 2.00 & 1.63 - 2.46 & <0.001 \\ {\rm (CRT \ dose \ 0.2500 \ Gy} & 2.00 & 1.63 - 2.46 & <0.001 \\ {\rm (CRT \ dose \ 0.2500 \ Gy} & 2.00 & 1.63 - 2.47 & <0.001 \\ {\rm (CRT \ dose \ 0.2500 \ Gy} & 2.00 & 1.63 - 2.47 & <0.001 \\ {\rm (CRT \ dose \ 0.2500 \ Gy} & 2.00 & 1.63 - 2.47 & <0.001 \\ {\rm (CRT \ dose \ 0.2500 \ Gy} & 2.00 & 1.63 - 2.47 & <0.001 \\ {\rm (CRT \ dose \ 0.2500 \ Gy} & 2.00 & 1.63 - 2.47 & <0.001 \\ {\rm (CRT \ dose \ 0.2500 \ Gy} & 2.00 & 1.63 - 2.47 & <0.001 \\ {\rm (CRT \ dose \ 0.2500 \ Gy} & 2.00 & 1.63 - 2.47 & <0.001 \\ {\rm (CRT \ dose \ 0.2500 \ Gy} & 2.00 & 1.63 - 2.47 & <0.001 \\ {\rm (CRT \ dose \ 0.2500 \ Gy} & 2.00 & 1.63 - 2.47 & <0.001 \\ {\rm (CRT \ dose \ 0.2500 \ Gy} & 2.00 & 1.63 - 2.47 & <0.001 \\ {\rm (CRT \ dose \ 0.2500 \ Gy} & 2.00 & 1.63 - 2.47 & <0.001 \\ {\rm (CRT \ dose \ 0.2500 \ Gy} & 2.00 & 1.63 - 2.47 & <0.001 \\ {\rm (CRT \ dose \ 0.2500 \ Gy} & 2.00 & 1.63 - 2.47 & <0.001 \\ {\rm (CRT \ dose \ 0.2500 \ Gy} & 2.00 & 1.63 - 2.47 & <0.001 \\ {\rm (CRT \ dose \ 0.2500 \ Gy} & 2.00 & 1.63 - 2.47 & <0.001 \\ {\rm (CRT \ dose \ 0.2500 \ Gy} & 2.00 & 1.63 - 2.47 & <0.001 \\ {\rm (CRT \ dose \ 0.2500 \ Gy} & 2.00 & 1.63 - 2.47 & <0.001 \\ {\rm (CRT \ dose \ 0.2500 \ Gy} & 2.00 & 1.63 - 2.47 & <0.001 \\ {\rm (CRT \ dose \ 0.2500 \ Gy} & 2.00 & 0.99 - 1.02 & 0.48 \\ {\rm (ace) \ 0.2500 \ Gy} & 2.00 & 0.99 - 1.02 & 0.48 \\ {\rm (ace) \ 0.21 \ Genotype \ dosage \ 7.97 & 3.56 - 17.84 & 4.57^*10^7 \\ {\rm (ace) \ 0.21 \ Genotype \ dosage \ 7.51 & 3.39 - 1.06 & 0.40 \\ {\rm (cate) \ 0.21 \ Genotyp$		Genotype*CRT	14.61	4.59 - 46.55	<0.001
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	CRT dose added (continuous) (whole cohort)	Genotype dosage	1.87	1.13 – 3.11	0.015
$ \begin{array}{llllllllllllllllllllllllllllllllllll$		CRT dose (cont.), per Gy	1.01	1.005 – 1.015	<0.001
CRT dose (cont.), per Gy 1.01 1.003 – 1.013 0.002 Genotype*CRT dose (cont.) 1.06 1.03 – 1.09 <001	Interaction with CRT dose added (continuous) (whole cohort)	Genotype dosage	0.76	0.37 – 1.57	0.46
Genotype*CRT dose (cont.) 1.03 1.03 1.03 40.01 CRT dose added (categorized) (whole cohor) Genotype dosage 1.82 1.09 3.021 Interaction with CRT dose added (categorized) (whole cohort) Genotype dosage 1.01 0.28 3.63 4.001 Interaction with CRT dose added (categorized) Genotype dosage 1.01 0.28 3.63 4.001 Interaction with CRT dose added (categorized) CRT dose >2500cGy 2.00 1.63 2.44 4.001 CRT dose >2500cGy 2.00 1.63 2.44 4.001 4.001 Genotype*CRT dose 2.00 1.63 2.44 4.001 Orgory dosage 1.01 1.00 1.03 4.001 Genotype dosage 7.97 3.56 1.01 4.021 Magion 1 1.01 1.00 1.03 4.021 Region 1 1.01 1.00 1.03 4.021 Region 2 1.00 1.00 1.03 4.021 Region 1 1.01 1.00 1.0		CRT dose (cont.), per Gy	1.01	1.003 - 1.013	0.002
CRT dose added (categorized) (whole cohort) Genotype dosage CRT dose 0-2500cGy 1.09 - 3.03 0.021 Interaction with CRT dose added (categorized) Genotype dosage 1.01 $0.28 - 3.64$ 0.98 (whole cohort) CRT dose 0-2500cGy 2.00 $1.63 - 2.47$ <0.001		Genotype*CRT dose	1.06	1.03 – 1.09	<0.001
CKT use added (categorized)Genotype dosage1.021.030.001CRT dose 0-2500CGy2.641.983.53<0.001	CPT does added (categorized) (whole cohort)	Concturne desege	1.82	1.00 3.03	0.021
$ \begin{array}{c} \mbox{CRT dose 0-2500 Gy} & 2.00 & 1.03 - 2.40 & 0.001 \\ \mbox{CRT dose >2500 Gy} & 2.64 & 1.98 - 3.53 & <0.001 \\ \mbox{(whole cohort)} & & & & & & & & & & & & & & & & & & &$	CIVI dose added (categorized) (whole conort)	CPT daga 0.2500 aCrr	2.00	1.09 - 5.03	.0.001
$ \begin{array}{c} \mbox{CRT dose 32,500 Cdy} & 2.04 & 1.98 - 5.53 & 50001 \\ \mbox{Interaction with CRT dose added (categorized)} & \mbox{Genotype dosage} & 1.01 & 0.28 - 3.64 & 0.98 \\ \mbox{(whole cohort)} & \mbox{CRT dose 0-2500 cGy} & 2.00 & 1.63 - 2.47 & <0.001 \\ \mbox{CRT dose >2500 cGy} & 2.40 & 1.78 - 3.23 & <0.001 \\ \mbox{Genotype*CRT dose} & 1.07 & 0.25 - 4.61 & 0.93 \\ \mbox{0-2500 cGy} & & & & & & & & & & & & & & & & & & &$		CRT dose 0-2300cGy	2.00	1.09 - 2.40	<0.001
$ \begin{array}{c} \mbox{Interaction with CK1 dose added (categorized)} & \mbox{Centry e dosage} & 1.01 & 0.26 - 3.04 & 0.98 \\ \mbox{(whole cohort)} & \mbox{CRT dose 0-2500cGy} & 2.00 & 1.63 - 2.47 & <0.001 \\ \mbox{CRT dose 2500cGy} & 2.40 & 1.78 - 3.23 & <0.001 \\ \mbox{Genotype*CRT dose} & 1.07 & 0.25 - 4.61 & 0.93 \\ \mbox{0-2500cGy} & \mbox{0-200} & 0-$	Internetion with CPT data added (according)	CRI dose >2,000CGy	2.04	1.98 - 3.93	<0.001
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	(whole cohort)	Genotype dosage	1.01	0.28 - 5.64	0.98
CRT dose >2500cGy2.40 $1.78 - 3.23$ <0.001Genotype*CRT dose 0-2500cGy 1.07 $0.25 - 4.61$ 0.93 Brain radiotherapy region added (CRT+ only)Genotype*CRT dose>2500cGy 1.51 $2.24 - 69.86$ 0.004 Brain radiotherapy region added (CRT+ only)Genotype dosage 7.97 $3.56 - 17.84$ $4.57*10^7$ Region 1 1.01 $1.00 - 1.02$ 0.23 Region 2 1.00 $0.99 - 1.02$ 0.48 Region 2 1.00 $0.99 - 1.02$ 0.48 Region 3 1.01 $1.00 - 1.03$ 0.043 Brain radiotherapy region 2 dose added (categorized) (CRT+ only)Genotype dosage 7.21 $3.25 - 15.96$ $1.13*10^6$ BMI added (continuous) (CRT+ only)Genotype dosage 7.51 $3.39 - 16.67$ $7.05*10^7$ BMI (cont.) 1.01 $0.99 - 1.04$ 0.21 Overweight added (CRT+ only)Genotype dosage 7.34 $3.31 - 16.30$ $9.68*10^7$ Obesity added (CRT+ only)Genotype dosage 7.66 $3.45 - 16.98$ $5.41*10^7$ Outcome BMI (cont.) (CRT+ only)*Genotype dosage 7.66 $3.45 - 16.98$ $5.41*10^7$ Outcome BMI (cont.) (CRT+ only)*Genotype dosage 7.66 $3.45 - 16.98$ $5.41*10^7$ Outcome overweight (CRT+ only)Genotype dosage 7.66 $3.45 - 16.98$ $5.41*10^7$ Outcome overweight (CRT+ only)Genotype dosage 7.66 $3.45 - 16.98$ $5.41*10^7$ Outcome overweight (CRT+ only)Genotype dosage 7.66 $3.45 - 16.98$		CRT dose 0-2500cGy	2.00	1.63 - 2.47	<0.001
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		CRT dose >2500cGy	2.40	1.78 – 3.23	<0.001
Genotype*CRT dose>2500cGy 12.51 2.24 - 69.86 0.004 Brain radiotherapy region added (CRT+ only) Genotype dosage 7.97 3.56 - 17.84 4.57*10 ⁻⁷ Region 1 1.01 1.00 - 1.02 0.23 Region 2 1.00 0.99 - 1.02 0.48 Region 3 1.01 1.00 - 1.03 0.043 Brain radiotherapy region 2 dose added (categorized) (CRT+ only) Genotype dosage 7.21 3.25 - 15.96 1.13*10 ⁶ BMI added (continuous) (CRT+ only) Region 2 dose >2500cGy 1.50 1.10 - 2.04 0.0096 BMI added (continuous) (CRT+ only) Genotype dosage 7.51 3.39 - 16.67 7.05*10 ⁻⁷ BMI (cont.) 1.01 0.99 - 1.04 0.21 Overweight added (CRT+ only) Genotype dosage 7.34 3.31 - 16.30 9.68*10 ⁻⁷ Overweight added (CRT+ only) Genotype dosage 7.66 3.45 - 16.98 5.41*10 ⁻⁷ Obesity added (CRT+ only)* Genotype dosage 7.66 3.45 - 16.98 5.41*10 ⁻⁷ Outcome BMI (cont.) (CRT+ only)* Genotype dosage 1.61		Genotype*CRT dose 0-2500cGy	1.07	0.25 - 4.61	0.93
Brain radiotherapy region added (CRT+ only)Genotype dosage 7.97 $3.56 - 17.84$ $4.57*10^{-7}$ Region 1 1.01 $1.00 - 1.02$ 0.23 Region 2 1.00 $0.99 - 1.02$ 0.48 Region 3 1.01 $1.00 - 1.03$ 0.043 Region 4 0.99 $0.98 - 1.01$ 0.40 Brain radiotherapy region 2 dose added (categorized) (CRT+ only)Genotype dosage 7.21 $3.25 - 15.96$ $1.13*10^{-6}$ BMI added (continuous) (CRT+ only)Region 2 dose >2500cGy 1.50 $1.10 - 2.04$ 0.0096 BMI added (continuous) (CRT+ only)Genotype dosage 7.51 $3.39 - 16.67$ $7.05*10^{-7}$ BMI (cont.) 1.01 $0.99 - 1.04$ 0.21 Overweight added (CRT+ only)Genotype dosage 7.34 $3.31 - 16.30$ $9.68*10^{-7}$ Obesity added (CRT+ only)Genotype dosage 7.66 $3.45 - 16.98$ $5.41*10^{-7}$ Obesity added (CRT+ only)Genotype dosage 7.66 $3.45 - 16.98$ $5.41*10^{-7}$ Outcome BMI (cont.) (CRT+ only)*Genotype dosage 7.66 $3.45 - 16.98$ $5.41*10^{-7}$ Outcome overweight (CRT+ only)*Genotype dosage 7.66 $3.45 - 16.98$ $5.41*10^{-7}$ Outcome overweight (CRT+ only)*Genotype dosage 1.93 $0.73 - 5.14$ 0.16 Outcome overweight (CRT+ only)Genotype dosage 1.93 $0.73 - 5.14$ 0.16 Outcome overweight (CRT+ only)Genotype dosage 1.93 $0.73 - 5.14$ 0.16		Genotype*CRT dose>2500cGy	12.51	2.24 - 69.86	0.004
Region 11.01 $1.00 - 1.02$ 0.23 Region 2 1.00 $0.99 - 1.02$ 0.48 Region 3 1.01 $1.00 - 1.03$ 0.043 Region 4 0.99 $0.98 - 1.01$ 0.40 Brain radiotherapy region 2 dose added (categorized) (CRT+ only)Genotype dosage 7.21 $3.25 - 15.96$ $1.13*10^6$ BMI added (continuous) (CRT+ only)Region 2 dose >2500cGy 1.50 $1.10 - 2.04$ 0.0096 BMI added (continuous) (CRT+ only)Genotype dosage 7.51 $3.39 - 16.67$ $7.05*10^7$ Overweight added (CRT+ only)Genotype dosage 7.34 $3.31 - 16.30$ $9.68*10^7$ Overweight added (CRT+ only)Genotype dosage 7.66 $3.45 - 16.98$ $5.41*10^7$ Obesity added (CRT+ only)Genotype dosage 7.66 $3.45 - 16.98$ $5.41*10^7$ Obesity added (CRT+ only)*Genotype dosage 7.66 $3.45 - 16.98$ $5.41*10^7$ Outcome BMI (cont.) (CRT+ only)*Genotype dosage 7.66 $3.45 - 16.98$ $5.41*10^7$ Outcome overweight (CRT+ only)Genotype dosage 7.66 $3.45 - 16.98$ $5.41*10^7$ Outcome overweight (CRT+ only)*Genotype dosage 1.93 $0.73 - 5.14$ 0.16 Outcome overweight (CRT+ only)Genotype dosage 1.93 $0.73 - 5.14$ 0.19 Outcome overweight (CRT+ only)Genotype dosage 1.93 $0.73 - 5.14$ 0.19	Brain radiotherapy region added (CRT+ only)	Genotype dosage	7.97	3.56 - 17.84	4.57*10 ⁻⁷
$Region 2$ 1.00 $0.99 - 1.02$ 0.48 Region 3 1.01 $1.00 - 1.03$ 0.043 Region 4 0.99 $0.98 - 1.01$ 0.40 Brain radiotherapy region 2 dose added (categorized) (CRT+ only)Genotype dosage 7.21 $3.25 - 15.96$ $1.13*10^6$ BMI added (continuous) (CRT+ only)Region 2 dose >2500cGy 1.50 $1.10 - 2.04$ 0.0096 BMI added (continuous) (CRT+ only)Genotype dosage 7.51 $3.39 - 16.67$ $7.05*10^7$ BMI (cont.) 1.01 $0.99 - 1.04$ 0.21 Overweight added (CRT+ only)Genotype dosage 7.34 $3.31 - 16.30$ $9.68*10^7$ Obesity added (CRT+ only)Genotype dosage 7.66 $3.45 - 16.98$ $5.41*10^7$ Obesity added (CRT+ only)Genotype dosage 7.66 $3.45 - 16.98$ $5.41*10^7$ Outcome BMI (cont.) (CRT+ only)*Genotype dosage $8eta=1.84$ $-0.74 - 4.43$ 0.16 Outcome overweight (CRT+ only)Genotype dosage 1.93 $0.73 - 5.14$ 0.19 Outcome shorizm (CRT+ only)Genotype dosage 1.93 $0.73 - 5.14$ 0.19		Region 1	1.01	1.00 - 1.02	0.23
C_{r} C_{r} 1.01 $1.00 - 1.03$ 0.043 Region 3Region 4 0.99 $0.98 - 1.01$ 0.40 Brain radiotherapy region 2 dose added (categorized) (CRT+ only)Genotype dosage 7.21 $3.25 - 15.96$ $1.13*10^6$ BMI added (continuous) (CRT+ only)Region 2 dose >2500cGy 1.50 $1.10 - 2.04$ 0.0096 BMI added (continuous) (CRT+ only)Genotype dosage 7.51 $3.39 - 16.67$ $7.05*10^7$ BMI (cont.) 1.01 $0.99 - 1.04$ 0.21 Overweight added (CRT+ only)Genotype dosage 7.34 $3.31 - 16.30$ $9.68*10^7$ Obesity added (CRT+ only)Genotype dosage 7.66 $3.45 - 16.98$ $5.41*10^7$ Obesity added (CRT+ only)Genotype dosage 7.66 $3.45 - 16.98$ $5.41*10^7$ Outcome BMI (cont.) (CRT+ only)*Genotype dosage $8eta=1.84$ $-0.74 - 4.43$ 0.16 Outcome overweight (CRT+ only)Genotype dosage 1.93 $0.73 - 5.14$ 0.19 Outcome overweight (CRT+ only)Genotype dosage 1.93 $0.73 - 5.14$ 0.19		Region 2	1.00	0.99 – 1.02	0.48
$r_{Region 4}$ 0.99 $0.98 - 1.01$ 0.40 Brain radiotherapy region 2 dose added (categorized) (CRT+ only)Genotype dosage 7.21 $3.25 - 15.96$ $1.13*10^{-6}$ BMI added (continuous) (CRT+ only)Region 2 dose >2500cGy 1.50 $1.10 - 2.04$ 0.0096 BMI added (continuous) (CRT+ only)Genotype dosage 7.51 $3.39 - 16.67$ $7.05*10^{-7}$ BMI (cont.) 1.01 $0.99 - 1.04$ 0.21 Overweight added (CRT+ only)Genotype dosage 7.34 $3.31 - 16.30$ $9.68*10^{-7}$ Obesity added (CRT+ only)Genotype dosage 7.66 $3.45 - 16.98$ $5.41*10^{-7}$ Obesity added (CRT+ only)Genotype dosage 7.66 $3.45 - 16.98$ $5.41*10^{-7}$ Obesity 1.41 $1.03 - 1.92$ 0.027 Outcome BMI (cont.) (CRT+ only)*Genotype dosage 1.93 $0.73 - 5.14$ 0.19 Outcome overweight (CRT+ only)Genotype dosage 1.93 $0.73 - 5.14$ 0.19		Region 3	1.01	1.00 - 1.03	0.043
Brain radiotherapy region 2 dose added (categorized) (CRT+ only) Genotype dosage 7.21 $3.25 - 15.96$ $1.13*10^{-6}$ Region 2 dose >2500cGy 1.50 $1.10 - 2.04$ 0.0096 BMI added (continuous) (CRT+ only) Genotype dosage 7.51 $3.39 - 16.67$ $7.05*10^{-7}$ BMI (cont.) 1.01 $0.99 - 1.04$ 0.21 Overweight added (CRT+ only) Genotype dosage 7.34 $3.31 - 16.30$ $9.68*10^{-7}$ Overweight 1.68 $1.17 - 2.42$ 0.005 Obesity added (CRT+ only) Genotype dosage 7.66 $3.45 - 16.98$ $5.41*10^{-7}$ Obesity added (CRT+ only) Genotype dosage 7.66 $3.45 - 16.98$ $5.41*10^{-7}$ Obesity 1.41 $1.03 - 1.92$ 0.027 Outcome BMI (cont.) (CRT+ only)* Genotype dosage $Beta=1.84$ $-0.74 - 4.43$ 0.16 Outcome overweight (CRT+ only) Genotype dosage 1.93 $0.73 - 5.14$ 0.19 Outcome overweight (CRT+ only) Genotype dosage 1.93 $0.73 - 5.14$ 0.16		Region 4	0.99	0.98 - 1.01	0.40
Region 2 dose >2500cGy 1.50 $1.10 - 2.04$ 0.0096 BMI added (continuous) (CRT+ only) Genotype dosage 7.51 $3.39 - 16.67$ $7.05*10^{-7}$ BMI (cont.) 1.01 $0.99 - 1.04$ 0.21 Overweight added (CRT+ only) Genotype dosage 7.34 $3.31 - 16.30$ $9.68*10^{-7}$ Obesity added (CRT+ only) Genotype dosage 7.66 $3.45 - 16.98$ $5.41*10^{-7}$ Obesity added (CRT+ only) Genotype dosage 7.66 $3.45 - 16.98$ $5.41*10^{-7}$ Obesity 1.41 $1.03 - 1.92$ 0.027 Outcome BMI (cont.) (CRT+ only)* Genotype dosage $8eta=1.84$ $-0.74 - 4.43$ 0.16 Outcome overweight (CRT+ only) Genotype dosage 1.93 $0.73 - 5.14$ 0.19 Outcome overweight (CRT+ only) Genotype dosage 1.93 $0.73 - 5.14$ 0.19	Brain radiotherapy region 2 dose added (categorized) (CRT+ only)	Genotype dosage	7.21	3.25 - 15.96	1.13*10 ⁻⁶
BMI added (continuous) (CRT+ only) Genotype dosage 7.51 $3.39 - 16.67$ $7.05*10^7$ BMI (cont.) 1.01 $0.99 - 1.04$ 0.21 Overweight added (CRT+ only) Genotype dosage 7.34 $3.31 - 16.30$ $9.68*10^7$ Obesity added (CRT+ only) Genotype dosage 7.66 $3.45 - 16.98$ $5.41*10^7$ Obesity added (CRT+ only) Genotype dosage 7.66 $3.45 - 16.98$ $5.41*10^7$ Obesity 1.41 $1.03 - 1.92$ 0.027 Outcome BMI (cont.) (CRT+ only)* Genotype dosage Beta=1.84 $-0.74 - 4.43$ 0.16 Outcome overweight (CRT+ only) Genotype dosage 1.93 $0.73 - 5.14$ 0.19 Outcome overweight (CRT+ only) Genotype dosage 1.93 $0.73 - 5.14$ 0.19	(Region 2 dose >2500cGv	1.50	1.10 - 2.04	0.0096
Driving added (continuous) (crit + only) Genotype dosage 7.31 5.55 7.65	BMI added (continuous) (CRT+ only)	Genotype dosage	7 51	3 39 - 16 67	7.05*10 ⁻⁷
Overweight added (CRT+ only) Genotype dosage 7.34 $3.31 - 16.30$ $9.68*10^7$ Overweight 1.68 $1.17 - 2.42$ 0.005 Obesity added (CRT+ only) Genotype dosage 7.66 $3.45 - 16.98$ $5.41*10^7$ Obesity 1.41 $1.03 - 1.92$ 0.027 Outcome BMI (cont.) (CRT+ only)* Genotype dosage Beta= 1.84 $-0.74 - 4.43$ 0.16 Outcome overweight (CRT+ only) Genotype dosage 1.93 $0.73 - 5.14$ 0.19 Outcome shoring (CRT+ only) Genotype dosage 1.93 $0.73 - 5.14$ 0.19	Divir added (continuous) (citer + omy)	BMI (cont.)	1.01	0.99 - 1.04	0.21
Overweight 1.68 $1.17 - 2.42$ 0.005 Obesity added (CRT+ only) Genotype dosage 7.66 $3.45 - 16.98$ $5.41*10^7$ Obesity 1.41 $1.03 - 1.92$ 0.027 Outcome BMI (cont.) (CRT+ only)* Genotype dosage Beta= 1.84 $-0.74 - 4.43$ 0.16 Outcome overweight (CRT+ only) Genotype dosage 1.93 $0.73 - 5.14$ 0.19 Outcome overweight (CRT+ only) Genotype dosage 1.93 $0.73 - 5.14$ 0.19	Overweight added (CRT+ only)	Genotype dosage	7 34	3.31 - 16.30	9.68*10 ⁻⁷
Obesity added (CRT+ only) Genotype dosage 7.66 $3.45 - 16.98$ $5.41*10^7$ Obesity 1.41 $1.03 - 1.92$ 0.027 Outcome BMI (cont.) (CRT+ only)* Genotype dosage Beta= 1.84 $-0.74 - 4.43$ 0.16 Outcome overweight (CRT+ only) Genotype dosage 1.93 $0.73 - 5.14$ 0.19 Outcome overweight (CRT+ only) Genotype dosage 1.93 $0.73 - 5.14$ 0.19	Overweight added (Orer + Only)	Overweight	1.68	1.17 - 2.42	0.005
Observe and (CRT+ only)* Genotype dosage 7.00 5.42 = 10.58 5.41 10 Observe and 1.41 $1.03 - 1.92$ 0.027 Outcome BMI (cont.) (CRT+ only)* Genotype dosage Beta=1.84 $-0.74 - 4.43$ 0.16 Outcome overweight (CRT+ only) Genotype dosage 1.93 $0.73 - 5.14$ 0.19 Outcome overweight (CRT+ only) Genotype dosage 1.93 $0.73 - 5.14$ 0.19	Obesity added (CRT+ only)	Genotype dosage	7.66	3.45 - 16.98	5 41*10 ⁻⁷
Outcome BMI (cont.) (CRT+ only)*Genotype dosageBeta= 1.84 $-0.74 - 4.43$ 0.16 Outcome overweight (CRT+ only)Genotype dosage 1.93 $0.73 - 5.14$ 0.19 Outcome overweight (CRT+ only)Genotype dosage 1.93 $0.73 - 5.14$ 0.19	Obesity added (CIXI + Offiy)	Obesity	1 /1	1.03 1.02	0.027
Outcome overweight (CRT+ only)Genotype dosage 1.93 $0.73 - 5.14$ 0.19 Outcome overweight (CRT+ only)Genotype dosage 1.93 $0.73 - 5.14$ 0.19 Outcome observe (CRT+ only)Genotype dosage 1.93 $0.73 - 5.14$ 0.19	Outcome BMI (cont.) (CPT. only)*	Construne desease	1.41 Bota-1.04	1.03 - 1.92	0.02/
Outcome overweight (CRT+ only) Genotype dosage $1.95 0.75 - 5.14 0.19$	Outcome Divit (cont.) (CRT+ only)	Conotype dosage	1.02	-0./4 - 4.43	0.10
	Outcome obesity (CRT+ only)	Genotype dosage	1.99	0.75 - 5.14 0.56 2.52	0.19

Supplemental table 5. Additional regression analyses

Additional regression models were built to study the interaction between genotype and CRT and the influence of BMI.

* Linear regression model, so beta reported instead of odds ratio.

All models are also adjusted for age, sex and the first four genetic principal components

GWAS	MAF	Beta	SE	OR	Lower 95%CI	Upper 95% CI	P-value
CRT+	0.012	2.82	0.49	16.7768507	6.42	43.83	1.18E-08
Overall	0.010	0.69	0.29	1.99371553	1.13	3.52	0.018
CRT -	0.0099*	-0.59	0.42	0.55432728	0.24	1.26	0.17
Abd RT +	0.012	0.54	0.50	1.71600686	0.64	4.57	0.27
Abd RT -	0.0099*	0.63	0.31	1.87761058	1.02	3.45	0.043
Asparaginase +	0.010	0.68	0.71	1.97387773	0.49	7.94	0.34
Asparaginase -	0.010	0.71	0.32	2.03399126	1.09	3.81	0.027
Prednisone +	0.012	0.83	0.45	2.29331874	0.95	5.54	0.066
Prednisone -	0.0097*	0.51	0.34	1.66529119	0.86	3.24	0.13

Supplemental table 6. Significant locus in other GWASs

*Excluded by Rvtest due to MAF<0.01, analysis was rerun in R (Wald test instead of score test)



Supplemental figure 1. Flow charts of the CCSS Original, SJLIFE and CCSS Expansion cohorts, indicating included and excluded subjects.

Exclusion because of absence of genetic data had various reasons, including non-consent, unusable samples (old or not enough DNA), consented but did not supply a sample.



Supplemental figure 2. Four brain regions for which dosimetry has been calculated in the CCSS cohort and included in post-GWAS analysis.



Supplemental figure 3. QQ plots of the initial discovery analysis and the CRT+ analysis.

QQ plots of the overall analysis (left) and CRT+ stratified analysis (right). Separate curves are shown for four categories of minor allele frequency (MAF).



Supplemental figure 4. LocusZoom of significant variant rs114017774 on chromosome 2.

REFERENCES

- 1. Gatta G, Botta L, Rossi S, et al. Childhood cancer survival in Europe 1999–2007: results of EUROCARE-5—a population-based study. The Lancet Oncology. 2014;15(1):35-47.
- 2. Armstrong GT, Liu Q, Yasui Y, et al. Late mortality among 5-year survivors of childhood cancer: a summary from the Childhood Cancer Survivor Study. J Clin Oncol. 2009;27(14):2328-38.
- 3. Mach F, Baigent C, Catapano AL, et al. 2019 ESC/EAS Guidelines for the management of dyslipidaemias: lipid modification to reduce cardiovascular risk. Eur Heart J. 2019.
- 4. Alberti KG, Eckel RH, Grundy SM, et al. Harmonizing the metabolic syndrome: a joint interim statement of the International Diabetes Federation Task Force on Epidemiology and Prevention; National Heart, Lung, and Blood Institute; American Heart Association; World Heart Federation; International Atherosclerosis Society; and International Association for the Study of Obesity. Circulation. 2009;120(16):1640-5.
- Chao C, Xu L, Bell E, Cooper R, Mueller L. Long-term Health Outcomes in Survivors of Childhood Cancer Diagnosed Between 1990 and 2000 in a Large US Integrated Health Care System. J Pediatr Hematol Oncol. 2016;38:123-30.
- 6. Felicetti F, D'Ascenzo F, Moretti C, et al. Prevalence of cardiovascular risk factors in long-term survivors of childhood cancer: 16 years follow up from a prospective registry. Eur J Prev Cardiol. 2015;22(6):762-70.
- 7. Armstrong GT, Oeffinger KC, Chen Y, et al. Modifiable risk factors and major cardiac events among adult survivors of childhood cancer. J Clin Oncol. 2013;31(29):3673-80.
- Kero AE, Madanat-Harjuoja LM, Jarvela LS, Malila N, Matomaki J, Lahteenmaki PM. Cardiovascular medication after cancer at a young age in Finland: A nationwide registry linkage study. Int J Cancer. 2016;139(3):683-90.
- 9. Saultier P, Auquier P, Bertrand Y, et al. Metabolic syndrome in long-term survivors of childhood acute leukemia treated without hematopoietic stem cell transplantation: an L.E.A. study. Haematologica. 2016;101(12):1603-10.
- Nottage KA, Ness KK, Li C, Srivastava D, Robison LL, Hudson MM. Metabolic syndrome and cardiovascular risk among long-term survivors of acute lymphoblastic leukaemia - From the St. Jude Lifetime Cohort. Br J Haematol. 2014;165(3):364-74.
- 11. van Waas M, Neggers SJ, Raat H, van Rij CM, Pieters R, van den Heuvel-Eibrink MM. Abdominal radiotherapy: a major determinant of metabolic syndrome in nephroblastoma and neuroblastoma survivors. PLoS One. 2012;7(12):e52237.
- Mostoufi-Moab S, Seidel K, Leisenring WM, et al. Endocrine Abnormalities in Aging Survivors of Childhood Cancer: A Report From the Childhood Cancer Survivor Study. J Clin Oncol. 2016;34(27):3240-7.
- Wilson CL, Liu W, Yang JJ, et al. Genetic and clinical factors associated with obesity among adult survivors of childhood cancer: A report from the St. Jude Lifetime Cohort. Cancer. 2015;121(13):2262-70.
- 14. Chemaitilly W, Cohen LE. Diagnosis of Endocrine Disease: Endocrine late-effects of childhood cancer and its treatments. Eur J Endocrinol. 2017;176(4):R183-R203.
- 15. van Dorp W, Blijdorp K, Laven JS, et al. Decreased ovarian function is associated with obesity in very long-term female survivors of childhood cancer. Eur J Endocrinol. 2013;168(6):905-12.
- 16. Pluimakers VG, van Waas M, Neggers S, van den Heuvel-Eibrink MM. Metabolic syndrome as cardiovascular risk factor in childhood cancer survivors. Crit Rev Oncol Hematol. 2019;133:129-41.
- 17. Gramatges MM, Bhatia S. Evidence for Genetic Risk Contributing to Long-Term Adverse Treatment Effects in Childhood Cancer Survivors. Annu Rev Med. 2018;69:247-62.

- Clemens E, van der Kooi ALF, Broer L, et al. The influence of genetic variation on late toxicities in childhood cancer survivors: A review. Critical Reviews in Oncology/Hematology. 2018;126:154-67.
- 19. Klarin D, Damrauer SM, Cho K, et al. Genetics of blood lipids among ~300,000 multi-ethnic participants of the Million Veteran Program. Nat Genet. 2018;50(11):1514-23.
- 20. Hoffmann TJ, Theusch E, Haldar T, et al. A large electronic-health-record-based genome-wide study of serum lipids. Nat Genet. 2018;50(3):401-13.
- 21. Willer CJ, Schmidt EM, Sengupta S, et al. Discovery and refinement of loci associated with lipid levels. Nat Genet. 2013;45(11):1274-83.
- 22. Teslovich TM, Musunuru K, Smith AV, et al. Biological, clinical and population relevance of 95 loci for blood lipids. Nature. 2010;466(7307):707-13.
- 23. Liu DJ, Peloso GM, Yu H, et al. Exome-wide association study of plasma lipids in >300,000 individuals. Nat Genet. 2017;49(12):1758-66.
- 24. England J, Drouin S, Beaulieu P, et al. Genomic determinants of long-term cardiometabolic complications in childhood acute lymphoblastic leukemia survivors. BMC Cancer. 2017;17(1):751.
- 25. Robison LL, Mertens AC, Boice JD, et al. Study design and cohort characteristics of the Childhood Cancer Survivor Study: a multi-institutional collaborative project. Med Pediatr Oncol. 2002;38(4):229-39.
- 26. Genomes Project C, Auton A, Brooks LD, et al. A global reference for human genetic variation. Nature. 2015;526(7571):68-74.
- 27. Howell CR, Bjornard KL, Ness KK, et al. Cohort Profile: The St. Jude Lifetime Cohort Study (SJLIFE) for paediatric cancer survivors. Int J Epidemiol. 2021;50(1):39-49.
- Hudson MM, Ehrhardt MJ, Bhakta N, et al. Approach for Classification and Severity Grading of Long-term and Late-Onset Health Events among Childhood Cancer Survivors in the St. Jude Lifetime Cohort. Cancer Epidemiol Biomarkers Prev. 2017;26(5):666-74.
- 29. St.-Jude-Children's-Research-Hospital. CCSS Baseline and Follow-Up Questionnaires. Website: https://ccssstjudeorg/tools-and-documents/questionnaires/baseline-and-follow-upquestionnaireshtml. 2018.
- 30. Purcell S, Cherny S, Sham P. Genetic Power Calculator: design of linkage and association genetic mapping studies of complex traits. Bioinformatics. 2003;19(1):149-50.
- 31. Zhan X, Hu Y, Li B, Abecasis GR, Liu DJ. RVTESTS: an efficient and comprehensive tool for rare variant association analysis using sequence data. Bioinformatics. 2016;32(9):1423-6.
- 32. van Waas M, Neggers SJ, Uitterlinden AG, et al. Treatment factors rather than genetic variation determine metabolic syndrome in childhood cancer survivors. Eur J Cancer. 2013;49(3):668-75.
- 33. Brooke RJ, Im C, Wilson CL, et al. A High-risk Haplotype for Premature Menopause in Childhood Cancer Survivors Exposed to Gonadotoxic Therapy. J Natl Cancer Inst. 2018;110(8):895-904.
- 34. van der Kooi ALF, van Dijk M, Broer L, et al. Possible modification of BRSK1 on the risk of alkylating chemotherapy-related reduced ovarian function. Hum Reprod. 2021;36(4):1120-33.
- Im C, Qin N, Wang Z, et al. Generalizability of "GWAS Hits" in Clinical Populations: Lessons from Childhood Cancer Survivors. Am J Hum Genet. 2020;107(4):636-53.
- Lupo PJ, Brown AL, Arroyo VM, et al. DNA methylation and obesity in survivors of pediatric acute lymphoblastic leukemia: A report from the Childhood Cancer Survivor Study. Genes Chromosomes Cancer. 2019;58(1):52-9.
- 37. R Core Team (2020). R: A language and environment for statistical computing. R Foundation for Statistical Computing, Vienna, Austria. URL https://www.R-project.org/.

- Rangamaran VR, Uppili B, Gopal D, Ramalingam K. EasyQC: Tool with Interactive User Interface for Efficient Next-Generation Sequencing Data Quality Control. J Comput Biol. 2018;25(12):1301-11.
- 39. Winkler TW, Kutalik Z, Gorski M, Lottaz C, Kronenberg F, Heid IM. EasyStrata: evaluation and visualization of stratified genome-wide association meta-analysis data. Bioinformatics. 2015;31(2):259-61.
- 40. Lumley T. rmeta: Meta-analysis 2018 [Available from: https://cran.r-project.org/web/packages/ rmeta/index.html.
- 41. Kent WJ, Sugnet CW, Furey TS, et al. The human genome browser at UCSC. Genome Res. 2002;12(6):996-1006.
- 42. Pruim RJ, Welch RP, Sanna S, et al. LocusZoom: regional visualization of genome-wide association scan results. Bioinformatics. 2010;26(18):2336-7.
- 43. Watanabe K, Taskesen E, van Bochoven A, Posthuma D. Functional mapping and annotation of genetic associations with FUMA. Nat Commun. 2017;8(1):1826.
- 44. Roadmap Epigenomics C, Kundaje A, Meuleman W, et al. Integrative analysis of 111 reference human epigenomes. Nature. 2015;518(7539):317-30.
- 45. Buniello A, MacArthur JAL, Cerezo M, et al. The NHGRI-EBI GWAS Catalog of published genome-wide association studies, targeted arrays and summary statistics 2019. Nucleic Acids Res. 2019;47(D1):D1005-D12.
- 46. Carithers LJ, Ardlie K, Barcus M, et al. A Novel Approach to High-Quality Postmortem Tissue Procurement: The GTEx Project. Biopreserv Biobank. 2015;13(5):311-9.
- 47. Bult CJ, Blake JA, Smith CL, Kadin JA, Richardson JE, Mouse Genome Database G. Mouse Genome Database (MGD) 2019. Nucleic Acids Res. 2019;47(D1):D801-D6.
- 48. Dickinson ME, Flenniken AM, Ji X, et al. High-throughput discovery of novel developmental phenotypes. Nature. 2016;537(7621):508-14.
- 49. Sherlock M, Toogood AA. Sensitivity of anterior pituitary hormones to irradiation. Expert Rev Endocrinol Metab. 2006;1(5):633-49.
- 50. Crowne E, Gleeson H, Benghiat H, Sanghera P, Toogood A. Effect of cancer treatment on hypothalamic–pituitary function. The Lancet Diabetes & Endocrinology. 2015;3(7):568-76.
- 51. Machiela MJ, Chanock SJ. LDlink: a web-based application for exploring population-specific haplotype structure and linking correlated alleles of possible functional variants. Bioinformatics. 2015;31(21):3555-7.
- 52. Lauren J, Airaksinen MS, Saarma M, Timmusk T. A novel gene family encoding leucine-rich repeat transmembrane proteins differentially expressed in the nervous system. Genomics. 2003;81(4):411-21.
- 53. Kathiresan S, Manning AK, Demissie S, et al. A genome-wide association study for blood lipid phenotypes in the Framingham Heart Study. BMC Med Genet. 2007;8 Suppl 1:S17.
- 54. Siddiqui TJ, Tari PK, Connor SA, et al. An LRRTM4-HSPG complex mediates excitatory synapse development on dentate gyrus granule cells. Neuron. 2013;79(4):680-95.
- 55. Schaffer AE, Breuss MW, Caglayan AO, et al. Biallelic loss of human CTNNA2, encoding alphaN-catenin, leads to ARP2/3 complex overactivity and disordered cortical neuronal migration. Nat Genet. 2018;50(8):1093-101.
- 56. Cheng H, Sewda A, Marquez-Luna C, et al. Genetic architecture of cardiometabolic risks in people living with HIV. BMC Med. 2020;18(1):288.
- 57. Koyama S, Ito K, Terao C, et al. Population-specific and trans-ancestry genome-wide analyses identify distinct and shared genetic risk loci for coronary artery disease. Nat Genet. 2020;52(11):1169-77.

- 58. Pulit SL, Stoneman C, Morris AP, et al. Meta-analysis of genome-wide association studies for body fat distribution in 694 649 individuals of European ancestry. Hum Mol Genet. 2019;28(1):166-74.
- Hoffmann TJ, Choquet H, Yin J, et al. A Large Multiethnic Genome-Wide Association Study of Adult Body Mass Index Identifies Novel Loci. Genetics. 2018;210(2):499-515.
- 60. Cook SA, Bronson RT, Donahue LR, Ben-Arie N, Davisson MT. Cerebellar deficient folia (cdf): a new mutation on mouse chromosome 6. Mamm Genome. 1997;8(2):108-12.
- 61. International Mouse Phenotyping Consortium. REG3G knockout mouse phenotype [Available from: https://www.mousephenotype.org/data/genes/MGI:109406.
- Wang L, Fouts DE, Starkel P, et al. Intestinal REG3 Lectins Protect against Alcoholic Steatohepatitis by Reducing Mucosa-Associated Microbiota and Preventing Bacterial Translocation. Cell Host Microbe. 2016;19(2):227-39.
- 63. Unno M, Nata K, Noguchi N, et al. Production and characterization of Reg knockout mice: reduced proliferation of pancreatic beta-cells in Reg knockout mice. Diabetes. 2002;51 Suppl 3:S478-83.
- 64. Rosén T, Edén S, Larson G, Wilhelmsen L, Bengtsson B-Å. Cardiovascular risk factors in adult patients with growht hormone disease. Acta Endocrinologica. 1993;129:195-200.
- 65. de Boer H, Blok GJ, Voerman HJ, Phillips M, Schouten JA. Serum Lipid Levels in Growth Hormone-Deficient Men. Metabolism. 1994;43(2):199-203.
- 66. Darzy KH, Shalet SM. Hypopituitarism following Radiotherapy Revisited. Endocr Dev. 2009;15:1-24.
- 67. Chemaitilly W, Li Z, Huang S, et al. Anterior hypopituitarism in adult survivors of childhood cancers treated with cranial radiotherapy: a report from the St Jude Lifetime Cohort study. J Clin Oncol. 2015;33(5):492-500.
- 68. Belle FN, Weiss A, Schindler M, et al. Overweight in childhood cancer survivors: the Swiss Childhood Cancer Survivor Study. Am J Clin Nutr. 2018;107:3-11.
- 69. van Santen HM, Geskus RB, Raemaekers S, et al. Changes in body mass index in long-term childhood cancer survivors. Cancer. 2015;121(23):4197-204.
- Brown AL, Lupo PJ, Danysh HE, Okcu MF, Scheurer ME, Kamdar KY. Prevalence and Predictors of Overweight and Obesity Among a Multiethnic Population of Pediatric Acute Lymphoblastic Leukemia Survivors: A Cross-Sectional Assessment. J Pediatr Hematol Oncol. 2016;38(6):429-36.
- 71. Xiao R, Boehnke M. Quantifying and correcting for the winner's curse in genetic association studies. Genet Epidemiol. 2009;33(5):453-62.
- 72. Makela KM, Seppala I, Hernesniemi JA, et al. Genome-wide association study pinpoints a new functional apolipoprotein B variant influencing oxidized low-density lipoprotein levels but not cardiovascular events: AtheroRemo Consortium. Circ Cardiovasc Genet. 2013;6(1):73-81.
- Armenian SH, Gibson CJ, Rockne RC, Ness KK. Premature Aging in Young Cancer Survivors. J Natl Cancer Inst. 2019;111(3):226-32.
- 74. Tung JY, Do CB, Hinds DA, et al. Efficient replication of over 180 genetic associations with self-reported medical data. PLoS One. 2011;6(8):e23473.
- 75. Day FR, Hinds DA, Tung JY, et al. Causal mechanisms and balancing selection inferred from genetic associations with polycystic ovary syndrome. Nat Commun. 2015;6:8464.

Chapter 7

Perspectives on follow-up care and research for childhood cancer survivors: results from an international SIOP meetthe-expert questionnaire in Kyoto, 2018

M.E. Bakker*, V.G. Pluimakers*, J.E. van Atteveld, S.J.C.M.M. Neggers, M.M. van den Heuvel-Eibrink, S. Sato, K. Yamashita, C. Kiyotani, Y. Ishida, M. Maeda

Japanese Journal of Clinical Oncology, 2021 Oct 5;51(10):1554-1560

ABSTRACT

Introduction

Survival of childhood cancer has increased over the past decades. This has led to the development of strategies aiming to enhance follow-up care and research, for which priorities may vary globally. We explored perspectives of an international healthcare workers panel.

Methods

Attendants of a meet-the-expert session on childhood cancer survivorship at the 2018 SIOP conference completed a survey about their view on important follow-up care and research aspects for survivors below and over 18 years. We analyzed overarching categories and subtopics, and compared Asian versus European and North American healthcare workers.

Results

Fifty-eight participants from different medical specialties (67.2% pediatric oncologists) and continents (48.3% Asia, 39.7% Europe/North America) responded. Follow-up care priorities for survivors below and over 18 years included physical care (39.3% \leq 18 years, 35.9% >18 years) and healthcare structure (29.4%, 26.0%). Physical care was also the most important research aspect for both age groups (52.5%, 50.7%). Psychological support was the most frequently reported subtopic. Asian clinicians (n=22) primarily prioritized physical care aspects of follow-up care, whereas European/North American (n=19) clinicians underscored the importance of healthcare structure.

Conclusion

Physical care is the most important aspect of survivorship care and research according to clinicians from several continents. Asian and European/North American respondents shared most priorities, however, healthcare structure was a more important category for European/North American clinicians. The most common subtopic was psychological support, underlining also the need to involve psychologists in follow-up.

MINI ABSTRACT

Physical care is the most important survivor follow-up aspect, according to clinicians from several continents. Asian and European/North American respondents shared most priorities. Healthcare structure was more important for European/North American clinicians.

INTRODUCTION

Over the past decades, childhood cancer survival rates have increased, from less than 30% in the sixties to current rates exceeding 80% in developed countries ⁽¹⁾. This is due to improvement of chemotherapy, radiotherapy, surgery, stem cell transplantation, and immunotherapy, as well as better treatment stratification and enhanced supportive care.

However, childhood cancer survivors can suffer from serious long-term treatmentrelated side effects, leading to a significant symptom burden and increased mortality rates ⁽²⁻⁴⁾. It has been shown that 70% of survivors develop at least one side effect, and 40% develop at least one severe side effect ^(5, 6). A recent study even showed that, at the age of 50, 96% of survivors have developed one or more severe chronic health conditions ⁽⁷⁾. These morbidities – physical sequelae such as secondary malignancies, cardiovascular disease, metabolic syndrome, adverse bone health, fertility problems, chronic lung and kidney disease, hearing loss, frailty, but also psychological side effects – are of detrimental influence on later life.

Increased awareness of late effects and knowledge on risk factors have led to the development of surveillance guidelines for survivors in different parts of the world, including the USA, Canada, European countries, and Japan⁽⁸⁻¹⁰⁾. Goals of these care and research strategies are optimal surveillance, development of interventions, and adjustments in treatment protocols for newly diagnosed childhood cancer patients. Although surveillance guidelines are being harmonized, different cultural values may lead to varying perspectives on which part of survivor care is most important, and what the focus of survivorship research should be. In addition, discrepancies in available resources, access to healthcare and insurance, as well as expertise to organize late effects care, may give rise to differences in follow-up care and research facilities across countries and continents. In some countries, follow-up care is uncommon.

Studies comparing perspectives on what is important for improving follow-up in different geographical areas, including Asia, are not yet available. The aim of this observational study was to evaluate survivorship care and research priorities across participants from different parts of the world.

METHODS

Attendants of a meet the expert (MTE) session on childhood cancer survivorship at the 2018 conference of the International Society of Pediatric Oncology (SIOP) in Kyoto, Japan, were invited to complete a survey created by the hosts of the MTE (MM, MMvdHE, YI, CK, KY, VP, and JvA; Supplemental Figure 1). This survey included characteristics of the respondents and continent of residence. Furthermore, respondents were asked how much of their employment time is spent on late effects care and research. Information regarding access to and setting of follow-up care in their respective country was retrieved.

In four open-ended questions, the most important aspects (maximum of 4 answers) of care and research for childhood and adult survivorship were evaluated. We did not request to rank these aspects. In this way, priorities were gathered for the following four areas: 1) follow-up care for survivors under 18 years old; 2) follow-up care for survivors over 18 years old; 3) follow-up research for survivors under 18 years old; and 4) research for survivors over 18 years old. In addition, the respondents were invited to explain what they considered the most important obstacle in establishing or improving survivorship care. All included respondents provided written permission for reporting on the outcome of the survey.

The answers to each of the open-ended questions were analyzed in two ways: in five overarching categories, and as separate, uncategorized subtopics (Supplemental Figure 2). The overarching categories included 'physical care', 'psychological care', and 'quality of life' (all components of care), a category 'healthcare structure', and a remaining category 'other'. Two authors (M-CB and VP) independently categorized the answers. Disagreements were resolved by discussion with each other or with the other authors. In the case of persistent ambiguity, an answer was placed in two categories. In the primary analysis, we focused on the overarching categories was calculated as a percentage of the total number of answers. In the secondary analysis, in which we focused on the respondents, the frequency of the uncategorized subtopics was calculated as a percentage of the total number of respondents. Finally, the separate answers to the four questions combined were ranked according to frequency, to provide an overview of all reported subtopics.

All items of the survey were analyzed for the total cohort. We also broadly assessed shared and differing opinions by the largest continental subsets of respondents, i.e. European/North American versus Asian clinicians. We grouped European and North American respondents, because these are Western high income countries with the longest experience in survivor follow-up care.

RESULTS

Participants

Sixty-seven participants of the MTE completed the questionnaire. Nine participants that did not provide written consent were excluded. The remaining 58 respondents included a variety of professions, ages, and nationalities (Table 1). The two largest groups were Asian (n=22) and European/North American clinicians (n=19). The other respondents included clinicians from the Middle East (n=3), Middle and South America (n=2), and Africa (n=1), as well as people with a different relationship to survivors: researchers (n=3), survivors (n=2), a parent, and 'other' (n=5).

The majority of respondents were pediatric oncologists (n=39). Other healthcare professions included nurses (n=3), a pediatrician, a social worker, a pediatric psychologist, a radiologist, and a pharmacist. Most respondents were 30 to 50 years old (n=29), or older than 50 years (n=25). Most respondents were female (n=40).

Thirty-nine respondents spent 1% to 30% of their employment on late effects care, 9 respondents spent 31% to 60%, and 7 respondents spent no time on late effects care. Thirty respondents reported to spend no employment time on late effects research and 22 respondents spent 1% to 30%. Almost all respondents answered that survivors in their country had access to survivorship care, either in a late effects clinic (n=25) or in a non-specialized clinic (n=30).

Priority categories and subtopics of follow-up care and research

Aspects of care for survivors under the age of 18 years

A total of 201 topics were reported (Table 2a). Physical care was the most frequently prioritized category (n=79, 39.3%). This was followed by health care structure (n=59, 29.4%), psychological care (n=39, 19.4%), quality of life (n=17, 8.5%), and other (n=7, 3.5%).

When analyzing all subtopics separately, the most frequently reported priorities were psychological support (answered by 32.8% of respondents), education (15.5%), and growth (12.1%) (Table 3).

Aspects of care for survivors over the age of 18 years

For adult survivors, a similar pattern was observed (Table 2b). Also for these survivors, physical care was the most frequently prioritized category (n=65/181 topics, 35.9%). This was followed by healthcare structure (n=47, 26.0%), psychological care (n=39, 21.5%), quality of life (n=26, 14.4%), and other (n=4, 2.2%).

The most frequently reported subtopics included psychological support (36.2% of respondents), fertility (29.3%), and continuing study/work (15.5%).

Aspects of research for survivors under the age of 18 years

A total of 160 subtopics was reported for this age group (Table 2c). The majority fell into the physical care category (n=84, 52.5%), followed by psychological care (n=37, 23.1%), other (n=15, 9.4%), healthcare structure (n=13, 8.1%), and quality of life (n=11, 8.1%).

The most frequently reported subtopics included psychological support (29.3% of respondents), second malignancy, genetics, and neurocognitive (all 12.1%).

Aspects of research for survivors over the age of 18 years

For survivors over 18 years old, the research priorities were similar (Table 2d). Of a total of 152 reported subtopics, 77 (50.7%) were related to physical care, followed by psychological care (n=30, 19.7%), other (n=19, 12.5%), quality of life (n=14, 9.2%), and healthcare structure (n=12, 7.9%).

The most frequently reported subtopics were fertility (34.5% of respondents), psychological support (24.1%), subsequent malignancy, and endocrinology (both 10.3%).

Overview of the frequency of subtopics

When the answers to the four questions were combined, psychological support was the most frequently prioritized subtopic of follow-up care and research for survivors (n=71, 10.2% of all answers) (Figure 1). This was followed by fertility (n=48, 6.9%), continuing study/work (n=30, 4.3%), and social support (n=29, 4.2%).

Comparison between Asian and European/North American clinicians

The two largest groups of respondents were Asian clinicians (n=22) and European/ North American clinicians (n=19), which enabled a descriptive, broad comparison of the answers reported by these groups of respondents. A detailed description of the two groups is supplied in Supplementary table 1. In both groups, most respondents were pediatric oncologists, involved in late effects care for some part of their employment. All but two respondents from Asia reported that survivors in their countries have access to survivorship care, and half of each group indicated that this care is embedded in specialized follow-up clinics.

The Asian clinicians primarily prioritized physical care aspects of follow-up care for both age groups of survivors, whereas the European/North American clinicians underscored the importance of healthcare structure. In the analysis of the uncategorized subtopics, psychological support was the most frequently answered subtopic for both groups of clinicians for survivors below as well as over 18 years.

The main research priority category concerned physical care for Asian as well as European/North American respondents. Psychological support was the most frequently reported subtopic for survivors below 18 years, although for Asian respondents this first place was shared with cardiology. For adult survivors, both groups of respondents predominantly reported fertility. For Asian respondents this was again a shared first place, with psychological support.

In the combined analysis, it is clear that psychological support was the most frequently reported aspect in survivor follow-up for the Asian as well as European/North American respondents, followed by fertility (Supplemental figure 3). Psychological support was mentioned 31 times (11.3% of a total of 274 answers) by Asian respondents and 32 times (14.4% of 222 answers) by European/North American. Fertility was reported 22 times (8.0%) and 21 (9.5%) times, respectively. Social support and continuing study/ work were also answered frequently by both groups. Cardiology, second malignancy and endocrinology were mentioned often by Asian respondents, but not by European/North American respondents.

Obstacles in survivorship care

Thirty-nine participants provided one or more obstacles in establishing or improving survivorship care. Commonly reported obstacles included lack of financial resources (n=10) and lack of manpower (n=9). Other obstacles were logistic difficulties, lack of knowledge of the importance of follow-up, lack of cooperation, lack of time, difficulties in organizing a multidisciplinary team, lack of support by the hospital, lack of communication with primary care, and difficulties in reaching survivors. None of the participants answered that there are no obstacles.

DISCUSSION

We gathered views from clinicians on follow-up care and research priorities for childhood cancer survivors on an international level, based on a survey distributed at a meet-the-expert session on childhood cancer survivorship on the 2018 SIOP conference.

Enhancement of components of care (i.e. physical care, psychological care, and quality of life) was regarded the most important challenge in follow-up care for survivors by the total group of respondents. This was felt to be relevant for survivors under as well as over 18 years old. Improving physical care was also the most important aspect of follow-up research, for both juvenile and adult survivors. This is likely a result of the many persisting knowledge gaps in the field of late effects surveillance ⁽⁹⁾. Research initiatives continuously discover new insights on the risks of specific treatments based on large cohort studies, including questionnaire investigations and recruitment studies such as the Childhood Cancer Survivor Study, St Jude Lifetime (both from the USA), the British Childhood Cancer Survivor Study, the DCCSS LATER cohort (The Netherlands), and the Japanese NCCHD Lifetime cohort. Psychological late effects are often not the first priority in care and research pursued by oncologists. Since the majority of our participants were oncologists, this topic may be less frequently addressed.

Nevertheless, when the uncategorized subtopics for care and research were combined, psychological support was the most frequently issued subtopic in the total group of respondents. This illustrates that psychological support is also an important topic for physicians. Still, this is a broad term, so involvement of psychologists in late effects care and research is desired, to address tailored support for survivors (e.g., anxiety, depression, concentration, empowerment, relationships, psychosexuality). By asking respondents' visions for survivors under and over 18 years old, we were able to analyze whether age was an important factor for determining the focus. In general, this was not the case: respondents answered largely similar for both age groups. One exception was fertility, which was particularly mentioned for adult survivors. This seems logical because fertility becomes a more important issue when survivors reach reproductive age.

The distribution of this survey on the international SIOP conference provided us the opportunity to collect and compare visions of an international group of respondents. Still, this is a selected cohort consisting of people who decided to attend the MTE and mainly comprised of pediatric oncologists. These factors may limit generalizability of our results to larger groups of healthcare professionals from the different countries

and continents. Replication in a larger, non-selected cohort is therefore an interesting future step.

We broadly compared the results from Asian and European/North American respondents. Whereas physical care was addressed most frequently by the Asian respondents, enhancement of healthcare structure was more prominently mentioned by European/North American respondents. We realize that these differences may be due to selection bias of attendants of the MTE, and also due to an age difference, as the Asian group was younger. Additionally, the priority for healthcare structure among European/North American respondents may also be due to the fact that in Europe, the USA and Canada, physicians have focused for a longer time on setting up late effects clinics, and are more involved in developing follow-up guidelines, which were unavailable in Japan until 2013 ^(8, 9, 11, 12). Since then, there has been an ongoing development to address cancer survivorship issues in clinical care, and the development of follow-up guidelines has continued in Japan⁽¹²⁻¹⁵⁾. As a result, a general Japanese long-term follow-up guideline, as well as specific guidelines for after stem cell transplantation, and for fertility preservation have become available. Given the relative recency of these developments, it is reasonable that improving physical care in follow-up had highest priority among Asian respondents. This emphasizes the need to implement these guidelines and disseminate knowledge. Additionally, the fact that these guidelines are only available in Japanese language limits accessibility to the international public. Therefore, we opt to involve Japan, and possibly other countries, in international consensus meetings on childhood cancer survivorship.

To our knowledge, there is no other survey comparing Asian and European/North American perspectives of survivorship care and research among clinicians. Opinions of the two groups have been studied separately, although only a few of these studies included Asian physicians ^(11, 12, 16-18). Similar to our study, survey-based studies among European or American physicians revealed aspects of healthcare structure improvement, such as specialized clinics and a smooth transfer after reaching adulthood ⁽¹⁶⁻¹⁸⁾. In contrast, the psychological aspect was not highlighted in any of these studies.

We also asked participants about obstacles in establishing and improving survivorship care and research in their country. Several obstacles were mentioned, including lack of manpower, insufficient awareness among survivors of the importance of follow-up, and lack of cooperation, time and money. Similar obstacles were identified in surveys among US, UK and Swiss physicians ^(11, 16, 17, 19, 20). Interestingly, in the light of financial obstacles, a recent study by Kaal et al. found that coordinated survivor care can lead to a reduction in healthcare costs of 300-900 Canadian dollar per patient ⁽²¹⁾.

In our study, two survivors and one parent were among the respondents. Unfortunately, this group is too small to draw firm conclusions. In future research, it remains of particular interest to include the opinion of more parents and survivors about important aspects of follow-up care and research.

Because the questions in our survey were open-ended, respondents were able to give their opinions in their own words. Although this is a good aspect, we experienced some difficulties with categorizing answers. For instance, some answers were not readable and could therefore not be included in our analysis. More importantly, we realize that categorization depends on the authors' interpretation of the answers. Two reviewers independently categorized the answers, and discussed disagreements with the other authors. Persistent ambiguous terms were placed in two categories, for example, the term 'puberty' may refer to physical changes during puberty, or psychological aspects of puberty. This way, we aimed to minimize the influence of the reviewers' interpretation of answers.

In conclusion, our survey revealed priorities of clinicians from several continents regarding aspects of childhood cancer survivor follow-up care and research. We identified shared as well as differing opinions between Asian and European/North American clinicians. Important priorities were physical care and healthcare structure, the latter being more prominent among European/North American respondents. The most frequent uncategorized answer was psychological support, underlining the need to involve psychologists in follow-up care. Therefore, specialized follow-up clinics are preferred, in order to deliver tailored care in a structured way. Although this may be challenging given the identified obstacles, the results of our survey emphasize the need for a structured approach, covering physical as well as psychological care. Our findings can aid healthcare workers in improving surveillance guidelines, adjusted to local settings and with good accessibility. Including experts from various continents, thereby including perspectives from various cultural angles, can lead to the international harmonization and optimization of follow-up, to provide survivors worldwide with optimal quality of survival.

Conflicts of interest:

None.

Acknowledgment:

Marie-Christine Bakker is a medical student participating in the Honours programme of the Faculty of Medicine, University Medical Center Utrecht, The Netherlands.

TABLES AND FIGURE

Question	Answer	Number of respondents	Percentage of total (N=58)
I am	Pediatric oncologist	39	67.2
	Nurse	3	5.2
	Pediatrician	1	1.7
	Social worker pediatric oncology	1	1.7
	Pediatric psychologist	1	1.7
	Radiologist	1	1.7
	Pharmacist	1	1.7
	Researcher	3	5.2
	Survivor	2	3.4
	Parent	1	1.7
	Other	5	8.6
Age	<30 years	4	6.9
	30-50 years	29	50.0
	>50 years	25	43.1
Gender	Female	40	69.0
I live/work in	Asia	28	48.3
	Europe	19	32.8
	USA/Canada	4	6.9
	Middle East	3	5.2
	Middle or South America	3	5.2
	Africa	1	1.7
I spend of my	0%	7	12.1
employment on late effects	1-30%	39	67.2
care	31-60%	9	15.5
	>60%	2	3.4
	Not answered	1	1.7
I spend of my	0%	30	51.7
employment on late effects	1-30%	22	37.9
research	31-60%	3	5.2
	>60%	2	3.4
	Not answered	1	1.7
Childhood cancer survivors	Yes, in a late effects clinic	25	43.1
in my country have access to	Yes, but not in a late effects clinic	30	51.7
survivorship care	No	1	1.7
	Not answered	2	3.4

Table 1. Description of study population

Category	Number of answers	Percentage of total number of answers (N=201)
Components of care	135	67.2
Physical care	79	39.3
Psychological care	39	19.4
Quality of life	17	8.5
Health care structure	59	29.4
Other	7	3.5
Genetics	1	0.5

Table 2a. Important aspects of late effects care for childhood cancer survivors ≤18 years

Table 2b. Important aspects of late effects care for childhood cancer survivors >18 years

Category	Number of answers	Percentage of total number of answers (N=181)
Components of care	130	71.8
Physical care	65	35.9
Psychological care	39	21.5
Quality of life	26	14.4
Health care structure	47	26.0
Other	4	2.2
Genetics	1	0.6

Table 2c. Important aspects of late effects research for childhood cancer survivors ≤18 years

Category	Number of answers	Percentage of total number of answers (N=160)
Components of care	132	82.5
Physical care	84	52.5
Psychological care	37	23.1
Quality of life	11	6.9
Health care structure	13	8.1
Other	15	9.4
Genetics	7	4,4
Category	Number of answers	Percentage of total number of answers (N=152)
-----------------------	-------------------	--
Components of care	121	79.6
Physical care	77	50.7
Psychological care	30	19.7
Quality of life	14	9.2
Health care structure	12	7.9
Other	19	12.5
Genetics	7	4.6

The war important appeets of fate enterts for entrantood entreet sur fronts for feat
--

Table 3. Most frequent answers to	open-ended questions
-----------------------------------	----------------------

Category	Most frequent answers	Number of respondents that gave answer	Percentage of total number of respondents (N=58)
Late effects care ≤18 years	Psychological support	19	32.8
	Education	9	15.5
	Growth	7	12.1
Late effects care >18 years	Psychological support	21	36.2
	Fertility	17	29.3
	Continuing study/work	9	15.5
Late effects research ≤18 years	Psychological support	17	29.3
	Second malignancy	7	12.1
	Genetics, inheritance	7	12.1
	Neurocognitive	7	12.1
Late effects research >18 years	Fertility	20	34.5
	Psychological support	14	24.1
	Second malignancy	6	10.3
	Endocrinology	6	10.3



Most frequent answers

Figure 1. Top 15 most frequent answers on all open-ended questions combined

SUPPLEMENTAL MATERIAL

Question	Answer	Asian respondents (N=22)	European/American respondents (N=19)
I am	Pediatric oncologist	19 (86.4%)	14 (73.7%)
	Nurse	0	3 (15.8%)
	Pediatrician	1 (4.5%)	0
	Social worker pediatric oncology	1 (4.5%)	0
	Pediatric psychologist	0	1 (5.3%)
	Radiologist	0	1 (5.3%)
	Pharmacist	1 (4.5%)	0
Age	<30 years	1 (4.5%)	0
	30-50 years	14 (63.6%)	6 (31.6%)
	>50 years	7 (31.8%)	13 (68.4%)
Gender	Female	13 (59.1%)	15 (78.9%)
I live/work in	Asia	22 (100%)	-
	Europe	-	15 (78.9%)
	USA/Canada	-	4 (21.1%)
I spend of my employment on	0%	0	2 (10.5%)
late effects care	1-30%	18 (81.8%)	13 (68.4%)
	31-60%	4 (18.2%)	3 (15.8%)
	>60%	0	1 (5.3%)
I spend of my employment on late effects research	0%	12 (54.5%)	9 (47.4%)
	1-30%	10 (45.5%)	8 (42.1%)
	31-60%	0	1 (5.3%)
	>60%	0	1 (5.3%)
Childhood cancer survivors	Yes, in a late effects clinic	9 (40.9%)	10 (52.6%)
in my country have access to	Yes, but not in a late effects clinic	11 (50.0%)	9 (47.4%)
survivorship care	No	1 (4.5%)	0
	Not answered	1 (4.5%)	0

Supplemental table 1. Description of the Asian and European/American healthcare workers in the study population

```
50<sup>TH</sup> CONGRESS OF THE INTERNATIONAL
SOCIETY OF PAEDIATRIC ONCOLOGY
KYOTO, JAPAN
NOVEMBER 16-19, 2018 Industriant and
 ST (D) P
                       1
Survey about optimal survivor care:
I am:
        Survivor
                                                                                     D Parent
        Nurse
                                                                                     D Researcher
        Pediatric oncologist
                                                                                     D Other physician,
        Other....
My age is:
         <30 years
        31-50 years
        >50 years
My gender is:
        Male
        Female
I live/work in:
        Asia
                                                                                     o Africa
                                                                                     o USA/Canada
        Europe

    Australia/NZ

        Middle/South America
        Middle East
Childhood cancer survivors in my country have access to survivorship care: -
        Yes, in a late effects clinic
        Yes, but not in a late effects clinic
        No
The most important obstacle to establish/improve/receive optimal survivorship care in my setting is:
I spend ......% of my employment on late effects care:
        0%/not applicable
        1-30%
        31-60%
        >60%
I spend .....% of my employment on late effects research:
        0%/ not applicable
        1-30%
        31-60%
        >60%
```

Supplemental figure 1: The questionnaire.

For me, the 4 most important topics that characterize excellent late of	effects care after childhood cancer are:
For survivors aged 5-18 years:	
Χ	
X	
X	
x	
For me, the 4 most important priorities for late effects research after	er childhood cancer are:
For survivors aged 5-18 years:	
X	
Χ	
X	
x	
x	
x	
For me, the 4 most important priorities for late effects <u>research</u> after a dult survivors, aged >18 years: XX	er childhood cancer are:
x	
I give permission to the organizers of the meeting to use this survey for analyses and scientific report	100
Signature,	SOUTH OF ALCONDUCTION AND A CONTRACTOR
Date, 17/18 November 2018	MANEMERS IN CO., STOLE AND



п 1

Genetics

Other:

Transition to adult care

Supplemental figure 3. Top 8 most frequent answers by Asian and European/American respondents, on all questions regarding both late effects care and research.



Most frequent answers

REFERENCES

- Howlader N, Noone AM, Krapcho M, et al. SEER Cancer Statistics Review, 1795-2017, National Cancer Institute, Bethesda, MD, https://seer.cancer.gov/csr/1975_2017/, based on November 2019 SEER data submission, posted to the SEER web site, April 2020.
- 2. Armstrong GT, Liu Q, Yasui Y, et al. Late mortality among 5-year survivors of childhood cancer: a summary from the Childhood Cancer Survivor Study. J Clin Oncol. 2009;27(14):2328-38.
- Bagnasco F, Caruso S, Andreano A, et al. Late mortality and causes of death among 5-year survivors of childhood cancer diagnosed in the period 1960-1999 and registered in the Italian Off-Therapy Registry. Eur J Cancer. 2019;110:86-97.
- 4. Mertens AC, Yasui Y, Neglia JP, et al. Late mortality experience in five-year survivors of childhood and adolescent cancer: the Childhood Cancer Survivor Study. J Clin Oncol. 2001;19(13):3163-72.
- 5. Geenen MM, Cardous-Ubbink MC, Kremer LCM, et al. Medical Assessment of Adverse Health Outcomes in Long-term Survivors of Childhood Cancer. JAMA. 2007;297(24):2705-15.
- 6. Oeffinger KC, Mertens AC, Sklar CA, et al. Chronic Health Conditions in Adult Survivors of Childhood Cancer. N Engl J Med. 2006;355(15):1572-82.
- 7. Bhakta N, Liu Q, Ness KK, et al. The cumulative burden of surviving childhood cancer: an initial report from the St Jude Lifetime Cohort Study (SJLIFE). Lancet. 2017;390(10112):2569-82.
- Children's Oncology Group. Long-Term Follow-Up Guidelines for Survivors of Childhood, Adolescent, and Young Adult Cancers [31 July 2020]. Available from: http://www. survivorshipguidelines.org/.
- 9. International Guideline Harmonization Group for Late Effects of Childhood Cancer. Long-term follow-up guidelines [31 July 2020]. Available from: https://www.ighg.org/guidelines/.
- Stichting Kinderoncologie Nederland. Richtlijn follow-up na kinderkanker [31 July 2020]. Available from: https://www.skion.nl/voor-patienten-en-ouders/late-effecten/533/richtlijnfollow-up-na-kinderkanker/.
- Henderson TO, Hlubocky FJ, Wroblewski KE, Diller L, Daugherty CK. Physician preferences and knowledge gaps regarding the care of childhood cancer survivors: a mailed survey of pediatric oncologists. J Clin Oncol. 2010;28(5):878-83.
- 12. Ishida Y, Takahashi M, Maru M, et al. Physician preferences and knowledge regarding the care of childhood cancer survivors in Japan: a mailed survey of the Japanese Society of Pediatric Oncology. Jpn J Clin Oncol. 2012;42(6):513-21.
- 13. Takahashi M. Cancer survivorship: current status of research, care, and policy in Japan. Jpn J Clin Oncol. 2016;46(7):599-604.
- 14. Miyoshi Y, Yorifuji T, Shimizu C, et al. A nationwide questionnaire survey targeting Japanese pediatric endocrinologists regarding transitional care in childhood, adolescent, and young adult cancer survivors. Clin Pediatr Endocrinol. 2020;29(2):55-62.
- 15. Matsuoka YJ, Okubo R, Shimizu Y, et al. Developing the structure of Japan's cancer survivorship guidelines using an expert panel and modified Delphi method. J Cancer Surviv. 2020;14(3):273-83.
- 16. Michel G, Gianinazzi ME, Vetsch J, et al. Physicians' experience with follow-up care of childhood cancer survivors challenges and needs. Swiss Med Wkly. 2017;147:w14457.
- 17. Suh E, Daugherty CK, Wroblewski K, et al. General internists' preferences and knowledge about the care of adult survivors of childhood cancer: a cross-sectional survey. Ann Intern Med. 2014;160(1):11-7.
- Mertens AC, Cotter KL, Foster BM, et al. Improving health care for adult survivors of childhood cancer: recommendations from a delphi panel of health policy experts. Health Policy. 2004;69(2):169-78.

- 19. Oeffinger KC, Wallace WH. Barriers to follow-up care of survivors in the United States and the United Kingdom. Pediatr Blood Cancer. 2006;46(2):135-42.
- 20. Zebrack BJ, Eshelman DA, Hudson MM, et al. Health care for childhood cancer survivors: insights and perspectives from a Delphi panel of young adult survivors of childhood cancer. Cancer. 2004;100(4):843-50.
- 21. Kaal KJ, McGrail K, Harrison M, McBride M. Patient Sharing and Cost of Care among Long-Term Survivors of Childhood, Adolescent and Young Adult Cancer in British Columbia (BC), Canada. Pediatric Blood & Cancer. 2020;67(S4):Supplement: Abstracts from the 52th Congress of the International Society of Paediatric Oncology (SIOP) Virtual Congress, October 14–7, 2020. Abstract 0279/#1353.

Chapter 8

Discussion and future perspectives

DISCUSSION AND FUTURE PERSPECTIVES

Over the past decades, curing children with cancer has become increasingly feasible. Nevertheless, on the long term, in adult survivors of childhood cancer excess morbidity and mortality due to cancer treatment are observed. Metabolic syndrome, which is a risk factor for early death due to cardio- and cerebrovascular disease and diabetes mellitus, is more prevalent among childhood cancer survivors (CCS), even at a very young age. Therefore, survivors need to be closely monitored, but metabolic syndrome and also the separate components can be underdiagnosed with the current criteria in (subgroups of) survivors. In addition, knowledge on prevalence as well as on clinical and genetic risk factors, based on large national cohort studies, is lacking. The studies described in this thesis aimed to address these challenges with regard to metabolic syndrome and the components overweight and dyslipidemia.

Determinants of developing metabolic syndrome, and better metabolic syndrome (bio)markers in childhood cancer survivors

Effective follow-up includes proactive screening of CCS at risk of developing metabolic syndrome, as well as avoidance of unnecessary screening of those who are not. Established risk factors, as identified in our literature review (*Chapter 1*), include brain tumors, cranial and abdominal radiotherapy, and alkylating chemotherapy. Knowledge is still incomplete, as most studies so far are based on small and selected survivor cohorts, have insufficient treatment data, are questionnaire-based, or have short follow-up. Additionally, more reliable metabolic syndrome (bio)markers are required, most importantly because waist circumference underestimates adiposity in survivors who received abdominal radiotherapy. Consequently, this may underdiagnose metabolic syndrome. During this thesis project, data from a national cohort was acquired that will increase insight in these subjects.

In the preparation phase for the national cohort study, a single-center study among 103 survivors of nephroblastoma and neuroblastoma (described in *Chapter 2*) was performed, showing that abdominal radiotherapy was significantly associated with occurrence of metabolic syndrome and the components high triglycerides, low HDL cholesterol and arterial hypertension. The mechanisms by which abdominal radiotherapy leads to metabolic syndrome are multifactorial and not yet fully elucidated. It has been shown that radiation to the pancreas area, most importantly the tail where the Langerhans islet density is highest, leads to insulin resistance and metabolic syndrome in a dose dependent manner^(1, 2). Females experience an additional risk, as radiation damage to the ovaries can lead to premature ovarian insufficiency (POI) and subsequent metabolic syndrome risk^(3, 4). This is because estrogen deficiency in POI , in addition to a directly increased cardiovascular disease risk⁽⁵⁾, leads to body

fat redistribution with increased central adiposity, reduced pancreatic insulin synthesis and secretion and beta cell survival, and increased low-density lipoprotein cholesterol (LDL) and apolipoprotein B (apoB) production in the liver^(6,7). Furthermore, radiation to the liver contributes to the dyslipidemia component of metabolic syndrome⁽⁸⁾. Abdominal adipose tissue may also be affected by irradiation. Previous *in vitro* and mouse experiments have suggested that irradiation could trigger fat accumulation in adipose tissue⁽⁹⁾, and that increase of fibrous adipose tissue – which radiation can induce – is associated with insulin resistance, but also with a favorable plasma lipid profile⁽¹⁰⁾. We could hypothesize that fibrous abdominal adipose tissue has less potential to serve as a fat reservoir, with negative consequences including hypertrophy of remaining adipocytes, and intra-abdominal and liver fat accumulation. Future research may therefore focus on how adipose tissue is affected by irradiation, and on identifying mechanisms by which this is associated with overweight, altered lipid metabolism and metabolic syndrome in CCS.

In order to further unravel the multifactorial manner in which abdominal radiotherapy may lead to metabolic syndrome, data on radiation dose received by different organs and tissues will be more informative than the prescribed field and dose. This may even allow analysis of organ parts and structures, which has revealed insights in radiationinduced diabetes, stroke and cardiac disease in recent survivor studies^(2, 11, 12). In the DCCSS-LATER study, such dosimetry data are currently analyzed. Based on this, dosimetry based surveillance strategies could be developed, which could reduce the amount of diagnostic testing⁽¹³⁾. In addition, such analyses may not be pursued only in survivors, but already start during and from the end of treatment, to identify early damage, to better understand the mechanisms, and to use interventions at an early stage.

Another next step may be to build prediction models that quantify the risk increase for treatment-related factors. To establish these risk scores, larger cohorts are required, enabling the concurrent analysis of several factors. Such prediction models need to be validated in independent cohorts. Prediction models were recently developed to predict ischemic heart disease and stroke in survivors⁽¹⁴⁾. A model predicting metabolic syndrome can be even more clinically relevant, as this is the preceding condition. Several studies have suggested that CCS with metabolic syndrome have a higher risk of developing cardiovascular disease and diabetes mellitus than the general population, due to an additive or even synergistic effect of factors including cardioand vasculotoxic therapies, comorbidities including endocrinopathies, and sedentary lifestyle⁽¹⁵⁻¹⁸⁾. Using prediction models for metabolic syndrome (components) could therefore enable earlier intervention with potential great benefits. Still, it needs to be proven that intervention also avoids cardiovascular disease and diabetes mellitus in CCS. This includes medical interventions, e.g., the use of GLP1 agonists for weight loss, and lifestyle interventions, including diet, smoking and physical exercise⁽¹⁹⁾. Such survivor specific intervention programs are currently being developed⁽²⁰⁻²²⁾. This may become challenging, as in the general population, it has been shown that hormonal, metabolic and neurochemical adaptations in metabolic syndrome hamper the effectiveness of interventions⁽²³⁾.

In the aforementioned single-center study, we also noticed that almost none of the survivors had the metabolic syndrome component "increased waist circumference", which is remarkable when taking the normal population into account. This may be explained by the historic administration of abdominal or flank radiotherapy, leading to a damaged and underdeveloped waist region. This underlines the troublesome situation that, while abdominal irradiation is a risk factor for metabolic syndrome, at the same time it hampers its assessment, as three of the four remaining diagnostic criteria are required. We therefore suggest that replacing the overweight criterion with an alternative component, may be an option. In a previous study, replacement of waist circumference with body composition, expressed as fat% on dual-energy X-ray absorptiometry (DXA), revealed a more accurate prevalence of metabolic syndrome⁽⁸⁾. We investigated the potential use of biomarkers for this purpose, among the nephroand neuroblastoma survivors in our single center study. Four serum biomarkers appeared to be potentially useful candidates to assess metabolic syndrome for this purpose: adiponectin, LDL cholesterol, apoB and uric acid.

Additionally, we summarized the available literature for the use of biomarkers in improving diagnosis and prediction of metabolic syndrome (*Chapter 3*). Five out of nine reviewed biomarkers were identified to be useful: uric acid, leptin, high sensitivity C-reactive protein (hsCRP), adiponectin and apoB. However, studies among survivors were scarce and often had high risk of bias. Based on general population studies high and moderate quality evidence for the use of these five biomarkers was found. Evidence from this older general population may be applicable to young-adult childhood cancer survivors, because accelerated aging was observed in several other studies, with frailty levels in 30 year old survivors comparable to 65 year old adults from the normal population⁽²⁴⁻²⁶⁾. Even so, it could be that survivor-specific comorbidity influences the pathophysiology of metabolic syndrome and therefore the relationship between biomarker and metabolic syndrome.

Advantages of biomarkers include that they are mostly easy and cheap to determine on a routine basis in clinical practice. We feel that LDL may be suboptimal as additional criterion, because the components "high triglycerides" and "low highdensity lipoprotein" already capture most of this effect. ApoB is also a lipid marker but evidence is increasing that it is a more accurate marker of atherogenic risk⁽²⁷⁻²⁹⁾. Measurement of hyperuricemia is a true separate component and may therefore be interesting. The same accounts for measurement of chronic inflammation with hsCRP. We suggest that leptin and particularly adiponectin, for which most evidence was found, may be the most interesting surrogate markers in survivors with unreliable waist circumference, given their relation with adipose tissue.

To further study the usefulness of these biomarkers, these need to be assessed in a prospective study, as additional metabolic syndrome components, in larger cohorts of survivors of heterogeneous malignancies. Furthermore, although reference values for these biomarkers are available, a stricter threshold value may be required when analyzed concurrently with the correlated metabolic syndrome components. This is illustrated in current metabolic syndrome definitions, that use a lower threshold for hypertension than commonly used when blood pressure is measured solely (130/85 vs 140/90mmHg)⁽³⁰⁻³²⁾. In the DCCSS-LATER study we are currently evaluating the use of biomarkers for this purpose.

We also studied the potential value of vascular ultrasound among the nephro- and neuroblastoma survivors, in enhancing metabolic syndrome diagnosis. Although vascular ultrasound can identify early vascular consequences of metabolic syndrome, we were unable to find a clear benefit. This is different from a few previous studies in adult CCS, that included other cancer types and treatments including testicular cancer and mediastinal and total body irradiation (TBI)⁽³³⁻³⁶⁾. It could be that the survivors in our study were still too young to develop vascular effects. Recent evidence from the general population suggests short-term progression of subclinical, ultrasound assessed, atherosclerosis between the age of 40 and $50^{(37)}$. The median age of the nephro- and neuroblastoma survivors was 30, so if accelerated vascular aging occurs, and if CCS develop similar vascular consequences, these effects may be identified in their next decade of life. Such a vascular study might also include carotid or femoral artery plaque burden on arterial ultrasound, as well as coronary artery calcium (CAC) score on CT scan of the heart. These modalities are more predictive of cardio- and cerebrovascular events than other radiologic measurements, according to the European Society of Cardiology (ESC) guideline on dyslipidemia⁽³⁸⁾.

Prevalence and determinants of overweight in childhood cancer survivors and the influence of measurement modalities

The DCCSS-LATER Metabolic syndrome study aims to provide knowledge on clinical and genetic determinants of metabolic syndrome and the separate components, and the diagnostic value of DXA scan and biomarkers, in a nationwide study cohort that comprises the first treated (1963-2002) survivors in The Netherlands (*Chapter 4*). This

thesis describes the results of the overweight component and the results of the total picture of metabolic syndrome are currently under analysis. Overweight occurred in almost half of all long-term survivors. The increase in overweight prevalence per age group was steeper in survivors compared to the general population. This may suggest a more rapid increase in overweight when survivors age, indicating the need for follow-up on the very long-term for all survivors. Or it may be an effect of the earlier treatment eras, when children with cancer received more harmful therapies, e.g., cranial radiotherapy (CrRT). This would imply that only treatment-based subgroups require follow-up. Given these contrary implications, a longitudinal follow-up study until elderly age will be relevant.

Overweight at diagnosis, CrRT and growth hormone deficiency (GHD) were associated with long-term overweight in this study. Overweight at diagnosis may in part reflect a genetic predisposition. In current additional analyses this genetic variation is also studied. As the relationship between GHD and overweight is bidirectional⁽³⁹⁾, it cannot vet be concluded that GHD is a risk factor for overweight development, because of the cross-sectional design of the study. A longitudinal study from childhood cancer treatment onwards would be required to fully elucidate which of the two occurred first. Additionally, hypogonadism may play a role in overweight development. Hypogonadism has been reported to be associated with overweight in previous studies⁽⁴⁾, and it often co-occurs with GHD in survivors after treatment with cranial radiotherapy⁽⁴⁰⁻⁴²⁾. Data on hypogonadism in our study cohort will become available. Hence, in future analyses it will be interesting to include data on gonadal status, to further elucidate these mechanisms. Future analyses could also benefit from the use of dosimetry data of parts of the brain. Whereas with full CrRT the hypothalamus and pituitary gland will receive the entire dose, with local radiotherapy these important structures may or may not be in the field. These nuances could not be incorporated yet in the current analyses.

Prevalence of overweight using several assessment methods was compared, of which the DXA scan was most sensitive. It identified overweight in an additional 30% of survivors, as compared to BMI, particularly in those treated with abdominal irradiation, TBI, anthracyclines and platinum chemotherapy. For abdominal irradiation and TBI this higher discrepancy rate between BMI and DXA has been reported in previous studies^(8, 43) but anthracycline and platinum chemotherapy treated survivors were never identified to be at risk, so far. This more sensitive method may therefore be relevant in these subgroups. In a previous study in Dutch survivors, there was an association between anthracyclines and low BMI, but fat% was not measured⁽⁴⁴⁾. In future studies it will be interesting to unravel why altered body composition seems to occur in these survivors. Sarcopenia frequently occurs among CCS⁽⁴⁵⁾ and could

be an explanation why anthropometric measurements that include lean mass, such as BMI, underestimate overweight. It is known that anthracyclines can induce cardiac myocyte cell death, leading to heart failure⁽⁴⁶⁾. This could lead to decreased exercise ability and less muscle mass, although in our analyses we adjusted for physical activity. Toxicity of anthracyclines so far appears to be limited to the heart, but it could be that also the skeletal myocytes are damaged, leading to sarcopenia and subsequent underestimation of overweight. Such a mechanism was observed in a rat study, which showed that doxorubicin reduced fiber size and satellite cell and capillary density of the soleus muscle⁽⁴⁷⁾. The question remains whether this accounts for CCS as well. As for platinum chemotherapy, this can induce skeletal muscle wasting⁽⁴⁸⁾. While this muscle damage and dysfunction during treatment has been frequently reported, a long-term effect has not been described. Therefore, whether anthracyclines and platinum lead to clinically relevant altered body composition remain to be demonstrated.

In this study we also observed that high molecular weight-adiponectin did not seem to add diagnostic value in assessing overweight. This was not what we expected based on our systematic review, although in this review we looked at metabolic syndrome as endpoint, not overweight specifically. Less than five percent of the national cohort had low adiponectin, so overweight development in CCS may be less correlated with adiponectin than in the general population. TBI and CrRT were associated with low adiponectin, and adiponectin levels were inversely associated with high BMI and high waist circumference. This does suggest its potential as a surrogate marker for overweight, but perhaps only in subgroups of CCS. Therefore, this remains interesting for further analysis, as we are currently performing in this cohort.

Genetic polymorphisms associated with dyslipidemia in survivors

Similarly treated CCS can have a different risk to develop metabolic syndrome and separate components. This may be due to environmental factors but also to genetic susceptibility. For the dyslipidemia component, only one candidate gene study in survivors has been performed so far, which identified three loci to be relevant in acute lymphoblastic leukemia (ALL) survivors (in *APOB, BAD* and *OGFOD3*)⁽⁴⁹⁾. By genome-wide association approach, numerous loci associated with dyslipidemia have been identified in the general population^(50, 51).

We performed a genome-wide association study (*Chapter 6*) in the Childhood Cancer Survivor Study cohort, which identified variant rs114017774 on chromosome 2 as a potential genetic variant associated with dyslipidemia, specifically in cranially irradiated survivors. After meta-analysis of the results of this cohort and the St Jude Lifetime cohort a higher risk of developing dyslipidemia was observed, with OR 11.30 (95%CI=5.03-25.40, p= $4.46*10^{-9}$). Because of the low minor allele frequency

of the variant rs114017774, our results remain based on a very small sample size, of only heterozygous subjects. Therefore, additional replication and meta-analysis in other cohorts of cranially irradiated survivors is required. We are currently performing a replication analysis using the Dutch CCS cohort, to further validate this finding. The potential candidate genes affected by rs114017774 are LRRTM4 and CTNNA2, although further functional evidence is required. No expression quantitative trait locus between the top locus and genes in the area was identified. These candidate genes were identified based on chromatin interactions. Future experiments will need to show whether this locus, which is in a non-coding region, influences expression of one of these genes, in relevant tissue (i.e., brain)⁽⁵²⁾. Also, additional evidence for development of dyslipidemia phenotype is required. LRRTM4 and CTNNA2 are expressed in the brain and involved in development and maintenance of the nervous system and synaptic functioning. Genetic variants inside these genes were associated with high BMI and coronary artery disease, which may suggest a link with metabolic phenotypes. Knock-out mice showed decreased brain synapse formation and growth and also decreased body growth^(53, 54), but knock-out mice in general show a phenotype involving reduced weight⁽⁵⁵⁾. Growth hormone deficiency would also show a phenotype with increased fat but this was not described in these mice. Mouse models with knock-out of the REG3 gene, for which chromatin interactions with rs114017774 were also observed, did develop dyslipidemia⁽⁵⁶⁾. However, this gene is mostly expressed in the pancreas, so one would expect to find this association in the abdominal radiotherapy exposed subgroup analysis, which was not the case. Based on current evidence, we may hypothesize that survivors carrying the variant rs114017774 identified in our GWAS, when treated with cranial radiotherapy, are more vulnerable to aberrant functioning of the hormonal pathways arising in the hypothalamus. Hence, at increased risk of developing dyslipidemia (and other metabolic syndrome components).

A remarkable finding in our analyses was that radiotherapy to the frontal brain region appeared to be even more associated with dyslipidemia than radiotherapy to the hypothalamic and pituitary region. This is a good example of how dosimetry data of organ structures can yield more detailed observations. An association between frontal lobe radiation damage and dyslipidemia was not reported in literature before. Among other functions, the frontal lobe is involved in impulse regulation⁽⁵⁷⁾. Although highly speculative, it could be that dysfunction of impulse control due to genetic predisposition and cranial radiotherapy leads to an unhealthier lifestyle and consequent dyslipidemia.

Additional genetic variants may be discovered by increasing power using a genomewide meta-analysis of multiple survivor cohorts. Unlike one might expect, it has been shown that with a large enough sample size a self-reported phenotype, including high cholesterol, is very well able to serve as outcome in GWAS^(58, 59). So for this strategy cohorts with either self-reported or lab data could both be included.

This study for the first time replicated the protective variant rs676210 in the apoB encoding APOB gene, which was identified in the aforementioned ALL study⁽⁴⁹⁾, in a more heterogeneous cohort of adult CCS. This general population variant alters the structure of apoB in a way that decreases oxidation of LDL cholesterol⁽⁶⁰⁾. In a large GWAS this variant was associated with lower triglycerides, LDL cholesterol and total cholesterol levels and with higher HDL cholesterol⁽⁵⁰⁾. The two other general population variants replicated in the ALL study, rs2286615 in the BAD gene and rs62079523 in the OGFOD3, were not replicated. BAD encodes a protein that has both a pro-apoptotic function and is involved in the insulin secretion pathway⁽⁶¹⁾. Dysfunction of BAD caused by rs2286615 is thought to have an impact on insulin resistance, and the authors of the study in ALL survivors suggest a subsequent effect on dyslipidemia development. For OGFOD3, which encodes a part of an enzyme involved in iron binding, no specific role in lipid metabolism is known. The authors included it as a candidate gene based on their gene ontology search terms and it increased dyslipidemia risk but the mechanism remained unclear. For the variant in BAD it may be interesting to follow-up insulin-resistant survivors and study whether in this apparent susceptible group rs2286615 carriers develop dyslipidemia faster.

Studying the effect of known polymorphisms in the general population associated with metabolic syndrome components in CCS is interesting in the context of accelerated aging⁽²⁵⁾. In CCS, these genetic predispositions can become relevant at an earlier age. A next step is to determine the effect of multiple general population genes predisposing to dyslipidemia in survivors, based on a polygenic risk score. This may also be relevant because of the polygenic nature of dyslipidemia, where multiple genes have small effects⁽⁶²⁾. This approach was recently performed for another metabolic syndrome component, hypertension⁽⁶³⁾, and for severe obesity (BMI≥40kg/m²)⁽⁶⁴⁾. The study on hypertension revealed that survivors with the lowest polygenic risk score still had a higher hypertension risk compared to the general population, suggesting treatment-related risk. Survivors with the highest polygenic risk score had a 2.5 fold higher hypertension risk compared to the lowest, and the authors determined that this polygenic risk score contributed about one quarter of the hypertension risk. The study on severe obesity revealed that including the polygenic risk score to risk prediction models based on cancer treatment and lifestyle factors greatly improved performance of the model, identifying four times more high-risk survivors. We are currently working on such a polygenic risk score for dyslipidemia. In addition to further unraveling the genetic susceptibility to dyslipidemia development in survivors, we will focus on

the other metabolic syndrome components as phenotypes, metabolic syndrome as a whole, but also other relevant outcomes including GHD, sarcopenia and frailty. Apart from two GWASs among survivors, which revealed genetic polymorphisms potentially modifying obesity after CrRT exposure^(65, 66), these studies are so far not available. These analyses are currently ongoing in the DCCSS-LATER cohort and in collaboration with international study groups.

Towards personalized follow-up for individual childhood cancer survivors worldwide

Research strategies to fill gaps of knowledge as discussed here, and summarized in Table 1, can continuously aid in assessing personalized risk for survivors. For metabolic sequelae it is of importance to take into account all components of metabolic syndrome, and this can be taken one step further. Several studies have observed the co-occurrence of multiple late effects, emphasizing the total burden of disease that survivors sometimes face⁽⁶⁷⁻⁶⁹⁾. Endocrinopathies after hypothalamic-pituitary damage may explain co-occurrence of metabolic syndrome and other endocrine sequelae, such as growth failure, infertility, osteoporosis and hypothyroidism. Furthermore, a detailed analysis in the SILIFE cohort showed that metabolic syndrome components often co-occurred with pulmonary function deficits, secondary neoplasms, hearing loss and recurrent infections⁽⁶⁸⁾. In future research, these mechanisms leading to a high cumulative late effects burden, why certain effects cluster, and how this affects quality of life, may further be unraveled. Another step in this regard includes, for therapies that come with a high late effects burden, attempts to adjust treatment protocols, while maintaining anti-tumor efficacy. Replacement of CrRT in ALL with improved systemic and intrathecal chemotherapy, and reduction of anthracyclines in renal tumors are successful examples of this⁽⁷⁰⁻⁷²⁾. In addition, innovative abdominal radiotherapy techniques, using lower and more targeted dose (highly conformal target-volume delineation, Intensity-Modulated Radiation Therapy and Volumetric Modulated Arc Therapy), were recently shown to decrease scatter to abdominal organs and are currently implemented^(73, 74). Studies have shown hopeful improvements in health outcome and life expectancy for more recently treated survivors⁽⁷⁵⁻⁷⁷⁾. On the other hand, late metabolic consequences of novel therapies such as immunotherapy remain to be investigated⁽⁷⁸⁾.

Survivors in The Netherlands can benefit from coordinated follow-up care that is delivered in specialized late effects clinics, according to comprehensive national guidelines⁽⁷⁹⁾. The survey described in *Chapter 7*, which explored perspectives on survivor care and research priorities among health care professions from several continents, revealed agreement that not only physical care, but also psychological support is essential. Furthermore, several obstacles were identified that physicians in

other countries face in setting up structured CCS care. Collaborations, support and sharing of knowledge and resources are required to make survivor follow-up a success story worldwide.

Goal	How
Personalized risk stratification	Use of <i>dosimetry based data</i> to express radiotherapy related risk of developing metabolic syndrome components, from the period of treatment on, in a longitudinal setting.
	<u>Genetic predisposition</u> to metabolic syndrome components and the syndrome as a whole. This concerns identification of <u>survivor-specific genetic variants</u> related to and interacting with cancer treatment, for which power can be enhanced by performing <u>genome-wide</u> <u>meta-analyses of multiple cohorts</u> . Additionally, <u>polygenic risk scores</u> based on relevant variants in the <u>general population</u> can be determined.
	Study the risk of currently applied <i><u>novel therapies</u></i> to determine the risk for more recently treated survivors
	Development of <u>prediction models</u> that calculate the combined risk of multiple factors, including therapies, genetic factors and comorbidities including growth hormone deficiency and hypogonadism
Improved diagnosis of metabolic syndrome	Continuous follow-up in <i>aging survivors</i> to study whether vascular ultrasound and adiponectin are useful diagnostic tools
	Explore the potential of the <i>biomarkers</i> adiponectin, uric acid, apoB, hsCRP and leptin by studying their value as <i>additional metabolic syndrome component</i>
Unravel pathophysiological mechanisms	Use of <i>dosimetry data</i> to further refine how organs and organ parts, including the brain and adipose tissue, are affected by abdominal and cranial irradiation
	<i>Longitudinal follow-up</i> from diagnosis, end of treatment towards long-term survivorship, to gain insight in the role of treatment-related determinants as well as comorbidities in the development of metabolic syndrome components
	To determine the <i>biological mechanism</i> by which anthracyclines and platinum chemotherapy may disguise overweight
	Functional <i>in vitro or in vivo</i> evidence that the variant rs114017774 influences the expression of <i>LRRTM4</i> or <i>CTNNA2</i> , and that this leads to dyslipidemia
Implementation in clinical care	<i>Dosimetry based</i> and <i>prediction model based surveillance</i> to enable early intervention for those at risk and to avoid unnecessary screening
	Development of a <u>surveillance guideline</u> for metabolic syndrome
	Surveillance including early <u>medical and lifestyle intervention</u> , and follow-up to find out

Table 1. Summary of gaps of knowledge and future perspectives regarding metabolic syndrome in childhood cancer survivors

REFERENCES

- de Vathaire F, El-Fayech C, Ben Ayed FF, et al. Radiation dose to the pancreas and risk of diabetes mellitus in childhood cancer survivors: a retrospective cohort study. The Lancet Oncology. 2012;13(10):1002-10.
- Friedman DN, Moskowitz CS, Hilden P, et al. Radiation Dose and Volume to the Pancreas and Subsequent Risk of Diabetes Mellitus: A Report from the Childhood Cancer Survivor Study. J Natl Cancer Inst. 2020;112(5):525-32.
- 3. Chemaitilly W, Mertens AC, Mitby P, et al. Acute ovarian failure in the childhood cancer survivor study. J Clin Endocrinol Metab. 2006;91(5):1723-8.
- 4. van Dorp W, Blijdorp K, Laven JS, et al. Decreased ovarian function is associated with obesity in very long-term female survivors of childhood cancer. Eur J Endocrinol. 2013;168(6):905-12.
- Roeters van Lennep JE, Heida KY, Bots ML, Hoek A, collaborators of the Dutch Multidisciplinary Guideline Development Group on Cardiovascular Risk Management after Reproductive D. Cardiovascular disease risk in women with premature ovarian insufficiency: A systematic review and meta-analysis. Eur J Prev Cardiol. 2016;23(2):178-86.
- 6. Anagnostis P, Christou K, Artzouchaltzi AM, et al. Early menopause and premature ovarian insufficiency are associated with increased risk of type 2 diabetes: a systematic review and metaanalysis. Eur J Endocrinol. 2019;180(1):41-50.
- 7. Wellons M. Cardiovascular disease and primary ovarian insufficiency. Semin Reprod Med. 2011;29(4):328-41.
- van Waas M, Neggers SJ, Raat H, van Rij CM, Pieters R, van den Heuvel-Eibrink MM. Abdominal radiotherapy: a major determinant of metabolic syndrome in nephroblastoma and neuroblastoma survivors. PLoS One. 2012;7(12):e52237.
- 9. Jo SK, Seol MA, Park HR, Jung U, Roh C. Ionising radiation triggers fat accumulation in white adipose tissue. Int J Radiat Biol. 2011;87(3):302-10.
- Sun K, Tordjman J, Clement K, Scherer PE. Fibrosis and adipose tissue dysfunction. Cell Metab. 2013;18(4):470-7.
- El-Fayech C, Haddy N, Allodji RS, et al. Cerebrovascular Diseases in Childhood Cancer Survivors: Role of the Radiation Dose to Willis Circle Arteries. Int J Radiat Oncol Biol Phys. 2017;97(2):278-86.
- 12. Shrestha S, Bates JE, Liu Q, et al. Radiation therapy related cardiac disease risk in childhood cancer survivors: Updated dosimetry analysis from the Childhood Cancer Survivor Study. Radiother Oncol. 2021;163:199-208.
- 13. Cohen-Cutler S, Olch A, Wong K, et al. Surveillance for radiation-related late effects in childhood cancer survivors: The impact of using volumetric dosimetry. Cancer Med. 2021;10(3):905-13.
- Chow EJ, Chen Y, Hudson MM, et al. Prediction of Ischemic Heart Disease and Stroke in Survivors of Childhood Cancer. J Clin Oncol. 2018;36(1):44-52.
- 15. Armstrong GT, Oeffinger KC, Chen Y, et al. Modifiable risk factors and major cardiac events among adult survivors of childhood cancer. J Clin Oncol. 2013;31(29):3673-80.
- Friedman DN, Tonorezos ES, Cohen P. Diabetes and Metabolic Syndrome in Survivors of Childhood Cancer. Horm Res Paediatr. 2019;91(2):118-27.
- Lipshultz ER, Chow EJ, Doody DR, et al. Cardiometabolic Risk in Childhood Cancer Survivors: A Report from the Children's Oncology Group. Cancer Epidemiol Biomarkers Prev. 2022;31(3):536-42.
- Hoffman MC, Mulrooney DA, Steinberger J, Lee J, Baker KS, Ness KK. Deficits in physical function among young childhood cancer survivors. J Clin Oncol. 2013;31(22):2799-805.

- Grundy SM, Cleeman JI, Daniels SR, et al. Diagnosis and management of the metabolic syndrome: an American Heart Association/National Heart, Lung, and Blood Institute Scientific Statement. Circulation. 2005;112(17):2735-52.
- 20. van Kalsbeek RJ, van der Pal HJH, Hjorth L, et al. The European multistakeholder PanCareFollowUp project: novel, person-centred survivorship care to improve care quality, effectiveness, cost-effectiveness and accessibility for cancer survivors and caregivers. Eur J Cancer. 2021;153:74-85.
- 21. Chow EJ, Doody DR, Di C, et al. Feasibility of a behavioral intervention using mobile health applications to reduce cardiovascular risk factors in cancer survivors: a pilot randomized controlled trial. J Cancer Surviv. 2021;15(4):554-63.
- 22. Ketterl TG, Ballard S, Bradford MC, et al. Feasibility and acceptability of a home-based resistance training intervention in adolescent and young adult hematopoietic cell transplant survivors. Pediatr Blood Cancer. 2021;68(11):e29206.
- 23. Bluher M. Obesity: global epidemiology and pathogenesis. Nat Rev Endocrinol. 2019;15(5):288-98.
- 24. Ness KK, Armstrong GT, Kundu M, Wilson CL, Tchkonia T, Kirkland JL. Frailty in childhood cancer survivors. Cancer. 2015;121(10):1540-7.
- 25. Armenian SH, Gibson CJ, Rockne RC, Ness KK. Premature Aging in Young Cancer Survivors. J Natl Cancer Inst. 2019;111(3):226-32.
- 26. van Atteveld JE, de Winter DTC, Pluimakers VG, et al. Frailty and sarcopenia within the earliest national Dutch childhood cancer survivor cohort: a DCCSS-LATER 2 Study. Submitted. 2022.
- 27. Barter PJ, Ballantyne CM, Carmena R, et al. Apo B versus cholesterol in estimating cardiovascular risk and in guiding therapy: report of the thirty-person/ten-country panel. J Intern Med. 2006;259(3):247-58.
- Johannesen CDL, Mortensen MB, Langsted A, Nordestgaard BG. Apolipoprotein B and Non-HDL Cholesterol Better Reflect Residual Risk Than LDL Cholesterol in Statin-Treated Patients. J Am Coll Cardiol. 2021;77(11):1439-50.
- 29. Sniderman AD, Thanassoulis G, Glavinovic T, et al. Apolipoprotein B Particles and Cardiovascular Disease: A Narrative Review. JAMA Cardiol. 2019;4(12):1287-95.
- 30. Alberti KG, Eckel RH, Grundy SM, et al. Harmonizing the metabolic syndrome: a joint interim statement of the International Diabetes Federation Task Force on Epidemiology and Prevention; National Heart, Lung, and Blood Institute; American Heart Association; World Heart Federation; International Atherosclerosis Society; and International Association for the Study of Obesity. Circulation. 2009;120(16):1640-5.
- 31. National Cholesterol Education Program Expert Panel on Detection E, and Treatment of High Blood Cholesterol in Adults (Adult Treatment Panel III). Third Report of the National Cholesterol Education Program (NCEP) Expert Panel on Detection, Evaluation, and Treatment of High Blood Cholesterol in Adults (Adult Treatment Panel III). 2002.
- 32. World Health Organization. Hypertension 2021 [Available from: https://www.who.int/news-room/fact-sheets/detail/hypertension.
- 33. Vatanen A, Sarkola T, Ojala TH, et al. Radiotherapy-related arterial intima thickening and plaque formation in childhood cancer survivors detected with very-high resolution ultrasound during young adulthood. Pediatr Blood Cancer. 2015;62(11):2000-6.
- 34. Broberg O, Ora I, Wiebe T, Weismann CG, Liuba P. Characterization of Cardiac, Vascular, and Metabolic Changes in Young Childhood Cancer Survivors. Front Pediatr. 2021;9:764679.
- 35. Krawczuk-Rybak M, Tomczuk-Ostapczuk M, Panasiuk A, Goscik E. Carotid intima-media thickness in young survivors of childhood cancer. J Med Imaging Radiat Oncol. 2017;61(1):85-92.

- Stelwagen J, Lubberts S, Steggink LC, et al. Vascular aging in long-term survivors of testicular cancer more than 20 years after treatment with cisplatin-based chemotherapy. Br J Cancer. 2020;123(11):1599-607.
- 37. Lopez-Melgar B, Fernandez-Friera L, Oliva B, et al. Short-Term Progression of Multiterritorial Subclinical Atherosclerosis. J Am Coll Cardiol. 2020;75(14):1617-27.
- 38. Mach F, Baigent C, Catapano AL, et al. 2019 ESC/EAS Guidelines for the management of dyslipidaemias: lipid modification to reduce cardiovascular risk. Eur Heart J. 2019.
- 39. Savastano S, Di Somma C, Barrea L, Colao A. The complex relationship between obesity and the somatropic axis: the long and winding road. Growth Horm IGF Res. 2014;24(6):221-6.
- 40. Darzy KH, Shalet SM. Hypopituitarism following Radiotherapy Revisited. Endocr Dev. 2009;15:1-24.
- 41. Sherlock M, Toogood AA. Sensitivity of anterior pituitary hormones to irradiation. Expert Rev Endocrinol Metab. 2006;1(5):633-49.
- 42. Crowne E, Gleeson H, Benghiat H, Sanghera P, Toogood A. Effect of cancer treatment on hypothalamic–pituitary function. The Lancet Diabetes & Endocrinology. 2015;3(7):568-76.
- Inaba H, Yang J, Kaste SC, et al. Longitudinal changes in body mass and composition in survivors of childhood hematologic malignancies after allogeneic hematopoietic stem-cell transplantation. J Clin Oncol. 2012;30(32):3991-7.
- 44. Brouwer CA, Gietema JA, Vonk JM, et al. Body mass index and annual increase of body mass index in long-term childhood cancer survivors; relationship to treatment. Support Care Cancer. 2012;20(2):311-8.
- 45. Guida JL, Ahles TA, Belsky D, et al. Measuring Aging and Identifying Aging Phenotypes in Cancer Survivors. J Natl Cancer Inst. 2019;111(12):1245-54.
- 46. Singal PK, Iliskovic N. Doxorubicin-induced cardiomyopathy. N Engl J Med. 1998;339(13):900-5.
- 47. D'Lugos AC, Fry CS, Ormsby JC, et al. Chronic doxorubicin administration impacts satellite cell and capillary abundance in a muscle-specific manner. Physiol Rep. 2019;7(7):e14052.
- 48. Conte E, Bresciani E, Rizzi L, et al. Cisplatin-Induced Skeletal Muscle Dysfunction: Mechanisms and Counteracting Therapeutic Strategies. Int J Mol Sci. 2020;21(4).
- England J, Drouin S, Beaulieu P, et al. Genomic determinants of long-term cardiometabolic complications in childhood acute lymphoblastic leukemia survivors. BMC Cancer. 2017;17(1):751.
- 50. Teslovich TM, Musunuru K, Smith AV, et al. Biological, clinical and population relevance of 95 loci for blood lipids. Nature. 2010;466(7307):707-13.
- 51. Liu DJ, Peloso GM, Yu H, et al. Exome-wide association study of plasma lipids in >300,000 individuals. Nat Genet. 2017;49(12):1758-66.
- 52. Cano-Gamez E, Trynka G. From GWAS to Function: Using Functional Genomics to Identify the Mechanisms Underlying Complex Diseases. Front Genet. 2020;11:424.
- 53. Siddiqui TJ, Tari PK, Connor SA, et al. An LRRTM4-HSPG complex mediates excitatory synapse development on dentate gyrus granule cells. Neuron. 2013;79(4):680-95.
- 54. Cook SA, Bronson RT, Donahue LR, Ben-Arie N, Davisson MT. Cerebellar deficient folia (cdf): a new mutation on mouse chromosome 6. Mamm Genome. 1997;8(2):108-12.
- 55. Reed DR, Lawler MP, Tordoff MG. Reduced body weight is a common effect of gene knockout in mice. BMC Genet. 2008;9:4.
- 56. International Mouse Phenotyping Consortium. REG3G knockout mouse phenotype [Available from: https://www.mousephenotype.org/data/genes/MGI:109406.

- 57. Sebastian A, Jung P, Krause-Utz A, Lieb K, Schmahl C, Tuscher O. Frontal dysfunctions of impulse control a systematic review in borderline personality disorder and attention-deficit/ hyperactivity disorder. Front Hum Neurosci. 2014;8:698.
- 58. Tung JY, Do CB, Hinds DA, et al. Efficient replication of over 180 genetic associations with self-reported medical data. PLoS One. 2011;6(8):e23473.
- 59. Day FR, Hinds DA, Tung JY, et al. Causal mechanisms and balancing selection inferred from genetic associations with polycystic ovary syndrome. Nat Commun. 2015;6:8464.
- 60. Makela KM, Seppala I, Hernesniemi JA, et al. Genome-wide association study pinpoints a new functional apolipoprotein B variant influencing oxidized low-density lipoprotein levels but not cardiovascular events: AtheroRemo Consortium. Circ Cardiovasc Genet. 2013;6(1):73-81.
- 61. Danial NN, Walensky LD, Zhang CY, et al. Dual role of proapoptotic BAD in insulin secretion and beta cell survival. Nat Med. 2008;14(2):144-53.
- 62. Brunham LR, Trinder M. Polygenic risk scores for the diagnosis and management of dyslipidemia. Curr Opin Endocrinol Diabetes Obes. 2022;29(2):95-100.
- 63. Sapkota Y, Li N, Pierzynski J, et al. Contribution of Polygenic Risk to Hypertension Among Long-Term Survivors of Childhood Cancer. JACC CardioOncol. 2021;3(1):76-84.
- 64. Sapkota Y, Qiu W, Dixon SB, et al. Genetic risk score enhances the risk prediction of severe obesity in adult survivors of childhood cancer. Nat Med. 2022;28(8):1590-8.
- 65. Wilson CL, Liu W, Yang JJ, et al. Genetic and clinical factors associated with obesity among adult survivors of childhood cancer: A report from the St. Jude Lifetime Cohort. Cancer. 2015;121(13):2262-70.
- 66. Richard MA, Brown AL, Belmont JW, et al. Genetic variation in the body mass index of adult survivors of childhood acute lymphoblastic leukemia: A report from the Childhood Cancer Survivor Study and the St. Jude Lifetime Cohort. Cancer. 2021;127(2):310-8.
- 67. Oeffinger KC, Mertens AC, Sklar CA, et al. Chronic Health Conditions in Adult Survivors of Childhood Cancer. N Engl J Med. 2006;355(15):1572-82.
- 68. Bhakta N, Liu Q, Ness KK, et al. The cumulative burden of surviving childhood cancer: an initial report from the St Jude Lifetime Cohort Study (SJLIFE). Lancet. 2017;390(10112):2569-82.
- 69. Geenen MM, Cardous-Ubbink MC, Kremer LCM, et al. Medical Assessment of Adverse Health Outcomes in Long-term Survivors of Childhood Cancer. JAMA. 2007;297(24):2705-15.
- Clarke M, Gaynon P, Hann I, et al. CNS-directed therapy for childhood acute lymphoblastic leukemia: Childhood ALL Collaborative Group overview of 43 randomized trials. J Clin Oncol. 2003;21(9):1798-809.
- 71. Jeha S, Pei D, Choi J, et al. Improved CNS Control of Childhood Acute Lymphoblastic Leukemia Without Cranial Irradiation: St Jude Total Therapy Study 16. J Clin Oncol. 2019;37(35):3377-91.
- 72. Pritchard-Jones K, Bergeron C, de Camargo B, et al. Omission of doxorubicin from the treatment of stage II-III, intermediate-risk Wilms' tumour (SIOP WT 2001): an open-label, non-inferiority, randomised controlled trial. Lancet. 2015;386(9999):1156-64.
- 73. Mul J, Seravalli E, Bosman ME, et al. Estimated clinical benefit of combining highly conformal target volumes with Volumetric-Modulated Arc Therapy (VMAT) versus conventional flank irradiation in pediatric renal tumors. Clin Transl Radiat Oncol. 2021;29:20-6.
- 74. Janssens GO, Melchior P, Mul J, et al. The SIOP-Renal Tumour Study Group consensus statement on flank target volume delineation for highly conformal radiotherapy. Lancet Child Adolesc Health. 2020;4(11):846-52.
- 75. Gibson TM, Mostoufi-Moab S, Stratton KL, et al. Temporal patterns in the risk of chronic health conditions in survivors of childhood cancer diagnosed 1970-99: a report from the Childhood Cancer Survivor Study cohort. Lancet Oncol. 2018;19(12):1590-601.

- 76. Yeh JM, Ward ZJ, Chaudhry A, et al. Life Expectancy of Adult Survivors of Childhood Cancer Over 3 Decades. JAMA Oncol. 2020;6(3):350-7.
- 77. Kilsdonk E, van Dulmen-den Broeder E, van Leeuwen FE, et al. Late Mortality in Childhood Cancer Survivors according to Pediatric Cancer Diagnosis and Treatment Era in the Dutch LATER Cohort. Cancer Invest. 2022:1-12.
- 78. Gebauer J, Higham C, Langer T, Denzer C, Brabant G. Long-Term Endocrine and Metabolic Consequences of Cancer Treatment: A Systematic Review. Endocr Rev. 2019;40(3):711-67.
- 79. Stichting Kinderoncologie Nederland. Richtlijn follow-up na kinderkanker [31 July 2020]. Available from: https://www.skion.nl/voor-patienten-en-ouders/late-effecten/533/richtlijnfollow-up-na-kinderkanker/.

SUMMARY

Over the past decades, cure of children with cancer has become increasingly feasible and current survival rates exceed eighty percent. However, this is in fact only partly a success story as excess multimorbidity and mortality due to cancer treatment are observed in survivors throughout adulthood. Metabolic syndrome, which is a risk factor for early death due to cardio- and cerebrovascular disease and diabetes mellitus, is more prevalent among childhood cancer survivors (CCS), even at a very young age. Therefore, survivors need to be closely monitored, but metabolic syndrome and the separate components can be underdiagnosed with the current criteria in (subgroups of) survivors. Additionally, knowledge on national prevalence as well as on clinical and genetic risk factors of metabolic syndrome and its separate components remains incomplete. This thesis project therefore aimed to identify better methods and biomarkers to increase metabolic syndrome diagnosis, and to describe prevalence and clinical and genetic determinants of metabolic syndrome and the components overweight and dyslipidemia.

Our literature review on prevalence and determinants of metabolic syndrome in CCS, identified that a history of brain tumor, cranial or abdominal radiotherapy, and alkylating chemotherapy can be considered as established variables, associated with occurrence of metabolic syndrome. Most studies so far, however, are based on small and selected cohorts, have insufficient treatment data, are questionnaire-based, or have short follow-up. Therefore, the national prevalence of metabolic syndrome and the separate components was unclear, and for several hypothesized risk factors, including overweight at diagnosis and treatment with corticosteroids, evidence was inconclusive. Additionally, more reliable metabolic syndrome (bio)markers are required, most importantly because waist circumference underestimates overweight in survivors who received abdominal radiotherapy. Consequently, this may underdiagnose metabolic syndrome. Also, a potential role for genetic susceptibility is suspected, as there can be a difference in occurrence and severity of metabolic syndrome and separate components in comparably treated survivors. However, so far, no replicated genetic polymorphisms have been found in CCS.

In our single-center cohort study among 103 survivors of nephroblastoma and neuroblastoma, we observed that metabolic syndrome was present in 14% of survivors, already at a young age, and that 33% had at least two components. Abdominal radiotherapy was significantly associated with metabolic syndrome and the components high triglycerides, low high-density-lipoprotein cholesterol and hypertension. We also noticed that almost none of the survivors had the metabolic

syndrome component "increased waist circumference", which is remarkable when taking the general population into account. This may be explained by the historic administration of abdominal radiotherapy, and it illustrates the need for additional (bio)markers to enhance metabolic syndrome diagnosis. In this study we observed that four serum biomarkers may be useful for this purpose: adiponectin, low-densitylipoprotein cholesterol, apolipoprotein B (apoB) and uric acid. We also found that vascular ultrasonography, which can detect early vascular consequences of metabolic syndrome, was not of additional benefit to improve metabolic syndrome diagnosis.

In our systematic literature review on metabolic syndrome biomarkers we identified five out of nine reviewed biomarkers as useful for the diagnosis and prediction of metabolic syndrome, i.e., uric acid, leptin, high sensitivity C-reactive protein, adiponectin and apoB. The level of evidence we found for these biomarkers was high or moderate, mainly based on general population studies. Studies among survivors are scarce and often had high risk of bias, so assessment of the use of these biomarkers in CCS is still required.

We described the methodology of the DCCSS-LATER Metabolic syndrome study on prevalence and clinical and genetic determinants of metabolic syndrome and the separate components, and on the diagnostic value of DXA scan and biomarkers. This is a nationwide Dutch cohort of the first treated (1963-2002) CCS, consisting of 2,338 long-term CCS. In this thesis we presented the first results of the overweight component. Overweight occurred in almost half of all long-term CCS. Compared to general population data, there was a significantly higher prevalence for overweight among women aged 50+ and for morbid obesity among men aged 50+. Overweight, and in particular obesity and morbid obesity, were most prevalent among survivors of central nervous system tumors. Overweight at diagnosis, cranial radiotherapy and growth hormone deficiency were associated with long-term overweight, but the historic use of corticosteroids was not. We also compared prevalence of overweight using several assessment methods. We found that waist-circumference-based methods classified a substantial number of survivors with normal BMI (i.e., incorrectly classified as healthy weight by BMI) as overweight. DXA scan was the most sensitive method. It identified overweight in an additional 30% of survivors, and even 40% in the subgroup of abdominally irradiated survivors. This was particularly the case in survivors treated with abdominal radiotherapy, total body irradiation, anthracyclines and platinum chemotherapy. In this study we also observed that high molecular weight-adiponectin did not seem to add diagnostic value in assessing overweight.

In the first genome-wide association study (GWAS) on the metabolic syndrome component dyslipidemia in three large cohorts of CCS – the CCSS Original, St

Jude Lifetime (SJLIFE) and CCSS Expansion cohorts - none of nine SNPs that were identified in discovery analysis of the entire cohort, replicated. Next, we identified rs114017774 on chromosome 2 as a potential genetic variant for dyslipidemia, specifically in cranially irradiated CCS. Meta-analysis of the CCSS Original and SJLIFE cohorts revealed a pooled odds ratio of 11.30 (95% confidence interval 5.03-25.40, $p=4.5x10^{-9}$), but this locus did not individually replicate in the CCSS Expansion or SJLIFE cohorts. This genetic variant had an interaction with cranial radiotherapy (genotypexCRT OR=14.61, 95%CI=4.59-56.55) independent of BMI. It is not reported in large general population GWASs on dyslipidemia, potentially suggesting a survivor-specific effect. Based on available sources we hypothesized that this genetic variant could potentially alter LRRTM4 or CTNNA2 (two genes expressed in the brain and involved in nervous system development and functioning) expression or function, making survivors more susceptible to hypothalamic damage caused by cranial radiotherapy, thereby increased dyslipidemia risk. Additional replication and functional validation are needed to further explore the relevance of this variant. Furthermore, we replicated the protective variant rs676210 in the APOB gene, which was identified in a previous study among survivors of acute lymphoblastic leukemia (ALL), in the CCSS Original cohort. Replication was significant in the ALL subgroup as well as the total cohort, indicating that this general population variant is relevant in survivors as well.

Surveillance standards for CCS may differ across the world, due to the (un)availability of guidelines, differences in cultural values, and discrepancies in available resources and expertise. We explored perspectives on survivor care and research priorities among clinicians from several continents and show that, regardless of country of origin, psychological support, next to several aspects of physical care, was regarded as a priority. This emphasizes the need for a well-organized approach, covering not only physical but also psychological care. Furthermore, the survey identified several obstacles that physicians in other countries face in setting up structured care for CCS.

NEDERLANDSE SAMENVATTING

De afgelopen decennia werd het in toenemende mate mogelijk kinderen met kanker te genezen, waardoor de overleving momenteel meer dan tachtig procent bedraagt. Echter, dit is slechts ten dele een succesverhaal, aangezien multimorbiditeit en mortaliteit overmatig voorkomen bij overlevers van kinderkanker (survivors). Dit is ten gevolge van de kankerbehandeling en treedt niet alleen in de eerste jaren na genezing op, maar loopt door tot de volwassen leeftijd. Metabool syndroom, een risicofactor voor vroege sterfte door hart- en vaatziekten en diabetes mellitus, komt vaker voor bij survivors, en al op jonge leeftijd. Het is daarom van belang goed toezicht te houden op het ontwikkelen hiervan, maar de diagnose van metabool syndroom en de losse componenten kan met de standaard criteria worden gemist in (subgroepen van) survivors. Daarbij komt dat de kennis op het gebied van nationale prevalentie en van klinische en genetische risicofactoren voor metabool syndroom en de losse componenten nog incompleet is. Het doel van dit proefschrift was om betere methodes en biomarkers te identificeren voor het diagnosticeren van metabool syndroom, en om de prevalentie en klinische en genetische determinanten te beschrijven van metabool syndroom en van de componenten overgewicht en dyslipidemie.

Onze literatuur review over prevalentie en determinanten van metabool syndroom in survivors toonde dat van een hersentumor, schedel- of buikbestraling en alkylerende chemotherapie in de voorgeschiedenis een duidelijke associatie is vastgesteld met het ontstaan van metabool syndroom. Niettemin zijn de meeste studies tot dusverre gebaseerd op kleine en geselecteerde cohorten, beschikken deze over onvoldoende gegevens van de kinderkankerbehandeling, zijn deze gebaseerd op enkel vragenlijsten, en is de follow-up kort. Daarom is de nationale prevalentie van metabool syndroom en van de losse componenten nog onbekend, en voor verschillende gehypothetiseerde risicofactoren, zoals overgewicht bij diagnose en behandeling met corticosteroïden, is het bewijs nog niet sluitend. Ook zijn betrouwbaardere (bio)markers voor metabool syndroom nodig, hoofdzakelijk omdat de buikomtrek een onderschatting geeft van overgewicht in survivors die behandeld zijn met abdominale radiotherapie. Derhalve kan in deze survivors de diagnose metabool syndroom worden gemist. Voorts wordt een mogelijke rol voor genetische predispositie verondersteld, omdat survivors die eenzelfde kankerbehandeling hebben ondergaan in verschillende mate te maken kunnen krijgen met metabool syndroom en de losse componenten. Vooralsnog zijn echter geen gerepliceerde genetische polymorfismen geïdentificeerd in survivors.

In onze single-center cohort studie in 103 survivors van nefroblastoom en neuroblastoom vonden we dat 14% van de survivors metabool syndroom had, al op

jonge leeftijd, en dat 33% ten minste twee metabool syndroom componenten had. Abdominale radiotherapie was significant geassocieerd met metabool syndroom en met de componenten hoge triglyceriden, laag high-density-lipoproteïne cholesterol en hypertensie. We zagen ook dat bijna geen enkele survivor de metabool syndroom component "grote buikomtrek" had, hetgeen opvallend is wanneer we de algemene populatie in ogenschouw nemen. Dit wordt mogelijk verklaard door behandeling met abdominale radiotherapie, en illustreert de behoefte aan betere (bio)markers om metabool syndroom te diagnosticeren. In deze studie observeerden we dat vier serum biomarkers nuttig kunnen zijn voor dit doeleinde: adiponectine, low-densitylipoproteïne cholesterol, apolipoproteïne B (apoB) en urinezuur. We vonden ook dat echografie van bloedvaten, wat vroege afwijkingen aan de vaten ten gevolge van metabool syndroom kan vaststellen, niet van toegevoegde waarde was voor het stellen van de diagnose metabool syndroom.

In onze systematische literatuur review over metabool syndroom biomarkers bleken vijf van de negen onderzochte biomarkers bruikbaar voor de diagnose en predictie van metabool syndroom, te weten urinezuur, leptine, high sensitivity C-reactief proteïne, adiponectine en apoB. Dit bewijs werd beoordeeld als van hoge of gemiddelde kwaliteit, en is vooral gebaseerd op studies in de algemene bevolking. Studies in survivors zijn schaars en hadden vaak een hoog risico op bias. Het is dus nog noodzakelijk om in survivors vast te stellen of deze biomarkers daadwerkelijk bruikbaar zijn.

We beschrijven in dit proefschrift de methodologie van de DCCSS-LATER Metabool syndroom studie, naar de prevalentie en klinische en genetische determinanten van metabool syndroom en van de losse componenten, en naar de diagnostische waarde van de DXA-scan en van biomarkers. Voor deze studie is het nationale cohort opgeroepen van de eerste in Nederland behandelde (tussen 1963 en 2002) survivors, hetgeen heeft geresulteerd in 2.338 deelnemende lange-termijn survivors. Dit proefschrift beschrijft de eerste resultaten van deze studie, op het gebied van de component overgewicht. Bijna de helft van alle lange-termijn survivors had overgewicht. Vergeleken met de algemene bevolking was de prevalentie van overgewicht significant hoger onder vrouwen van vijftig jaar en ouder, en was de prevalentie van morbide obesitas significant hoger onder mannen van vijftig jaar en ouder. Overgewicht, en voornamelijk obesitas en morbide obesitas, kwamen het meest voor onder survivors die een tumor van het centraal zenuwstelsel hadden gehad. Overgewicht bij kankerdiagnose, schedelbestraling en groeihormoondeficiëntie waren geassocieerd met overgewicht op de lange termijn, maar behandeling met corticosteroïden was dat niet. We hebben ook de prevalentie van overgewicht volgens verschillende methodes vergeleken. We vonden dat de methodes die gebruik maken van de buikomtrek overgewicht vaststelden in een substantieel aantal survivors met een normaal BMI (van wie het gewicht dus op basis van het BMI incorrect als gezond was geclassificeerd). DXA-scan was de meest gevoelige methode. Deze methode stelde overgewicht vast in 30% meer survivors, en zelfs in 40% in de subgroep van buikbestraalde survivors. Deze discrepantie werd eveneens vaker gezien in survivors die behandeld waren met totale lichaamsbestraling, anthracyclines en platinumhoudende chemotherapie. We zagen in deze studie ook dat high molecular weight-adiponectine geen toegevoegde waarde leek te hebben om overgewicht te diagnosticeren.

In de eerste genoombrede associatiestudie (GWAS) naar de metabool syndroom component dyslipidemie in drie grote survivor cohorten - de CCSS Original, St Jude Lifetime (SJLIFE) en CCSS Expansion cohorten - repliceerde geen van de negen SNPs die in de discovery analyse van het gehele CCSS Original cohort waren geïdentificeerd. Vervolgens identificeerden we rs114017774, gelegen op chromosoom 2, als een mogelijke genetische variant bijdragend aan het ontwikkelen van dyslipidemie, specifiek in schedelbestraalde survivors. Meta-analyse van de CCSS Original en SILIFE cohorten toonde een gepoolde odds ratio van 11,30 (95% betrouwbaarheidsinterval 5,03-25,40 p=4,5x10⁻⁹), maar dit locus repliceerde niet los in de CCSS Expansion en SILIFE cohorten. Aanwezigheid van deze genetische variant toonde een interactie met craniale radiotherapie (genotypexCRT OR=14,61, 95%BI=4,59-56,55), welke onafhankelijk was van BMI. Deze variant wordt niet gerapporteerd in grote GWASs naar dyslipidemie in de algemene bevolking, hetgeen een survivor-specifiek effect kan suggereren. Gebaseerd op reeds beschikbare bronnen hypothetiseren we dat deze genetische variant mogelijk de expressie of functie verandert van LRRTM4 of CTNNA2 (twee genen die tot expressie komen in het brein en betrokken zijn bij ontwikkeling en functie van het zenuwstelsel), waarbij survivors vatbaarder zijn voor schade aan de hypothalamus ten gevolge van craniale radiotherapie, en daarbij een verhoogd risico hebben op dyslipidemie. Verdere replicatie en functionele validatie zijn noodzakelijk om de mogelijke relevantie van deze genetische variant verder te exploreren. In deze studie repliceerden we eveneens in het CCSS Original cohort de beschermende variant rs676210 in het APOB gen, welke eerder geassocieerd met dyslipidemie was bevonden in een studie onder survivors van acute lymfatische leukemie (ALL). Replicatie was significant in zowel de ALL subgroep als in het gehele cohort, wat indiceert dat deze variant uit de algemene bevolking tevens relevant is in survivors.

De standaarden voor follow-up van survivors kunnen wereldwijd verschillen, door het al dan niet beschikbaar zijn van richtlijnen hiervoor, door verschillen in culturele waardes, en door discrepanties in beschikbare middelen en expertise. We verkenden perspectieven op zorg voor survivors en onderzoeksprioriteiten onder zorgmedewerkers afkomstig van verschillende continenten en zagen dat, ongeacht land van oorsprong, psychologische ondersteuning, naast verschillende aspecten van fysieke zorg, als prioriteit werd bestempeld. Dit onderschrijft de noodzaak van een goed georganiseerde aanpak, waarbij niet alleen aandacht voor somatische klachten is, maar ook voor psychologische zorg. Deze enquête stelde ook obstakels vast waar clinici in verschillende landen mee te maken hebben in hun poging gestructureerde zorg voor survivors op te zetten.
CURRICULUM VITAE

Vincent Pluimakers was born in Utrecht, The Netherlands, on February 25, 1991. He graduated from secondary school (Christelijk Gymnasium, Utrecht) in 2008. This was followed by his study at the faculty of medicine of Utrecht University and the University Medical Center Utrecht, for which he obtained his degree in 2016. After working as an internal medicine resident not in training for one year (Gelderse Vallei, Ede) he joined the research group of prof. dr. van den Heuvel-



Eibrink for his PhD project at the Princess Maxima Center for Pediatric Oncology, also under supervision of dr. Neggers and dr. Janssens. The results of his research are presented in this thesis. In September 2021, he started his residency training in internal medicine in Gelre, Apeldoorn and University Medical Center Utrecht. He lives in Utrecht with his wife and their two children.

PHD PORTFOLIO

Name:	Vincent Pluimakers	
PhD period:	March 2017 – December 2022	
Research school:	Clinical and Translational Oncology, Utrecht Universit	
Department:	Princess Maxima Center for Pediatric Oncology	
Promotor:	Prof. dr. M.M. van den Heuvel-Eibrink	
Copromotors:	Dr. S.J.C.M.M. Neggers	
	Dr. G.O.R. Janssens	

PhD training	Year	ECTS
Courses		
NFU Basiscursus Regelgeving en Orgnisatie voor Klinisch Onderzoekers	2017	3.0
Clinical and Translational Oncology course, Utrecht University	2017	2.0
Basic Human Genetics, Erasmus MC	2018	1.0
Genome wide association studies, Erasmus MC	2019	3.0
SNPs and human diseases, Erasmus MC	2019	3.0
Writing a scientific paper, Utrecht University	2019	4.0
Practical Biostatistics, Amsterdam UMC	2019	4.0
Statistical learning, Stanford University	2020	4.0
Best practices for writing reproducible code, Utrecht University	2021	1.5
The art of presenting science, Utrecht University	2021	4.0
Regression analysis, Leiden UMC	2021	3.0
Seminars		
Presentations at research seminars of Princess Maxima Center and Erasmus MC	2017-21	0.4
Conferences		
International Symposium on Late Complications after Childhood Cancer, Atlanta	2017	2.0
PanCare, Prague	2018	1.5
International Society of Pediatric Oncology (SIOP), Kyoto	2018	2.0
International Society of Pediatric Oncology (SIOP), Lyon (poster discussion)	2019	2.5
International Society of Pediatric Oncology (SIOP), digital	2020	1.5
American Society of Hematology (ASH), hybrid	2020	2.0
American Society of Hematology (ASH), hybrid (poster presentation)	2021	2.0
Other activities		
Supervision of bachelor honours programme student	2018-20	6.0
Article peer review for Cancer and European Journal of Endocrinology	2020	1.0

LIST OF PUBLICATIONS

THIS THESIS

<u>Pluimakers VG</u>, Van Waas M, Neggers SJCMM, Van den Heuvel-Eibrink MM. Metabolic syndrome as cardiovascular risk factor in childhood cancer survivors. *Critical Reviews in Oncology/Hematology, 2019 Jan;133:129-141*

<u>Pluimakers VG</u>, Van Waas M, (et al.), Van den Heuvel-Eibrink MM, Neggers SJCMM. Metabolic syndrome detection with biomarkers in childhood cancer survivors. *Endocrine Connections, 2020 Jun;9:1-11*

<u>Pluimakers VG</u>, Van Santen SS, (et al.), Van den Heuvel-Eibrink MM, Neggers SJCMM. Can biomarkers be used to improve diagnosis and prediction of metabolic syndrome in childhood cancer survivors? A systematic review. *Obesity Reviews, 2021 Nov;22(11):e13312.*

<u>Pluimakers VG</u>, Fiocco M, (et al.), Van den Heuvel-Eibrink MM, Neggers SJCMM. Metabolic syndrome parameters, determinants, and biomarkers in adult survivors of childhood cancer: protocol for the Dutch Childhood Cancer Survivor Study on Metabolic Syndrome (Dutch LATER METS). *JMIR Research Protocols, 2021 Jan;10(1):e21256*

<u>Pluimakers VG</u>, Van Atteveld JE, (et al.), Van den Heuvel-Eibrink MM, Neggers SJCMM. Prevalence, risk factors and optimal way to determine overweight, obesity and morbid obesity, in the first Dutch cohort of 2,338 very long-term survivors of childhood cancer: a DCCSS-LATER study. *Submitted*

<u>Pluimakers VG</u>, Broer L, (et al.), Yasui Y, Van den Heuvel-Eibrink MM. Genetic susceptibility to treatment related dyslipidemia in adult long-term survivors of childhood cancer: a report from the Childhood Cancer Survivor Study and the St Jude Lifetime cohorts.

In preparation

Bakker ME, <u>Pluimakers VG</u>, (et al.), Ishida Y, Maeda M. Perspectives on follow-up care and research for childhood cancer survivors: results from an international SIOP meet-the-expert questionnaire in Kyoto, 2018.

Japanese Journal of Clinical Oncology, 2021 Oct 5;51(10):1554-1560

OTHER PROJECTS

Van Atteveld JE, De Winter DTC, <u>Pluimakers VG</u>, (et al.), Van den Heuvel-Eibrink MM, Dutch LATER study group. Risk and determinants of low and very low bone mineral density and fractures in a national cohort of Dutch adult childhood cancer survivors (DCCSS-LATER): a cross-sectional study. *Lancet Diabetes & Endocrinology*, 2023 Jan;11(1):21-32

Van Atteveld JE, De Winter DTC, <u>Pluimakers VG</u>, (et al.), Van den Heuvel-Eibrink MM, Dutch LATER study group. Frailty and sarcopenia within the earliest national Dutch childhood cancer survivor cohort (DCCSS-LATER): a cross-sectional study. *Lancet Healthy Longevity, 2023 Apr;4(4):e155-e165*

Bolier M, <u>Pluimakers VG</u>, (et al.), Van den Heuvel-Eibrink MM, Neggers SJCMM. Prevalence and determinants of dyslipidemia in a national cohort of 2,338 Dutch childhood cancer survivors: a DCCS-LATER Study. *Submitted*

Man WH, van Dongen TMA, Venekamp RP, <u>Pluimakers VG</u>, (et al.), Schilder AGM, Bogaert D. Respiratory microbiota predicts clinical disease course of acute otorrhea in children with tympanostomy tubes.

The Pediatric Infectious Disease Journal, 2019 Jun;38(6):e116-e125

DANKWOORD

Utrecht, mei 2023

Hèhè, zes jaar en twee maanden sinds de start mag ik dan toch eindelijk dit laatste onderdeel van mijn proefschrift op papier zetten. Het was een avontuur dat mij bracht in het Erasmus, het UMCU, het WKZ en het indrukwekkende Prinses Máxima Centrum dat haar deuren opende; ik had het voorrecht Atlanta, Praag en het prachtige Kyoto te bezoeken. Het was ook een mentale reis, van mezelf tegenkomen en wedergeboortes, toen ik de taaie aspecten van het onderzoek doen leerde kennen. Er was de eindfase waar ik zelf haast metabool syndroom van kreeg en waarin ik mij bij vlagen Sisyphus met zijn onmogelijke opgave waande. Ik ben blij en opgelucht dat ik nu met gepaste trots door het eindresultaat scroll en kan terugblikken op alle mooie aspecten. Wat ik allemaal heb geleerd over late metabolic consequences of childhood cancer. De toffe mensen waarmee ik heb mogen samenwerken. Mezelf verrijken met kennis in diepgaande cursussen. Informatie die verborgen ligt in een dataset naar het daglicht brengen, een hele dag ploeterend op een script tot het eureka moment waarop alles werkt. Al doende de medische wetenschap een heel klein beetje verder proberen te helpen. Echte weekenden hebben, en de tijd om een keer een koffie te pakken en te bedenken waar je naartoe wil met je leven.

Uiteraard is dit ook en vooral het moment om mensen te bedanken. De komende pagina's zijn voor jullie.

Te beginnen met mijn promotor, prof. dr. Van den Heuvel-Eibrink. Beste Marry, toen onze paden elkaar kruisten na mijn onderzoekstage in het WKZ raakte ik direct door je geënthousiasmeerd, en hoewel het uiteindelijk een heel ander project werd dan we die eerste keer bespraken ben ik blij dat we er toen voor zijn gegaan. Je hebt een inspirerende, tomeloze ambitie en een voor een promovendus ontzettend prettige bereikbaarheid en snelheid van manuscript controles, zelfs op vakantie. Ik waardeer je inzet om met iedere onderzoeker in je groep kansen te zoeken om te komen waar die wil, en je vaardigheden om te netwerken op borrels zijn indrukwekkend. Veel dank voor de leuke dingen die je organiseerde voor de groep, zoals de retreat op Schiermonnikoog, het pizza bakken, de verscheidene dinertjes en de fietstocht door Kyoto. Ik kijk uit naar onze verdere samenwerking.

Mijn co-promotor dr. Neggers, beste Sebastian, bij onze kennismaking beging ik de kolderieke vergissing te vragen of je als co-promotor een soort knecht van Marry was, maar dat heb je me gelukkig vergeven. Dank voor je rake less-is-more commentaren op manuscripten, voor je terloops gebrachte levenslessen en voor je gevleugelde "Komt goed". Je laat mensen zelfstandig werken terwijl je altijd een oogje in het zeil hebt. Je was er op de juiste momenten, om even een ijsje te eten of te lunchen, en vooral ook toen de voortgang van mijn project in het nauw kwam.

Mijn co-promotor dr. Janssens, beste Geert, toen Jenneke en ik je ontmoetten in een restaurant in Kyoto bleek je ons reeds te hebben opgezocht in PubMed en had je geconcludeerd dat er nog wel ruimte voor verbetering was. Ik denk dat we die verbetering zeker hebben waargemaakt in de papers waaraan we samen hebben gewerkt. Dank voor je uitgebreide spoedcursussen radiotherapie. Je scherpe blik op abstracts en manuscripten, om de boodschap zo helder mogelijk op papier te krijgen. Je verfrissende en omdenkende relativeringen, zoals positief zijn over het niet winnen van een prijs, omdat dat betekent dat er nóg beter onderzoek gedaan is voor onze patiënten.

Beste leden van de beoordelingscommissie, prof. dr. Visseren, prof. dr. Van der Lelij, prof. dr. Minnema, prof. dr. Uitterlinden en prof. dr. Van Noesel, hartelijk bedankt dat jullie de tijd hebben genomen mijn manuscript te lezen. Het is een eer dat jullie erover met mij in discussie willen. Dr. Yasui, it is an honor that you are part of the doctoral examination committee. Ik kijk uit naar de verdediging, I look forward to the defense.

Dit kan ik zeggen omdat ik mij geflankeerd weet door twee uit het juiste hout gesneden paranimfen. Jenneke, onze complementaire samenwerking van de afgelopen jaren bezegelen we uiteraard met elkaars paranimfschap. Sinds de memorabele uitputtingsslag naar het congres in Atlanta hebben we alle PhD-pieken en -dalen gedeeld. We pepten elkaar op als we om en om betwijfelden of we ons proefschrift ooit af zouden krijgen. We hadden gedeelde dus halve smart als we ons weer eens hadden laten misleiden door laaghangend fruit. Je hebt een jaloersmakende werkdrive, en als ik verzandde in een R-script hield je me bij de les met een scherp "Maar wat wil je hier nou mee zeggen?". Jordy, toen je je avontuurlijke academische hart volgde naar Australië sprak ik het voornemen uit je als paranimf te vragen als je weer in Nederland zou zijn tijdens mijn verdediging, en ik ben verheugd dat dit is uitgekomen nu we weer vlakbij elkaar wonen. Het zal wel even wennen zijn om in serieuzere context dan vroegah in rokkostuum naast elkaar te staan. Voor de verdere dankbetuiging verwijs ik je naar een paar alinea's verder.

Beste Marjolein, omdat er wat tijd zat tussen het einde van jouw promotietraject en het begin van het mijne hebben we helaas niet veel samengewerkt. Veel dank voor al het werk dat je had verricht aan de stukken waar ik als jouw opvolger mee verder ging, en die nu als hoofdstuk 1 en 2 in mijn proefschrift prijken.

Beste deelnemers aan de LATER-studie, zonder jullie was het grootste deel van dit proefschrift er niet. Het was indrukwekkend om te horen over jullie ervaringen met kinderkanker en late effecten. Jullie motivatie en gedrevenheid om mee te doen aan onderzoek voor toekomstige patiënten en survivors waren een inspiratie.

Veel dank gaat uit naar iedereen die aan de LATER-studie heeft gewerkt. Wat een klus hebben we met zijn allen verricht. In willekeurige volgorde...Maaike, Felice, Gerda, Andrica, Jenneke, Inge, Esmee, Selvetta, Lisa, Shauny, Else, Tarik, Heleen, Kim, Ardine, Marry, Manita, Marjo, Suzanne, Dominique, Luciënne, Manon, Jolande, uiteraard het CB - Margriet, Nynke, Lennart en Anke - voor de coördinatie, maar ook alle planners, poli-assistentes, prikkers, functielaboranten, werkstudenten, andere promovendi, de PI's en teams uit andere ziekenhuizen. Met ons studieteam hebben we drie keer de studielogistiek uitgevogeld op de poli's. We hebben talloze brieven gepost, buizen gestickerd en dagprogramma's gemaakt. We onderzochten deelnemers op een stoel of achter een kamerscherm toen er ruimtegebrek was. Er was het eerste ECG-apparaat waarvan de zuignappen al losschoten voordat de laatste erop zat. We worstelden met vragen over parkeerkosten en eigen risico. We zwoegden op query's van het CB als er een monitor visit aankwam. Er was het pijnlijke realisatiemoment dat we materiaal hadden vernietigd dat bewaard had moeten worden. De deadlines en te halen targets hingen soms als een zwaard van Damocles boven ons hoofd. Tijdens dit alles hielden we de moed erin met elkaar en was er altijd ruimte voor een praatje en een lach en daarom hebben we er elke keer een geoliede machine van weten te maken en de klus geklaard.

Mijn kamergenoten waren ook onmisbaar. We begonnen als de Q02-matties, in een UMCU-uithoek met precaire koffievoorziening tot we gelukkig ons eigen Nespresso apparaat aanschaften, en waar we nog wel eens het hoofd moesten bieden aan een stoel gooiende chirurg met territoriumdrift. Daarna mochten we ons intrede doen in het Máxima, eerst knus op de LATER-poli en vervolgens in het penthouse in het research gebouw. Dank voor de gezelligheid op de kamer, voor de aandacht voor elkaars onderzoeks- en privézaken, voor de frequente vrijmibo's. Tijdens Covid was het samen thuiswerken met Jenneke en Joeri waardevol. Het was leuk, leerzaam en stimulerend om met een aantal van jullie - Selvetta, Jenneke, Demi en Melissa - samen te werken aan papers.

I am grateful to Greg Armstrong, Smita Bhatia, Yutaka Yasui and the other collaborators from the USA for the opportunity to work together on the dyslipidemia GWAS project with the data from your survivor cohorts.

Ik ben alle co-auteurs dankbaar voor het meewerken aan de inhoud van mijn proefschrift, in het bijzonder Caspar Looman en Marta Fiocco voor de uitgebreide, geduldige sessies met uitleg over statistiek, en Linda Broer voor die over GWAS.

De inspanningen van Monique, Radha en Marieke voor het altijd weer vinden van een afspraakmogelijkheid in drukke agenda's mogen niet onbelicht blijven - veel dank hiervoor.

Ik had tot voor kort nooit gedacht een robot te benoemen in mijn dankwoord, maar het gebeurt toch: ik dank de fascinerende AI van Midjourney voor het genereren van het gave plaatje op de omslag.

Dan is het nu tijd om vrienden en familie wat verdiende veren in de reet te steken.

Beste Guaps, veel dank voor alle mooie borrels en feesten, de vakanties, festivals, dicksweekends, clubweekends, de met Mario smash football en FIFA gevulde avondjes, het huisgenoten zijn, het BVC zijn, het losgaan op drum and bass, het losgaan op mijn bruiloft, de interesse in elkaars hoogtepunten en problemen, uiteraard de reeds in de zomer geplande kerstdiners. Deze sinds de studententijd en zelfs al basisschooltijd gedeelde momenten vormen een fijn fundament nu ieder wat meer zijn eigen weg begint te gaan, met een deel nog in Utrecht, terwijl anderen de vleugels hebben uitgeslagen naar Amsterdam, het Gooi en de topografische rariteiten Terkaple en Gronsveld. Ook ik moet wat vaker verstek laten gaan nu ik het burgerleven met jong gezin leid, getuige bijvoorbeeld de geboorte van Bram die zich tijdens ons kerstdiner aandiende. Er gaan nog vele gezellige avonturen volgen, en ik ben overtuigd dat, ondanks Jelles voorspelling, een guapscompleet spoedig weer zal komen.

Wouter, mijn interesse voor het doen van een promotieonderzoek werd aangewakkerd toen ik stage deed in je onderzoeksgroep. Je was de afgelopen jaren altijd enthousiast paraat als vraagbaak voor mijn R-problemen en dat was niet alleen buitengewoon prettig maar ook erg leuk om samen over te sparren. Net zo belangrijk zijn onze privé afspraken met Nynke voor een lekkere borrel en een bordspelletje.

Goede buren, wat ben ik blij dat ik de afgelopen jaren heb leren kennen wat er wordt bedoeld met deze term in het spreekwoord. Skihutfeestjes met de babyfoons op het aanrecht, Squashfligem-avondjes, koken voor elkaar, tuinborrels terwijl we toekijken hoe onze kinderen met elkaar spelen, de praatjes op straat en bij het ophalen van pakketjes waarmee we elkaar overladen - het voelt hier echt als thuiskomen.

Successupporters, het is elke keer weer een genot om met jullie een middag of avond UUUUUU te scanderen op het mooiste plekkie op aard, en na het juichen voor Utregs sambavoetbal, het uiteraard louter bemoedigend toezingen van de tegenstander en het complimenteren van de scheidsrechter met diens voortreffelijke beslissingen, in een staat van catharsis huiswaarts te fietsen, opgeladen voor een nieuwe werkweek.

Beste Thanos, we hebben in de jaren dat ik les bij je had tijdens mijn promotieonderzoek een prachtige muzikale reis afgelegd. Het was heerlijk om samen in een stuk te duiken en tot in detail te bediscussiëren wat de componist bedoeld heeft. Het vormde ook een prettige afleiding tijdens het pittigste stuk van mijn promotietraject. Ik wil zeker spoedig weer contact met je opnemen om het muziek maken te hervatten.

De Apeldoornse collega's die ik de afgelopen anderhalf jaar heb leren kennen dank ik voor het gunnen van voldoende opstarttijd om weer te wennen aan kliniekwerk na een promotietraject. Voor het aanhoren van mijn frustraties over vruchteloze pogingen mijn discussie te schrijven tijdens een nachtdienst en van andere worstelingen met het afronden van mijn proefschrift. Voor de heerlijke werksfeer en de gezellige carpoolritjes. En bovenal voor het Apenshot drinken op de Internistendagen.

Lieve pap en mam, bedankt dat jullie altijd klaar staan voor mij: hulp bij verhuizen en klussen, onderdak toen de verbouwing uit de hand liep, het frequente, flexibele en soms acute oppassen welke voor Maartje en mij mogelijk maken ons werk te doen en wat tijd voor elkaar te maken, de studeerplek op zolder waar ik op "papa"dag de laatste analyses van mijn onderzoek uitvoerde terwijl jullie beneden op Nine pasten. Ik heb het gevoel dat de komst van Nine en Bram in de familie en onze gezamenlijke liefde voor hen onze relatie heeft verdiept. Vorig jaar was het even spannend…ik hoop dat we nog lang met elkaar mogen genieten.

Marieke en Maurits, hoewel jullie ruim de leeftijd voorbij zijn dat ik mijzelf al heel wat vond blijft het soms gek om mijn "kleine" zusje en broertje aan de slag te zien. Samenwonen, werken, een boek publiceren, een ton ophalen voor een goed doel, een huis bouwen, emigreren. Het twee-tegen-één sarren cq de hersens inslaan van weleer heeft plaatsgemaakt voor een fatsoenlijke brussen-relatie, en dat we zo nu en dan iets doen met alleen ons drieën vind ik erg leuk. Hopelijk weten we dit straks trans-Atlantisch op een manier vol te houden. Ook de al lang niet meer koude kant, Dorus en Charlotte, dient genoemd te worden. Wanneer we met zijn allen samenzijn heb ik een echt gezellig familiegevoel. Kristien, bedankt voor het altijd warme welkom in Delft, sinds ik mij alweer ruim twaalf jaar geleden voor het eerst meldde. Voor hoe ontzettend leuk je als dappere solo-oma speelt met Nine en Bram, en voor je persoonlijke boodschappen op kaartjes. Koen en Andreia, ieder van ons had promotie sores, maar we hebben nu toch alledrie de eindstreep gehaald. Bedankt en obrigado voor de verscheidene met oud-Hollands Padvinder en ander vermaak gevulde schoonfamilieavondjes. Beste Dick, vorig jaar kreeg ik jouw in 1980 verdedigde proefschrift onder ogen, een mooie blik terug in de tijd. We dachten verschillend over cholesterol maar vonden elkaar in klassieke muziek en je liet me kennismaken met sushi en haute cuisine. Je was kort en krachtig de immens trotse opa van Nine. Het is ontzettend jammer dat je niet bij mijn verdediging bent.

In welke volgorde bedank je de belangrijkste mensen in je leven? Gelukkig heeft de wetenschap een oplossing gevonden voor deze kwestie: de gedeelde laatste plaats.

Lieve Bram, de echte reden dat ik zo lang over mijn proefschrift heb gedaan is natuurlijk dat ik jou erin wilde vermelden. Ik heb geen idee hoe kort of lang dit je kenmerkende eigenschappen gaan zijn, maar wat geniet ik van je knuffelbaarheid en van je vertederende gekir als ik je heen en weer schud. Ik waardeer je slaaprijke nachten en hoe je (meestal) bescheiden op je beurt wacht als je honger hebt terwijl ik bezig ben met je zus. Je herstelvermogen als ze je al te enthousiast een kopstoot verkoopt of een boek in je gezicht gooit. Ik kan niet wachten tot je de wereld verder gaat verkennen.

Lieve Nine, wat smelt ik van het dolle enthousiasme waarmee je me, bedekt in yoghurt, begroet als ik thuiskom na een werkdag. Ik geniet van hoe je net zo razend vrolijk kan worden van een schommel, dierenplaatje of iets dat blauw!-blauw!-blauw! is. Van hoe ik mijn lachen niet kan inhouden als ik je eigenlijk moet corrigeren om een streek. Van hoe liefdevol en vrij van jaloezie je bent naar Bram. Van je dansmoves die nu al beter zijn dan de mijne. Van het samen op lieveheersbeestjesmissie gaan. Ik kijk uit naar alle avonturen die nog komen.

Lieve Maart, wat bof ik ontzettend met jou. Bedankt voor je liefde, steun, gezelligheid, vertrouwen, empathie, vergevingsgezindheid, openheid, opvoedplezier en onze borden kaartspelletjeshobby - al twaalfenhalf jaar waarvan bijna vier in de echt, wat een indrukwekkende en gelukmakende cijfers! Bedankt dat je met tot de orde riep toen ik op het dieptepunt van mijn promotietraject zat maar hier niks over deelde. Ik ben blij dat we hetzelfde werk hebben, niet alleen vanwege de gedeelde interesse waardoor we over SGLT2-remmers kunnen praten tijdens het avondeten, maar vooral vanwege het wederzijdse begrip dat soms het thuisfront wordt verwaarloosd. Ik geniet van hoe we van ons huis een steeds fijner plekje aan het maken zijn. Ik grinnik om onze pogingen een strikt veganistische, milieubewuste yuppenleefstijl na te streven, die steevast gevolgd worden door een Bourgondisch borrelexces wanneer het vlees zwak is. Ik hou van groepsknuffels met onze kids, maar ook van samen af en toe nog lekker ouderwets uit de band springen op een feestje. We hebben Zuid-Amerika, Azië en Nieuw-Zeeland verkend, en zitten nu met net zoveel genoegen met ons gezinnetje op een camping aan de Moezel - oké, bijna dan. Wat is het heerlijk om met jou door het leven te gaan! :-D